

Entropy methods for Martingales

Y. Nishiyama

1991 MSC Classification: 60F05, 60G44, 62G20, 62M09. ISBN 90 6196 490 3 NUGI-code: 811

Copyright C2000, Stichting Mathematisch Centrum, Amsterdam Printed in the Netherlands

Preface

This is a revised version of my thesis at the University of Utrecht which appeared in May 1998. Let me quote here the preface for the original thesis.

December 1999, in Tokyo

Yoichi Nishiyama

Preface for my thesis

Let me begin with stating my personal history; readers who are not interested may skip to the next paragraph. In the spring of 1992, I was looking for a subject for my master's thesis. I took an interest in censoring problems, and learned that O.O. Aalen's paper in 1978 is the origin of the martingale approach to those problems. The approach has been one of the most active areas in statistics since the early 80's. (Needless to say, R.D. Gill's pioneering monograph in 1980 is also important.) Having read Aalen's paper, I made a conjecture, my first as a statistician, concerning the weak convergence of Nelson-Aalen's estimator in the multiplicative intensity model of point processes with general marks, where the estimator is considered as a set-indexed stochastic process. Unfortunately (or fortunately?), I was not able to solve it in my master's thesis, which consequently dealt with another problem. (At that time, I didn't know the prominent paper by R.M. Dudley in 1978; this means that I didn't know anything about the modern theory of empirical processes.) However, the conjecture brought me the motivation of my current research subject — how to manage entropy methods, which have been developed mainly for i.i.d. empirical processes, in the framework of martingales. The result up to the present is this thesis. An answer to the conjecture is presented in Section 4.1.

I would like to express my greatest gratitude to Prof. R.D. Gill for his advice, comments, kindness, patience and encouragement. He has always been the first reader of my drafts during the last two years, and gave me useful advice every time. His enthusiasm really accelerated my study. Also, although my stay in Utrecht was not originally intended to end up in a Doctor's degree, he has kindly given me this opportunity. It would be a great honor for me to succeed in obtaining a degree at the prestigious University of Utrecht.

I am really grateful to my supervisor in Osaka, Prof. N. Inagaki, for his general statistical advice and constant encouragement; without him, I might not be a statistician. I would like to thank Dr. N. Yoshida for stimulating discussion at many stages of my work since I moved to Tokyo; without him, my statistics might be much weaker. My thanks also go to: Dr. S. Aki for his lectures on empirical processes in the winter of 1993–1994, which inspired me to do this project; Prof. A.W. van der Vaart for a discussion in February 1997 which taught me the importance of maximal inequalities; Prof. S.A. van de Geer for a discussion in November 1997 which improved my understanding of *M*-estimation. I express my gratitude to Profs. N. Ikeda, K. Isii, S. Shirahata, M. Fukushima, R. Shimizu, K. Hirano, T. Matsunawa, Y. Ogata, S. Eguchi, J. Jacod, Yu.A. Kutoyants and B.Y. Levit for their advice and encouragement.

My work in Utrecht from June 1996 to May 1998 has been supported by a JSPS Fellowship for Research Abroad from the Japan Society for the Promotion of Science. These two years living here have made a strong positive impact on my study, and I hope that my scientific fruits may have achieved the Society's high expectations. I also owe a great deal to the Institute of Statistical Mathematics for allowing such a long period of leave, and for support from the Tokyo side; and to the kind hospitality of the Mathematical Institute at the University of Utrecht, which has made my period of work here exciting beyond my expectations. My special thanks go to the secretarial staff and computer managers both in Tokyo and Utrecht, and to the stochastics group in Utrecht, for their help and kindness; in particular, to Damien White for correcting grammatical errors and awkward sentenses contained in a draft of the preface, and to Erik van Zwet for translating the summary ¹ into Dutch.

April 1998, in Utrecht

Yoichi Nishiyama

¹Deleted from the CWI tract version

Contents

•	Pre	ace	i
1	Intr	oduction	1
	1.1	Overview	1
	1.2	General Notations and Remarks	7
2	Maz	imal Inequalities 10)
	2.1	Preliminaries	0
	2.2	Multivariate Point Processes	2
	2.3	Martingale Difference Arrays	4
	2.4	Continuous Local Martingales	9
	2.A	Notes	3
3	Wea	k Convergence Theorems 34	4
	3.1	Preliminaries	4
	3.2	Multivariate Point Processes	5
	3.3	Martingale Difference Arrays	1
	3.4	Continuous Local Martingales	3
	3.A	Notes	5
4	Inte	gral Estimators 4'	7
	4.1	Multiplicative Intensity Model	7
		4.1.1 Asymptotic Normality	7
		4.1.2 Asymptotic Efficiency	0
	4.2	Continuous Semimartingales with Non-linear Covariates	4
		4.2.1 Preliminaries \ldots 5-	4
		4.2.2 Asymptotic Normality	6
		4.2.3 Asymptotic Efficiency	0
	4.3	Counting Processes with Non-linear Covariates	3

		4.3.1 Preliminaries	64
		4.3.2 Asymptotic Normality	66
		4.3.3 Asymptotic Efficiency	68
	4.A	Notes	70
5	$M-\mathbf{I}$	Estimators: General Criterion and Euclidean Parameters	71
	5.1	General Criterion	71
	5.2	Gaussian White Noise Model	74
		5.2.1 Criterion for Rate of Convergence	74
		5.2.2 Examples	79
		5.2.3 Remarks for Non-Gaussian Cases	84
	5.3	Multiplicative Intensity Model	84
	5.A	Notes	88
6	Nor	n-parametric Maximum Likelihood Estimators	89
	6.1	Gaussian White Noise Model	89
	6.2	Multiplicative Intensity Model	93
	6.3	Counting Processes with Non-linear Covariates	00
	6.4	Diffusion-type Processes	06
	6.A	Notes	12
7	${ m Mis}$	cellanies 1	13
	7.1	Local Random Fields of Kernel Estimators	13
		7.1.1 I.I.D. Case	13
		7.1.2 Estimation of Mode \ldots 1	20
		7.1.3 Remarks for Non-I.I.D. Cases	24
	7.2	Log-likelihood Ratio Random Fields	25
		7.2.1 Results	25
		7.2.2 Proofs	28
	7.3	Model Checking for a Non-linear Times Series	32
	7.A	Notes	35
•	\mathbf{Bib}	liography 1	36

Chapter 1

Introduction

1.1 Overview

The purpose of this study is to develop entropy methods, which were first introduced for empirical processes of i.i.d. data, in order to handle some martingales with applications to statistical inference for stochastic processes. Among a lot of directions of statistical applications of the methods, we are concerned with two main themes in this monograph. The following brief description of them is intended to illustrate also the motivation of our work.

Theme 1: Asymptotic normality and efficiency in ℓ^{∞} -spaces. Let (E, \mathcal{E}) be a measurable space. Let $\{Z_i\}_{i\in\mathbb{N}}$ be a sequence of *E*-valued i.i.d. random variables with the common law *P*, and let Ψ be a subset of $\mathcal{L}^2(P) = \mathcal{L}^2(E, \mathcal{E}, P)$ with an envelope function belonging to $\mathcal{L}^2(P)$. We are interested in estimating the $\ell^{\infty}(\Psi)$ -valued unknown parameter $P = (P(\psi)|\psi \in \Psi)$ given by $P(\psi) = \int_E \psi(z)P(dz)$; a natural estimator is the empirical process $\mathbb{P}^n = (\mathbb{P}^n(\psi)|\psi \in \Psi)$ given by $\mathbb{P}^n(\psi) = n^{-1}\sum_{i=1}^n \psi(Z_i)$. The Donsker property is then nothing else than the asymptotic normality of \mathbb{P}^n about P; if the class Ψ is *P*-Donsker then the residual process $\sqrt{n}(\mathbb{P}^n - P)$ converges weakly in $\ell^{\infty}(\Psi)$ to a Brownian bridge indexed by Ψ . A sufficient condition for the class Ψ to be a Donsker class is that the class satisfies an integrability condition for metric entropy with $\mathcal{L}^2(P)$ -bracketing, given by Ossiander (1987), which we will recall later in this chapter.

When we have the asymptotic normality of an estimator, the next interest should be to show its asymptotic efficiency. For this purpose, a general procedure based on the Le Cam theory for finite- or infinite-dimensional parameters goes as follows (see e.g. Chapter 3.11 of van der Vaart and Wellner (1996)):

- (i) show the *local asymptotic normality* of a model;
- (ii) show the *differentiability* of an unknown parameter;
- (iii) show that the sequence of proposed estimators converges weakly to the distribu-

tion of the efficient bound specified in terms of some factors appearing in the steps (i) and (ii).

Then, the asymptotic efficiency in the sense of the local asymptotic minimax theorem with certain loss functions follows from the weak convergence shown at the step (iii) and the continuous mapping theorem. Furthermore if the proposed estimator can be shown to be *regular*, then the asymptotic efficiency in the sense of the convolution theorem is also fulfilled.

Van der Vaart and Wellner (1996) illustrated the usefulness of this approach by a discussion about the asymptotic efficiency of the estimator \mathbb{P}^n for the unknown parameter P (see their Section 3.11.1 for the details); the Donsker theorems for empirical processes are applied at the step (iii) above. On the other hand, a merit of the Le Cam theory is that, as seen in the step (i), the i.i.d. setup has been generalized up to local asymptotic normality. It is thus meaningful to present some new limit theorems, which should be useful at step (iii), in order to make full use of the general approach. In particular, such theorems in ℓ^{∞} -spaces were given mostly for row-independent arrays in the 80's, and have been recently established also for stationary sequences (see Notes to Chapter 3). We consider this subject in some martingale contexts.

Theme 2: Rate of convergence of *M*-estimators. Let $\theta \rightsquigarrow \gamma(\theta)$ be a deterministic process with parameter θ in a set Θ . Suppose that we are interested in estimating a maximum point θ_0 of the function $\theta \rightsquigarrow \gamma(\theta)$. If $\theta \rightsquigarrow \gamma(\theta)$ is well approximated by a stochastic process $\theta \rightsquigarrow \Gamma^n(\theta)$, a natural estimator would be a maximum point $\hat{\theta}^n$ of the latter, that is, an *M*-estimator with respect to the criterion process $\theta \rightsquigarrow \Gamma^n(\theta)$.

In the case of i.i.d. data, those processes are typically given by $\gamma(\theta) = P(\psi_{\theta})$ and $\Gamma^{n}(\theta) = \mathbb{P}^{n}(\psi_{\theta})$, where $\{\psi_{\theta} : \theta \in \Theta\}$ is a given class of elements of $\mathcal{L}^{1}(P)$ indexed by Θ . When the data is a sample from a density p_{θ} with respect to a measure on (E, \mathcal{E}) , the maximum likelihood estimator is an *M*-estimator for $\psi_{\theta} = \log p_{\theta}$. On the other hand, when $E = \Theta = \mathbb{R}$, if we set $\psi_{\theta} = 1_{[\theta-a,\theta+a]}$ for a constant a > 0, then the maximum point θ_{0} of $\theta \rightsquigarrow \gamma(\theta) = P([\theta - a, \theta + a])$ is something like a mode of the unknown distribution P.

Such *M*-estimation procedures allowing (Θ, d) to be a general metric space have been studied in recent years. A general approach to derive the rate of convergence requires the following (see Theorem 3.2.5 of van der Vaart and Wellner (1996) for the details; a version of the theorem, with some modifications, is given also in this monograph, namely Theorem 5.1.2):

(1.1.1)
$$\gamma(\theta) - \gamma(\theta_0) \le -d(\theta, \theta_0)^2$$
 in a neighborhood of θ_0 ;

<u>1.1. Overview</u>

(1.1.2)
$$E^* \sup_{d(\theta,\theta_0) \le \delta} |(\Gamma^n - \gamma)(\theta) - (\Gamma^n - \gamma)(\theta_0)| \le \phi^n(\delta) \text{ for small } \delta > 0$$

Here, $\delta \rightsquigarrow \phi^n(\delta)$ is an appropriate non-decreasing function. When we have checked those conditions, by choosing some constants $r_n > 0$ which satisfy $\phi^n(r_n^{-1}) \leq r_n^{-2}$, we can conclude that $r_n d(\hat{\theta}^n, \theta_0) = O_{P^*}(1)$ for *M*-estimators $\hat{\theta}^n = \operatorname{argmax}_{\theta \in \Theta} \Gamma^n(\theta)$. The crucial point of this approach is how to get a moment inequality for the residual processes $\theta \rightsquigarrow (\Gamma^n - \gamma)(\theta)$ as in (1.1.2). In the case of i.i.d. data mentioned above, the residual $(\Gamma^n - \gamma)(\theta)$ equals $(\mathbb{P}^n - P)(\psi_\theta)$, and the function $\phi^n(\delta)$ is typically of the form $\phi^n(\delta) =$ $n^{-1/2}\varphi(\delta)$ for a function $\delta \rightsquigarrow \varphi(\delta)$ not depending on *n*; the function $\varphi(\delta) = \delta$ leads to the standard rate $r_n = n^{1/2}$, while $\varphi(\delta) = \sqrt{\delta}$ does to the "cube root asymptotics" $r_n = n^{1/3}$. It should be noted, however, that this method possesses a good potential to be applied in much broader situations. As a matter of fact, some authors have already taken this kind of approach in non-i.i.d. settings, for instance regression models, but most of them are based on some maximal inequalities for i.i.d. empirical processes. With this aim in mind, we develop moment inequalities to obtain a bound (1.1.2) when the residual $(\Gamma^n - \gamma)(\theta)$ is the terminal variable of a martingale.

To handle martingales, we introduce a quantity called "quadratic modulus" in Chapter 2, which plays a key role in this work. For the sake of intuitive explanation, let us recall Ossiander's central limit theorem for i.i.d. sequences under the entropy condition for L^2 -bracketing, and next see how to generalize it to a dependent case; the idea of the quantity naturally appears there.

Let (E, \mathcal{E}) be a measurable space. Let $\{Z_i\}_{i \in \mathbb{N}}$ be a sequence of E-valued i.i.d. random variables with the common law P. Let Ψ be a subset of $\mathcal{L}^2(P)$ with an envelope function belonging to $\mathcal{L}^2(P)$. For every $\varepsilon \in (0, 1]$ choose $N(\varepsilon)$ pairs of elements of $\mathcal{L}^2(P)$, namely, $[l^{\varepsilon,k}, u^{\varepsilon,k}], k = 1, ..., N(\varepsilon)$, such that for every $\psi \in \Psi$ the relation $l^{\varepsilon,k} \leq \psi \leq u^{\varepsilon,k}$ holds for some k and that

(1.1.3)
$$\sqrt{\int_E |u^{\varepsilon,k}(z) - l^{\varepsilon,k}(z)|^2 P(dz)} \le \varepsilon.$$

Ossiander's theorem says that if this bracketing procedure can be accomplished with

(1.1.4)
$$\int_0^1 \sqrt{\log N(\varepsilon)} d\varepsilon < \infty.$$

then the sequence of stochastic processes $\psi \rightsquigarrow X^n(\psi)$ defined by

$$X^{n}(\psi) = \sqrt{n}(\mathbb{P}^{n} - P)(\psi)$$
$$= \frac{1}{\sqrt{n}} \sum_{i=1}^{n} \left\{ \psi(Z_{i}) - \int_{E} \psi(z) P(dz) \right\}$$

converges weakly in $\ell^{\infty}(\Psi)$ to a Brownian bridge indexed by Ψ .

Now, let $\{Z_i\}_{i\in\mathbb{N}}$ be an arbitrary sequence of *E*-valued random variables, and denote by P_i the conditional law of Z_i given $\mathcal{F}_{\flat-\infty} = \sigma\{Z_1, ..., Z_{i-1}\}$ (\mathcal{F}_0 is the null σ -field). We are interested in the sequence of stochastic processes $\psi \rightsquigarrow X^n(\psi)$ defined by

$$X^{n}(\psi) = \frac{1}{\sqrt{n}} \sum_{i=1}^{n} \left\{ \psi(Z_i) - \int_{E} \psi(z) P_i(dz) \right\}.$$

Consider the bracketing procedure as above with (1.1.3) replaced by

$$\sqrt{\int_{E} |u^{\varepsilon,k}(z) - l^{\varepsilon,k}(z)|^2 P_i(dz)} \le K_i \varepsilon \quad \text{almost surely},$$

where K_i is a random variable, not depending on ε and k, that is \mathcal{F}_{i-1} -measurable; since the left hand side is random in the present case, we have allowed the random coefficient K_i in the right hand side. Then, a result given in Chapter 3 (Theorem 3.2.2 or 3.3.1) says that if the entropy condition (1.1.4) is satisfied and if the sequence of random variables \overline{K}^n defined by

$$\overline{K}^n = \sqrt{\frac{1}{n} \sum_{i=1}^n |K_i|^2}$$

is bounded in probability, then the asymptotic tightness of the processes $\psi \rightsquigarrow X^n(\psi)$ follows from the finite-dimensional convergence and a Lindeberg condition on an envelope function of Ψ . Ossiander's theorem can be thought as the case of $P_i \equiv P$ and $K_i \equiv 1$. Some quantities "quadratic modulus", which we will define for three kinds of martingales in Chapter 2, have the same spirit as the random variables \overline{K}^n ; a closer explanation might be that

"quadratic modulus" =
$$\sup_{\varepsilon \in (0,1]} \max_{1 \le k \le N(\varepsilon)} \frac{\sqrt{n^{-1} \sum_{i=1}^{n} \int_{E} |u^{\varepsilon,k}(z) - l^{\varepsilon,k}(z)|^2 P_i(dz)}}{\varepsilon}$$

Since this random variable depends awkwardly on the choice of the brackets, and moreover since we will treat also random weight functions, we will actually define the quantities in a slightly different way based on a series of finite partitions of Ψ , avoiding the explicit construction of pairs $[l^{\varepsilon,k}, u^{\varepsilon,k}]$ in the definition of brackets.

The entropy methods were at first recognized to be useful to statisticians chiefly through efforts to seek for sharper and/or more general versions of uniform laws of large numbers and central limit theorems for empirical processes especially in multidimensional cases. However, some recent works have shown that a core part of them, namely, chaining and bracketing techniques controlled in terms of entropies, can be

<u>1.1. Overview</u>

applied also to other problems in statistics which are not directly connected to those limit theorems; a good example is M-estimation (recall Theme 2). From our point of view, an important advantage of the methods is that some of the techniques work well also for the dependent case above on the set $\{\overline{K}^n \leq L\}$ for a given constant L > 0. Hence, some problems of statistical inference for stochastic processes can be solved by handling some truncations such that the complements like $\{\overline{K}^n > L\}$ are asymptotically negligible for a fixed, large constant L. This is the basic outline of the approach which we frequently take in this work.

Chapters 2 and 3 are concerned with some abstract martingales, while the remaining chapters deal with concrete models in statistics. To be more precise, the stochastic processes treated in the former chapters are some classes $(X^{\psi}|\psi \in \Psi)$ of martingales, indexed by an arbitrary set Ψ , in the sense that each coordinate process $t \rightsquigarrow X_t^{\psi}$ for every ψ is an \mathbb{R} -valued martingale. We consider the following three situations: (i) each coordinate process $t \rightsquigarrow X_t^{\psi}$ is represented as a stochastic integral, namely,

$$\begin{split} X_t^{\psi} &= W^{\psi} * (\mu - \nu)_t \\ &= \int_{[0,t] \times E} W^{\psi}(\omega, s, z)(\mu(\omega; ds, dz) - \nu(\omega; ds, dz)), \end{split}$$

where $W^{\psi} = W^{\psi}(\omega, t, z)$ is a predictable function on $\Omega \times \mathbb{R}_+ \times E$, μ is an *E*-valued multivariate point process, and ν is the predictable compensator of μ ; (ii) each process $t \rightsquigarrow X_t^{\psi}$ is a partial sum process of a discrete time martingale, namely,

$$X_t^{\psi} = \sum_{i=1}^{\sigma_t} \xi_i^{\psi},$$

where $\{\xi_i^{\psi}\}_{i\in\mathbb{N}}$ is a discrete time martingale and $(\sigma_t)_{t\in\mathbb{R}_+}$ is an increasing family of finite stopping times; (iii) each process $t \rightsquigarrow X_t^{\psi}$ is a continuous local martingale. There are three reasons why we choose the martingales as the objects of our study. First, the Bernstein inequality, which is a basic tool in the i.i.d. case, is already provided also in the framework of martingales with the modification that a truncation based on the predictable quadratic variation is introduced (Lemma 2.1.1). The second reason is that we can take advantage of the well-developed martingale central limit theorems to establish the finite-dimensional convergence in our situation. Last, but not least, the martingale is a vital concept in analyzing a rich class of statistical models, including the multiplicative intensity model for survival data, Markov chains, the Gaussian white noise model, and diffusion processes derived from stochastic differential equations.

The organization of the monograph is as follows. In Chapter 2, we introduce the quantities "quadratic modulus" and "exponential modulus" for the three kinds of mar-

tingales above, and establish maximal inequalities, namely, some bounds for

$$E \sup_{t} \sup_{\psi,\phi \in \Psi} |X_t^{\psi} - X_t^{\phi}| \mathbf{1}_B$$

with the truncation by the set B in terms of the modulus. Those inequalities are not asymptotic estimates, and are applied not only for the proofs of weak convergence theorems in Chapter 3 but also as a crucial tool to derive the rate of convergence of Mestimators in Chapters 5 and 6. As for the case (iii) of continuous local martingales, we study also the continuity of the sample paths along the direction of parameter $\psi \sim X_t^{\psi}$.

Chapter 3 is devoted to weak convergence theorems for the three kinds of martingales. An essential part of the proofs is the asymptotic tightness, which is established by using the maximal inequalities in Chapter 2. As we have mentioned above, the sufficient condition that we present is that the quadratic modulus is bounded in probability and that an entropy condition of the type (1.1.4) is satisfied. In particular, natural generalizations of Jain-Marcus' and Ossiander's central limit theorems are presented. The entropy condition for the cases (i) and (ii) above is analogous to that for L^2 -bracketing, while the case (iii) is based on the metric entropy condition without bracketing. The results of this chapter are repeatedly applied to derive the asymptotic distribution of estimators in the subsequent chapters.

Some results concerning Theme 1 are given in Chapter 4. We consider the multiplicative intensity model for point processes with general marks, and derive the asymptotic normality and efficiency in ℓ^{∞} -spaces of a generalized Nelson-Aalen estimator. An interesting difference from the i.i.d. case, where the L^2 -bracketing condition is optimal, is that an L^p -bracketing condition with $p \in [2, \infty]$ is sometimes preferable; this fact is valid also for other problems in this monograph, and the multiplicative intensity model provides a good illustration. We also study two non-linear models, of continuous semimartingales and of counting processes, both with time-dependent covariates.

Theme 2, the *M*-estimation procedure, is studied in Chapters 5 and Chapter 6 stressing non-standard rates of convergence. First, in Section 5.1, we present a general criterion for rate of convergence. A difference from known results in this area is that a kind of "twice differentiability" of criterion functions is generalized to a "*p*-times differentiability", that is, " $d(\theta, \theta_0)^2$ " appearing in (1.1.1) is replaced by " $d(\theta, \theta_0)^p$ ". Sections 5.2 and 5.3 are concerned with some estimation problems of Euclidean parameters in the Gaussian white noise model and the multiplicative intensity model, respectively. Jump point estimation, among other things, is considered in those models.

Chapter 6 is devoted to the study of rate of convergence of non-parametric maximum likelihood estimators. The models considered there are the Gaussian white noise model,

the multiplicative intensity model, a counting process model with non-linear covariates, and the diffusion type processes. The third model above contains a discussion about the Lexis diagram, which is important in the context of survival analysis.

The last chapter contains three independent topics. Since the setups of these problems are simple, this chapter, as well as Section 4.1, perhaps gives a guideline of the usage of the weak convergence theorems in Chapter 3. Except for Subsection 7.1.2, we do not use any results presented in Chapters 4, 5 and 6. In Section 7.1, we study the asymptotics of local random fields of kernel type estimators. The results are applied to the problem of estimating the mode of a density function; we derive the asymptotic behavior of an estimator defined as the argmax of kernel density estimator by using also the general theorem for M-estimators presented in Section 5.1. Section 7.2 is devoted to deriving the asymptotic behavior of log-likelihood ratio random fields in a general discrete-time statistical experiment with abstract parameters. An application to Markov chains is also discussed. In Section 7.3, we study a testing problem for a non-parametric regression model with dependent noise.

1.2 General Notations and Remarks

- (1) $\mathbb{R} = (-\infty, \infty); \mathbb{R}_+ = [0, \infty); \mathbb{N} = \{1, 2, ...\}; \mathbb{N}_0 = \{0\} \cup \mathbb{N}; \mathbb{Z} = \{\text{integers}\}; \mathbb{Q} = \{\text{rational numbers}\}.$
- (2) We denote by | · | the Euclidean norm, even in the multi-dimensional case. We denote by Leb(B) the Lebesgue measure of a Borel measurable subset B of a Euclidean space.
- (3) The inequality " $x \leq y$ " $(x, y \in [0, \infty)$) means that there exists a universal constant C > 0 such that $x \leq Cy$.
- (4) $\ell^{\infty}(\Psi)$ denotes the set of all bounded functions defined on a set Ψ . We denote by $\|\cdot\|_{\infty}$ the supremum-norm on $\ell^{\infty}(\Psi)$.
- (5) For every $p \in [1, \infty]$, we denote by $\mathcal{L}^p(E, \mathcal{E}, \lambda)$ the set of all *p*-integrable functions defined on a measure space $(E, \mathcal{E}, \lambda)$ (when $p = \infty$, it means the set of all λ essentially bounded functions), and by $L^p(E, \mathcal{E}, \lambda)$ the equivalent classes of elements of $\mathcal{L}^p(E, \mathcal{E}, \lambda)$. These are often abbreviated to \mathcal{L}^p and L^p , respectively. This kind of notational abbreviations of spaces are given section-wisely.
- (6) $\operatorname{Card}(\Psi)$ denotes the number of the elements of a set Ψ , allowing ∞ .

- (7) $\operatorname{Diam}(\Psi, \rho)$ denotes the diameter of a set Ψ with respect to a semimetric ρ .
- (8) When a semimetric space (S, ρ) is given, we denote by B_(S,ρ)(x; ε) the closed ball with center x ∈ S and ρ-radius ε > 0; when there is no danger of confusion, it is sometimes denoted by B_ρ(x; ε) or even by B(x; ε). This notational abbreviation is also given section-wisely.
- (9) When a semimetric space (S, ρ) and a subset Ψ of S are given, we denote by $N(\Psi, \rho; \varepsilon)$ the minimum number of closed balls with ρ -radius $\varepsilon > 0$ which cover Ψ , allowing ∞ . The centers of the balls need not belong to Ψ .
- (10) Let S be a linear space of \mathbb{R} -valued functions ψ defined on a set, and let a seminorm $\|\cdot\|$ on S be given. For a given pair $l, u \in S$, we denote $[l, u] = \{\psi \in S : l \leq \psi \leq u\}$. Such [l, u] is called a $(\|\cdot\|, \varepsilon)$ -bracket in S if $\|u - l\| \leq \varepsilon$. For a given class $\Psi \subset S$, the bracketing number $N_{[]}(\Psi, \|\cdot\|; \varepsilon)$ is the minimum number of $(\|\cdot\|, \varepsilon)$ -brackets which cover Ψ ; that is, the smallest $N \in \mathbb{N} \cup \{\infty\}$ such that: there exists $l^k, u^k \in S$, k = 1, ..., N, such that $\Psi \subset \bigcup_{k=1}^{N} [l^k, u^k]$ and that $\|u^k - l^k\| \leq \varepsilon$ for all k.
- (11) Let $\alpha, H > 0$ be given, and denote by $\underline{\alpha}$ the greatest integer strictly smaller than α . Let a bounded, convex subset E in \mathbb{R}^d with nonempty interior E^i be given. We denote by $C^{\alpha}_H(E)$ the set of all continuous functions $f : E \to \mathbb{R}$ such that $\|f\|_{\alpha} \leq H$, where

$$\|f\|_{\alpha} = \max_{k \leq \underline{\alpha}} \sup_{x \in E^{i}} |D^{k}f(x)| + \max_{k = \underline{\alpha}} \sup_{x, y \in E^{i} \atop x \neq y} \frac{|D^{k}f(x) - D^{k}f(y)|}{|x - y|^{\alpha - \underline{\alpha}}}$$

with the notations $k_{\cdot} = \sum_{i=1}^{d} k_i$ and

$$D^k = \frac{\partial^{k}}{\partial x_1^{k_1} \cdots \partial x_d^{k_d}}$$

for every vector $k = (k_1, ..., k_d)$ of d non-negative integers. It is well-known that there exists a constant K > 0 depending only on α and d such that

$$\log N(C_H^{\alpha}(E), \|\cdot\|_{\infty}; \varepsilon) \le K \cdot \operatorname{Leb}(E_+) \left(\frac{H}{\varepsilon}\right)^{d/\alpha} \qquad \forall \varepsilon > 0$$

where $E_+ = \{x : |x - E| < 1\}$ (see, e.g., Theorem 2.7.1 of van der Vaart and Wellner (1996)).

(12) A random semimetric ρ on a set T is a collection $\{\rho(\omega; \cdot, \cdot) : \omega \in \Omega\}$ of semimetric on T indexed by a probability space (Ω, \mathcal{F}, P) , although we do not require any measurability. We often denote a random semimetric by ρ and a (non-random) semimetric by ρ .

- (13) The words increasing and decreasing mean "non-decreasing" and "non-increasing", respectively (the situation where we should use the words strictly increasing or strictly decreasing does not appear in the monograph).
- (14) We follow the standard definitions and notations of the martingale theory, which can be found in the book by Jacod and Shiryaev (1987).
- (15) We refer to Part 1 of van der Vaart and Wellner (1996) for the weak convergence theory which does not require the measurability of random sequences. In particular, see their Chapter 1.2 for the definitions of the notations E*, E_{*}, P* and P_{*} that mean the outer integral, inner integral, outer probability and inner probability, respectively. Let (X, d) be a metric space, and for every n ∈ N let Xⁿ be a mapping from a probability space (Ωⁿ, Fⁿ, Pⁿ) to X. We denote by "Xⁿ ⇒ X in X" the weak convergence of Xⁿ to a tight, Borel measurable random element X taking values in X; by "Xⁿ → c" the convergence in P^{n*}-probability to a non-random element c of X; by "Xⁿ → c" the convergence in Pⁿ-probability (in this case Xⁿ is assumed to be Borel measurable).

Chapter 2

Maximal Inequalities

2.1 Preliminaries

This chapter is devoted to getting some bounds for expectation of supremum of martingales up to a universal constant; throughout we use the notation " \leq " given in (3) of Section 1.2. The present section prepares two things, namely, quotation of two known inequalities which are used in Sections 2.2, 2.3 and 2.4, and introduction of two definitions which are necessary to formulate a quantity "quadratic modulus" in Sections 2.2 and 2.3. Thus, readers who are interested only in continuous local martingales studied in Section 2.4 may skip the latter.

First, let us state two lemmas which are well-known. The first one is the Bernstein inequality for martingales with bounded jumps; see e.g. Section 4.13 of Liptser and Shiryaev (1989) for the proof. The second one, which is used in connection with the former, is an adaptation of Lemma 2.2.10 of van der Vaart and Wellner (1996).

Lemma 2.1.1 Let $t \rightsquigarrow X_t$ be an \mathbb{R} -valued, locally square-integrable martingale such that $X_0 = 0$ and that $|\Delta X| \leq a$ for a constant $a \geq 0$, and τ a bounded stopping time. Then, it holds that for every $\Gamma > 0$

$$P\left(\sup_{t\in[0,\tau]}|X_t|>\varepsilon, \ \langle X,X\rangle_{\tau}\leq\Gamma\right)\leq 2\exp\left(-\frac{\varepsilon^2}{2(a\varepsilon+\Gamma)}\right)\qquad\forall\varepsilon>0.$$

Lemma 2.1.2 Let $N \in \mathbb{N}$ and let $X_1, ..., X_N$ be arbitrary \mathbb{R} -valued random variables. Assume that for a measurable set B and some constants $a \ge 0$ and $\Gamma > 0$

$$P(|X_i| > \varepsilon, B) \le 2 \exp\left(-\frac{\varepsilon^2}{2(a\varepsilon + \Gamma)}\right) \qquad \forall \varepsilon > 0, \ \forall i = 1, ..., N.$$

Then, it holds that

$$E \max_{1 \le i \le N} |X_i| \mathbb{1}_B \lesssim a \log(1+N) + \sqrt{\Gamma \log(1+N)}.$$

2.1. Preliminaries

Combining these inequalities, we can easily get the following.

Corollary 2.1.3 Let $N \in \mathbb{N}$. Let $t \rightsquigarrow X_t = (X_t^1, ..., X_t^N)$ be an \mathbb{R}^N -valued, locally square-integrable martingale such that $X_0^i = 0$ and that $|\Delta X^i| \leq a$ for a constant $a \geq 0$, and let τ be a finite stopping time. Then, for any constant K satisfying

$$a\sqrt{\log(1+N)} \le K,$$

it holds that

$$E \sup_{t \in [0,\tau]} \max_{1 \le i,j \le N} |X_t^i - X_t^j| \mathbf{1}_B \lesssim K\sqrt{\log(1+N)}$$

where

$$B = \left\{ \max_{1 \le i, j \le N} \sqrt{\langle X^i - X^j, X^i - X^j \rangle_{\tau}} \le K \right\}.$$

The purpose of this chapter is to study what happens in the case of " $N = \infty$ ". We consider this problem for three kinds of martingales in Sections 2.2, 2.3 and 2.4, respectively.

Next, let us give two definitions for Sections 2.2 and 2.3.

Definition 2.1.4 Let $(\mathcal{X}, \mathcal{A}, \lambda)$ be a σ -finite measure space. For a given mapping $Z : \mathcal{X} \to \mathbb{R} \cup \{\infty\}$, we denote by $[Z]_{\mathcal{A},\lambda}$ any \mathcal{A} -measurable function $U : \mathcal{X} \to \mathbb{R} \cup \{\infty\}$ such that:

(i) $U \ge Z$ holds identically;

(ii) $\tilde{U} \geq U$ holds λ -almost everywhere, for every \mathcal{A} -measurable function \tilde{U} such that $\tilde{U} \geq Z$ holds λ -almost everywhere.

The existence of such a random variable $[Z]_{\mathcal{A},\lambda}$ and its uniqueness up to a λ -negligible set can be shown by using Lemma 1.2.1 of van der Vaart and Wellner (1996).

Definition 2.1.5 Let Ψ be an arbitrary set such that $Card(\Psi) = \infty$.

 $\Pi = \{\Pi(\varepsilon)\}_{\varepsilon \in (0,\Delta_{\Pi}]}, \text{ where } \Delta_{\Pi} \in (0,\infty) \cap \mathbb{Q}, \text{ is called a Decreasing series of Finite Partitions (DFP) of <math>\Psi$ if it satisfies the following (i), (ii) and (iii):

(i) each $\Pi(\varepsilon) = \{\Psi(\varepsilon; k) : 1 \leq k \leq N_{\Pi}(\varepsilon)\}$ is a finite partition of Ψ , that is, $\Psi = \bigcup_{k=1}^{N_{\Pi}(\varepsilon)} \Psi(\varepsilon; k)$ and $\Psi(\varepsilon; k_1) \cap \Psi(\varepsilon; k_2) = \emptyset$ whenever $k_1 \neq k_2$;

(ii)
$$N_{\Pi}(\Delta_{\Pi}) = 1$$
 and $\lim_{\varepsilon \downarrow 0} N_{\Pi}(\varepsilon) = \infty$;

(iii) $N_{\Pi}(\varepsilon) \ge N_{\Pi}(\varepsilon')$ whenever $\varepsilon \le \varepsilon'$.

 $\Pi = \{\Pi(\varepsilon)\}_{\varepsilon \in (0,\Delta_{\Pi}]}, \text{ where } \Delta_{\Pi} \in (0,\infty) \cap \mathbb{Q}, \text{ is called a Nested series of Finite} \\ Partitions (NFP) of \Psi \text{ if it satisfies the above (i) and (ii) and the following (iii'):} \\ (\text{iii'}) \Pi(\varepsilon) \text{ is a refinement of } \Pi(\varepsilon') \text{ whenever } \varepsilon \leq \varepsilon'.$

The ε -entropy $H_{\Pi}(\varepsilon)$ and the modified ε -entropy $H_{\Pi}(\varepsilon)$ of a DFP Π are defined by:

$$\begin{aligned} H_{\Pi}(\varepsilon) &= \sqrt{\log N_{\Pi}(\varepsilon)}; \\ \tilde{H}_{\Pi}(\varepsilon) &= \sqrt{\log(1+N_{\Pi}(\varepsilon))}. \end{aligned}$$

Notice that any NFP is a DFP. Although the converse is not true, we can sometimes construct a new NFP from a given DFP, due to Lemma 2.2.2 given later, without loss of generality for our purpose. Notice also that for any DFP Π

$$\int_0^{\delta} \tilde{H}_{\Pi}(\varepsilon) d\varepsilon \le \delta \sqrt{\log 2} + \int_0^{\delta} H_{\Pi}(\varepsilon) d\varepsilon \qquad \forall \delta \in (0, \Delta_{\Pi}]$$

if the integral on the right hand side is finite; in fact, it holds that

(2.1.1)
$$\sqrt{\log(1+N)} \le \sqrt{\log 2N} \le \sqrt{\log 2} + \sqrt{\log N} \quad \forall N \ge 1.$$

2.2 Multivariate Point Processes

Let (E, \mathcal{E}) be a Blackwell space, and let $\mathbf{B} = (\Omega, \mathcal{F}, \mathbf{F} = (\mathcal{F}_t)_{t \in \mathbb{R}_+}, P)$ be a stochastic basis. We put $\tilde{\Omega} = \Omega \times \mathbb{R}_+ \times E$ and $\tilde{\mathcal{P}} = \mathcal{P} \otimes \mathcal{E}$, where \mathcal{P} is the predictable σ -field on $\Omega \times \mathbb{R}_+$. For a given predictable function W on $\tilde{\Omega}$ and a given random measure μ on $\mathbb{R}_+ \times E$, we denote by $W * \mu$ the integral process defined as the path-wise Lebesgue-Stieltjes integral: for every $t \in \mathbb{R}_+$,

$$W * \mu_t(\omega) = \begin{cases} \int_{[0,t] \times E} W(\omega, s, x) \mu(\omega; ds, dx) & \text{if } \int_{[0,t] \times E} |W(\omega, s, x)| \mu(\omega; ds, dx) < \infty, \\ \infty & \text{otherwise.} \end{cases}$$

(See Section II.1a of Jacod and Shiryaev (1987) for the detail).

Let μ be an *E*-valued multivariate point process. Let ν be a "good" version of the predictable compensator of μ (thus $\nu(\omega; \{t\} \times E) \leq 1$). We introduce the Doléans measure M_{ν}^{P} on $(\tilde{\Omega}, \tilde{\mathcal{P}})$, which is $\tilde{\mathcal{P}}$ - σ -finite, given by

$$M^P_{\nu}(d\omega, dt, dx) = P(d\omega)\nu(\omega; dt, dx)$$

(See Section II.1b, Definition III.1.23 and III.3.15 of Jacod and Shiryaev (1987)).

Let $\mathcal{W} = \{W^{\psi} : \psi \in \Psi\}$ be a family of predictable functions on Ω indexed by an arbitrary set Ψ . We give a definition, using the notation of Definition 2.1.4, which plays the key role in our context.

Definition 2.2.1 The predictable envelope \overline{W} of $\mathcal{W} = \{W^{\psi} : \psi \in \Psi\}$ is defined by

$$\overline{W} = \left[\sup_{\psi \in \Psi} |W^{\psi}| \right]_{\tilde{\mathcal{P}}, M_{\nu}^{P}}$$

For a given DFP Π of Ψ , the quadratic Π -modulus $\|\mathcal{W}\|_{\Pi}$ of $\mathcal{W} = \{W^{\psi} : \psi \in \Psi\}$ is defined as the $[0, \infty]$ -valued increasing process $t \rightsquigarrow \|\mathcal{W}\|_{\Pi, t}$ given by

$$\|\mathcal{W}\|_{\Pi,t} = \sup_{\varepsilon \in (0,\Delta_{\Pi}] \cap \mathbb{Q}} \max_{1 \le k \le N_{\Pi}(\varepsilon)} \frac{\sqrt{|W(\Psi(\varepsilon;k))|^2 * \nu_t}}{\varepsilon} \qquad \forall t \in \mathbb{R}_+,$$

where

(2.2.1)
$$W(\Psi') = \left[\sup_{\psi,\phi\in\Psi'} |W^{\psi} - W^{\phi}|\right]_{\tilde{\mathcal{P}},M^{P}_{\nu}} \quad \forall \Psi' \subset \Psi.$$

For a given DFP Π of Ψ and a given constant a > 0, the exponential (Π, a) -modulus $\|\mathcal{W}\|_{\Pi}^{\mathcal{E}_a}$ of $\mathcal{W} = \{W^{\psi} : \psi \in \Psi\}$ is defined as the $[0, \infty]$ -valued increasing process $t \rightsquigarrow \|\mathcal{W}\|_{\Pi,t}^{\mathcal{E}_a}$ given by

$$\|\mathcal{W}\|_{\Pi,t}^{\mathcal{E}_a} = \sup_{\varepsilon \in (0,\Delta_{\Pi}] \cap \mathbb{Q}} \max_{1 \le k \le N_{\Pi}(\varepsilon)} \frac{\sqrt{(\mathcal{E}_a(W(\Psi(\varepsilon;k)))) * \nu_t}}{\varepsilon} \qquad \forall t \in \mathbb{R}_+.$$

where

(2.2.2)
$$\mathcal{E}_a(x) = \begin{cases} 2a^2 (\exp(a^{-1}x) - 1 - a^{-1}x) & \forall x \in [0, \infty), \\ \infty & x = \infty, \end{cases}$$

and $W(\Psi')$ is defined by (2.2.1) for every $\Psi' \subset \Psi$.

It is clear that there exist some increasing versions of $t \rightsquigarrow \|\mathcal{W}\|_{\Pi,t}$ and $t \rightsquigarrow \|\mathcal{W}\|_{\Pi,t}^{\mathcal{E}_a}$ uniquely up to a *P*-negligible set, respectively. It holds that $\|\mathcal{W}\|_{\Pi,t} \leq \|\mathcal{W}\|_{\Pi,t}^{\mathcal{E}_a}$ almost surely, since $|x|^2 \leq \mathcal{E}_a(x)$. Notice also that all of \overline{W} , $\|\mathcal{W}\|_{\Pi}$ and $\|\mathcal{W}\|_{\Pi}^{\mathcal{E}_a}$ depend on \mathbf{F} , *P* and ν , through $\tilde{\mathcal{P}}$ and M_{ν}^{P} . One may find that the exponential modulus above is based on the "Bernstein norm" (see 324 page of van der Vaart and Wellner (1996) for the i.i.d. case; a discrete martingale version is contained in van de Geer (1997)).

Lemma 2.2.2 For any DFP Π of Ψ such that $\int_0^{\Delta_{\Pi}} H_{\Pi}(\varepsilon) d\varepsilon < \infty$, there exists a NFP Π' such that:

$$\begin{aligned} \Delta_{\Pi'} &= \Delta_{\Pi}; \\ \int_{0}^{\Delta_{\Pi'}} H_{\Pi'}(\varepsilon) d\varepsilon &\leq 4 \int_{0}^{\Delta_{\Pi}} H_{\Pi}(\varepsilon) d\varepsilon; \\ \int_{0}^{\Delta_{\Pi'}} \tilde{H}_{\Pi'}(\varepsilon) d\varepsilon &\leq 4 \int_{0}^{\Delta_{\Pi}} \tilde{H}_{\Pi}(\varepsilon) d\varepsilon; \\ \|\mathcal{W}\|_{\Pi',t} &\leq \|\mathcal{W}\|_{\Pi,t} \quad \forall t \in \mathbb{R}_{+}. \end{aligned}$$

Proof. For every $\varepsilon \in (0, \Delta_{\Pi}]$, let us define

$$\Pi'(\varepsilon) = \bigvee_{i_0 \le j \le i} \Pi(2^{-j}) \quad \text{if} \quad \varepsilon \in [2^{-i}, 2^{-i+1}) \cap (0, \Delta_{\Pi}], \quad i \ge i_0,$$

where $i_0 = \min\{i \in \mathbb{Z} : 2^{-i} \leq \Delta_{\Pi}\}$. Then, the constructed $\Pi' = \{\Pi'(\varepsilon)\}_{\varepsilon \in (0,\Delta_{\Pi'}]}$ is a NFP such that $\Delta_{\Pi'} = \Delta_{\Pi}$. Further, since $N_{\Pi'}(2^{-i}) \leq \prod_{i_0 \leq j \leq i} N_{\Pi}(2^{-j})$ we have

$$\int_{0}^{\Delta_{\Pi'}} H_{\Pi'}(\varepsilon) d\varepsilon \leq \sum_{i=i_{0}}^{\infty} 2^{-i} H_{\Pi'}(2^{-i})$$

$$\leq \sum_{i=i_{0}}^{\infty} 2^{-i} \sum_{j=i_{0}}^{i} H_{\Pi}(2^{-j})$$

$$= \sum_{j=i_{0}}^{\infty} H_{\Pi}(2^{-j}) \sum_{i=j}^{\infty} 2^{-i}$$

$$= 4 \sum_{j=i_{0}}^{\infty} 2^{-j-1} H_{\Pi}(2^{-j})$$

$$\leq 4 \int_{0}^{\Delta_{\Pi}} H_{\Pi}(\varepsilon) d\varepsilon.$$

The same argument is valid also for the modified entropies. The last inequality is trivial from the construction of Π' .

Supposing that there exists a version of \overline{W} such that $\overline{W} * \nu_t(\omega) < \infty$ for all $\omega \in \Omega$ and $t \in \mathbb{R}_+$, let us define the random variables X_t^{ψ} and $X_t^{a,\psi}$ by

(2.2.3)
$$X_t^{\psi} = W^{\psi} * (\mu - \nu)_t \qquad \forall t \in \mathbb{R}_+ \ \forall \psi \in \Psi$$

and

(2.2.4)
$$X_t^{a,\psi} = W^{\psi} \mathbf{1}_{\{\overline{W} \le a\}} * (\mu - \nu)_t \qquad \forall t \in \mathbb{R}_+ \ \forall \psi \in \Psi \ \forall a > 0,$$

respectively. Then, the process $t \rightsquigarrow X_t^{\psi}$ and the process $t \rightsquigarrow X_t^{a,\psi}$ is a locally squareintegrable martingale on **B**, both of which have finite variation. (see Lemma I.3.10 and Proposition II.1.28 of Jacod and Shiryaev (1987)). The following theorem gives some maximal inequalities for these processes in terms of $\|\mathcal{W}\|_{\Pi}$.

Theorem 2.2.3 Let μ be an *E*-valued multivariate point process defined on a stochastic basis **B**, and ν a "good" version of the predictable compensator of μ . Let $\mathcal{W} = \{W^{\psi} : \psi \in \Psi\}$ be a family of predictable functions on $\tilde{\Omega}$, indexed by an arbitrary set Ψ , such that $\overline{W} * \nu_t(\omega) < \infty$ for all $\omega \in \Omega$ and $t \in \mathbb{R}_+$ for a version of predictable envelope \overline{W} of \mathcal{W} . Let τ be a finite stopping time. Then, we have the following (i) and (ii).

(i) It holds for any NFP Π of Ψ and any constants $\delta \in (0, \Delta_{\Pi}]$ and K > 0 that

$$E^* \sup_{t \in [0,\tau]} \sup_{\substack{1 \le k \le N_{\Pi}(\delta) \\ \psi, \phi \in \Psi(\delta;k)}} |X_t^{a,\psi} - X_t^{a,\phi}| \mathbf{1}_{\{||\mathcal{W}||_{\Pi,\tau} \le K\}} \lesssim K \int_0^{\delta} \tilde{H}_{\Pi}(\varepsilon) d\varepsilon,$$

where the random variables $X_t^{a,\psi}$ are defined by (2.2.4) with $a = a(\delta, K) = \delta K / \tilde{H}_{\Pi}(\delta/2)$. (ii) It holds for any DFP Π of Ψ and any constants K, L > 0 that

$$E^* \sup_{t \in [0,\tau]} \sup_{\psi,\phi \in \Psi} |X_t^{\psi} - X_t^{\phi}| \mathbf{1}_{\{||\mathcal{W}||_{\Pi,\tau} \le K, |\overline{W}|^2 * \nu_{\tau} \le L\}} \lesssim K \int_0^{\Delta_{\Pi}} \tilde{H}_{\Pi}(\varepsilon) d\varepsilon + \frac{L}{\Delta_{\Pi} K},$$

where the random variables X_t^{ψ} are defined by (2.2.3).

Theorem 2.2.4 Consider the same situation as Theorem 2.2.3. It holds for any random semimetric ϱ on Ψ , any NFP Π of Ψ and any constants $\delta \in (0, \Delta_{\Pi}]$ and K > 0 that

$$E^* \sup_{t \in [0,\tau]} \sup_{\substack{\psi,\phi \in \Psi\\ \varrho(\psi,\phi) \le \delta}} |X_t^{a,\psi} - X_t^{a,\phi}| \mathbb{1}_{\{||\mathcal{W}||_{\Pi,\tau} \le K\} \cap B} \lesssim K \int_0^\delta \tilde{H}_{\Pi}(\varepsilon) d\varepsilon,$$

where

(2.2.5)
$$B = \left\{ \sup_{\substack{\psi,\phi \in \Psi\\ \varrho(\psi,\phi) \le \delta}} \frac{\sqrt{\langle X^{a,\psi} - X^{a,\phi}, X^{a,\psi} - X^{a,\phi} \rangle_{\tau}}}{\delta} \le K \right\}$$

and the random variables $X_t^{a,\psi}$ are defined by (2.2.4) with $a = a(\delta, K) = \delta K / \tilde{H}_{\Pi}(\delta/2)$.

Proof of Theorem 2.2.3 (i). Fix any $\delta, K > 0$; we may assume $\delta \in \mathbb{Q}$ without loss of generality. For every integer $p \ge 0$, we set

$$a_p = 2^{-p+1} \delta K / \tilde{H}_{\Pi}(2^{-p-1} \delta).$$

Next, choosing an element $\psi_{p,k}$ from each partitioning set $\Psi(2^{-p}\delta;k)$ such that

$$\{\psi_{p,k}: 1 \le k \le N_{\Pi}(2^{-p}\delta)\} \subset \{\psi_{p+1,k}: 1 \le k \le N_{\Pi}(2^{-(p+1)}\delta)\},\$$

we define for every $\psi \in \Psi$

$$\begin{cases} \pi_p \psi = \psi_{p,k}, \\ \Pi_p \psi = \Psi(2^{-p}\delta;k), \end{cases} \quad \text{if } \psi \in \Psi(2^{-p}\delta;k). \end{cases}$$

For every integer $q \ge 1$, we introduce the stopping time

$$\tau_q = \inf\left\{t \in \mathbb{R}_+ : \nu([0,t] \times E) > \frac{|\tilde{H}_{\Pi}(2^{-q-2}\delta)|^2}{16} - 1\right\} \wedge \tau.$$

Since $\nu([0,\tau] \times E) < \infty$ almost surely and $\lim_{\varepsilon \downarrow 0} N_{\Pi}(\varepsilon) = \infty$, it holds that $\tau_q \uparrow \tau$ as $q \to \infty$ almost surely. Hence it is enough to show that

$$(2.2.6) \quad E^* \sup_{t \in [0,\tau_q]} \sup_{\psi \in \Psi} |X_t^{a,\psi} - X_t^{a,\pi_0\psi}| \mathbf{1}_{\{\|\mathcal{W}\|_{\Pi,\tau} \le K\}} \lesssim K \int_0^\delta \tilde{H}_{\Pi}(\varepsilon) d\varepsilon \qquad \forall q \ge 1,$$

where $a = a(\delta, K)$.

Let us now fix any integer $q \geq 1$, and denote $\tau = \tau_q$ [there should be no danger of confusion]. For every p = 0, 1, ..., q, we consider the predictable functions $W(\Pi_p \psi)$ on $\tilde{\Omega}$ defined by (2.2.1). Since $\Pi = {\Pi(\varepsilon)}_{\varepsilon \in (0, \Delta_{\Pi}]}$ is nested, it follows from Definition 2.1.4 that

(2.2.7)
$$2\overline{W} \ge W(\Pi_0 \psi) \ge W(\Pi_1 \psi) \ge \dots \ge W(\Pi_q \psi),$$

 M^P_{ν} -almost everywhere. Defining the values on the exceptional sets as zero, we can choose some versions such that the above inequality holds *identically*. Notice also that $W(\Pi_p \psi) = W(\Pi_p \phi)$ holds identically, whenever $\psi, \phi \in \Psi(2^{-q}\delta; k)$ for some k. Next, let us introduce the following predictable functions on $\tilde{\Omega}$:

$$\begin{split} A_{p}(\psi) &= \mathbf{1}_{\{W(\Pi_{0}\psi) \leq a_{0},...,W(\Pi_{p-1}\psi) \leq a_{p-1},W(\Pi_{p}\psi) \leq a_{p}\}}, \quad p = 0, 1, ..., q; \\ B_{p}(\psi) &= \mathbf{1}_{\{W(\Pi_{0}\psi) \leq a_{0},...,W(\Pi_{p-1}\psi) \leq a_{p-1},W(\Pi_{p}\psi) > a_{p}\}}, \quad p = 1, ..., q; \\ B_{0}(\psi) &= \mathbf{1}_{\{W(\Pi_{0}\psi) > a_{0}\}}. \end{split}$$

It is important that $A_p(\psi)$ and $B_p(\psi)$ depend on ψ only through the subsets $\Pi_0\psi, ..., \Pi_p\psi$ of Ψ . Next observe the identity

$$(2.2.8) W^{\psi} - W^{\pi_0 \psi} = (W^{\psi} - W^{\pi_0 \psi}) B_0(\psi) + \sum_{p=1}^q (W^{\psi} - W^{\pi_p \psi}) B_p(\psi) + (W^{\psi} - W^{\pi_q \psi}) A_q(\psi) + \sum_{p=1}^q (W^{\pi_p \psi} - W^{\pi_{p-1} \psi}) A_{p-1}(\psi).$$

Since $a_0 = 2a(\delta, K)$, we have $B_0(\psi) \leq \mathbf{1}_{\{\overline{W} > a(\delta, K)\}}$. Hence we obtain

$$\sup_{t \in [0,\tau]} \sup_{\psi \in \Psi} |X_t^{a(\delta,K),\psi} - X_t^{a(\delta,K),\pi_0\psi}| \le (I_1) + (I_2) + (II_1) + (II_2) + (III),$$

where

$$(I_{1}) = \sup_{\psi \in \Psi} \sum_{p=1}^{q} W(\Pi_{p}\psi)B_{p}(\psi) * \mu_{\tau},$$

$$(I_{2}) = \sup_{\psi \in \Psi} \sum_{p=1}^{q} W(\Pi_{p}\psi)B_{p}(\psi) * \nu_{\tau},$$

$$(II_{1}) = \sup_{\psi \in \Psi} W(\Pi_{q}\psi)A_{q}(\psi) * \mu_{\tau},$$

$$(II_{2}) = \sup_{\psi \in \Psi} W(\Pi_{q}\psi)A_{q}(\psi) * \nu_{\tau},$$

$$(III) = \sup_{t \in [0,\tau]} \sup_{\psi \in \Psi} \sum_{p=1}^{q} \left| (W^{\pi_{p}\psi} - W^{\pi_{p-1}\psi})A_{p-1}(\psi) * (\mu - \nu)_{t} \right|.$$

2.2. Multivariate Point Processes

Further, it holds that $(I_1) \leq (I'_1) + (I_2)$ where

$$(I'_1) = \sup_{\psi \in \Psi} \sum_{p=1}^q |W(\Pi_p \psi) B_p(\psi) * (\mu - \nu)_\tau|,$$

and that $(II_1) \leq (II'_1) + (II_2)$ where

$$(II'_{1}) = \sup_{\psi \in \Psi} |W(\Pi_{q}\psi)A_{q}(\psi) * (\mu - \nu)_{\tau}|.$$

Hereafter we will obtain bounds for the terms $(I'_1), (I_2), (II'_1), (II_2)$ and (III).

Estimation of (I_2) and (II_2) . We can easily see that

$$(I_{2}) \leq \sup_{\psi \in \Psi} \sum_{p=1}^{q} \frac{1}{a_{p}} |W(\Pi_{p}\psi)|^{2} B_{p}(\psi) * \nu_{\tau}$$

$$\leq \max_{1 \leq p \leq q} \sup_{\psi \in \Psi} \frac{|W(\Pi_{p}\psi)|^{2} B_{p}(\psi) * \nu_{\tau}}{|2^{-p}\delta|^{2}} \cdot \sum_{p=1}^{q} \frac{|2^{-p}\delta|^{2}}{a_{p}}$$

$$\leq K \sum_{p=1}^{q} 2^{-p-1} \delta \tilde{H}_{\Pi}(2^{-p-1}\delta) \quad \text{on the set} \quad \{\|W\|_{\Pi,\tau} \leq K\}.$$

On the other hand, it follows from Schwarz's inequality that

$$(II_2) \leq \sup_{\psi \in \Psi} \sqrt{|W(\Pi_q \psi)|^2 * \nu_\tau} \cdot \sqrt{\nu([0,\tau] \times E)}$$
$$\leq 2^{-q} \delta K \cdot \frac{\tilde{H}_{\Pi}(2^{-q-2}\delta)}{4}.$$

Hence we have

$$E|(I_2) + (II_2)|\mathbf{1}_{\{||\mathcal{W}||_{\Pi,\tau} \le K\}} \le K \sum_{p=1}^{q+1} 2^{-p-1} \delta \tilde{H}_{\Pi}(2^{-p-1}\delta)$$
$$\le 2K \int_0^\delta \tilde{H}_{\Pi}(\varepsilon) d\varepsilon.$$

Estimation of (I'_1) , (II'_1) and (III). Let us consider the term (I'_1) . We will apply the Bernstein inequality (Lemma 2.1.1) to the processes

$$t \rightsquigarrow M_t = W(\Pi_p \psi) B_p(\psi) * (\mu - \nu)_t.$$

It follows from

$$0 \le W(\Pi_p \psi) B_p(\psi) \le W(\Pi_{p-1} \psi) B_p(\psi) \le a_{p-1}$$

that $|\Delta M| \leq a_{p-1}$; it is also clear that

$$\langle M, M \rangle_{\tau} \leq |W(\Pi_p \psi)|^2 B_p(\psi) * \nu_{\tau}$$

$$\leq |2^{-p} \delta K|^2 \quad \text{on the set} \quad \{ \|\mathcal{W}\|_{\Pi, \tau} \leq K \}.$$

Thus we have

$$P\left(\sup_{t\in[0,\tau]}|W(\Pi_{p}\psi)B_{p}(\psi)*(\mu-\nu)_{t}|>\varepsilon, \ \|\mathcal{W}\|_{\Pi,\tau}\leq K\right)$$
$$\leq 2\exp\left(-\frac{\varepsilon^{2}}{2(a_{p-1}\varepsilon+|2^{-p}\delta K|^{2})}\right) \qquad \forall \varepsilon>0.$$

Hence it follows from Lemma 2.1.2 that

$$E \sup_{\psi \in \Psi} \sup_{t \in [0,\tau]} |W(\Pi_p \psi) B_p(\psi) * (\mu - \nu)_t| \mathbf{1}_{\{||\mathcal{W}||_{\Pi,\tau} \leq K\}}$$

$$\lesssim a_{p-1} |\tilde{H}_{\Pi}(2^{-p}\delta)|^2 + 2^{-p} \delta K \tilde{H}_{\Pi}(2^{-p}\delta)$$

$$\leq 5K \cdot 2^{-p} \delta \tilde{H}_{\Pi}(2^{-p}\delta),$$

where it should be noted that " $\sup_{\psi \in \Psi}$ " appearing on the left hand side is actually " $\max_{1 \le k \le N_{\Pi}(2^{-p})}$ ". We therefore obtain

$$\begin{split} E|(I_1')|\mathbf{1}_{\{\|\mathcal{W}\|_{\Pi,\tau} \leq K\}} &\lesssim 5K \sum_{p=1}^q 2^{-p} \delta \tilde{H}_{\Pi}(2^{-p} \delta) \\ &\leq 5K \int_0^\delta \tilde{H}_{\Pi}(\varepsilon) d\varepsilon. \end{split}$$

Exactly the same calculation as for (I'_1) yields some bounds for (II'_1) and (III), which lead to the inequality (2.2.6).

Proof of Theorem 2.2.3 (ii). Due to Lemma 2.2.2, it suffices to show the assertion in the case of Π being a NFP. We extend the given NFP $\Pi = {\Pi(\varepsilon)}_{\varepsilon \in (0,\Delta_{\Pi}]}$ to $\Pi = {\Pi(\varepsilon)}_{\varepsilon \in (0,2\Delta_{\Pi}]}$ where $N_{\Pi}(\varepsilon) = 1$ for all $\varepsilon \in [\Delta_{\Pi}, 2\Delta_{\Pi}]$. In order to apply the assertion (i) with $\delta = 2\Delta_{\Pi}$, we consider the truncated processes $X_t^{a,\psi}$ with $a = a(2\Delta_{\Pi}, K) = 2\Delta_{\Pi}K/\sqrt{\log 2}$; notice that

$$\sup_{t\in[0,\tau]} \sup_{\psi,\phi\in\Psi} |X_t^{\psi} - X_t^{\phi}| \leq \sup_{t\in[0,\tau]} \sup_{\psi,\phi\in\Psi} |X_t^{a,\psi} - X_t^{a,\phi}| +2\overline{W}\mathbf{1}_{\{\overline{W}>a\}} * \mu_{\tau} + 2\overline{W}\mathbf{1}_{\{\overline{W}>a\}} * \nu_{\tau}.$$

First we have

$$\overline{W}\mathbf{1}_{\{\overline{W}>a\}} * \nu_{\tau} \leq \frac{|\overline{W}|^2 * \nu_{\tau}}{a} \\ \leq \frac{L}{a} \quad \text{on the set} \quad \{|\overline{W}|^2 * \nu_{\tau} \leq L\}.$$

Next, let us introduce the predictable time

$$S = \inf\{t \in \mathbb{R}_+ : |\overline{W}|^2 * \nu_t > L\}.$$

Take an announcing sequence $\{S_n\}$ for S (see I.2.16 of Jacod and Shiryaev (1987)). Since $0 \leq S_n < S$ almost surely on the set $\{S > 0\}$, it holds that $|\overline{W}|^2 * \nu_{S_n} \leq L$ almost surely. Thus it follows also from Doob's stopping theorem that

$$E\overline{W}\mathbf{1}_{\{\overline{W}>a\}}*\mu_{S_n\wedge T_m} = E\overline{W}\mathbf{1}_{\{\overline{W}>a\}}*\nu_{S_n\wedge T_m}$$
$$\leq \frac{E|\overline{W}|^2*\nu_{S_n\wedge T_m}}{a} \leq \frac{L}{a}$$

where $\{T_m\}$ is a localizing sequence for the local martingale $t \rightsquigarrow \overline{W} \mathbf{1}_{\{\overline{W} > a\}} * (\mu - \nu)_t$. By letting $n, m \to \infty$, we obtain $E\overline{W} \mathbf{1}_{\{\overline{W} > a\}} * \mu_S \leq L/a$. The predictable time S appearing in this inequality can be replaced by τ on the set $\{|\overline{W}|^2 * \nu_\tau \leq L\}$.

Hence it follows from the assertion (i) with $\delta = 2\Delta_{\Pi}$ that

$$E^* \sup_{t \in [0,\tau]} \sup_{\psi,\phi \in \Psi} |X_t^{\psi} - X_t^{\phi}| \mathbf{1}_{\{||\mathcal{W}||_{\Pi,\tau} \leq K, |\overline{\mathcal{W}}|^2 * \nu_{\tau} \leq L\}}$$

$$\lesssim K \int_0^{2\Delta_{\Pi}} \tilde{H}_{\Pi}(\varepsilon) d\varepsilon + 4 \cdot \frac{L}{2\Delta_{\Pi} K / \sqrt{\log 2}}$$

$$\leq 2 \left\{ K \int_0^{\Delta_{\Pi}} \tilde{H}_{\Pi}(\varepsilon) d\varepsilon + \frac{L}{\Delta_{\Pi} K} \right\}.$$

Proof of Theorem 2.2.4. We use the notations introduced in the first paragraph of the proof of Theorem 2.2.3 (p = 0 only). Notice that

$$|X_t^{a,\psi} - X_t^{a,\phi}| \le |X_t^{a,\psi} - X_t^{a,\pi_0\psi}| + |X_t^{a,\phi} - X_t^{a,\pi_0\phi}| + |X_t^{a,\pi_0\psi} - X_t^{a,\pi_0\phi}|$$

and thus

$$\sup_{\varrho(\psi,\phi) \le \delta} |X_t^{a,\psi} - X_t^{a,\phi}| \le 2 \sup_{\substack{1 \le k \le N_{\Pi}(\delta) \\ \psi,\phi \in \Psi(\delta;k)}} |X_t^{a,\psi} - X_t^{a,\phi}| + \sup_{\varrho(\psi,\phi) \le \delta} |X_t^{a,\pi_0\psi} - X_t^{a,\pi_0\phi}|.$$

The second term on the right hand side equals

$$\max_{1 \le k_1, k_2 \le N_{\Pi}(\delta)} |X_t^{a, \pi_{0, k_1}} - X_t^{a, \pi_{0, k_2}}| 1_{A_{k_1, k_2}},$$

where

$$A_{k_1,k_2} = \left\{ \omega \in \Omega : \quad \exists \psi, \phi \in \Psi \text{ s.t. } \left\{ \begin{array}{c} \pi_0 \psi = \pi_{0,k_1} \\ \pi_0 \phi = \pi_{0,k_2} \end{array} \right\} \text{ and } \varrho(\psi,\phi)(\omega) \le \delta \right\}.$$

Here, notice that for every $\psi, \phi \in \Psi$

$$\begin{aligned} \langle X^{a,\pi_0\psi} - X^{a,\pi_0\phi}, X^{a,\pi_0\psi} - X^{a,\pi_0\phi} \rangle_{\tau} &\leq 3 \langle X^{a,\pi_0\psi} - X^{a,\psi}, X^{a,\pi_0\psi} - X^{a,\psi} \rangle_{\tau} \\ &+ 3 \langle X^{a,\pi_0\phi} - X^{a,\phi}, X^{a,\pi_0\phi} - X^{a,\phi} \rangle_{\tau} \\ &+ 3 \langle X^{a,\psi} - X^{a,\phi}, X^{a,\psi} - X^{a,\phi} \rangle_{\tau} \end{aligned}$$

on the set $\{ \| \mathcal{W} \|_{\Pi,\tau} \leq K \} \cap B$, where *B* is given by (2.2.5). Thus, for every $k_1, k_2 = 1, ..., N_{\Pi}(\delta)$ and every $\omega \in \{ \| \mathcal{W} \|_{\Pi,\tau} \leq K \} \cap B \cap A_{k_1,k_2}$, by choosing some appropriate $\psi = \psi(\omega, k_1, k_2)$ and $\phi = \phi(\omega, k_1, k_2)$, we get

$$\langle X^{a,\pi_{0,k_1}} - X^{a,\pi_{0,k_2}}, X^{a,\pi_{0,k_1}} - X^{a,\pi_{0,k_2}} \rangle_{\tau} \le 9\delta^2 K^2.$$

Thus it follows from Lemma 2.1.1 that

$$P\left(\sup_{t\in[0,\tau]} |X_t^{a,\pi_{0,k_1}} - X_t^{a,\pi_{0,k_2}}| 1_{A_{k_1,k_2}} > \varepsilon, \ \|\varrho\|_{\Pi} \le K\right) \le 2\exp\left(-\frac{\varepsilon^2}{2(2a\varepsilon + 9K^2\delta^2)}\right)$$

for every $\varepsilon > 0$. Hence, we obtain from Lemma 2.1.2 that

$$\begin{split} E \sup_{t \in [0,\tau]} \sup_{\varrho(\psi,\phi) \le \delta} |X_t^{a,\pi_0\psi} - X_t^{a,\pi_0\phi}| \mathbf{1}_{||\mathcal{W}||_{\Pi,\tau} \le K\} \cap B} \\ \lesssim 2a \log(1 + N_{\Pi}(\delta)^2) + 3\delta K \sqrt{\log(1 + N_{\Pi}(\delta)^2)} \\ \le 4K\delta \frac{\log(1 + N_{\Pi}(\delta))}{\sqrt{\log(1 + N_{\Pi}(\delta/2))}} + 3\sqrt{2}\delta K \sqrt{\log(1 + N_{\Pi}(\delta))} \end{split}$$

This, together with (i) of Theorem 2.2.3, yields the assertion.

So far we have been concerned with the truncated processes $X^{a,\psi} = W^{\psi} 1_{\{\overline{W} \leq a\}} * (\mu - \nu)$ (except for (ii) of Theorem 2.2.3). This means that the predictable functions of integrands should be uniformly bounded, and this assumption is sometimes too strong. However, as is explained in Chapter 3.4 of van der Vaart and Wellner, it can be replaced by a moment assumption of exponential order (see, in particular, their Lemmas 3.4.2 and 3.4.3 which are concerned with the i.i.d. case); the key tool for this purpose is the extended Bernstein inequality (e.g., their Lemma 2.2.11). In our situation, we can make use of a martingale version of the inequality given by van de Geer (1995b, Lemma 2.2); we will quote it as Lemma 2.2.6 below. Let us begin with giving a version of Theorem 2.2.3.

Lemma 2.2.5 Consider the same situation as Theorem 2.2.3. It holds for any NFP Π of Ψ and any constants $\delta \in (0, \Delta_{\Pi}]$ and K > 0 that

$$\begin{split} E^* \sup_{t \in [0,\tau]} \sup_{\substack{1 \le k \le N_{\Pi}(\delta) \\ \psi, \phi \in \Psi(\delta;k)}} |(W^{\psi} - W^{\phi}) * (\mu - \nu)_t| \mathbf{1}_{\{||\mathcal{W}||_{\Pi,\tau} \le K\}} \\ \lesssim K \int_0^{\delta} \tilde{H}_{\Pi}(\varepsilon) d\varepsilon + E^* \sup_{t \in [0,\tau]} \max_{1 \le k \le N_{\Pi}(\delta)} |W(\Psi(\delta;k)) * (\mu - \nu)_t| \mathbf{1}_{\{||\mathcal{W}||_{\Pi,\tau} \le K\}} \end{split}$$

(recall the notation $W(\Psi')$ defined by (2.2.1) for every $\Psi' \subset \Psi$).

Proof. Recall the proof of (i) of Theorem 2.2.3; it suffices to consider the first term on the right hand side of (2.2.8). Notice that

$$(W^{\psi} - W^{\pi_0 \psi})B_0(\psi) * (\mu - \nu)_t \le W(\Pi_0 \psi)B_0(\psi) * (\mu - \nu)_t + 2W(\Pi_0 \psi)B_0(\psi) * \nu_t.$$

The second term on the right hand side is bounded by

$$\begin{aligned} \frac{2}{a_0} |W(\Pi_0 \psi)|^2 * \nu_\tau &\leq \frac{H_{\Pi}(\delta/2)}{\delta K} (\delta K)^2 \quad \text{on the set } \{ \|\mathcal{W}\|_{\Pi,\tau} \leq K \} \\ &\leq 2K \int_0^{\delta/2} \tilde{H}_{\Pi}(\varepsilon) d\varepsilon. \end{aligned}$$

On the other hand, the first term is bounded by

$$|W(\Pi_0\psi) * (\mu - \nu)_t| + |W(\Pi_0\psi) \mathbf{1}_{\{W(\Pi_0\psi) \le a_0\}} * (\mu - \nu)_t|$$

and it follows from Corollary 2.1.3 that

$$E^* \sup_{t \in [0,\tau]} \sup_{\psi \in \Psi} |W(\Pi_0 \psi) \mathbf{1}_{\{W(\Pi_0 \psi) \le a_0\}} * (\mu - \nu)_t |\mathbf{1}_{\{\|\mathcal{W}\|_{\Pi,\tau} \le K\}} \lesssim \delta K \tilde{H}_{\Pi}(\delta)$$

The assertion follows from these inequalities.

Thus the problem is how to manage the second term on the right hand side of Lemma 2.2.5. As we announced above, this can be solved by means of the following lemma that is an easy consequence of Lemma 2.2 of van de Geer (1995b). From now on, we will assume that $\nu(\{t\} \times E) = 0$ for every $t \in \mathbb{R}_+$ for simplicity.

Lemma 2.2.6 Let μ be an *E*-valued multivariate point process which has the predictable compensator ν such that $\nu(\omega; \{t\} \times E) = 0$ for all $\omega \in \Omega$ and $t \in \mathbb{R}_+$. Let *W* be a predictable function, and suppose that for a given constant a > 0 it holds that $\exp(a^{-1}|W|) * \nu_t(\omega) < \infty$ for all $\omega \in \Omega$ and $t \in \mathbb{R}_+$. Let τ be a finite stopping time. Then, it holds for every $\Gamma > 0$

$$P\left(\sup_{t\in[0,\tau]}|W*(\mu-\nu)_t|>\varepsilon, \ (\mathcal{E}_a(|W|))*\nu_{\tau}\leq\Gamma\right)\leq 2\exp\left(-\frac{\varepsilon^2}{2(a\varepsilon+\Gamma)}\right)\qquad\forall\varepsilon>0.$$

Lemma 2.2.7 Consider the same situation as Theorem 2.2.3. For a given constant a > 0, suppose also that for every $\omega \in \Omega$ it holds that $\nu(\omega; \{t\} \times E) = 0$ and that $\exp(a^{-1}\overline{W}) * \nu_t(\omega) < \infty$ for all $t \in [0, \tau(\omega)]$. Then, it holds that for any NFP Π of Ψ and any constants $\delta \in (0, \Delta_{\Pi}]$ and K > 0 that

$$E^* \sup_{t \in [0,\tau]} \sup_{\substack{1 \le k \le N_{\Pi}(\delta) \\ \psi, \phi \in \Psi(\delta;k)}} |(W^{\psi} - W^{\phi}) * (\mu - \nu)_t| \mathbf{1}_{\{\|\mathcal{W}\|_{\Pi,\tau} \le K\} \cap B}$$

$$\lesssim K \int_0^{\delta} \tilde{H}_{\Pi}(\varepsilon) d\varepsilon + a \log(1 + N_{\Pi}(\delta)),$$

where

$$B = \left\{ \max_{1 \le k \le N_{\Pi}(\delta)} \frac{\sqrt{(\mathcal{E}_{2a}(W(\Psi(\delta; k)))) * \nu_{\tau}}}{\delta} \le K \right\}$$

Proof. Lemmas 2.2.6 and 2.1.2 yield that

$$E^* \sup_{t \in [0,\tau]} \sup_{\psi \in \Psi} |W(\Psi(\delta;k)) * (\mu - \nu)_t| \mathbf{1}_B \lesssim \delta K \tilde{H}_{\Pi}(\delta) + a \log(1 + N_{\Pi}(\delta)).$$

which implies the assertion.

While the argument δ in the above lemma is arbitrary, it sometimes suffices to consider a specific range of δ . In the context of *M*-estimation studied in Chapter 6, we will use it in the following form that is reasonably weak and simple. It should be noted that one may get different versions by going back to Lemmas 2.2.6 and 2.2.7; even removing the assumption that $\nu(\omega; \{t\} \times E) = 0$ is also possible.

Theorem 2.2.8 Consider the same situation as Theorem 2.2.3. For a given constant a > 0, suppose also that for every $\omega \in \Omega$ it holds that $\nu(\omega; \{t\} \times E) = 0$ and that $\exp(a^{-1}\overline{W}) * \nu_t(\omega) < \infty$ for all $t \in [0, \tau(\omega)]$. Let Π be an arbitrary NFP of Ψ .

(i) For any constants $\delta \in (0, \Delta_{\Pi}]$ and K > 0 satisfying

(2.2.9)
$$a \int_0^{\delta} \tilde{H}_{\mathrm{II}}(\varepsilon) d\varepsilon \le K \delta^2,$$

it holds that

$$E^* \sup_{t \in [0,\tau]} \sup_{\substack{1 \le k \le N_{\Pi}(\delta) \\ \psi, \phi \in \Psi(\delta;k)}} |X_t^{\psi} - X_t^{\phi}| \mathbb{1}_{\{\|\mathcal{W}\|_{\Pi,\tau}^{\mathcal{E}_{2a}} \le K\}} \lesssim K \int_0^{\delta} \tilde{H}_{\Pi}(\varepsilon) d\varepsilon.$$

(ii) If a given random semimetric ρ on Ψ satisfies that

$$\sqrt{(\mathcal{E}_{2a}(|W^{\psi} - W^{\phi}|)) * \nu_{\tau}} \le \varrho(\psi, \phi) \qquad \forall \psi, \phi \in \Psi \qquad P_* \text{-almost surely,}$$

then, for any constants $\delta \in (0, \Delta_{\Pi}]$ and K > 0 satisfying (2.2.9), it holds that

$$E^* \sup_{t \in [0,\tau]} \sup_{\substack{\psi,\phi \in \Psi\\ \varrho(\psi,\phi) \le K\delta}} |X_t^{\psi} - X_t^{\phi}| \mathbf{1}_{\{\|\mathcal{W}\|_{\Pi,\tau}^{\mathcal{E}_{2a}} \le K\} \cap \{\|\varrho\|_{\Pi} \le K\}} \lesssim K \int_0^{\varepsilon} \tilde{H}_{\Pi}(\varepsilon) d\varepsilon,$$

where

$$\|\varrho\|_{\Pi} = \sup_{\varepsilon \in (0,\Delta_{\Pi}]} \max_{1 \le k \le N_{\Pi}(\varepsilon)} \sup_{\psi, \phi \in \Psi(\varepsilon;k)} \frac{\varrho(\psi,\phi)}{\varepsilon}.$$

22

Proof. The second term of Lemma 2.2.7 is bounded by $a\delta^{-2} \left| \int_0^{\delta} \tilde{H}_{\Pi}(\varepsilon) d\varepsilon \right|^2$, which is bounded by $K \int_0^{\delta} \tilde{H}_{\Pi}(\varepsilon) d\varepsilon$ whenever $a\delta^{-2} \int_0^{\delta} \tilde{H}_{\Pi}(\varepsilon) d\varepsilon \leq K$. Thus we have obtained the first assertion.

For the proof of the second assertion, we use the notations introduced in the first paragraph of the proof of Theorem 2.2.3 (p = 0 only). The method of the proof is quite similar to that of Theorem 2.2.4. Notice that

$$|X_t^{\psi} - X_t^{\phi}| \le |X_t^{\psi} - X_t^{\pi_0 \psi}| + |X_t^{\phi} - X_t^{\pi_0 \phi}| + |X_t^{\pi_0 \psi} - X_t^{\pi_0 \phi}|$$

and thus

$$\sup_{\varrho(\psi,\phi)\leq K\delta} |X_t^{\psi} - X_t^{\phi}| \leq 2 \sup_{\substack{1\leq k\leq N_{\Pi}(\delta)\\\psi,\phi\in\Psi(\delta;k)}} |X_t^{\psi} - X_t^{\phi}| + \sup_{\varrho(\psi,\phi)\leq K\delta} |X_t^{\pi_0\psi} - X_t^{\pi_0\phi}|.$$

The second term on the right hand side equals

$$\max_{1 \le k_1, k_2 \le N_{\Pi}(\delta)} |X_t^{\pi_{0,k_1}} - X_t^{\pi_{0,k_2}}| \mathbf{1}_{A_{k_1,k_2}},$$

where

$$A_{k_1,k_2} = \left\{ \omega \in \Omega : \quad \exists \psi, \phi \in \Psi \text{ s.t. } \left\{ \begin{array}{c} \pi_0 \psi = \pi_{0,k_1} \\ \pi_0 \phi = \pi_{0,k_2} \end{array} \right\} \text{ and } \varrho(\psi,\phi)(\omega) \le K\delta \right\}.$$

Here notice that for every $\psi, \phi \in \Psi$

$$\sqrt{\left(\mathcal{E}_{2a}(|W^{\pi_{0}\psi} - W^{\pi_{0}\phi}|)\right) * \nu_{\tau}} \leq \varrho(\pi_{0}\psi, \pi_{0}\phi) \\
\leq \varrho(\pi_{0}\psi, \psi) + \varrho(\pi_{0}\phi, \phi) + \varrho(\psi, \phi) \\
\leq 2K\delta + \varrho(\psi, \phi)$$

on the set $\{\|\varrho\|_{\Pi} \leq K\}$. Thus, for every $k_1, k_2 = 1, ..., N_{\Pi}(\delta)$, and every $\omega \in \{\|\varrho\|_{\Pi} \leq K\} \cap A_{k_1,k_2}$, by choosing some appropriate $\psi = \psi(k_1,k_2,\omega)$ and $\phi = \phi(k_1,k_2,\omega)$, we can get

$$\sqrt{\left(\mathcal{E}_{2a}(|W^{\pi_{0,k_{1}}} - W^{\pi_{0,k_{2}}}|)\right) * \nu_{\tau}} \le 3K\delta$$

Thus Lemma 2.2.6 yields that

$$P^*\left(\sup_{t\in[0,\tau]} |X_t^{\pi_{0,k_1}} - X_t^{\pi_{0,k_2}}| \mathbf{1}_{A_{k_1,k_2}} > \varepsilon, \ \|\varrho\|_{\Pi} \le K\right) \le 2\exp\left(-\frac{\varepsilon^2}{2(2a\varepsilon + 9K^2\delta^2)}\right)$$

for every $\varepsilon > 0$. Hence it follows from Corollary 2.1.2 that

$$E^* \sup_{t \in [0,\tau]} \sup_{\varrho(\psi,\phi) \le K\delta} |X_t^{\pi_0\psi} - X_t^{\pi_0\phi}| 1_{\{\|\varrho\|_{\Pi} \le K\}}$$

$$\lesssim 2a \log(1 + N_{\Pi}(\delta)^2) + 3K\delta\sqrt{\log(1 + N_{\Pi}(\delta)^2)}$$

$$\leq 4a \log(1 + N_{\Pi}(\delta)) + 3\sqrt{2}K\delta\sqrt{\log(1 + N_{\Pi}(\delta))}.$$

This, together with the first assertion of the theorem, yields the second.

2.3 Martingale Difference Arrays

Let a discrete-time stochastic basis $\mathbf{B} = (\Omega, \mathcal{F}, \mathbf{F}, P)$ be given, where (Ω, \mathcal{F}, P) is a probability space and $\mathbf{F} = \{\mathcal{F}_i\}_{i \in \mathbb{N}_0}$ is an increasing sequence of sub- σ -fields of \mathcal{F} indexed by $\mathbb{N}_0 = \{0\} \cup \mathbb{N}$. Let Ψ be an arbitrary set.

Definition 2.3.1 $\{\xi_i\}_{i\in\mathbb{N}} = \{(\xi_i^{\psi}|\psi\in\Psi)\}_{i\in\mathbb{N}} \text{ is called an } \ell^{\infty}(\Psi)\text{-valued martingale dif$ ference array on**B**if:

(i) ξ_i is a mapping from Ω to $\ell^{\infty}(\Psi)$ for every $i \in \mathbb{N}$;

(ii) $\{\xi_i^{\psi}\}_{i\in\mathbb{N}}$ is an \mathbb{R} -valued martingale difference array on **B** for every $\psi \in \Psi$.

It is required in (ii) that ξ_i^{ψ} is \mathcal{F}_i -measurable and $E_{i-1}\xi_i^{\psi} = 0$ almost surely, for every $\psi \in \Psi$, where E_{i-1} denotes the \mathcal{F}_{i-1} -conditional expectation; the exceptional sets may depend on ψ . Notice also that we do not require any measurability of the $\ell^{\infty}(\Psi)$ -valued random element ξ_i .

Based on the notation of Definition 2.1.4, we make the following definition.

Definition 2.3.2 The adapted envelope $\{\overline{\xi}_i\}_{i\in\mathbb{N}}$ of $\{\xi_i\}_{i\in\mathbb{N}}$ is defined by

$$\overline{\xi}_i = \left[\sup_{\psi \in \Psi} |\xi_i^{\psi}| \right]_{\mathcal{F}_i, P} \quad \forall i \in \mathbb{N}.$$

For a given DFP Π of Ψ , the quadratic Π -modulus $\|\xi\|_{\Pi}$ of $\{\xi_i\}_{i\in\mathbb{N}}$ is defined as the $\mathbb{R}_+ \cup \{\infty\}$ -valued increasing process $\{\|\xi\|_{\Pi,i}\}_{i\in\mathbb{N}}$ given by

$$\|\xi\|_{\Pi,i} = \sup_{\varepsilon \in (0,\Delta_{\Pi}] \cap \mathbb{Q}} \max_{1 \le k \le N_{\Pi}(\varepsilon)} \frac{\sqrt{\sum_{j=1}^{i} E_{j-1} |\xi_j(\Psi(\varepsilon;k))|^2}}{\varepsilon} \qquad \forall i \in \mathbb{N}$$

where

(2.3.1)
$$\xi_i(\Psi') = \left[\sup_{\psi,\phi\in\Psi'} |\xi_i^{\psi} - \xi_i^{\phi}|\right]_{\mathcal{F}_i,P} \quad \forall i \in \mathbb{N} \quad \forall \Psi' \subset \Psi.$$

For a given DFP Π of Ψ and a given constant a > 0, the exponential (Π, a) -modulus $\|\xi\|_{\Pi}^{\mathcal{E}_a}$ of $\{\xi_i\}_{i\in\mathbb{N}}$ is defined as the $[0,\infty]$ -valued increasing process $\{\|\xi\|_{\Pi,i}^{\mathcal{E}_a}\}_{i\in\mathbb{N}}$ given by

$$\|\xi\|_{\Pi,i}^{\mathcal{E}_a} = \sup_{\varepsilon \in (0,\Delta_{\Pi}] \cap \mathbb{Q}} \max_{1 \le k \le N_{\Pi}(\varepsilon)} \frac{\sqrt{\sum_{j=1}^{i} E_{j-1} \mathcal{E}_a(\xi_j(\Psi(\varepsilon;k)))}}{\varepsilon} \qquad \forall i \in \mathbb{N},$$

where $\mathcal{E}_a(x)$ is defined by (2.2.2) and $\xi_i(\Psi')$ is defined by (2.3.1) for every $\Psi' \subset \Psi$.

Theorem 2.3.3 Let $\{\xi_i\}_{i\in\mathbb{N}}$ be an $\ell^{\infty}(\Psi)$ -valued martingale difference array, and let σ be a finite stopping time, both of which are defined on a discrete-time stochastic basis **B**. Then, we have the following (i) and (ii).

(i) For any NFP Π of Ψ and any constants $\delta \in (0, \Delta_{\Pi}]$ and K > 0,

$$E^* \max_{1 \le m \le \sigma} \sup_{\substack{1 \le k \le N_{\Pi}(\delta) \\ \psi, \phi \in \Psi(\delta; k)}} \left| \sum_{i=1}^{m} (\xi_i^{a, \psi} - \xi_i^{a, \phi}) \right| \mathbf{1}_{\{\|\xi\|_{\Pi, \sigma} \le K\}} \lesssim K \int_0^{\delta} \tilde{H}_{\Pi}(\varepsilon) d\varepsilon$$

where $\xi_i^{a,\psi} = \xi_i^{\psi} \mathbf{1}_{\{\overline{\xi}_i \leq a\}}$ with $a = a(\delta, K) = \delta K / \tilde{H}_{\Pi}(\delta/2)$. (ii) For any DFP Π of Ψ and any constants K, L > 0,

$$E^* \max_{1 \le m \le \sigma} \sup_{\psi, \phi \in \Psi} \left| \sum_{i=1}^m (\xi_i^{\psi} - \xi_i^{\phi}) \right| \mathbf{1}_{\{\|\xi\|_{\Pi, \sigma} \le K, \sum_{i=1}^\sigma E_{i-1}|\overline{\xi}_i|^2 \le L\}} \lesssim K \int_0^{\Delta_{\Pi}} \tilde{H}_{\Pi}(\varepsilon) d\varepsilon + \frac{L}{\Delta_{\Pi} K}.$$

The result above is similar to Theorem 2.2.3, although the proof needs a careful discussion about the choice of versions of conditional expectations. It gives us the analogue of Theorem 2.2.4.

Theorem 2.3.4 Consider the same situation as Theorem 2.3.3. It holds for any random semimetric ρ on Ψ , any NFP Π of Ψ and any constants $\delta \in (0, \Delta_{\Pi}]$ and K > 0 that

$$E^* \max_{1 \le m \le \sigma} \sup_{\substack{1 \le k \le N_{\Pi}(\delta) \\ \psi, \phi \in \Psi(\delta; k)}} \left| \sum_{i=1}^m (\xi_i^{a, \psi} - \xi_i^{a, \phi}) \right| 1_{\{\|\xi\|_{\Pi, \sigma} \le K\} \cap B} \lesssim K \int_0^\delta \tilde{H}_{\Pi}(\varepsilon) d\varepsilon,$$

where

(2.3.2)
$$B = \left\{ \sup_{\substack{\psi, \phi \in \Psi\\\varrho(\psi, \phi) \le \delta}} \frac{\sqrt{\sum_{i=1}^{\sigma} E_{i-1} |\xi_i^{a, \psi} - \xi_i^{a, \phi}|^2}}{\delta} \le K \right\}$$

and where $\xi_i^{a,\psi} = \xi_i^{\psi} \mathbf{1}_{\{\overline{\xi}_i \leq a\}}$ with $a = a(\delta, K) = \delta K / \tilde{H}_{\Pi}(\delta/2)$.

Proof of Theorem 2.3.3. Fix any $\delta, K > 0$, and define a_p, π_p and Π_p for every integer $p \ge 0$ in the same way as the first paragraph of the proof of Theorem 2.2.3 (i). For every integer $q \ge 1$ we introduce the finite stopping time

$$\sigma_q = \inf\left\{i \in \mathbb{N} : i > \frac{|\tilde{H}_{\Pi}(2^{-q-2}\delta)|^2}{16} - 1\right\} \wedge \sigma.$$

Then, we have $\sigma_q \uparrow \sigma$ as $q \to \infty$ almost surely. Hence it is enough for getting the assertion (i) to show that

$$(2.3.3) E^* \max_{1 \le m \le \sigma_q} \sup_{\psi \in \Psi} \left| \sum_{i=1}^m (\xi_i^{a,\psi} - \xi_i^{a,\pi_0\psi}) \right| \mathbf{1}_{\{\|\xi\|_{\Pi,\sigma \le K\}}} \lesssim K \int_0^\delta \tilde{H}_{\Pi}(\varepsilon) d\varepsilon \qquad \forall q \ge 1,$$

where $a = a(\delta, K)$.

Let us now fix any integer $q \ge 1$, and choose some appropriate versions of $\overline{\xi}_i$ and $\xi_i(\Pi_p \psi), \ p = 0, 1, ..., q$ (recall the argument about (2.2.7)). We define:

$$\begin{aligned} A_{i,p}(\psi) &= \mathbf{1}_{\{\xi_{i}(\Pi_{0}\psi) \leq a_{0},...,\xi_{i}(\Pi_{p-1}\psi) \leq a_{p-1},\xi_{i}(\Pi_{p}\psi) \leq a_{p}\}}, \quad p = 0, 1, ..., q; \\ B_{i,p}(\psi) &= \mathbf{1}_{\{\xi_{i}(\Pi_{0}\psi) \leq a_{0},...,\xi_{i}(\Pi_{p-1}\psi) \leq a_{p-1},\xi_{i}(\Pi_{p}\psi) > a_{p}\}}, \quad p = 1, ..., q; \\ B_{i,0}(\psi) &= \mathbf{1}_{\{\xi_{i}(\Pi_{0}\psi) > a_{0}\}}. \end{aligned}$$

Next observe the identity

$$\xi_{i}^{\psi} - \xi_{i}^{\pi_{0}\psi} = (\xi_{i}^{\psi} - \xi_{i}^{\pi_{0}\psi})B_{i,0}(\psi) + \sum_{p=1}^{q} (\xi_{i}^{\psi} - \xi_{i}^{\pi_{p}\psi})B_{i,p}(\psi) + (\xi_{i}^{\psi} - \xi_{i}^{\pi_{q}\psi})A_{i,q}(\psi) + \sum_{p=1}^{q} (\xi_{i}^{\pi_{p}\psi} - \xi_{i}^{\pi_{p-1}\psi})A_{i,p-1}(\psi).$$

Taking the \mathcal{F}_{i-1} -conditional expectations of all terms, we obtain

$$(2.3.4) \quad (0 =) \quad E_{i-1}\xi_i^{\psi} - E_{i-1}\xi_i^{\pi_0\psi} = E_{i-1}(\xi_i^{\psi} - \xi_i^{\pi_0\psi})B_{i,0}(\psi) + \sum_{p=1}^q E_{i-1}(\xi_i^{\psi} - \xi_i^{\pi_p\psi})B_{i,p}(\psi) + E_{i-1}(\xi_i^{\psi} - \xi_i^{\pi_q\psi})A_{i,q}(\psi) + \sum_{p=1}^q E_{i-1}(\xi_i^{\pi_p\psi} - \xi_i^{\pi_{p-1}\psi})A_{i,p-1}(\psi)$$

almost surely. Further, it holds that

(2.3.5)
$$|E_{i-1}(\xi_i^{\psi} - \xi_i^{\pi_0 \psi})B_{i,0}(\psi)| \le E_{i-1}\xi_i(\Pi_0 \psi)B_{i,0}(\psi)$$

almost surely, and that

(2.3.6)
$$|E_{i-1}(\xi_i^{\psi} - \xi_i^{\pi_p \psi})B_{i,p}(\psi)| \le E_{i-1}\xi_i(\Pi_p \psi)B_{i,p}(\psi) \le a_{p-1}, \quad p = 1, ..., q,$$

almost surely, and that

(2.3.7)
$$|E_{i-1}(\xi_i^{\pi_p\psi} - \xi_i^{\pi_{p-1}\psi})A_{i,p-1}(\psi)| \le a_{p-1}$$

almost surely. Here we choose versions of conditional expectations as follows: first choose some versions of the terms $E_{i-1}\xi_i(\Pi_p(\psi))B_{i,p}(\psi)$ of (2.3.5) and (2.3.6), which are non-negative and fulfill the second inequalities of (2.3.6), identically; next, on the exceptional sets of (2.3.4), (2.3.5), (2.3.6) and (2.3.7), we define the values of all other conditional expectations as zero. Then, the values of $E_{i-1}\xi_i(\Pi_p\psi)B_{i,p}(\psi)$ and $E_{i-1}(\xi_i^{\pi_p\psi}-\xi_i^{\pi_{p-1}\psi})A_{i,p-1}(\psi)$ depend on ψ only through $\Pi_0\psi, \ldots, \Pi_p\psi$ and $\pi_{p-1}\psi, \pi_p\psi$, respectively, while (2.3.4), (2.3.5), (2.3.6) and (2.3.7) hold *identically* for all $\psi \in \Psi$.

Since $a_0 = 2\delta K \tilde{H}_{\Pi}(\delta/2)$ we have $B_{i,0}(\psi) \mathbf{1}_{\{\overline{\xi}_i \leq a(\delta,K)\}} = 0$. This implies that

$$E^* \left(\sup_{\psi \in \Psi} \left| E_{i-1}(\xi_i^{\psi} - \xi_i^{\pi_0 \psi}) B_{i,0}(\psi) \right| \mathbf{1}_{\{\overline{\xi}_i \le a(\delta, K)\}} \right)$$

$$\leq E \left(\sup_{\psi \in \Psi} \left| E_{i-1} \xi_i(\Pi_0 \psi) B_{i,0}(\psi) \right| \mathbf{1}_{\{\overline{\xi}_i \le a(\delta, K)\}} \right) \qquad \text{since } (2.3.5) \text{ holds identically}$$

$$\leq 2E \left(\overline{\xi}_i B_{i,0}(\psi) \mathbf{1}_{\{\overline{\xi}_i \le a(\delta, K)\}} \right) = 0,$$

and thus

$$\sup_{\psi \in \Psi} \left| E_{i-1}(\xi_i^{\psi} - \xi_i^{\pi_0 \psi}) B_{i,0}(\psi) \right| \mathbf{1}_{\{\overline{\xi}_i \le a(\delta, K)\}} = 0$$

almost surely. Hence we obtain

$$\max_{1 \le m \le \sigma_q} \sup_{\psi \in \Psi} \left| \sum_{i=1}^m (\xi_i^{a,\psi} - \xi_i^{a,\pi_0\psi}) \right| \le (I_1') + 2(I_2) + (II_1') + 2(II_2) + (III),$$

where

$$(I'_{1}) = \sup_{\psi \in \Psi} \sum_{p=1}^{q} \left| \sum_{i=1}^{\sigma_{q}} \{\xi_{i}(\Pi_{p}\psi)B_{i,p}(\psi) - E_{i-1}\xi_{i}(\Pi_{p}\psi)B_{i,p}(\psi)\} \right|,$$

$$(I_{2}) = \sup_{\psi \in \Psi} \sum_{p=1}^{q} \sum_{i=1}^{\sigma_{q}} E_{i-1}\xi_{i}(\Pi_{p}\psi)B_{i,p}(\psi),$$

$$(II'_{1}) = \sup_{\psi \in \Psi} \left| \sum_{i=1}^{\sigma_{q}} \{\xi_{i}(\Pi_{q}\psi)A_{i,q}(\psi) - E_{i-1}\xi_{i}(\Pi_{q}\psi)A_{i,q}(\psi)\} \right|,$$

$$(II_{2}) = \sup_{\psi \in \Psi} \sum_{i=1}^{\sigma_{p}} E_{i-1}\xi_{i}(\Pi_{q}\psi)A_{i,q}(\psi),$$

$$(III) = \max_{1 \leq m \leq \sigma_{q}} \sup_{\psi \in \Psi} \sum_{p=1}^{q} \left| \sum_{i=1}^{m} \{(\xi_{i}^{\pi_{p}\psi} - \xi_{i}^{\pi_{p-1}\psi})A_{i,p-1}(\psi) - \sum_{i=1}^{m} E_{i-1}(\xi_{i}^{\pi_{p}\psi} - \xi_{i}^{\pi_{p-1}\psi})A_{i,p-1}(\psi)\} \right|,$$

almost surely. To get (2.3.3), we can deal with terms $(I'_1), (I_2), (II'_1), (II_2)$ and (III) exactly in the same way as those of the proof of Theorem 2.2.3 (i).

The assertion (ii) can be proved in the same way as that of Theorem 2.2.3, paying attention to the choice of conditional expectations; introduce a continuous-time stochastic basis and repeat the argument with an announcing sequence (see page 14 and I.2.43 of Jacod and Shiryaev (1987)). \Box

Proof of Theorem 2.3.4. The result follows from the same argument as Theorem 2.2.4. \Box

When a maximal inequality not for the truncated $\xi_i^{a,\psi}$'s but for the original ξ_i^{ψ} 's is needed, one may follow exactly the same discussion as Lemmas 2.2.5, 2.2.7 and Theorem 2.2.8, replacing Lemma 2.2.6 by the following version of the extended Bernstein inequality due to van de Geer (1995b).

Lemma 2.3.5 Let $\xi = {\xi_i}_{i \in \mathbb{N}_0}$ be an \mathbb{R} -valued martingale difference array on a discrete stochastic basis **B**. Suppose that for a given constant a > 0 it holds that $E \exp(a^{-1}|\xi_i|) < \infty$ for every $i \in \mathbb{N}$. Let σ be a finite stopping time on **B**. Then, it holds for every $\Gamma > 0$

$$P\left(\max_{1\leq m\leq \sigma} \left|\sum_{i=1}^{m} \xi_i\right| > \varepsilon, \ \sum_{i=1}^{\sigma} E_{i-1}\mathcal{E}_a(|\xi_i|) \le \Gamma\right) \le 2\exp\left(-\frac{\varepsilon^2}{2(a\varepsilon+\Gamma)}\right) \qquad \forall \varepsilon > 0.$$

We state here the analogue of Theorem 2.2.8 only.

Theorem 2.3.6 Let $\{\xi_i\}_{i\in\mathbb{N}}$ be an $\ell^{\infty}(\Psi)$ -valued martingale difference array and let σ be a finite stopping time both of which defined on a discrete-time stochastic basis **B**. Suppose also that for a given constant a > 0 it holds that $E \exp(a^{-1}\overline{\xi_i}) < \infty$ for every $i \in \mathbb{N}$. Let Π be an arbitrary NFP of Ψ .

(i) For any constants $\delta \in (0, \Delta_{\Pi}]$ and K > 0 satisfying

(2.3.8)
$$a \int_0^\delta \tilde{H}_{\Pi}(\varepsilon) d\varepsilon \le K \delta^2$$

it holds that

$$E^* \max_{1 \le m \le \sigma} \sup_{\substack{\psi, \phi \in \Psi\\ \rho(\psi, \phi) \le \delta}} \left| \sum_{i=1}^m (\xi_i^{\psi} - \xi_i^{\phi}) \right| 1_{\{\|\xi\|_{\Pi, \sigma}^{\mathcal{E}_{2a}} \le K\}} \lesssim K \int_0^{\delta} \tilde{H}_{\Pi}(\varepsilon) d\varepsilon,$$

(ii) If a given semimetric ρ on Ψ satisfies that

$$\sqrt{\sum_{i=1}^{\sigma} E_{i-1} \mathcal{E}_{2a}(|\xi_i^{\psi} - \xi_i^{\phi}|)} \le \varrho(\psi, \phi) \qquad \forall \psi, \phi \in \Psi \qquad P_* \text{-almost surely},$$
then, for any constant $\delta \in (0, \Delta_{\Pi}]$ and K > 0 satisfying (2.3.8), it holds that

$$E^* \max_{1 \le m \le \sigma} \sup_{\substack{\psi, \phi \in \Psi\\ \varrho(\psi, \phi) \le K\delta}} \left| \sum_{i=1}^m (\xi_i^{\psi} - \xi_i^{\phi}) \right| \mathbf{1}_{\{\|\xi\|_{\Pi, \sigma}^{\mathcal{E}_{2a}} \le K\} \cap \{\|\varrho\|_{\Pi} \le K\}} \lesssim K \int_0^{\delta} \tilde{H}_{\Pi}(\varepsilon) d\varepsilon$$

where

$$\|\varrho\|_{\Pi} = \sup_{\varepsilon \in (0,\Delta_{\Pi}]} \max_{1 \le k \le N_{\Pi}(\varepsilon)} \sup_{\psi, \phi \in \Psi(\varepsilon;k)} \frac{\varrho(\psi,\phi)}{\varepsilon}$$

2.4 Continuous Local Martingales

Let $\mathbf{B} = (\Omega, \mathcal{F}, \mathbf{F} = (\mathcal{F}_t)_{t \in \mathbb{R}_+}, P)$ be a stochastic basis and (Ψ, ρ) a *proper* metric space. Let $X = (X^{\psi} | \psi \in \Psi)$ be a family of continuous local martingales defined on \mathbf{B} indexed by Ψ . We introduce a kind of "quadratic modulus" which fits in this situation.

Definition 2.4.1 A quadratic ρ -modulus $||X||_{\rho}$ of a family $X = (X^{\psi}|\psi \in \Psi)$ of continuous local martingales is defined as a $[0,\infty]$ -valued stochastic process $t \rightsquigarrow ||X||_{\rho,t}$ given by

$$\|X\|_{\rho,t} = \sup_{\substack{\psi,\phi\in\Psi\\\psi\neq\phi}} \frac{\sqrt{\langle X^{\psi} - X^{\phi}, X^{\psi} - X^{\phi} \rangle_t}}{\rho(\psi,\phi)} \qquad \forall t \in \mathbb{R}_+.$$

Since the set Ψ is not necessarily countable, the random element $||X||_{\rho,t}$ may not have any measurability. Moreover, although the predictable covariation $\langle X^{\psi}, X^{\phi} \rangle$ is uniquely determined up to a negligible set for every pair $\psi, \phi \in \Psi$, due to the same reason the quadratic ρ -modulus of X may not be unique even in the almost sure sense. However, we do not require its uniqueness because the assertion of the following theorem is valid for any choice of quadratic ρ -modulus of X.

Theorem 2.4.2 Let (Ψ, ρ) be a totally bounded metric space. Let $X = (X^{\psi}|\psi \in \Psi)$ be a family of continuous local martingales indexed by Ψ such that $X_0^{\psi} = 0$, and τ a finite stopping time, both of which are defined on a stochastic basis **B**.

Then, for any choice of quadratic ρ -modulus $||X||_{\rho}$ of X, it holds that for every $\delta, K > 0$

$$\sup_{\Psi^* \subset \Psi \atop \text{countable}} E^* \sup_{t \in [0,\tau]} \sup_{\psi,\phi \in \Psi^* \atop \rho(\psi,\phi) \le \delta} |X_t^{\psi} - X_t^{\phi}| \mathbf{1}_{\{\|X\|_{\rho,\tau} \le K\}} \lesssim K \int_0^{\delta} \sqrt{\log(1 + N(\Psi,\rho;\varepsilon))} d\varepsilon,$$

provided the integral on the right hand side is finite (the first supremum is taken over all countable subsets Ψ^* of Ψ).

Proof. Fix any countable subset Ψ^* of Ψ . Let $\{\Psi^m\}_{m\in\mathbb{N}}$ be a sequence of finite subsets of Ψ^* such that $\Psi^m \uparrow \Psi^*$ as $m \to \infty$. For every $m \in \mathbb{N}$ and $p \in \mathbb{Z}$, let us denote by q(m,p) the smallest integer such that q(m,p) > p and that each of closed balls with centers in Ψ^m and ρ -radius $2 \cdot 2^{-q(m,p)}$ contains exactly one point in Ψ^m . Then it is clear that $\operatorname{Card}(\Psi^m) \leq N(\Psi,\rho; 2^{-q(m,p)})$.

Next let us introduce some mappings $\pi_r^{m,p}: \Psi^m \to \Psi_r^{m,p}, p \leq r \leq q(m,p)$, defined by

$$\pi_r^{m,p} = \lambda_r^{m,p} \circ \lambda_{r+1}^{m,p} \circ \cdots \circ \lambda_{q(m,p)}^{m,p},$$

where the sets $\Psi_r^{m,p} \subset \Psi^m$ and the mappings $\lambda_r^{m,p} : \Psi^m \to \Psi_r^{m,p}$ should be specified in the following way. For $p \leq r < q(m,p)$, choose $\Psi_r^{m,p}$ and define $\lambda_r^{m,p}$ which satisfy the following two conditions: (i) $\operatorname{Card}(\Psi_r^{m,p}) \leq N(\Psi,\rho;2^{-r})$; (ii) $\rho(\psi,\lambda_r^{m,p}(\psi)) \leq 2 \cdot 2^{-r}$ for every $\psi \in \Psi^m$. For r = q(m,p), put $\Psi_{q(m,p)}^{m,p} = \Psi^m$ and denote by $\lambda_{q(m,p)}^{m,p}$ the identity mapping on Ψ^m .

In term of the mappings $\pi_r^{m,p}$ which have been introduced, we consider the *chaining* given as follows: for every $t \in \mathbb{R}_+$ and $\psi, \phi \in \Psi^m$

$$|X_t^{\psi} - X_t^{\phi}| \le (I) + (II)$$

where the terms on the right hand side are given by:

$$(I) = \sum_{r=p+1}^{q(m,p)} |X_t^{\pi_r^{m,p}(\psi)} - X_t^{\pi_{r-1}^{m,p}(\psi)}| + \sum_{r=p+1}^{q(m,p)} |X_t^{\pi_r^{m,p}(\phi)} - X_t^{\pi_{r-1}^{m,p}(\phi)}|;$$

(II) = $|X_t^{\pi_p^{m,p}(\psi)} - X_t^{\pi_p^{m,p}(\phi)}|.$

First let us consider the term (I). It follows from Lemma 2.1.1 that for every $\varepsilon, T > 0$

$$P\left(\sup_{t\in[0,\tau\wedge T]} |X_t^{\pi_r^{m,p}(\psi)} - X_t^{\pi_{r-1}^{m,p}(\psi)}| > \varepsilon, \ \|X\|_{\rho,\tau} \le K\right) \le 2\exp\left(-\frac{\varepsilon^2}{2\cdot 2^{-2r+4}K^2}\right),$$

and by letting $T \to \infty$ we can replace " $\tau \wedge T$ " by " τ " on the left hand side. Thus we obtain from Lemma 2.1.2 that

$$E \sup_{\psi \in \Psi^m} \sup_{t \in [0,\tau]} |X_t^{\pi_r^{m,p}(\psi)} - X_t^{\pi_{r-1}^{m,p}(\psi)}| \mathbb{1}_{\{\|X\|_{\rho,\tau} \le K\}} \lesssim 2^{-r} K \sqrt{\log(1 + N(\Psi,\rho;2^{-r}))}.$$

Next let us consider the term (H). Notice that

$$\begin{split} \rho(\pi_p^{m,p}(\psi),\pi_p^{m,p}(\phi)) \\ &\leq \sum_{r=p+1}^{q(m,p)} \rho(\pi_r^{m,p}(\psi),\pi_{r-1}^{m,p}(\psi)) + \sum_{r=p+1}^{q(m,p)} \rho(\pi_r^{m,p}(\phi),\pi_{r-1}^{m,p}(\phi)) + \rho(\psi,\phi) \end{split}$$

and the right hand side is not bigger than $33 \cdot 2^{-p}$ whenever $\rho(\psi, \phi) \leq 2^{-p}$. Hence it follows from Lemmas 2.1.1 and 2.1.2 that

$$\begin{split} E \sup_{\substack{\psi,\phi\in\Psi^{m}\\\rho(\psi,\phi)\leq 2^{-p}}} \sup_{t\in[0,\tau]} |X_{t}^{\pi_{p}^{m,p}(\psi)} - X_{t}^{\pi_{p}^{m,p}(\phi)}| \mathbf{1}_{\{\|X\|_{\rho,\tau}\leq K\}} \\ \lesssim 33 \cdot 2^{-p} K \sqrt{\log(1+N(\Psi,\rho;2^{-p})^{2})} \leq 33\sqrt{2} \cdot 2^{-p} K \sqrt{\log(1+N(\Psi,\rho;2^{-p}))}. \end{split}$$

To show the assertion of the theorem, for a given $\delta > 0$ choose $p \in \mathbb{Z}$ such that $2^{-p-1} < \delta \leq 2^{-p}$. Then, the estimates for the terms (I) and (II) yield that

$$\begin{split} E \sup_{\substack{\psi,\phi\in\Psi^m\\\rho(\psi,\phi)\leq\delta}} \sup_{t\in[0,\tau]} |X_t^{\psi} - X_t^{\phi}| \mathbf{1}_{\{\|X\|_{\rho,\tau}\leq K\}} \\ \lesssim \sum_{r=p}^{q(m,p)} 2^{-r} K \sqrt{\log(1+N(\Psi,\rho;2^{-r}))} \leq 2K \int_0^{2\delta} \sqrt{\log(1+N(\Psi,\rho;\varepsilon))} d\varepsilon. \end{split}$$

The proof is accomplished by letting $m \to \infty$.

One may sometimes encounter the question whether the paths $\psi \rightsquigarrow X_{\tau}^{\psi}$ and $(t, \psi) \rightsquigarrow X_{t}^{\psi}$ are continuous and/or bounded. Applying the result above, we can get two kinds of answers to this problem. The first one is concerned with the case where Ψ is countable.

Theorem 2.4.3 Consider the same situation as Theorem 2.4.2. Suppose also that Ψ is countable and that

$$P(\|X\|_{\rho,\tau} < \infty) = 1 \quad and \quad \int_0^1 \sqrt{\log N(\Psi,\rho;\varepsilon)} d\varepsilon < \infty.$$

Then, almost all paths of $\psi \rightsquigarrow X^{\psi}_{\tau}$ are uniformly ρ -continuous on Ψ ; moreover, they belong to $\ell^{\infty}(\Psi)$. Furthermore, when $\tau > 0$ is a constant, almost all paths of $(t, \psi) \rightsquigarrow X^{\psi}_t$ are uniformly $\tilde{\rho}$ -continuous on $[0, \tau] \times \Psi$, where $\tilde{\rho}((t, \psi), (s, \phi)) = |t - s| \lor \rho(\psi, \phi)$; moreover, they belong to $\ell^{\infty}([0, \tau] \times \Psi)$.

Proof. It follows from Theorem 2.4.2 that for every $i \in \mathbb{N}$ there exists $\delta_i > 0$ such that

$$E \sup_{t \in [0,\tau]} \sup_{\substack{\psi,\phi \in \Psi\\\rho(\psi,\phi) \le \delta_i}} |X_t^{\psi} - X_t^{\phi}| \mathbb{1}_{\{\|X\|_{\rho,\tau} \le K\}} \lesssim K \cdot 4^{-i}.$$

Here, we set

$$A_i = \{ \sup_{\rho(\psi,\phi) \le \delta_i} \sup_{t \in [0,\tau]} |X_t^{\psi} - X_t^{\phi}| > 2^{-i} \} \qquad \forall i \in \mathbb{N}.$$

Then, since $\sum_{i} P(A_i \cap \{ \|X\|_{\rho,\tau} \leq K \}) < \infty$, it follows from the Borel-Cantelli lemma that $P(\limsup_{i} A_i \cap \{ \|X\|_{\rho,\tau} \leq K \}) = 0$ for every K > 0. Noting $\bigcup_{K \in \mathbb{N}} \{ \|X\|_{\rho,\tau} \leq K \}$

K = { $||X||_{\rho,\tau} < \infty$ }, we obtain that $P(\limsup_i A_i) = 0$, which implies the uniform continuity. Since (Ψ, ρ) is totally bounded, almost all pathes are bounded. \Box

When Ψ is uncountable, the following gives a sufficient condition for the existence of a continuous version of $\psi \rightsquigarrow X_{\tau}^{\psi}$.

Theorem 2.4.4 Consider the same situation as Theorem 2.4.2. Suppose also that it holds for a choice of quadratic ρ -modulus $||X||_{\rho,\tau}$ of X that

$$P\left([\|X\|_{\rho,\tau}]_{\mathcal{F}_{\tau},P} < \infty\right) = 1 \quad and \quad \int_{0}^{1} \sqrt{\log N(\Psi,\rho;\varepsilon)} d\varepsilon < \infty.$$

Then, there exists a family $\{\widetilde{X}(\psi) : \psi \in \Psi\}$ of \mathcal{F}_{τ} -measurable random variables such that $\widetilde{X}(\psi) = X^{\psi}_{\tau}$ almost surely for every $\psi \in \Psi$ and that almost all paths of $\psi \rightsquigarrow \widetilde{X}(\psi)$ are uniformly ρ -continuous; moreover, they belong to $\ell^{\infty}(\Psi)$. (Such a process $\psi \rightsquigarrow \widetilde{X}(\psi)$ is called a ρ -continuous version of $\psi \rightsquigarrow X^{\psi}_{\tau}$.)

Proof. Consider the \mathcal{F}_{τ} -measurable partition $\Omega = \bigcup_{K \in \mathbb{N} \cup \{\infty\}} \Omega(K)$ given by

$$\Omega(K) = \left\{ [\|X\|_{\rho,\tau}]_{\mathcal{F}_{\tau},P} \in [K-1,K) \right\} \in \mathcal{F}_{\tau} \qquad \forall K \in \mathbb{N}$$

$$\Omega(\infty) = \left\{ [\|X\|_{\rho,\tau}]_{\mathcal{F}_{\tau},P} = \infty \right\} \in \mathcal{F}_{\tau},$$

and define the process $\psi \rightsquigarrow Y_K(\psi)$ by $Y_K(\psi) = X_\tau^{\psi} \mathbf{1}_{\Omega(K)}$ for every $K \in \mathbb{N}$. Notice that (Ψ, ρ) is separable. For every $K \in \mathbb{N}$, since $\psi \rightsquigarrow Y_K(\psi)$ is continuous in probability by Theorem 2.4.2, it admits a separable version $\psi \rightsquigarrow \widetilde{Y}_K(\psi)$; here, we may define $Y_K \equiv 0$ on the set $\Omega \setminus \Omega(K)$. In the same way as Theorem 2.4.3, we can show that almost all paths of $\psi \rightsquigarrow \widetilde{Y}_K(\psi)$ are uniformly ρ -continuous. Thus the process $\widetilde{X} = \sum_{K \in \mathbb{N}} \widetilde{Y}_K$ satisfies the required properties.

Notice that the constructed $\widetilde{X}(\psi)$ is not the terminal variable of a continuous local martingale any more. However, it is conjectured that such a construction, including also the parameter t, would be possible.

In Theorem 2.4.2, the requirement that ρ should be a *proper* metric on Ψ is strong for some applications. The following theorem is concerned with an adaptation to a (random) semimetric ρ which is "weaker" than the metric ρ ; the entropy number should be still computed with respect to the metric ρ . The proof is similar to (and easier than) that for (ii) of Theorem 2.2.8, hence is omitted.

Theorem 2.4.5 Let (Ψ, ρ) be a totally bounded metric space. Let $X = (X^{\psi}|\psi \in \Psi)$ be a family of continuous local martingales indexed by Ψ such that $X_0^{\psi} = 0$, and let τ be a finite stopping time, both of which are defined on a stochastic basis **B**. If a given random semimetric ρ on Ψ satisfies that

 $\sqrt{\langle X^{\psi} - X^{\phi}, X^{\psi} - X^{\phi} \rangle_{\tau}} \le \varrho(\psi, \phi) \qquad \forall \psi, \phi \in \Psi \qquad P_*\text{-almost surely},$

then it holds that for every $\delta, K > 0$

$$\sup_{\Psi^* \subset \Psi \atop \text{countable}} E^* \sup_{t \in [0,\tau]} \sup_{\psi, \phi \in \Psi^* \atop \varrho(\psi, \phi) \leq K\delta} |X_t^{\psi} - X_t^{\phi}| \mathbf{1}_{\{\|\varrho\|_{\rho} \leq K\}} \lesssim K \int_0^{\delta} \sqrt{\log(1 + N(\Psi, \rho; \varepsilon))} d\varepsilon,$$

where

$$\|\varrho\|_{\rho} = \sup_{\substack{\psi,\phi\in\Psi\\\psi\neq\phi}} \frac{\varrho(\psi,\phi)}{\rho(\psi,\phi)},$$

provided the integral on the right hand side is finite (the first supremum is taken over all countable subsets Ψ^* of Ψ).

2.A Notes

The martingale version of the Bernstein inequality (Lemma 2.1.1) is due to Freedman (1975) who dealt with the discrete-time case. The inequality requires that the jumps of a martingale are bounded, but this assumption has been replaced by a kind of higher order moment condition by van de Geer (1995b), which we quoted as Lemmas 2.2.6 and 2.3.5.

The usefulness of bounds for expectation of supremum was shown by Pollard (1989). See also Pollard (1990), Kim and Pollard (1990) and, for more details, van der Vaart and Wellner (1996). The inequalities given in Section 2.2 have the same nature as that of van de Geer (1995b, 1997) who derived a probability inequality with a different definition of brackets.

Related to Theorems 2.4.3 and 2.4.4, one can find a general theory of the regularity of sample paths in Chapter 11 of Ledoux and Talagrand (1991).

Chapter 3

Weak Convergence Theorems

3.1 Preliminaries

Let us quote here a tightness criterion for sequences of random elements taking values in ℓ^{∞} -spaces. The proof can be found in Chapter 1.5 of van der Vaart and Wellner (1996).

Theorem 3.1.1 Let T be an arbitrary set. For every $n \in \mathbb{N}$, let $(\Omega^n, \mathcal{F}^n, P^n)$ be a probability space and X^n a mapping from Ω^n to $\ell^{\infty}(T)$. Consider the following statements: (i) X^n converges weakly in $\ell^{\infty}(T)$ to a tight, Borel law;

(ii) every finite-dimensional marginal of $t \rightsquigarrow X^n(t)$ converges weakly to a (tight,) Borel law;

(iii) for every $\varepsilon, \eta > 0$ there exists a finite partition $\{T_k : 1 \le k \le N\}$ of T such that

$$\limsup_{n \to \infty} P^{n*} \left(\max_{1 \le k \le N} \sup_{t, s \in T_k} |X^n(t) - X^n(s)| > \varepsilon \right) \le \eta;$$

(iv) there exist a semimetric ρ on T such that (T, ρ) is totally bounded and that for every $\varepsilon, \eta > 0$ there exists $\delta > 0$ such that

$$\limsup_{n \to \infty} P^{n*} \left(\sup_{\substack{t, s \in T \\ \rho(t,s) \le \delta}} |X^n(t) - X^n(s)| > \varepsilon \right) \le \eta.$$

Then, there is the equivalence (i) \iff (ii) + (iii) \iff (ii) + (iv). Furthermore, if the finite-dimensional marginals of a process $t \rightsquigarrow X(t)$ have the same laws as those of the limits in (ii), then there exists a version \widetilde{X} of X such that $X^n \stackrel{P^n}{\Longrightarrow} \widetilde{X}$ in $\ell^{\infty}(T)$ and that almost all paths $t \rightsquigarrow \widetilde{X}(t)$ are uniformly ρ -continuous, where ρ is the semimetric appearing in (iv). Furthermore, if the finite-dimensional marginals of the process $t \rightsquigarrow$ X(t) are Gaussian, the semimetric ρ_2 defined by

$$\rho_2(t,s) = \sqrt{E|X(t) - X(s)|^2} \qquad \forall t, s \in T$$

satisfies the same properties as ρ .

Remark. No measurability of X^n has been assumed. On the other hand, in the latter part of the conclusion, all finite-dimensional marginals of the processes $t \rightsquigarrow X(t)$ and $t \rightsquigarrow \widetilde{X}(t)$ are implicitly assumed to be Borel measurable. Moreover, the assertion means that it is possible to choose a version \widetilde{X} that is Borel measurable in $(\ell^{\infty}(T), \|\cdot\|_{\infty})$.

3.2 Multivariate Point Processes

Let (E, \mathcal{E}) be a Blackwell space and Ψ an arbitrary set. For every $n \in \mathbb{N}$, let μ^n be an E-valued multivariate point process defined on a stochastic basis $\mathbf{B}^n = (\Omega^n, \mathcal{F}^n, \mathbf{F}^n = (\mathcal{F}^n_t)_{t \in \mathbb{R}_+}, P^n)$, and ν^n a "good" version of the predictable compensator of μ^n . Let $\mathcal{W}^n = \{W^{n,\psi} : \psi \in \Psi\}$ be a family of predictable functions on $\tilde{\Omega}^n = \Omega^n \times \mathbb{R}_+ \times E$ indexed by Ψ . Let a DFP Π of Ψ be given. Notice that $(E, \mathcal{E}), \Psi$ and Π do not depend on n, while all other objects are indexed by $n \in \mathbb{N}$ (we will discuss the case of DFP's Π^n varying with $n \in \mathbb{N}$ at the end of this section). In the same way as Section 2.2, we introduce the following notations:

- the predictable envelope \overline{W}^n of \mathcal{W}^n ;
- the quadratic Π -modulus $\|\mathcal{W}^n\|_{\Pi}$ of \mathcal{W}^n .

Further, let a finite stopping time τ^n on \mathbf{B}^n be given. Throughout this section, we shall assume:

(3.2.1) the process $t \rightsquigarrow \overline{W}^n * \nu_t^n$ takes values in $[0, \infty)$.

As in Section 2.2, we define the local martingales $t \rightsquigarrow X_t^{n,\psi}$ and the locally squareintegrable martingales $t \rightsquigarrow X^{n,a,\psi}$ on \mathbf{B}^n by

$$X_t^{n,\psi} = W^{n,\psi} * (\mu^n - \nu^n)_t \qquad \forall t \in \mathbb{R}_+ \ \forall \psi \in \Psi$$

and

$$X^{n,a,\psi}_t = W^{n,\psi} \mathbf{1}_{\{\overline{W} \le a\}} * (\mu^n - \nu^n)_t \qquad \forall t \in \mathbb{R}_+ \ \forall \psi \in \Psi \ \forall a > 0,$$

respectively. We will derive the asymptotic behavior of the processes $\psi \rightsquigarrow X_{\tau^n}^{n,\psi}$ and $(t,\psi) \rightsquigarrow X_t^{n,\psi}$, as $n \to \infty$.

Let us now introduce several conditions. The first one is the *Partitioning Entropy con*dition, which is a natural generalization of the metric entropy condition for L^2 -bracketing in the I.I.D. case:

 $[\mathbf{PE}]$ there exists a DFP Π of Ψ such that

$$\|\mathcal{W}^n\|_{\Pi,\tau^n} = O_{P^n}(1) \text{ and } \int_0^{\Delta_{\Pi}} H_{\Pi}(\varepsilon) d\varepsilon < \infty.$$

Notice that, due to Lemma 2.2.2, under $[\mathbf{PE}]$ we can always construct a new NFP Π which satisfies the displayed conditions. Next, we shall also consider two kinds of *Lindeberg conditions*:

 $\begin{aligned} \mathbf{[L1]} \ \overline{W}^{n} \mathbf{1}_{\{\overline{W}^{n} > \varepsilon\}} * \nu_{\tau^{n}}^{n} \xrightarrow{P^{n}} 0 \text{ for every } \varepsilon > 0; \\ \mathbf{[L2]} \ |\overline{W}^{n}|^{2} \mathbf{1}_{\{\overline{W}^{n} > \varepsilon\}} * \nu_{\tau^{n}}^{n} \xrightarrow{P^{n}} 0 \text{ for every } \varepsilon > 0. \\ \end{aligned}$ When we mention [L2], the assumption that

(3.2.2) the process $t \rightsquigarrow |\overline{W}^n|^2 * \nu_t^n$ takes values in $[0, \infty)$

is also implicitly imposed in addition to (3.2.1), and in this case the process $t \sim X_t^{n,\psi}$ is a locally square-integrable martingale on \mathbf{B}^n . It is trivial that [L2] implies [L1].

Next let us introduce some conditions prescribing the asymptotic behavior of the quadratic covariations. Let S be a subset of \mathbb{R}_+ , and suppose that the family $\{C_t^{(\psi,\phi)} : t \in \mathbb{R}_+, (\psi,\phi) \in \Psi^2\}$ of constants in the following satisfies that

(3.2.3)
$$t \rightsquigarrow C_t^{(\psi,\phi)}$$
 is continuous for every $(\psi,\phi) \in \Psi^2$:

[C1] $[X^{n,\psi}, X^{n,\phi}]_t \xrightarrow{P^n} C_t^{(\psi,\phi)}$ for every $t \in S$ and $(\psi, \phi) \in \Psi^2$; [C2] $\langle X^{n,\psi}, X^{n,\phi} \rangle_t \xrightarrow{P^n} C_t^{(\psi,\phi)}$ for every $t \in S$ and $(\psi, \phi) \in \Psi^2$; [C1_a] $[X^{n,a,\psi}, X^{n,a,\phi}]_t \xrightarrow{P^n} C_t^{(\psi,\phi)}$ for every $t \in S$ and $(\psi, \phi) \in \Psi^2$, for every a > 0; [C2_a] $\langle X^{n,a,\psi}, X^{n,a,\phi} \rangle_t \xrightarrow{P^n} C_t^{(\psi,\phi)}$ for every $t \in S$ and $(\psi, \phi) \in \Psi^2$, for every a > 0. Similarly to the remark following [L2], the assumption (3.2.2) is implicitly imposed when we mention [C2]. It is well-known that the quadratic covariations are given by

$$[X^{n,\psi}, X^{n,\phi}]_t = \sum_{s \le t} \Delta X^{n,\psi}_s \Delta X^{n,\phi}_s$$

and (under the assumption (3.2.2))

$$\langle X^{n,\psi}, X^{n,\phi} \rangle_t = (W^{n,\psi}W^{n,\phi}) * \nu_t^n - \sum_{s \le t} \widehat{W}_s^{n,\psi} \widehat{W}_s^{n,\phi}$$

where $\widehat{W}_t^{n,\psi}(\omega) = \int_E W^{n,\psi}(\omega,t,x)\nu(\omega;\{t\}\times dx)$, respectively.

Using the constants $\{C_t^{\psi,\phi}\}$ appeared in the conditions above, we set (formally)

(3.2.4)
$$\rho_2((t,\psi),(s,\phi)) = \sqrt{C_t^{(\psi,\psi)} + C_s^{(\phi,\phi)} - 2C_{t\wedge s}^{(\psi,\phi)}}$$

for every $(t, \psi), (s, \phi) \in \mathbb{R}_+ \times \Psi$. Any of **[C1]**, **[C2]**, **[C1**_a] or **[C2**_a] implies that the value under the square-root is non-negative for every $(t, \psi), (s, \phi) \in S \times \Psi$, hence the \mathbb{R}_+ -valued function ρ_2 is well-defined by the formula (3.2.4) at least on $(S \times \Psi)^2$. Further,

by virtue of (3.2.3), this is true also on $([0, \tau] \times \Psi)^2$ if S is a dense subset of the finite interval $[0, \tau]$ with τ being a constant.

The assertions in the following lemma are clear or rather well-known (see e.g. Theorem VIII.3.6 of Jacod and Shiryaev (1987) for the part (ii) below), but we state the proofs with minor modification to our situation.

 $\begin{array}{l} \textbf{Lemma 3.2.1} (i) \ The \ condition \ [\textbf{L1}] \ implies \ the \ following: \\ (i_1) \ \overline{W}^n \textbf{1}_{\{\overline{W}^n > \varepsilon\}} * \mu_{\tau^n}^n \xrightarrow{P^n} 0 \ for \ every \ \varepsilon > 0; \\ (i_2) \ \sup_{t \in [0, \tau^n]} \sup_{\psi \in \Psi} |X_t^{n, \psi} - X_t^{n, a, \psi}| \xrightarrow{P^{n*}} 0 \ for \ every \ a > 0; \\ (i_3) \ \sup_{t \in [0, \tau^n]} \Delta(\overline{W}^n * \mu_t^n) \xrightarrow{P^n} 0 \ and \ \sup_{t \in [0, \tau^n]} \Delta(\overline{W}^n * \nu_t^n) \xrightarrow{P^n} 0; \\ (i_4) \ \sup_{t \in [0, \tau^n]} \sup_{\psi \in \Psi} |\Delta X_t^{n, a, \psi}| \xrightarrow{P^{n*}} 0 \ for \ every \ a > 0. \end{array}$

(ii) Let $\tau^n \equiv \tau$ be a fixed constant, and suppose that S is a subset of the finite interval $[0, \tau]$. Then, under [L1] it holds that [C1] \Leftrightarrow [C1_a] \Leftrightarrow [C2_a]. Under [L2], the condition [C2] is also equivalent to any of them.

Proof. It follows from Lenglart's inequality that

$$P^{n}\left(\overline{W}^{n}\mathbf{1}_{\{\overline{W}^{n}>\varepsilon\}}*\mu_{\tau^{n}}^{n}\geq\eta\right)\leq\eta+P^{n}\left(\overline{W}^{n}\mathbf{1}_{\{\overline{W}^{n}>\varepsilon\}}*\nu_{\tau^{n}}^{n}\geq\eta^{2}\right)\qquad\forall\eta>0,$$

hence the condition [L1] implies (i_1) . The assertions (i_2) , (i_3) and (i_4) are immediate from (i_1) .

Next we show the part (ii) of the lemma. By polarization it is enough to consider the case $\phi = \psi$. Observe that

$$\begin{split} \left| [X^{n,\psi}, X^{n,\psi}]_t &- [X^{n,a,\psi}, X^{n,a,\psi}]_t \right| \\ &= \left| \sum_{s \le t} (\Delta X^{n,\psi}_s + \Delta X^{n,a,\psi}_s) (\Delta X^{n,\psi}_s - \Delta X^{n,a,\psi}_s) \right| \\ &\le 2 \sum_{s \le t} \left| \Delta W^{n,\psi} \mathbf{1}_{\{\overline{W} > a\}} * (\mu^n - \nu^n)_s \right| \quad \text{on the set } \Omega^n_1 \\ &\le 2 \left| \overline{W}^n \mathbf{1}_{\{\overline{W} > a\}} * \mu^n_\tau + \overline{W}^n \mathbf{1}_{\{\overline{W} > a\}} * \nu^n_\tau \right|, \end{split}$$

where $\Omega_1^n = \{\sup_{t \in [0,\tau]} |\Delta X_t^{n,\psi}| \le 1\} \cup \{\sup_{t \in [0,\tau]} |\Delta X_t^{n,a,\psi}| \le 1\}$. The assertion that **[C1]** \Leftrightarrow **[C1**_a] under **[L1]** is now derived from (i₁), (i₃) and (i₄).

To show the equivalence $[\mathbf{C1}_a] \Leftrightarrow [\mathbf{C2}_a]$ under $[\mathbf{L1}]$, fix any a > 0, and we set $Y^n = [X^{n,a,\psi}, X^{n,a,\psi}] - \langle X^{n,a,\psi}, X^{n,a,\psi} \rangle$. We will prove that $\sup_{s \in [0,t]} |\Delta Y_s^n| \xrightarrow{P^n} 0$ for every $t \in S$ under either $[\mathbf{L1}] + [\mathbf{C1}_a]$ or $[\mathbf{L1}] + [\mathbf{C2}_a]$. Since $X^{n,a,\psi}$ is a locally square-integrable martingale, we have that Y^n is a local martingale and that so is $|Y^n|^2 - [Y^n, Y^n]$

(see Proposition I.4.50 of Jacod and Shiryaev (1987)). Hence Lenglart's inequality yields that for every $\varepsilon, \eta > 0$

$$\begin{split} P^n \left(\sup_{s \in [0,t]} |Y_s^n|^2 \ge \varepsilon \right) &\leq \frac{1}{\varepsilon} \left(\eta + E^n \sup_{s \in [0,t]} \Delta[Y^n, Y^n]_s \right) + P^n([Y^n, Y^n]_t \ge \eta) \\ &\leq \frac{2\eta}{\varepsilon} + \left(\frac{16a^4}{\varepsilon} + 1 \right) P^n([Y^n, Y^n]_t \ge \eta), \end{split}$$

because $\Delta[Y^n, Y^n] = |\Delta Y^n|^2 \le (|\Delta[X^{n,a,\psi}, X^{n,a,\psi}]|^2 \vee |\Delta \langle X^{n,a,\psi}, X^{n,a,\psi} \rangle|^2) \le 16a^4$. Thus it suffices to show that

$$(3.2.5) [Y^n, Y^n]_t \xrightarrow{P^n} 0 \forall t \in S$$

under either $[\mathbf{L1}] + [\mathbf{C1}_a]$ or $[\mathbf{L1}] + [\mathbf{C2}_a]$.

Since the local martingale Y^n has finite variation, we have

$$[Y^n, Y^n]_t = \sum_{s \le t} |\Delta Y^n_s|^2$$

$$\leq \sum_{s \le t} |\Delta [X^{n,a,\psi}, X^{n,a,\psi}]_s|^2 + \sum_{s \le t} |\Delta \langle X^{n,a,\psi}, X^{n,a,\psi} \rangle_s|^2$$

$$\leq \alpha^n_t A^n_t + \beta^n_t B^n_t,$$

where

$$\begin{aligned} \alpha_t^n &= \sup_{s \in [0,t]} \Delta[X^{n,a,\psi}, X^{n,a,\psi}]_s, \qquad A_t^n &= [X^{n,a,\psi}, X^{n,a,\psi}]_t, \\ \beta_t^n &= \sup_{s \in [0,t]} \Delta\langle X^{n,a,\psi}, X^{n,a,\psi} \rangle_s, \qquad B_t^n &= \langle X^{n,a,\psi}, X^{n,a,\psi} \rangle_t. \end{aligned}$$

Using (i₃), we obtain that $\alpha_t^n \xrightarrow{P^n} 0$ and $\beta_t^n \xrightarrow{P^n} 0$ for every $t \in S$, under [L1]. On the other hand, Lenglart's inequality yields that

$$P^{n}(A_{t}^{n} \geq \varepsilon) \leq \frac{\eta + 2a^{2}}{\varepsilon} + P^{n}(B_{t}^{n} \geq \eta) \qquad \forall \varepsilon, \eta > 0$$

and that

$$P^{n}(B^{n}_{t} \geq \varepsilon) \leq \frac{\eta + 4a^{2}}{\varepsilon} + P^{n}(A^{n}_{t} \geq \eta) \qquad \forall \varepsilon, \eta > 0.$$

Hence $[\mathbf{C2}_a]$ implies that $A_t^n = O_{P^n}(1)$, and $[\mathbf{C1}_a]$ does that $B_t^n = O_{P^n}(1)$. The claim (3.2.5) has been established.

The equivalence that $[\mathbf{C2}] \Leftrightarrow [\mathbf{C2}_a]$ under $[\mathbf{L2}]$ follows from the inequality

$$\begin{split} |\langle X^{n,\psi}, X^{n,\psi} \rangle_t &- \langle X^{n,a,\psi}, X^{n,a,\psi} \rangle_t | \\ &\leq |\overline{W}^n|^2 \mathbf{1}_{\{\overline{W}^n > a\}} * \nu_\tau^n \\ &+ \sum_{t \leq \tau} \int_E 2\overline{W}^n(t,x) \nu(\{t\} \times dx) \int_E \overline{W}^n(t,x) \mathbf{1}_{\{\overline{W}^n(t,x) > a\}} \nu(\{t\} \times dx) \\ &\leq |\overline{W}^n|^2 \mathbf{1}_{\{\overline{W}^n > a\}} * \nu_\tau^n + 2\overline{W}^n \mathbf{1}_{\{\overline{W}^n > a\}} * \nu_\tau^n \quad \text{ on the set } \Omega_2^n, \end{split}$$

where $\Omega_2^n = \{\sup_{t \in [0,\tau]} \Delta(\overline{W}^n * \nu_t^n) \le 1\}.$

The first result of this section is concerned with the processes $\psi \rightsquigarrow X_{\tau^n}^{n,\psi}$.

Theorem 3.2.2 Consider the above situation with (3.2.1). Suppose that every finitedimensional marginal of $X_{\tau^n}^n = (X_{\tau^n}^{n,\psi}|\psi \in \Psi)$ converges weakly to a (tight,) Borel law, and also that the conditions [**PE**] and [**L1**] are satisfied. Then $X_{\tau^n}^n$ converges weakly in $\ell^{\infty}(\Psi)$ to a tight, Borel law.

The result above is a direct consequence of the next lemma, applying Theorem 3.1.1

Lemma 3.2.3 The conditions [**PE**] and [**L1**] imply that for every $\varepsilon, \eta > 0$ there exists a finite partition { $\Psi_k : 1 \le k \le N$ } of Ψ such that

$$\limsup_{n \to \infty} P^{n*} \left(\sup_{\substack{t \in [0,\tau^n]}} \sup_{\substack{1 \le k \le N \\ \psi, \phi \in \Psi_k}} |X_t^{n,\psi} - X_t^{n,\phi}| > \varepsilon \right) \le \eta$$

Proof. Take a NFP Π which satisfies the requirements of **[PE]**. Fix any $\varepsilon, \eta > 0$. First notice that for any $\delta \in (0, \Delta_{\Pi}]$ and K > 0

$$(3.2.6) \qquad P^{n*}\left(\sup_{\substack{t\in[0,\tau^n]\\\psi,\phi\in\Psi(\delta;k)}}\sup_{X_t^{n,a(\delta,K),\psi}}-X_t^{n,a(\delta,K),\phi}|>\varepsilon\right)\leq (I)+(II)$$

where the terms on the right hand side are given by:

$$(I) = P^{n}(\|\mathcal{W}^{n}\|_{\Pi,\tau^{n}} > K);$$

$$(II) = \frac{1}{\varepsilon} E^{n*} \sup_{\substack{t \in [0,\tau^{n}] \\ \psi, \phi \in \Psi(\delta;k)}} \sup_{\substack{1 \le k \le N_{\Pi}(\delta) \\ \psi, \phi \in \Psi(\delta;k)}} |X_{t}^{n,a(\delta,K),\psi} - X_{t}^{n,a(\delta,K),\phi}| \mathbf{1}_{\{\|\mathcal{W}^{n}\|_{\Pi,\tau^{n}} \le K\}};$$

where $a(\delta, K) = \delta K / \tilde{H}_{\Pi}(\delta/2)$. It follows from (i) of Theorem 2.2.3 that there exists a universal constant C > 0 such that

(3.2.7)
$$(II) \le C \cdot \frac{K}{\varepsilon} \int_0^\delta \tilde{H}_{\Pi}(\epsilon) d\epsilon$$

Now, the first condition of $[\mathbf{PE}]$ yields that there exists a constant $K = K_{\eta} > 0$ such that $\limsup_{n\to\infty} (I) \leq \eta/2$. Next, since $\tilde{H}_{\Pi}(\epsilon) \leq 1 + H_{\Pi}(\epsilon)$, the second condition of $[\mathbf{PE}]$ implies that we can choose a sufficiently small constant $\delta = \delta_{\varepsilon,\eta} > 0$ such that the right hand side of (3.2.7) is not bigger than $\eta/2$. Consequently, (i₂) of Lemma 3.2.1 with $a = a(\delta_{\varepsilon,\eta}, K_{\eta})$ yields the assertion.

The next result deals with the processes $(t, \psi) \rightsquigarrow X_t^{n,\psi}$.

Theorem 3.2.4 Consider the above situation with (3.2.1) where $\tau^n \equiv \tau$ is a fixed positive constant, and let S be a dense subset of the finite interval $[0, \tau]$ containing τ . Suppose that either [**PE**] + [**L1**] + [**C1**] or [**PE**] + [**L2**] + [**C2**] is satisfied. Then, it holds that $X^n \xrightarrow{P^n} X$ in $\ell^{\infty}([0, \tau] \times \Psi)$, where each finite-dimensional marginal of the process $(t, \psi) \rightsquigarrow X_t^{\psi}$ has the Gaussian distribution $N(0, \Sigma)$ with $\Sigma = \{\Sigma_{ij}\}$ given by $\Sigma_{ij} = C_{t_i \wedge t_j}^{(\psi_i, \psi_j)}$. Furthermore, the formula (3.2.4) defines a semimetric ρ_2 on $[0, \tau] \times \Psi$ such that $[0, \tau] \times \Psi$ is totally bounded with respect to ρ_2 , and that almost all paths of Xare uniformly ρ_2 -continuous.

The following lemma, which is rather well-known, is used to show the result above.

Lemma 3.2.5 Under [L1] + [C1], for every $\psi \in \Psi$ and every $\varepsilon, \eta > 0$ there exists $\delta > 0$ such that

$$\limsup_{n \to \infty} P^n \left(\sup_{\substack{t, s \in [0, \tau] \\ |t-s| \le \delta}} |X_t^{n, \psi} - X_s^{n, \psi}| > \varepsilon \right) \le \eta$$

Proof. Fix any $N \in \mathbb{N}$ for a while, and put $a = N^{-1}$. By (ii) of Lemma 3.2.1 we may assume $[\mathbf{L1}] + [\mathbf{C2}_a]$. It always holds that $C_0^{(\psi,\psi)} = 0$ and that $t \rightsquigarrow C_t^{(\psi,\psi)}$ is increasing, because so does $t \rightsquigarrow \langle X^{n,a,\psi}, X^{n,a,\psi} \rangle_t$. We may assume $C_{\tau}^{(\psi,\psi)} > 0$ without loss of generality. Since $t \rightsquigarrow C_t^{(\psi,\psi)}$ is continuous and S is dense in $[0,\tau]$, we can choose some points $\tau_i \in S$ (i = 1, ..., N) such that $C_{\tau_i}^{(\psi,\psi)} - C_{\tau_{i-1}}^{(\psi,\psi)} = C_{\tau}^{(\psi,\psi)} N^{-1}$, where $\tau_0 = 0$ and $\tau_N = \tau$. It follows from Lemma 2.1.1 that for every $\varepsilon > 0$

$$P^n\left(\sup_{t\in[\tau_{i-1},\tau_i]}|X_t^{n,a,\psi}-X_{\tau_{i-1}}^{n,a,\psi}|>\varepsilon,\ \Omega_N^n\right)\leq 2\exp\left(-\frac{\varepsilon^2}{2[\varepsilon a+2C_{\tau}^{(\psi,\psi)}N^{-1}]}\right),$$

where

$$\Omega_N^n = \bigcap_{i=1}^N \left\{ \langle X^{n,a,\psi}, X^{n,a,\psi} \rangle_{\tau_i} - \langle X^{n,a,\psi}, X^{n,a,\psi} \rangle_{\tau_{i-1}} \le 2C_{\tau}^{(\psi,\psi)} N^{-1} \right\}.$$

Hence we have

$$P^n\left(\max_{1\leq i\leq N}\sup_{t\in[\tau_{i-1},\tau_i]}|X_t^{n,a,\psi}-X_{\tau_{i-1}}^{n,a,\psi}|>\varepsilon,\ \Omega_N^n\right)\leq 2N\exp\left(-\frac{\varepsilon^2N}{2[\varepsilon+2C_{\tau}^{(\psi,\psi)}]}\right).$$

Here notice that $\lim_{n\to\infty} P^n(\Omega_N^n) = 1$. Choosing a large number N, and then letting $n \to \infty$, we can easily deduce the assertion from (i₂) of Lemma 3.2.1.

Proof of Theorem 3.2.4. Let us check the conditions of Theorem 3.1.1. First, Theorem VIII.3.11 of Jacod and Shiryaev (1987) says that either of $[\mathbf{L1}] + [\mathbf{C1}_a]$ or $[\mathbf{L2}] +$

 $[\mathbf{C2}_{a}]$ implies the finite-dimensional convergence of $X^{n,a}$ for any a > 0 (recall also (i₄) of Lemma 3.2.1). Thus the finite-dimensional convergence of X^{n} follows from (i₂) and (ii) of Lemma 3.2.1. The condition (iii) of Theorem 3.1.1 can be shown by means of Lemmas 3.2.3 and 3.2.5.

Let us close this section with discussing the case of DFP's Π^n varying with $n \in \mathbb{N}$. In this case, we shall check the condition (iv) of Theorem 3.1.1 instead of (iii). We thus introduce the following condition.

 $[\mathbf{PE}^n]$ there exists a semimetric ρ on Ψ such that (Ψ, ρ) is totally bounded, and for every $n \in \mathbb{N}$ there exists a DFP Π^n of Ψ such that:

$$\|\mathcal{W}^{n}\|_{\Pi^{n},\tau^{n}} = O_{P^{n}}(1); \qquad \lim_{\delta \downarrow 0} \limsup_{n \to \infty} \int_{0}^{\delta} H_{\Pi^{n}}(\varepsilon) d\varepsilon = 0;$$

$$\lim_{\sigma \to \infty} \lim_{\delta \downarrow 0} \sup_{n \to \infty} P^{n*} \left(\sup_{\substack{\psi, \phi \in \Psi\\ \rho(\psi, \phi) \le \delta}} \frac{\sqrt{|W^{n,\psi} - W^{n,\phi}|^{2} 1_{\{\overline{W}^{n} \le 1\}} * \nu_{\tau^{n}}^{n}}}{\delta} > K \right) = 0$$

We then have an analogue of Lemma 3.2.3.

 $\frac{1}{K}$

Lemma 3.2.6 The conditions $[\mathbf{PE}^n]$ and $[\mathbf{L1}]$ imply that for every $\varepsilon, \eta > 0$ there exists $\delta > 0$ such that

$$\limsup_{n \to \infty} P^{n*} \left(\sup_{\substack{t \in [0,\tau^n]}} \sup_{\substack{\psi, \phi \in \Psi\\ \rho(\psi,\phi) \le \delta}} |X_t^{n,\psi} - X_t^{n,\phi}| > \varepsilon \right) \le \eta.$$

Proof. Repeat the same argument as Lemma 3.2.3 using Theorem 2.2.4 instead of (i) of Theorem 2.2.3. $\hfill \Box$

Consequently, Theorems 3.2.2 and 3.2.4 hold also with $[\mathbf{PE}^n]$ instead of $[\mathbf{PE}]$. In particular, Theorem 3.2.4 under $[\mathbf{PE}^n] + [\mathbf{L2}] + [\mathbf{C2}]$ refines Theorem 2.2 of Nishiyama (1997); the condition (B_3) there has been removed.

3.3 Martingale Difference Arrays

We give some analogues of Theorems 3.2.2 and 3.2.4; those can be shown using Theorem 2.3.3 (i) instead of Theorem 2.2.3 (i) (or using Theorem 2.3.4 instead of Theorem 2.2.4), thus the proofs are omitted. Let Ψ be an arbitrary set. For every $n \in \mathbb{N}$, let $\{\xi_i^n\}_{i \in \mathbb{N}}$ be an $\ell^{\infty}(\Psi)$ -valued martingale difference array on a discrete-time stochastic basis $\mathbf{B}^n = (\Omega^n, \mathcal{F}^n, \mathbf{F}^n = \{\mathcal{F}_i^n\}_{i \in \mathbb{N}_0}, P^n).$

Let Π a DFP of Ψ , and let Π^n be a sequence of DPF's of Ψ . In the same way as Section 2.3, we introduce the following notations:

- the adapted envelope $\{\overline{\xi}_i^n\}_{i\in\mathbb{N}}$ of $\{\xi_i^n\}_{i\in\mathbb{N}}$;
- the quadratic Π -modulus $\{ \|\xi^n\|_{\Pi,i} \}_{i \in \mathbb{N}}$ of $\{\xi^n_i\}_{i \in \mathbb{N}}$;
- the quadratic Π^n -modulus $\{\|\xi^n\|_{\Pi^n,i}\}_{i\in\mathbb{N}}$ of $\{\xi^n_i\}_{i\in\mathbb{N}}$.

We shall always assume:

$$(3.3.1) E^n \overline{\xi}_i^n < \infty \forall i \in \mathbb{N}.$$

For a given finite stopping time σ^n , we make the following conditions: [**PE**'] there exists a DFP Π of Ψ such that

$$\|\xi^n\|_{\Pi,\sigma^n} = O_{P^n}(1) \quad \text{and} \quad \int_0^{\Delta_{\Pi}} H_{\Pi}(\varepsilon) d\varepsilon < \infty;$$

 $[\mathbf{PE}^{n'}]$ there exists a semimetric ρ on Ψ such that (Ψ, ρ) is totally bounded, and for every $n \in \mathbb{N}$ there exists a DFP Π^n of Ψ such that

$$\|\xi^n\|_{\Pi^n,\sigma^n} = O_{P^n}(1), \qquad \lim_{\delta \downarrow 0} \limsup_{n \to \infty} \int_0^\delta H_{\Pi^n}(\varepsilon) d\varepsilon = 0,$$

and

$$\lim_{K \to \infty} \limsup_{\delta \downarrow 0} \limsup_{n \to \infty} P^{n*} \left(\sup_{\substack{\psi, \phi \in \Psi\\\rho(\psi, \phi) \le \delta}} \frac{\sqrt{\sum_{i=1}^{\sigma^n} E_{i-1}^n (|\xi_i^{n, \psi} - \xi_i^{n, \phi}|^2 \wedge 1)}}{\delta} > K \right) = 0$$

 $\begin{aligned} \mathbf{[L1']} & \sum_{i=1}^{\sigma^n} E_{i-1}^n \overline{\xi}_i^n \mathbf{1}_{\{\overline{\xi}_i^n > \varepsilon\}} \xrightarrow{P^n} 0 \text{ for every } \varepsilon > 0; \\ \mathbf{[L2']} & \sum_{i=1}^{\sigma^n} E_{i-1}^n |\overline{\xi}_i^n|^2 \mathbf{1}_{\{\overline{\xi}_i^n > \varepsilon\}} \xrightarrow{P^n} 0 \text{ for every } \varepsilon > 0. \end{aligned}$ When we mention $\mathbf{[L2']}$, the assumption that

$$(3.3.2) E^n |\overline{\xi}_i^n|^2 < \infty \quad \forall i \in \mathbb{N}.$$

which is stronger than (3.3.1), is implicitly imposed. It is clear that [L2'] implies [L1'].

Theorem 3.3.1 Consider the above situation with (3.3.1). Suppose that every finitedimensional marginal of $X^n = (X^{n,\psi}|\psi \in \Psi)$ given by $X^{n,\psi} = \sum_{i=1}^{\sigma^n} \xi_i^{n,\psi}$ converges weakly to a (tight,) Borel law, and also that either of $[\mathbf{PE'}] + [\mathbf{L1'}]$ or $[\mathbf{PE}^{n'}] + [\mathbf{L1'}]$ is satisfied. Then X^n converges weakly in $\ell^{\infty}(\Psi)$ to a tight, Borel law.

Next, let us consider the process $(t,\psi) \rightsquigarrow X^{n,\psi}_t$ given by

(3.3.3)
$$X_t^{n,\psi} = \sum_{i=1}^{\sigma_t^n} \xi_i^{n,\psi} \qquad \forall t \in [0,\tau] \; \forall \psi \in \Psi$$

where $\tau > 0$ is a constant, and $(\sigma_t^n)_{t \in [0,\tau]}$ is a family of finite stopping times on \mathbf{B}^n such that $\sigma_0^n = 0$ and that each path $t \rightsquigarrow \sigma_t^n$ is increasing, càd, with jumps equal to 1. We introduce two kinds of conditions, in which the family $\{C_t^{(\psi,\phi)} : t \in \mathbb{R}_+, (\psi,\phi) \in \Psi^2\}$ of

constants should satisfy (3.2.3): $[\mathbf{C1}'] \sum_{i=1}^{\sigma_t^n} \xi_i^{n,\psi} \xi_i^{n,\phi} \xrightarrow{P^n} C_t^{(\psi,\phi)} \text{ for every } t \in S \text{ and } (\psi,\phi) \in \Psi^2;$ $[\mathbf{C2}'] \sum_{i=1}^{\sigma_t^n} E_{i-1}^n \xi_i^{n,\psi} \xi_i^{n,\phi} \xrightarrow{P^n} C_t^{(\psi,\phi)} \text{ for every } t \in S \text{ and } (\psi,\phi) \in \Psi^2.$ Similarly to the remark following $[\mathbf{L2}']$, the assumption (3.3.2) is implicitly imposed when we mention $[\mathbf{C2}']$.

Theorem 3.3.2 Let S be a dense subset of the finite interval $[0, \tau]$ containing τ . Consider the above situation with (3.3.1), and assume $[\mathbf{PE'}]$ or $[\mathbf{PE^{n'}}]$ with $\sigma^n = \sigma_{\tau}^n$. Suppose also that either $[\mathbf{L1'}] + [\mathbf{C1'}]$ or $[\mathbf{L2'}] + [\mathbf{C2'}]$ is satisfied. Then, the same conclusion as Theorem 3.2.4 holds for the sequence of processes $X^n = (X_t^{n,\psi}|(t,\psi) \in [0,\tau] \times \Psi)$ defined by (3.3.3).

Let us close this section with stating a generalization of Jain-Marcus' (1975) central limit theorem to the case of martingale difference arrays. We denote by $N(\Psi, \rho; \varepsilon)$ the ε -covering number of a semimetric space (Ψ, ρ) .

Proposition 3.3.3 Let (Ψ, ρ) be a totally bounded semimetric space. For every $n \in \mathbb{N}$, let $\{\xi_i^n\}_{i\in\mathbb{N}}$ be an $\ell^{\infty}(\Psi)$ -valued martingale difference array on a discrete-time stochastic basis \mathbf{B}^n such that

$$|\xi_i^{n,\psi} - \xi_i^{n,\phi}| \le K_i^n \rho(\psi,\phi) \qquad \forall \psi, \phi \in \Psi,$$

where $\{K_i^n\}_{i\in\mathbb{N}}$ is an \mathbb{R}_+ -valued adapted process. For given finite stopping time σ^n , a sufficient condition for $[\mathbf{PE}']$ is

$$\sum_{i=1}^{\sigma^n} E_{i-1}^n |K_i^n|^2 = O_{P^n}(1) \quad and \quad \int_0^1 \sqrt{\log N(\Psi,\rho;\varepsilon)} d\varepsilon < \infty.$$

3.4 Continuous Local Martingales

Let us begin with giving a definition.

Definition 3.4.1 A family $X = (X^{\psi}|\psi \in \Psi)$ of continuous local martingales indexed by a metric space (Ψ, ρ) is said to be ρ -separable if there exist a countable subset Ψ^* of Ψ and a negligible set $N \in \mathcal{F}$ such that for every $\varepsilon > 0$ and $\omega \in \Omega \setminus N$

$$X_t^{\psi}(\omega) \in \overline{\{X_t^{\phi}(\omega) : \phi \in \Psi^*, \ \rho(\psi, \phi) < \varepsilon\}} \qquad \forall t \in \mathbb{R}_+, \ \forall \psi \in \Psi,$$

where the closure is taken in $\mathbb{R} \cup \{-\infty, +\infty\}$.

When (Ψ, ρ) is separable, a sufficient condition for the ρ -separability is that almost all paths $\psi \rightsquigarrow X_t^{\psi}$ are ρ -continuous, but it is not always easy to check the continuity in general. On the other hand, it is clear that any family of continuous local martingales indexed by a countable set Ψ is ρ -separable (for any metric ρ on Ψ).

Let us now turn to the context of weak convergence. Let (Ψ, ρ) be a totally bounded proper metric space. For every $n \in \mathbb{N}$, let $X^n = (X^{n,\psi}|\psi \in \Psi)$ be a (not necessarily ρ -separable) family of continuous local martingales X^{ψ} such that $X_0^{\psi} = 0$ indexed by Ψ defined on a stochastic basis $\mathbf{B}^n = (\Omega^n, \mathcal{F}^n, \mathbf{F}^n = (\mathcal{F}_t^n)_{t \in \mathbb{R}_+}, P^n)$. Let τ^n be a finite stopping time on \mathbf{B}^n . We introduce the *Metric Entropy condition*. [**ME**] Given finite stopping time τ^n on \mathbf{B}^n ,

$$\|X^n\|_{\rho,\tau^n} = O_{P^{n*}}(1)$$
 and $\int_0^1 \sqrt{\log N(\Psi,\rho;\varepsilon)} d\varepsilon < \infty.$

Theorem 3.4.2 In the above situation, suppose that the family $X^n = (X^{n,\psi}|\psi \in \Psi)$ is ρ -separable and that $X^n_{\tau^n} = (X^{n,\psi}_{\tau^n}|\psi \in \Psi)$ takes values in $\ell^{\infty}(\Psi)$ almost surely. Suppose also that every finite-dimensional marginal of $X^n_{\tau^n}$ converges to a (tight,) Borel law, and that the condition [**ME**] is satisfied. Then $X^n_{\tau^n}$ converges weakly in $\ell^{\infty}(\Psi)$ to a tight, Borel law.

Proof. The assertion is immediate from Theorems 2.4.2 and 3.1.1. \Box

The result above generalizes Theorem 1 of Bae and Levental (1995b) who assumed the continuity of $(t, \psi) \rightsquigarrow X_t^{n, \psi}$. According to Theorem 2.4.3, when Ψ is countable and **[ME]** is assumed, a sufficient condition for $X_{\tau^n}^n = (X_{\tau^n}^{n,\psi}|\psi \in \Psi)$ to take values in $\ell^{\infty}(\Psi)$ almost surely is that $P^n(||X^n||_{\rho,\tau^n} < \infty) = 1$. If one encounters the situation where Ψ is uncountable and the ρ -separability of the family $X^n = (X^{n,\psi}|\psi \in \Psi)$ itself is difficult to show, the following result concerning a uniformly ρ -continuous version $\psi \rightsquigarrow \tilde{X}^n(\psi)$ of $\psi \sim X_{\tau^n}^{n,\psi}$ will be helpful. Recall that a sufficient condition for the existence of such a version is given by Theorem 2.4.4, and that $\tilde{X}^n(\psi)$ is not the terminal variable of a continuous local martingale.

Corollary 3.4.3 In the above situation, suppose that every finite-dimensional marginal of $X_{\tau^n}^n$ converges to a (tight,) Borel law, and that the condition [**ME**] is satisfied. Suppose also that it holds for a choice of the quadratic ρ -modulus $||X^n||_{\rho}$ that

$$P^{n}\left(\left[\|X^{n}\|_{\rho,\tau^{n}}\right]_{\mathcal{F}^{n}_{\tau^{n}},P^{n}}<\infty\right)=1\qquad\forall n\in\mathbb{N},$$

and choose a uniformly ρ -continuous version $\psi \rightsquigarrow \widetilde{X}^n(\psi)$ of the process $\psi \rightsquigarrow X^{n,\psi}_{\tau^n}$. Then \widetilde{X}^n converges weakly in $\ell^{\infty}(\Psi)$ to a tight, Borel law. If the limit of each finite-dimensional

<u>3.A. Notes</u>

marginal of $\psi \rightsquigarrow X^{n,\psi}_{\tau^n}$ coincides with that of a process $\psi \rightsquigarrow X(\psi)$, then there exists a uniformly ρ -continuous version \widetilde{X} of X such that $\widetilde{X}^n \stackrel{P^n}{\Longrightarrow} \widetilde{X}$ in $\ell^{\infty}(\Psi)$.

Proof. The finite-dimensional convergence of $\psi \rightsquigarrow \widetilde{X}^n(\psi)$ follows from that of $\psi \rightsquigarrow X_{\tau^n}^{n,\psi}$. Next choose a countable dense subset Ψ^* of Ψ . Then, by Theorem 3.4.2, the statement (iv) of Theorem 3.1.1 is satisfied for $(X_{\tau^n}^{n,\psi}|\psi \in \Psi^*)$, thus also for $(\widetilde{X}^n(\psi)|\psi \in \Psi)$.

Next we consider the process $(t, \psi) \sim X_t^{n,\psi}$. Given subset S of \mathbb{R}_+ , we make also the following condition, in which the family $\{C_t^{(\psi,\phi)} : t \in \mathbb{R}_+, (\psi,\phi) \in \Psi^2\}$ of constants should satisfy (3.2.3):

[C2] $\langle X^{n,\psi}, X^{n,\phi} \rangle_t \xrightarrow{P^n} C_t^{(\psi,\phi)}$ for every $t \in S$ and $(\psi,\phi) \in \Psi^2$.

Theorem 3.4.4 Let S be a dense subset of the a finite interval $[0, \tau]$ containing τ . In the above situation, suppose that the family $X^n = (X^{n,\psi}|\psi \in \Psi)$ is ρ -separable and that $(X_t^{n,\psi}|(t,\psi) \in [0,\tau] \times \Psi)$ takes values in $\ell^{\infty}([0,\tau] \times \Psi)$ almost surely. Assume [ME] with $\tau^n \equiv \tau$ and [C2]. Then, the same conclusion as Theorem 3.2.4 holds for X^n .

Proof. Repeat the same assertion as Lemma 3.2.5 to obtain under [C2] that for every $\psi \in \Psi$ and every $\varepsilon, \eta > 0$ there exists $\delta > 0$ such that

$$\limsup_{n \to \infty} P^n \left(\sup_{\substack{t, s \in [0, \tau] \\ |t-s| \le \delta}} |X_t^{n, \psi} - X_s^{n, \psi}| > \varepsilon \right) \le \eta.$$

This fact, together with Theorem 2.4.2, implies the asymptotic tightness. Thus, the assertion follows from Theorem 3.1.1 and the martingale central limit theorem. \Box

The result above refines Theorem 2.3 of Nishiyama (1997); the condition (C_3) there has been removed and the condition (C_5) has been weakened.

3.A Notes

The study of the Donsker theorems for i.i.d. empirical processes indexed by classes of sets or functions was initiated by the landmark paper by Dudley (1978), and was developed in the 80's. There are two types of sufficient conditions for the Donsker property, namely, the uniform entropy condition (Pollard (1982)) and the L^2 -bracketing condition (Ossiander (1987)); see van der Vaart and Wellner (1996) and references therein for refinement and generalizations up to row-independent arrays. In particular, it should be noted that Andersen et al. (1988) contains a result based on a weaker condition than L^2 -bracketing one in a row-independent case. In the recent years, some authors have considered to remove the assumption of independence: Arcone and Yu (1994) and Doukhan et al. (1995) for stationary sequences based on mixing conditions; Levental (1989), Bae (1995) and Bae and Levental (1995a) for stationary martingale sequences; Bae and Levental (1995b) for continuous martingales; Nishiyama (1997) for some continuous-time semimartingales. One can find the roots of the quantity "quadratic modulus" in the works by Bae and Levental cited above. A major part of the results in this chapter was originally presented by Nishiyama (1997), although some of the conditions have been refined as mentioned in the main text. The refinement is partly due to the use of the tightness criterion in terms of partitioning (i.e., (iii) of Theorem 3.1.1) rather than the well-known stochastic ρ -equicontinuity criterion. Van der Vaart and Wellner (1996) is apparently the first to present the partitioning criterion.

For the weak convergence of (finite-dimensional) semimartingales, one should consult the excellent book by Jacod and Shiryaev (1987) which we have referred many times.

Chapter 4

Integral Estimators

4.1 Multiplicative Intensity Model

Let (E, \mathcal{E}) be a Blackwell space on which a measure λ is defined. For every $n \in \mathbb{N}$, let μ^n be an *E*-valued multivariate point process defined on a filtered measurable space $(\Omega^n, \mathcal{F}^n, \mathbf{F}^n = (\mathcal{F}_t^n)_{t \in \mathbb{R}_+})$. Notice that μ^n can be identified with an *E*-marked point process $\{(T_i^n, Z_i^n); i \in \mathbb{N}\}$ through the equality

$$\mu^{n}(\omega; dt, dz) = \sum_{i} \varepsilon_{(T_{i}^{n}(\omega), Z_{i}^{n}(\omega))}(dt, dz),$$

where $0 < T_1^n < T_2^n < \cdots$ and each Z_i^n is an *E*-valued random variable. We suppose that the predictable compensator ν^n of μ^n under the probability measure P^n on $(\Omega^n, \mathcal{F}^n)$ is given by

$$\nu^{n}(\omega; dt, dz) = \alpha(t, z) Y^{n}(\omega, t, z) dt \lambda(dz).$$

where $\alpha(t,z)$ is a $[0,\infty)$ -valued measurable function on $\mathbb{R}_+ \times E$, and $Y^n(\omega,t,z)$ is a $[0,\infty)$ -valued predictable function on $\Omega^n \times \mathbb{R}_+ \times E$.

Let a constant $\tau > 0$ be given. Throughout this section, we always assume

$$\int_{[0,\tau]\times E} \alpha(t,z) dt \lambda(dz) < \infty;$$

then $\alpha(t, z)dt\lambda(dz)$ defines a finite measure on $[0, \tau] \times E$. We will use the following notation

$$\mathcal{L}^{p}(\alpha) = \mathcal{L}^{p}([0,\tau] \times E, \mathfrak{B}[\mathfrak{o},\tau] \otimes \mathcal{E}, \alpha(\mathfrak{t},\mathfrak{z})\mathfrak{d}\mathfrak{t}\lambda(\mathfrak{d}\mathfrak{z})) \qquad \forall \mathfrak{p} \in [\mathfrak{1},\infty].$$

We denote by $\|\cdot\|_{\mathcal{L}^p(\alpha)}$ the L^p -seminorm on $\mathcal{L}^p(\alpha)$.

4.1.1 Asymptotic Normality

Let Ψ be a subset of $\mathcal{L}^{2p}(\alpha)$ with an envelop function φ belonging to $\mathcal{L}^{2p}(\alpha)$, where $p \in [1, \infty]$ should be specified in connection with another assumption in Condition 4.1.1

below. Our goal is to estimate the functional $(t, \psi) \rightsquigarrow A(\psi)_t$ where

$$A(\psi)_t = \int_{[0,t]\times E} \psi(s,z)\alpha(s,z)ds\lambda(dz) \qquad \forall (t,\psi)\in [0,\tau]\times \Psi.$$

To do it, we introduce the generalized inverse Y^{n-} of Y^n defined by

$$Y^{n-}(\omega,t,z) = \frac{\mathbb{1}_{\{Y^n(\omega,t,z) \ge 1\}}}{Y^n(\omega,t,z)},$$

with the convention 0/0 = 0; we define also

$$I^n(\omega, t, z) = \mathbb{1}_{\{Y^n(\omega, t, z) \ge 1\}}.$$

It then holds that $Y^{n-}Y^n = I^n$. We propose the estimator $(t, \psi) \rightsquigarrow \widehat{A}^n(\psi)_t$ defined by

$$\widehat{A}^{n}(\psi)_{t}(\omega) = (\psi Y^{n-}) * \mu_{t}^{n}(\omega) \qquad \forall (t,\psi) \in [0,\tau] \times \Psi.$$

The main step is to derive the weak convergence of the residual $\sqrt{n}(\widehat{A}^n - \widetilde{A}^n)$, where $(t, \psi) \rightsquigarrow \widetilde{A}^n(\psi)_t$ is given by

$$\widetilde{A}^{n}(\psi)_{t}(\omega) = \int_{[0,t]\times E} \psi(s,z)I^{n}(\omega,s,z)\alpha(s,z)dt\lambda(dz) \qquad \forall (t,\psi)\in[0,\tau]\times\Psi.$$

We make the following condition.

Condition 4.1.1 For some $p, q \in [1, \infty]$ such that (1/p) + (1/q) = 1, and a measurable function y = y(t, z) on $[0, \tau] \times E$, which is bounded away from zero, it holds that:

(4.1.1)
$$\varphi \in \mathcal{L}^{2p}(\alpha) \quad and \quad \int_0^1 \sqrt{\log N_{[]}(\Psi, \|\cdot\|_{\mathcal{L}^{2p}(\alpha)}; \varepsilon)} d\varepsilon < \infty;$$

(4.1.2)
$$\|nY^{n-}\|_{\mathcal{L}^{q}(\alpha)} = O_{P^{n}}(1);$$

(4.1.3)
$$\left\|\varphi^2 \cdot |nY^{n-} - (1/y)|\right\|_{\mathcal{L}^1(\alpha)} = o_{P^n}(1).$$

This condition generalizes (8.4.1) of Andersen et al. (1993). Although the framework of this section does not contain empirical processes of i.i.d. data, one may find an interesting "difference" between the i.i.d. and the present cases. In the i.i.d. case, since the random elements nY^{n-} do not appear, the entropy condition for L^2 -bracketing is always optimal. In the present case, however, it is sometimes wise to seek for the entropy condition (4.1.1) with respect to a stronger semimetric when the requirement (4.1.2) can be checked only for $q < \infty$. For instance, when $E = \mathbb{R}$ and $\psi = 1_{(-\infty,z]}$, the bracketing entropy condition is satisfied with the L^{2p} -semimetric for any $p < \infty$, and thus (4.1.2) for an arbitrary q > 1 is sufficient. Compare the present Condition 4.1.1 with Conditions 4.1 and 4.2 of Nishiyama (1997) which were concerned only with the case of p = 1 and $q = \infty$.

Theorem 4.1.2 Suppose that a given class Ψ satisfies Condition 4.1.1. Then, it holds that $\sqrt{n}(\widehat{A}^n - \widetilde{A}^n) \stackrel{P^n}{\Longrightarrow} G$ in $\ell^{\infty}([0, \tau] \times \Psi)$, where $(t, \psi) \sim G_t^{\psi}$ is a zero-mean Gaussian process such that

$$EG_t^{\psi}G_s^{\phi} = \int_{[0,t\wedge s]\times E} \psi(u,z)\phi(u,z)\frac{\alpha(u,z)}{y(u,z)}du\lambda(dz) \qquad \forall (t,\psi), (s,\phi) \in [0,\tau] \times \Psi$$

and that almost all paths are uniformly ρ_2 -continuous on $[0, \tau] \times \Psi$, where

$$\rho_2((t,\psi),(s,\phi)) = \sqrt{E|G_t^{\psi} - G_s^{\phi}|^2} \qquad \forall (t,\psi), (s,\phi) \in [0,\tau] \times \Psi.$$

Further if

(4.1.4)
$$\|\varphi \cdot (1-I^n)\|_{\mathcal{L}^1(\alpha)} = o_{P^n}(n^{-1/2}),$$

then it also holds that $\sqrt{n}(\widehat{A}^n - A) \stackrel{P^n}{\Longrightarrow} G$ in $\ell^{\infty}([0, \tau] \times \Psi)$.

Proof. We will apply Theorem 3.2.4 to $\mathcal{W}^n = \{W^{n,\psi} : \psi \in \Psi\}$ given by $W^{n,\psi} = \sqrt{n\psi}Y^{n-}$. It indeed holds that $\sqrt{n}(\widehat{A}^n(\psi)_t - \widetilde{A}^n(\psi)_t) = W^{n,\psi} * (\mu^n - \nu^n)_t$.

First notice that

$$\begin{split} \langle W^{n,\psi}*(\mu^n-\nu^n), W^{n,\phi}*(\mu^n-\nu^n)\rangle_t \\ &= n\int_{[0,t]\times E}\psi(s,z)\phi(s,z)Y^{n-}(s,z)\alpha(s,z)ds\lambda(dz) \end{split}$$

Since

$$\begin{split} \left\| \int_{[0,t]\times E} \psi(s,z)\phi(s,z) \left\{ nY^{n-}(s,z) - \frac{1}{y(s,z)} \right\} \alpha(s,z)ds\lambda(dz) \\ &\leq \int_{[0,\tau]\times E} |\varphi(s,z)|^2 \left| nY^{n-}(s,z) - \frac{1}{y(s,z)} \right| \alpha(s,z)ds\lambda(dz) \\ &= \left\| \varphi^2 \cdot |nY^{n-} - (1/y)| \right\|_{\mathcal{L}^1(\alpha)}, \end{split}$$

the condition [C2] follows from (4.1.3). To show [L2], notice that $\overline{W}^n \leq \sqrt{n}\varphi Y^{n-}$ and that, when $p \in [1, \infty)$,

$$(4.1.5) \qquad \left(|\sqrt{n}\varphi Y^{n-}|^{2} \mathbb{1}_{\{\sqrt{n}\varphi Y^{n-}>\varepsilon\}} \right) * \nu_{\tau}^{n} \\ = \left\| \varphi^{2} \cdot nY^{n-} \mathbb{1}_{\{\sqrt{n}\varphi Y^{n-}>\varepsilon\}} \right\|_{\mathcal{L}^{1}(\alpha)} \\ \leq \left\| \varphi \right\|_{\mathcal{L}^{2p}(\alpha)} \cdot \left\| \varphi \cdot nY^{n-} \mathbb{1}_{\{\sqrt{n}\varphi Y^{n-}>\varepsilon\}} \right\|_{\mathcal{L}^{2p/(2p-1)}(\alpha)} \\ \leq \left\| \varphi \right\|_{\mathcal{L}^{2p}(\alpha)} \cdot \frac{1}{\sqrt{n}\varepsilon} \left\| \varphi \cdot nY^{n-} \right\|_{\mathcal{L}^{2p/(2p-1)}(\alpha)} \\ \leq \left\| \varphi \right\|_{\mathcal{L}^{2p}(\alpha)} \cdot \frac{1}{\sqrt{n}\varepsilon} \cdot \left\| \varphi \right\|_{\mathcal{L}^{2p}(\alpha)} \cdot \left\| nY^{n-} \right\|_{\mathcal{L}^{q}(\alpha)},$$

which converges in P^n -probability to zero by $\varphi \in \mathcal{L}^{2p}(\alpha)$ and (4.1.2). The case of $p = \infty$ and q = 1 is easier.

To check the condition [**PE**], for every $\varepsilon \in (0, 1]$, choose $(\|\cdot\|_{\mathcal{L}^{2p}}, \varepsilon)$ -brackets $[l^{\varepsilon,k}, u^{\varepsilon,k}]$ in \mathcal{L}^{2p} which cover Ψ . Introduce a DFP Π of Ψ induced from these brackets, that is, $\Pi(\varepsilon) = \{\Psi(\varepsilon; k) : 1 \leq k \leq N_{\Pi}(\varepsilon)\}$ is given by $\Psi(\varepsilon; k) = \{\psi \in \Psi : l^{\varepsilon,k} \leq \psi \leq u^{\varepsilon,k}\}$ with modification to make the partition disjoint. This can be done with $N_{\Pi}(\varepsilon) = N_{[]}(\Psi, \|\cdot\|_{\mathcal{L}^{2p}}; \varepsilon)$. Since

$$\begin{split} |\sqrt{n}(u^{\varepsilon,k} - l^{\varepsilon,k})Y^{n-}|^2 * \nu_{\tau}^n &= \left\| |u^{\varepsilon,k} - l^{\varepsilon,k}|^2 nY^{n-} \right\|_{\mathcal{L}^1(\alpha)} \\ &\leq \left\| |u^{\varepsilon,k} - l^{\varepsilon,k}|^2 \right\|_{\mathcal{L}^p(\alpha)} \cdot \|nY^{n-}\|_{\mathcal{L}^q(\alpha)} \\ &= \left\| u^{\varepsilon,k} - l^{\varepsilon,k} \right\|_{\mathcal{L}^{2p}(\alpha)}^2 \cdot \|nY^{n-}\|_{\mathcal{L}^q(\alpha)} \\ &\leq \varepsilon^2 \|nY^{n-}\|_{\mathcal{L}^q(\alpha)}, \end{split}$$

the quadratic Π -modulus $\|\mathcal{W}^n\|_{\Pi,\tau}$ is bounded by $\sqrt{\|nY^{n-}\|_{\mathcal{L}^q(\alpha)}}$, which is bounded in P^n -probability. This completes the proof.

4.1.2 Asymptotic Efficiency

Let us discuss the asymptotic efficiency of the estimator \widehat{A}^n following the general theory developed in Chapter 3.11 of van der Vaart and Wellner (1996). We set:

$$(4.1.6) \qquad \mathbb{H} = L^2([0,\tau] \times E, \mathfrak{B}[\mathfrak{o},\tau] \otimes \mathcal{E}, \frac{\alpha(\mathfrak{t},\mathfrak{z})}{\mathfrak{y}(\mathfrak{t},\mathfrak{z})} \mathfrak{d}\mathfrak{t}\lambda(\mathfrak{d}\mathfrak{z})); \\ H = L^\infty([0,\tau] \times E, \mathfrak{B}[\mathfrak{o},\tau] \otimes \mathcal{E}, \mathfrak{d}\mathfrak{t}\lambda(\mathfrak{d}\mathfrak{z})).$$

Here, the function y is the one appearing in Condition 4.1.1. We equip \mathbb{H} with the usual L^2 -inner product $\langle \cdot, \cdot \rangle_{\mathbb{H}}$. Since 1/y is assumed to be bounded, H is a linear subspace of the Hilbert space \mathbb{H} . Let $\mathbf{P}^n = \{P_h^n : h \in H\}$ be a family of probability measures on $(\Omega^n, \mathcal{F}^n)$ indexed by H.

Suppose that the predictable compensator of $N^{n,i}$ under the probability measure P_h^n is given by

$$\alpha_h^n(t,z)Y^n(t,z)dt,$$

where $\alpha_h^n = \alpha_h^n(t, z)$ is defined by

$$\alpha_h^n = \left(1 + \frac{h}{2\sqrt{ny}}\right)^2 \alpha.$$

Notice that $\alpha_0^n = \alpha$.

STEP I: LOCAL ASYMPTOTIC NORMALITY. Assume $P_h^n \ll P_0^n$. It is well-known that, under some conditions, a version of the log-likelihood ratio is given by

(4.1.7)
$$\log \frac{dP_h^n |\mathcal{F}_{\tau}^n}{dP_0^n |\mathcal{F}_{\tau}^n} = 2\left(\log\left|1 + \frac{h}{2\sqrt{ny}}\right|\right) * \mu_{\tau}^n - \left(\left|1 + \frac{h}{2\sqrt{ny}}\right|^2 - 1\right) * \nu_{\tau}^{n,0}$$

(see, e.g., Theorem III.5.43 of Jacod and Shiryaev (1987)).

Proposition 4.1.3 Suppose that

$$\left\| (1/y^2) \cdot |n^{-1}Y^n - y| \right\|_{\mathcal{L}^1(\alpha)} = o_{P_0^n}(1)$$

is satisfied, where y is the function appearing in Condition 4.1.1, and introduce the Hilbert space \mathbb{H} given by (4.1.6). Let C be a given subset of H, and suppose that $P_h^n \ll P_0^n$ and (4.1.7) hold for every $h \in C$. Then, it holds that for every $h \in C$

$$\log \frac{dP_h^n |\mathcal{F}_{\tau}^n}{dP_0^n |\mathcal{F}_{\tau}^n} = \Delta^n(h) - \frac{1}{2} \|h\|_{\mathbb{H}}^2 + \epsilon_n(h),$$

where

$$\Delta^{n}(h) = \left(\frac{h}{\sqrt{ny}}\right) * (\mu^{n} - \nu^{n,0})_{\tau}$$

and $\epsilon_n(h) = o_{P_0^n}(1)$. Furthermore, it also holds that $(\Delta^n(h_1), ..., \Delta^n(h_d)) \xrightarrow{P_0^n} N(0, \Sigma)$ where $\Sigma_{ij} = \langle h_i, h_j \rangle_{\mathbb{H}}$.

Proof. Since $|\log(1+x) - x + \frac{x^2}{2}| \le \frac{8}{3}x^3$ for all $x \in [-\frac{1}{2}, \frac{1}{2}]$, we have that for any $\varepsilon \in (0, 1]$,

$$\log \frac{dP_h^n |\mathcal{F}_{\tau}^n}{dP_0^n |\mathcal{F}_{\tau}^n} - \Delta^n(h) + D^n(h) + \tilde{D}^n(h) \bigg| \le \frac{16}{3} \varepsilon D^n(h)$$

whenever $\sup_{t,z} \frac{|h(t,z)|}{\sqrt{n}y(t,z)} \leq \varepsilon$, where:

$$D^{n}(h) = \left|\frac{h}{2\sqrt{ny}}\right|^{2} * \mu_{\tau}^{n};$$
$$\tilde{D}^{n}(h) = \left|\frac{h}{2\sqrt{ny}}\right|^{2} * \nu_{\tau}^{n,0}.$$

Notice that $\tilde{D}^n(h) \xrightarrow{P_0^n} \frac{1}{4} \|h\|_{\mathbb{H}}^2$. Also, using Lenglart's inequality, we have $D^n(h) - \tilde{D}^n(h) \xrightarrow{P_0^n} 0$. These facts imply the first assertion. The finite-dimensional convergence of $h \rightsquigarrow \Delta^n(h)$ follows easily from the martingale central limit theorem. \Box

STEP II: DIFFERENTIABILITY OF UNKNOWN PARAMETER. We consider the functional $(t, \psi) \sim A(\psi)_t$ as an unknown parameter taking values in the Banach space $(\ell^{\infty}([0, \tau] \times \Psi), \| \cdot \|_{\infty})$. We denote by $\ell^{\infty}([0, \tau] \times \Psi)^*$ the dual space of $\ell^{\infty}([0, \tau] \times \Psi)$. Introducing a sequence of operators $A^n : H \to \ell^{\infty}([0, \tau] \times \Psi)$, which should be regarded as a local sequence of A, we aim to get a derivative operator $\dot{A} : H \to \ell^{\infty}([0, \tau] \times \Psi)$ with rate r_n (= \sqrt{n} in the present case) which should satisfy

$$r_n \|A^n(h) - A^n(0) - \dot{A}(h)\|_{\infty} \to 0 \qquad \forall h \in H,$$

and its adjoint operator $\dot{A}^* : \ell^{\infty}([0,\tau] \times \Psi)^* \to \bar{H}$, where \bar{H} is the completion of H in \mathbb{H} (thus $\bar{H} = \mathbb{H}$ in the present case), which is determined by

$$\langle \dot{A}^* b^*, h \rangle_{\mathbb{H}} = b^* \dot{A}(h) \qquad \forall h \in H$$

for every $b^* \in \ell^{\infty}([0,\tau] \times \Psi)^*$.

Now, we define the sequence of operators $A^n: H \to \ell^{\infty}([0,\tau] \times \Psi)$ by

$$A^{n}(h)(\psi)_{t} = \int_{[0,t]\times E} \psi(s,z)\alpha_{h}^{n}(s,z)dt\lambda(dz).$$

Then the sequence A^n is differentiable with rate \sqrt{n} and its derivative $\dot{A}: H \to \ell^{\infty}([0,\tau] \times \Psi)$ is given by

$$\dot{A}(h)(t,\psi) = \langle 1_{[0,t]}\psi,h \rangle_{\mathbb{H}}.$$

We denote by $\pi_{t,\psi} : \ell^{\infty}([0,\tau] \times \Psi) \to \mathbb{R}$ the projection on the (t,ψ) -coordinate, which belongs to $\ell^{\infty}([0,\tau] \times \Psi)^*$. The above formula shows that $\dot{A}^*\pi_{t,\psi} = 1_{[0,t]}\psi$ for every $(t,\psi) \in [0,\tau] \times \Psi$, and this means that the process $(t,\psi) \rightsquigarrow G_t^{\psi}$ appearing in the limit of Theorem 4.1.2 satisfies that

$$EG_t^{\psi}G_s^{\phi} = \langle A^*\pi_{t,\psi}, A^*\pi_{s,\phi} \rangle_{\mathbb{H}} \qquad \forall (t,\psi), (s,\phi) \in [0,\tau] \times \Psi.$$

Since the law of the process $(t, \psi) \rightsquigarrow G_t^{\psi}$ is characterized by its finite-dimensional distributions, we can conclude that it coincides with the bound of asymptotic efficiency (see Theorem 3.11.2 of van der Vaart and Wellner (1996)).

STEP III: ASYMPTOTIC EFFICIENCY. In order to discuss the asymptotic efficiency in the sense of the convolution theorem, it remains to show that the estimator $(t, \psi) \rightsquigarrow \widehat{A}^n(\psi)_t$ for $(t, \psi) \rightsquigarrow A^n(h)(\psi)_t$ is regular. The following is an easy consequence of Theorem 4.1.2 and Proposition 4.1.3 which implies the contiguity.

Proposition 4.1.4 Suppose that a given class Ψ satisfies Condition 4.1.1 and the assumption (4.1.4) with $P^n = P_0^n$, and introduce the Hilbert space \mathbb{H} given by (4.1.6). Let

C be an arbitrary subset of H, and suppose that all assumptions of Proposition 4.1.3 are satisfied. Then, it holds that

$$\sqrt{n}(\widehat{A}^n - A^n(h)) \stackrel{P_h^n}{\Longrightarrow} G \qquad in \ \ell^{\infty}([0,\tau] \times \Psi) \qquad \forall h \in C,$$

where $(t, \psi) \sim G_t^{\psi}$ is the process appearing in the limit of Theorem 4.1.2.

Proof. The local asymptotic normality established in Proposition 4.1.3 implies that P_h^n and P_0^n are two-sided contiguous (see, e.g., Definition 3.10.1 and Example 3.10.6 of van der Vaart and Wellner (1996)), although $P_h^n \triangleleft P_0^n$ suffices for us. Hence we have that Condition 4.1.1 and the assumption (4.1.4) are satisfied with $P^n = P_h^n$ for all $h \in C$. We can obtain the assertion in the same way as Theorem 4.1.2.

Consequently, the estimator \widehat{A}^n has shown to be regular. It also holds for any bounded, continuous function $\ell : \ell^{\infty}([0,\tau] \times \Psi) \to [0,\infty)$ that

(4.1.8)
$$\sup_{I \subset C} \limsup_{n \to \infty} \sup_{h \in I} E_{h*}^n \ell\left(\sqrt{n}(\widehat{A}^n - A^n(h))\right) = E\ell(G),$$

where the supremum with respect to $I \subset C$ is taken over all finite subsets I of C. Summarizing the above discussion, we can conclude that:

Corollary 4.1.5 Suppose that all assumptions of Proposition 4.1.4 are satisfied. If C is a convex cone in \mathbb{H} such that its closed linear span coincides with \mathbb{H} , then the estimator \widehat{A}^n for A^n is asymptotically efficient in the sense of the convolution and the locally asymptotic minimax theorems with respect to bounded, continuous, subconvex loss functions.

See Theorems 3.11.2 and 3.11.5 of van der Vaart and Wellner (1996) for the convolution and the locally asymptotic minimax theorems. See also their Example 3.11.8 for the choice of loss functions in the latter theorem. In particular, when we choose a loss function $\ell : \ell^{\infty}([0,\tau] \times \Psi) \to [0,\infty)$ of the type $\ell(z) = \ell_0(||z||_{\infty})$, where $\ell_0 : [0,\infty) \to$ $[0,\infty)$ is a bounded, continuous, increasing function, their Theorem 3.11.5 says that: for any $T^n : \Omega^n \to \ell^{\infty}([0,\tau] \times \Psi)$ such that $T^n(t,\psi)$ is \mathcal{F}^n_{τ} -measurable for every $(t,\psi) \in$ $[0,\tau] \times \Psi$, it holds that

$$\sup_{I \subset C} \liminf_{n \to \infty} \sup_{h \in I} E_{h*}^n \ell\left(\sqrt{n}(T^n - A^n(h))\right) \geq E\ell(G)$$

where the supremum with respect to $I \subset C$ is taken over all finite subsets I of C. Recalling (4.1.8), we can see that the estimator \widehat{A}^n achieves this bound.

4.2 Continuous Semimartingales with Non-linear Covariates

Let $(E, \mathcal{E}, \lambda)$ be a finite measure space; this is the state space of covariate processes in the following model. In the *n*-th statistical experiment, we consider k_n continuous, adapted processes $X^{n,i}$ defined on a filtered measurable space $(\Omega^n, \mathcal{F}^n, \mathbf{F}^n = (\mathcal{F}_t^n)_{t \in \mathbb{R}_+})$. Suppose that $X^{n,i}$'s are special semimartingales under the probability measure P^n on $(\Omega^n, \mathcal{F}^n)$, and that their canonical decompositions are given by

$$dX_{t}^{n,i} = \alpha(t, Z_{t}^{n,i})Y_{t}^{n,i}dt + Y_{t}^{n,i}dB_{t}^{n,i} \qquad \forall i = 1, ..., k_{n},$$

where $\alpha(t, z)$ is an \mathbb{R} -valued $\mathfrak{B}(\mathbb{R}_+) \otimes \mathcal{E}$ -measurable function, $t \rightsquigarrow Z_t^{n,i}$'s are E-valued predictable processes, $t \rightsquigarrow Y_t^{n,i}$'s are $\{0,1\}$ -valued predictable processes, and $t \rightsquigarrow B_t^{n,i}$'s are orthogonal Brownian motions. It has implicitly been assumed that

$$\int_0^t |\alpha(s, Z_s^{n,i}(\omega))| Y_s^{n,i}(\omega) ds < \infty \qquad \forall \omega \in \Omega^n, \ \forall t \in \mathbb{R}_+$$

for every *i*. Notice that we do not assume the independence of $Z^{n,i}$'s and $Y^{n,i}$'s (thus $X^{n,i}$'s, too). We are interested in estimating the functional $(t,\psi) \rightsquigarrow A(\psi)_t$, which is given by

(4.2.1)
$$A(\psi)_t = \int_{[0,t]\times E} \psi(s,z)\alpha(s,z)ds\lambda(dz),$$

where ψ belongs to an appropriate class of measurable functions on $\mathbb{R}_+ \times E$.

4.2.1 Preliminaries

Let $E = \bigcup_m E_m^n$ be a partition of \mathcal{E} -measurable sets, that is at most countable. Set $\mathcal{E}^n = \sigma\{E_m^n : m = 1, 2, ...\}$. We denote $\tilde{\mathcal{P}}^n = \mathcal{P}^n \otimes \mathcal{E}$ and $\tilde{\mathcal{Q}}^n = \mathcal{P}^n \otimes \mathcal{E}^n$, where \mathcal{P}^n is the predictable σ -field on $\Omega^n \times \mathbb{R}_+$. We introduce a kind of "generalized inverse" Y^{n-} defined by

$$Y^{n-}(\omega,t,z) = \sum_{m} \frac{1_{\{\overline{Y}_{m}^{n}(\omega,t)>0\}}\lambda(E_{m}^{n})}{\overline{Y}_{m}^{n}(\omega,t)} \mathbb{1}_{\{z\in E_{m}^{n}\}}$$

with the convention 0/0 = 0, where

$$\overline{Y}_m^n(\omega,t) = \sum_{i=1}^{k_n} Y_t^{n,i}(\omega) \mathbb{1}_{\{Z_t^{n,i}(\omega) \in E_m^n\}}.$$

Here notice that (4.2.2)

$$0 \leq Y^{n-} \leq \sup_m \lambda(E^n_m) < \infty$$

Define also

$$I^{n}(\omega, t, z) = \sum_{m} \mathbb{1}_{\{\overline{Y}^{n}_{m}(\omega, t) > 0\}} \mathbb{1}_{\{z \in E^{n}_{m}\}}.$$

We denote by $\overline{\alpha}^n$ the $\mathfrak{B}(\mathbb{R}_+) \otimes \mathcal{E}^n$ -measurable majorant of α , and by $\underline{\alpha}^n$ the $\mathfrak{B}(\mathbb{R}_+) \otimes \mathcal{E}^n$ -measurable minorant of α . By using the notation of Definition 2.1.4, they are given by:

$$\begin{aligned} \overline{\alpha}^n &= & [\alpha]_{\mathfrak{B}(\mathbb{R}_+)\otimes\mathcal{E}^{\mathfrak{n}},\mathfrak{d}\mathfrak{t}\lambda(\mathfrak{d}\mathfrak{z})} ; \\ \underline{\alpha}^n &= & - [-\alpha]_{\mathfrak{B}(\mathbb{R}_+)\otimes\mathcal{E}^{\mathfrak{n}},\mathfrak{d}\mathfrak{t}\lambda(\mathfrak{d}\mathfrak{z})} \end{aligned}$$

The "generalized inverse" Y^{n-} will be useful through the following lemma.

Lemma 4.2.1 For any \mathbb{R} -valued, $\tilde{\mathcal{Q}}^n$ -measurable function W^n on $\Omega^n \times \mathbb{R}_+ \times E$, it holds that

$$\sum_{i=1}^{k_n} \int_0^t W^n(s, Z_s^{n,i}) Y^{n-}(s, Z_s^{n,i}) Y_s^{n,i} ds = \int_{[0,t] \times E} W^n(s, z) I^n(s, z) ds \lambda(dz) \qquad \forall t \in \mathbb{R}_+,$$

identically, provided the integrals are finite. Furthermore, it holds that

$$\begin{split} \left| \sum_{i=1}^{k_n} \int_0^t W^n(s, Z_s^{n,i}) \alpha(s, Z_s^{n,i}) Y^{n-}(s, Z_s^{n,i}) Y_s^{n,i} ds - \int_{[0,t] \times E} W^n(s, z) \alpha(s, z) I^n(s, z) ds \lambda(dz) \right| \\ & \leq \int_{[0,t] \times E} |W^n(s, z)| (\overline{\alpha}^n - \underline{\alpha}^n)(s, z) I^n(s, z) ds \lambda(dz) \quad \forall t \in \mathbb{R}_+, \end{split}$$

identically, provided the integrals are finite.

Proof. The first equality indeed holds since

$$\begin{split} &\sum_{i=1}^{k_n} \int_{[0,t]\times E} W^n(\omega,s,Z_s^{n,i}(\omega)) Y^{n-}(\omega,s,Z_s^{n,i}(\omega)) Y_s^{n,i}(\omega) ds \\ &= \int_{[0,t]\times E} \sum_m W^n(\omega,s,z_m^n) \mathbf{1}_{\{\overline{Y}_m^n(\omega,s)>0\}} \lambda(E_m^n) ds \\ &= \int_{[0,t]\times E} W^n(\omega,s,z) I^n(\omega,s,z) ds \lambda(dz), \end{split}$$

where z_m^n is any point of E_m^n . The second inequality is an easy consequence of the first. \Box

Let a constant $\tau > 0$ be given, and denote:

(4.2.3)
$$\mathcal{L}^p = \mathcal{L}^p([0,\tau] \times E, \mathfrak{B}[\mathfrak{o},\tau] \otimes \mathcal{E}, \mathfrak{dt}\lambda(\mathfrak{dg})) \quad \forall \mathfrak{p} \in [\mathfrak{1},\infty];$$

$$(4.2.4) \qquad \qquad \mathcal{L}^{p,n} = \mathcal{L}^{p}([0,\tau] \times E, \mathfrak{B}[\mathfrak{o},\tau] \otimes \mathcal{E}^{\mathfrak{n}}, \mathfrak{dt}\lambda(\mathfrak{dg})) \qquad \forall \mathfrak{p} \in [\mathfrak{1},\infty].$$

We denote by $\|\cdot\|_{\mathcal{L}^p}$ the L^p -seminorm on \mathcal{L}^p . For every $n \in \mathbb{N}$, we introduce the mapping $\pi^n : \mathcal{L}^1 \to \mathcal{L}^{1,n}$ by

(4.2.5)
$$\pi^{n}(\psi)(t,z) = \sum_{m} \frac{1_{\{\lambda(E_{m}^{n})>0\}}}{\lambda(E_{m}^{n})} \int_{E_{m}^{n}} \psi(t,w)\lambda(dw) \mathbb{1}_{\{z\in E_{m}^{n}\}}$$

with the convention 0/0 = 0.

Lemma 4.2.2 (i) For every $p \in [1, \infty]$, it holds that $\pi^n \mathcal{L}^p \subset \mathcal{L}^{p,n}$, and the mapping $\pi^n : \mathcal{L}^p \to \mathcal{L}^{p,n}$ is linear.

(ii) If $\psi, \phi \in \mathcal{L}^1$ and $\psi \leq \phi$, then $\pi^n(\psi) \leq \pi^n(\phi)$.

(iii) For every $p \in [1, \infty)$, it holds for any $\psi \in \mathcal{L}^1$ and any function $f : [0, \tau] \times E \to [0, \infty)$ which is $\mathfrak{B}[\mathfrak{o}, \tau] \otimes \mathcal{E}^n$ -measurable that

$$\int_{[0,\tau]\times E} |\pi^n(\psi)(t,z)|^p f(t,z) dt\lambda(dz) \le \int_{[0,\tau]\times E} |\psi(t,z)|^p f(t,z) dt\lambda(dz),$$

provided the right hand side is finite. (iv) It holds for any $\psi \in \mathcal{L}^{\infty}$ that $\|\pi^n(\psi)\|_{\mathcal{L}^{\infty}} \leq \|\psi\|_{\mathcal{L}^{\infty}}$.

Proof. The assertions (i), (ii) and (iv) are trivial, and (iii) follows from that

$$\begin{aligned} |\pi^{n}(\psi)(t,z)|^{p} &= \sum_{m} \frac{1_{\{\lambda(E_{m}^{n})>0\}}}{\lambda(E_{m}^{n})^{p}} \left| \int_{E_{m}^{n}} \psi(t,w)\lambda(dw) \right|^{p} \mathbf{1}_{\{z\in E_{m}^{n}\}} \\ &\leq \sum_{m} \frac{1_{\{\lambda(E_{m}^{n})>0\}}}{\lambda(E_{m}^{n})} \int_{E_{m}^{n}} |\psi(t,w)|^{p} \lambda(dw) \mathbf{1}_{\{z\in E_{m}^{n}\}}. \end{aligned}$$

4.2.2 A	Asymptotic	Normal	ity
---------	------------	--------	-----

Let $\tau > 0$ be a constant, and let Ψ be a countable class of $\mathfrak{B}[\mathfrak{o}, \tau] \otimes \mathcal{E}$ -measurable functions on $[0, \tau] \times E$. We will always assume

(4.2.6)
$$\sup_{\psi \in \Psi} \int_{[0,\tau] \times E} |\psi(t,z)| (|\overline{\alpha}^n| \vee |\underline{\alpha}^n|)(t,z) dt \lambda(dz) < \infty;$$

then, the functional $(t, \psi) \rightsquigarrow A(\psi)_t$ defined by (4.2.1) can be thought as an unknown parameter taking values in the space $\ell^{\infty}([0, \tau] \times \Psi)$. The main step in our approach is to estimate the modified unknown parameter $(t, \psi) \rightsquigarrow \widetilde{A}^n(\psi)_t$ given by

$$\widetilde{A}^{n}(\psi)_{t}(\omega) = \int_{[0,t]\times E} \pi^{n}(\psi)(s,z)I^{n}(\omega,s,z)ds\lambda(dz) \qquad \forall (t,\psi)\in[0,\tau]\times\Psi.$$

It follows from Lemma 4.2.2 that this also takes values in $\ell^{\infty}([0,\tau] \times \Psi)$ under the assumption (4.2.6). We will discuss later under which conditions this \widetilde{A}^n is indeed "close" to A. We propose the estimator $(t,\psi) \rightsquigarrow \widehat{A}^n(\psi)_t$ given by

$$\widehat{A}^{n}(\psi)_{t} = \sum_{i=1}^{k_{n}} \int_{0}^{t} \pi^{n}(\psi)(s, Z_{s}^{n,i}) Y^{n-}(s, Z_{s}^{n,i}) dX_{s}^{n,i} \qquad \forall (t, \psi) \in [0, \tau] \times \Psi$$

The stochastic integral appearing above is well-defined if

$$(4.2.7) \sup_{t \in [0,\tau]} |\pi^n(\psi)(t, Z_t^{n,i}(\omega))| Y^{n-}(\omega, t, Z_t^{n,i}(\omega)) < \infty \qquad \forall \omega \in \Omega^n, \ \forall i = 1, \dots, k_n$$

(see Theorem I.4.31 of Jacod and Shiryaev (1987)). We will always assume (4.2.7), and define $t \rightsquigarrow \widehat{A}^n(\psi)_t$ by any version of the stochastic integral, which is unique up to a P^n -negligible set. It is immediate from the definitions of π^n and Y^{n-} that a sufficient condition for (4.2.7) is that

$$\sup_{t\in[0,\tau]}\sup_{m}\int_{E_{m}^{n}}|\psi(t,z)|\lambda(dz)<\infty.$$

However, it is still unclear even under this condition that $(t, \psi) \rightsquigarrow \widehat{A}^n(\psi)_t$ takes values in $\ell^{\infty}([0, \tau] \times \Psi)$. This requirement will be shown to be fulfilled under (4.2.2) and (4.2.8) given below, by means of Theorem 2.4.3.

Condition 4.2.3 For some $p, q \in [1, \infty]$ such that (1/p) + (1/q) = 1, and a $\mathfrak{B}[\mathfrak{o}, \tau] \otimes \mathcal{E}$ measurable function y = y(t, z), which is bounded away from zero, it holds that $\Psi \subset \mathcal{L}^{2p}$,
and that $\|\cdot\|_{\mathcal{L}^{2p}}$ defines a proper metric on Ψ , and that:

(4.2.8)
$$\int_0^1 \sqrt{\log N(\Psi, \|\cdot\|_{\mathcal{L}^{2p}}; \varepsilon)} d\varepsilon < \infty;$$

(4.2.9)
$$||nY^{n-}||_{\mathcal{L}^q} = O_{P^n}(1);$$

(4.2.10)
$$\left\| |\pi^{n}(\psi)|^{2} \cdot |nY^{n-} - (1/y)| \right\|_{\mathcal{L}^{1}} = o_{P^{n}}(1) \quad \forall \psi \in \Psi;$$

(4.2.11)
$$\|\{\pi^n(\psi)\pi^n(\phi) - \psi\phi\} \cdot (1/y)\|_{\mathcal{L}^1} \to 0 \qquad \forall \psi, \phi \in \Psi;$$

(4.2.12)
$$\sup_{\psi \in \Psi} \|\pi^n(\psi) \cdot (\overline{\alpha}^n - \underline{\alpha}^n) \cdot I^n\|_{\mathcal{L}^1} = o_{P^n}(n^{-1/2}).$$

Compare this with Condition 4.1.1; the conditions (4.2.8), (4.2.9) and (4.2.10) correspond to (4.1.1) (4.1.2), and (4.1.3), respectively, while (4.2.11) and (4.2.12) are needed for the present situation.

Theorem 4.2.4 Suppose that a given countable class Ψ satisfies the assumptions (4.2.6) and (4.2.7) and Condition 4.2.3. Then, the random elements $(t, \psi) \rightsquigarrow \widehat{A}^n(\psi)_t$ and $(t, \psi) \rightsquigarrow \widetilde{A}^n(\psi)_t$ take values in $\ell^{\infty}([0, \tau] \times \Psi)$, P^n -almost surely, and it holds that $\sqrt{n}(\widehat{A}^n - \widetilde{A}^n) \stackrel{P^n}{\Longrightarrow} G$ in $\ell^{\infty}([0, \tau] \times \Psi)$, where $(t, \psi) \rightsquigarrow G_t^{\psi}$ is a zero-mean Gaussian process such that

$$EG_t^{\psi}G_s^{\phi} = \int_{[0,t\wedge s]\times E} \frac{\psi(u,z)\phi(u,z)}{y(u,z)} du\lambda(dz) \qquad \forall (t,\psi), (s,\phi) \in [0,\tau] \times \Psi$$

and that almost all paths are uniformly ρ_2 -continuous on $[0, \tau] \times \Psi$, where

$$\rho_2((t,\psi),(s,\phi)) = \sqrt{E|G_t^{\psi} - G_s^{\phi}|^2} \qquad \forall (t,\psi), (s,\phi) \in [0,\tau] \times \Psi.$$

Proof. Observe that

$$\sqrt{n}(\widehat{A}^{n}(\psi)_{t} - \widetilde{A}^{n}(\psi)_{t}) = M_{t}^{n,\psi} + N_{t}^{n,\psi}$$

where:

$$\begin{split} M_t^{n,\psi} &= \sqrt{n} \sum_{i=1}^{k_n} \int_0^t \pi^n(\psi)(s, Z_s^{n,i}) Y^{n-}(s, Z_s^{n,i}) Y_s^{n,i} dB_s^{n,i}; \\ N_t^{n,\psi} &= \sqrt{n} \sum_{i=1}^{k_n} \int_0^t \pi^n(\psi)(s, Z_s^{n,i}) \alpha(s, Z_s^{n,i}) Y^{n-}(s, Z_s^{n,i}) Y_s^{n,i} ds \end{split}$$

It follows from Lemmas 4.2.1 and 4.2.2 that the term $N^{n,\psi}$ is well-defined under the assumption (4.2.6), and that

$$\left|N_t^{n,\psi} - \sqrt{n}\widetilde{A}^n(\psi)_t\right| \le \sqrt{n} \int_{[0,t]\times E} |\pi^n(\psi)(s,z)|(\overline{\alpha}^n - \underline{\alpha}^n)(s,z)I^n(s,z)ds\lambda(dz),$$

which converges in P^n -probability to zero uniformly in $(t, \psi) \in [0, \tau] \times \Psi$ by (4.2.12).

On the other hand, $t \rightsquigarrow M_t^{n,\psi}$ is a continuous local martingale under the assumption (4.2.7). Since $B^{n,i}$'s are orthogonal Brownian motions, and since $|Y^{n,i}|^2 = Y^{n,i}$, we obtain from Lemma 4.2.1 that

$$\langle M^{n,\psi}, M^{n,\phi} \rangle_t = n \int_{[0,t] \times E} \pi^n(\psi)(s,z) \pi^n(\phi)(s,z) Y^{n-}(s,z) ds \lambda(dz).$$

It follows from Lemma 4.2.2 that

$$\begin{split} \langle M^{n,\psi} - M^{n,\phi}, M^{n,\psi} - M^{n,\phi} \rangle_{\tau} \\ &= n \left\| |\pi^{n}(\psi) - \pi^{n}(\phi)|^{2} Y^{n-} \right\|_{\mathcal{L}^{1}} \\ &\leq n \left\| |\psi - \phi|^{2} Y^{n-} \right\|_{\mathcal{L}^{1}} \\ &\leq n \| |\psi - \phi|^{2} \|_{\mathcal{L}^{p}} \cdot \| Y^{n-} \|_{\mathcal{L}^{q}} \\ &= n \| \psi - \phi \|_{\mathcal{L}^{2p}}^{2} \cdot \| Y^{n-} \|_{\mathcal{L}^{q}}. \end{split}$$

Thus we have that the quadratic $\|\cdot\|_{\mathcal{L}^{2p}}$ -modulus of $M^n = (M^{n,\psi}|\psi \in \Psi)$ is bounded by $\sqrt{n\|Y^{n-}\|_{\mathcal{L}^q}}$. Hence, Theorem 2.4.3 implies that almost all paths of $(t,\psi) \rightsquigarrow M_t^{n,\psi}$ take values in $\ell^{\infty}([0,\tau] \times \Psi)$, and thus so do those of $(t,\psi) \rightsquigarrow \widehat{A}^n(\psi)_t$.

In order to derive the weak convergence of the processes $(t, \psi) \rightsquigarrow M_t^{n,\psi}$, we will apply Theorem 3.4.4. The above computation of the quadratic $\|\cdot\|_{\mathcal{L}^{2p}}$ -modulus together with (4.2.9), yields [**ME**]. Next, notice that

$$\begin{split} \left| \int_{[0,t]\times E} \pi^{n}(\psi)(s,z)\pi^{n}(\phi)(s,z) \left\{ nY^{n-}(s,z) - \frac{1}{y(s,z)} \right\} ds\lambda(dz) \right| \\ &\leq \sqrt{\int_{[0,\tau]\times E} |\pi^{n}(\psi)(s,z)|^{2} \left| nY^{n-}(s,z) - \frac{1}{y(s,z)} \right| ds\lambda(dz)} \\ &\times \sqrt{\int_{[0,\tau]\times E} |\pi^{n}(\psi)(s,z)|^{2} \left| nY^{n-}(s,z) - \frac{1}{y(s,z)} \right| ds\lambda(dz)} \end{split}$$

and that

$$\begin{split} \left\| \int_{[0,t]\times E} \left\{ \pi^{n}(\psi)(s,z)\pi^{n}(\phi)(s,z) - \psi(s,z)\phi(s,z) \right\} \frac{1}{y(s,z)} ds\lambda(dz) \\ &\leq \| \left\{ \pi^{n}(\psi)\pi^{n}(\phi) - \psi\phi \right\} \cdot (1/y) \|_{\mathcal{L}^{1}} \,. \end{split}$$

Thus (4.2.10) and (4.2.11) imply [C2]. This finishes the proof.

In order to derive the weak convergence of $\sqrt{n}(\widehat{A}^n - A)$ rather than $\sqrt{n}(\widehat{A}^n - \widetilde{A}^n)$, we have to show that $\sqrt{n}(\widetilde{A}^n - A) \stackrel{P^n}{\Longrightarrow} 0$ in $\ell^{\infty}([0, \tau] \times \Psi)$. For this purpose, it suffices to check the following:

(4.2.13)
$$\sup_{\psi \in \Psi} \| (\pi^n(\psi) - \psi) \cdot \alpha \|_{\mathcal{L}^1} = o(n^{-1/2});$$

(4.2.14)
$$\sup_{\psi \in \Psi} \|\psi \cdot (1 - I^n) \cdot \alpha\|_{\mathcal{L}^1} = o_{P^n}(n^{-1/2}).$$

Example: Euclidean covariates

Set $E = [0, 1]^d$, and equip it with the Lebesgue measure. Suppose that $(t, z) \rightsquigarrow \alpha(t, z)$ is continuous on $[0, \tau] \times [0, 1]^d$ with respect to the (d + 1)-dimensional Euclidean metric. Consider the class of indicator functions $\Psi = \{1_{[0,u_1] \times \cdots \times [0,u_d]}(z) : u \in [0,1]^d\}$. This problem in the case of d = 1 was studied by McKeague and Utikal (1990), based on a classical criterion of tightness of sequences of stochastic processes with finite-dimensional parameters. Among the assumptions appearing above, the entropy condition (4.2.8) is satisfied for any $p < \infty$. If we choose a partition $[0,1]^d = \bigcup_m E_m^n$ of d-dimensional

rectangles with side length at most b_n , where $b_n n^{1/2} \to 0$, then the weak convergence of $\sqrt{n}(\widehat{A}^n - A)$ in $\ell^{\infty}([0,\tau] \times [0,1]^d)$ follows from that:

$$\|nY^{n-}\|_{\mathcal{L}^{q}} = O_{P^{n}}(1) \quad \text{for some } q > 1;$$
$$\|nY^{n-} - (1/y)\|_{\mathcal{L}^{1}} = o_{P^{n}}(1);$$
$$\|(1 - I^{n}) \cdot \alpha\|_{\mathcal{L}^{1}} = o_{P^{n}}(n^{-1/2}).$$

The first condition above refines (A3) of McKeague and Utikal (1990), which is corresponding to the case of q = 3.

4.2.3 Asymptotic Efficiency

Let us discuss the asymptotic efficiency of the estimator $(t, \psi) \sim \widehat{A}^n(\psi)_t$ along the general theory exposited in Chapter 3.11 of van der Vaart and Wellner (1996), again. We set:

$$(4.2.15) \qquad \mathbb{H} = L^{2}([0,\tau] \times E, \mathfrak{B}[\mathfrak{o},\tau] \otimes \mathcal{E}, \frac{1}{\mathfrak{y}(\mathfrak{t},\mathfrak{z})}\mathfrak{d}\mathfrak{t}\lambda(\mathfrak{d}\mathfrak{z})); \\ H = L^{\infty}([0,\tau] \times E, \mathfrak{B}[\mathfrak{o},\tau] \otimes \mathcal{E}, \mathfrak{d}\mathfrak{t}\lambda(\mathfrak{d}\mathfrak{z})); \\ H^{n} = L^{\infty}([0,\tau] \times E, \mathfrak{B}[\mathfrak{o},\tau] \otimes \mathcal{E}^{\mathfrak{n}}, \mathfrak{d}\mathfrak{t}\lambda(\mathfrak{d}\mathfrak{z})).$$

Here, the function y is the one which appears in Condition 4.2.3. We equip \mathbb{H} with the usual L^2 -inner product $\langle \cdot, \cdot \rangle_{\mathbb{H}}$. Notice that H is a linear subspace of the Hilbert space \mathbb{H} since 1/y is assumed to be bounded. Let $\mathbf{P}^n = \{P_h^n : h \in H\}$ be a family of probability measures on $(\Omega^n, \mathcal{F}^n)$ indexed by H. Define the mapping $\pi^n : H \to H^n$ by (4.2.5). Hereafter, we denote by $\pi^n(h)$ any function of the equivalent elements in H^n .

Suppose that the canonical decompositions of special semimartingales $X^{n,i}$ under the probability measure P_h^n are given by

(4.2.16)
$$dX_t^{n,i} = \alpha_h^n(t, Z_t^{n,i}) Y_t^{n,i} dt + Y_t^{n,i} dB_t^{n,h,i} \qquad \forall i = 1, ..., k_n,$$

where $\alpha_h^n = \alpha_h^n(t, z)$ is defined by

(4.2.17)
$$\alpha_h^n = \alpha + n^{-1/2} \pi^n (h/y),$$

and where $t \rightsquigarrow B_t^{n,h,i}$'s are orthogonal Brownian motions on $[0,\tau]$ under P_h^n . We should first see that the local model (4.2.16) is "well-defined" in the sense that it does not depend on the choice of a version of $\pi^n(h/y) \in H^n$. To see this, notice that, if $f,g \in$ $\mathcal{L}^{\infty}([0,\tau] \times E, \mathfrak{B}[\mathfrak{o},\tau] \otimes \mathcal{E}^n, \mathfrak{dt}\lambda(\mathfrak{dg}))$ such that f(t,z) = g(t,z) for $dt\lambda(dz)$ -almost all (t,z), then it holds that for every $\omega \in \Omega^n$

$$f(t, Z_t^{n,i}(\omega)) = g(t, Z_t^{n,i}(\omega))$$
 dt -almost all t .

Thus, for every $\omega \in \Omega^n$, the function $t \rightsquigarrow \alpha_h^n(t, Z_t^{n,i}(\omega))$ is uniquely determined by (4.2.17) up to a negligible set with respect to the Lebesgue measure on $[0, \tau]$, not depending on the choice of a version of $\pi^n(h/y) \in H^n$. Hence the model (4.2.16) is well-defined, and in particular, it holds that $\alpha_0^n(t, Z_t^{n,i}(\omega)) = \alpha(t, Z_t^{n,i}(\omega))$ for almost all t.

STEP I: LOCAL ASYMPTOTIC NORMALITY. It is well-known that, under some conditions, a version of the log-likelihood ratio is given by

$$(4.2.18) \qquad \log \frac{dP_h^n |\mathcal{F}_{\tau}^n}{dP_0^n |\mathcal{F}_{\tau}^n} = \frac{1}{\sqrt{n}} \sum_{i=1}^{k_n} \int_0^{\tau} \pi^n (h/y)(t, Z_t^{n,i}) Y_t^{n,i} dB_t^{n,0,i} \\ -\frac{1}{2n} \sum_{i=1}^{k_n} \int_0^{\tau} |\pi^n (h/y)(t, Z_t^{n,i})|^2 Y_t^{n,i} dt$$

(see, e.g., Theorem III.5.32 of Jacod and Shiryaev (1987)). Again, this representation does not depend on the choice of a version of $\pi^n(h/y) \in H^n$.

Proposition 4.2.5 Let C be an arbitrary subset of H. Suppose that the function y appearing in Condition 4.2.3 satisfies that

(4.2.19)
$$y \in \mathcal{L}^1$$
 and $\left\| \frac{I^n}{nY^{n-}} - y \right\|_{\mathcal{L}^1} = o_{P_0^n}(1)$

with the convention 0/0 = 0, and that

(4.2.20)
$$\|\pi^n(h_1/y)\pi^n(h_2/y)y - (h_1h_2/y)\|_{\mathcal{L}^1} \to 0 \quad \forall h_1, h_2 \in C,$$

and introduce the Hilbert space \mathbb{H} given by (4.2.15). Suppose also that $P_h^n \ll P_0^n$ and (4.2.18) hold for every $h \in C$. Then, it holds that for every $h \in C$

$$\log \frac{dP_h^n | \mathcal{F}_{\tau}^n}{dP_0^n | \mathcal{F}_{\tau}^n} = \Delta^n(h) - \frac{1}{2} \|h\|_{\mathbb{H}}^2 + \epsilon_n(h),$$

where

$$\Delta^{n}(h) = \frac{1}{\sqrt{n}} \sum_{i=1}^{k_{n}} \int_{0}^{\tau} \pi^{n}(h/y)(t, Z_{t}^{n,i}) Y_{t}^{n,i} dB_{t}^{n,0,i}$$

and $\epsilon_n(h) = o_{P_0^n}(1)$. Furthermore, it also holds that $(\Delta^n(h_1), ..., \Delta^n(h_d)) \stackrel{P_0^n}{\Longrightarrow} N(0, \Sigma)$ where $\Sigma_{ij} = \langle h_i, h_j \rangle_{\mathbb{H}}$.

Proof. Notice that $\Delta^n(h)$ is the terminal variable of the continuous local martingale $M^{n,h}$ given by

$$\begin{split} M_t^{n,h} &= \frac{1}{\sqrt{n}} \sum_{i=1}^{k_n} \int_0^t \pi^n (h/y)(s, Z_s^{n,i}) I^n(s, Z_s^{n,i}) Y_s^{n,i} dB_s^{n,0,i} \\ &= \frac{1}{\sqrt{n}} \sum_{i=1}^{k_n} \int_0^t \pi^n (h/y)(s, Z_s^{n,i}) \frac{I^n(s, Z_s^{n,i})}{Y^{n-}(s, Z_s^{n,i})} Y^{n-}(s, Z_s^{n,i}) Y_s^{n,i} dB_s^{n,0,i}. \end{split}$$

4. Integral Estimators

It follows from Lemma 4.2.1 that

$$\langle M^{n,h_1}, M^{n,h_2} \rangle_{\tau} = \int_{[0,\tau] \times E} \pi^n (h_1/y)(t,z) \pi^n (h_2/y)(t,z) \frac{I^n(t,z)}{nY^{n-}(t,z)} dt \lambda(dz).$$

Since

$$\begin{aligned} \left\| \int_{[0,\tau]\times E} \pi^n(h_1/y)(t,z)\pi^n(h_2/y)(t,z) \left\{ \frac{I^n(t,z)}{nY^{n-}(t,z)} - y(t,z) \right\} dt\lambda(dz) \\ &\leq \|\pi^n(h_1/y)\|_{\mathcal{L}^{\infty}} \cdot \|\pi^n(h_2/y)\|_{\mathcal{L}^{\infty}} \cdot \left\| \frac{I^n}{nY^{n-}} - y \right\|_{\mathcal{L}^1}, \end{aligned}$$

the finite-dimensional convergence of $h \rightsquigarrow \Delta^n(h)$ follows from (4.2.19) and (4.2.20) by means of the martingale central limit theorem.

On the other hand, it follows again from Lemma 4.2.1 that

$$\frac{1}{n}\sum_{i=1}^{k_n}\int_0^\tau |\pi^n(h/y)(t,Z_t^{n,i})|^2 Y_t^{n,i}dt = \int_{[0,\tau]\times E} |\pi^n(h/y)(t,z)|^2 \frac{I^n(t,z)}{nY^{n-}(t,z)}dt\lambda(dz),$$

which converges in P^n -probability to $||h||_{\mathbb{H}}^2$ by (4.2.19) and (4.2.20).

STEP II: DIFFERENTIABILITY OF UNKNOWN PARAMETER. The discussion here is similar to that at STEP II of Subsection 4.1.2. Recall the first paragraph there (we use exactly the same notation).

Now, we define the sequence of operators $A^n: H \to \ell^{\infty}([0,\tau] \times \Psi)$ by

$$A^{n}(h)(\psi)_{t} = \int_{[0,t]\times E} \psi(s,z) \left(\alpha(s,z) + n^{-1/2} \pi^{n}(h/y)(s,z) \right) ds \lambda(dz).$$

Under the assumption that

(4.2.21)
$$\sup_{(t,\psi)\in[0,\tau]\times\Psi} \left| \langle 1_{[0,t]}\psi, \pi^n(h/y)y - h \rangle_{\mathbb{H}} \right| \to 0 \qquad \forall h \in H,$$

the sequence A^n is differentiable with rate \sqrt{n} and its derivative $\dot{A}: H \to \ell^{\infty}([0,\tau] \times \Psi)$ is given by

$$\dot{A}(h)(t,\psi) = \langle 1_{[0,t]}\psi,h\rangle_{\mathbb{H}}.$$

We denote by $\pi_{t,\psi} : \ell^{\infty}([0,\tau] \times \Psi) \to \mathbb{R}$ the projection to the (t,ψ) -coordinate, which belongs to $\ell^{\infty}([0,\tau] \times \Psi)^*$ (do not confuse this with the mapping π^n given by (4.2.5)). The above formula shows that $\dot{A}^*\pi_{t,\psi} = \mathbb{1}_{[0,t]}\psi$ for every $(t,\psi) \in [0,\tau] \times \Psi$. By the same reason as STEP II of Subsection 4.1.2, the law of the limit process $(t,\psi) \rightsquigarrow G_t^{\psi}$ appearing in Theorem 4.2.4 coincides with the bound of asymptotic efficiency.

STEP III: ASYMPTOTIC EFFICIENCY. Let us show the regularity of the estimator $(t, \psi) \rightsquigarrow \widehat{A}^n(\psi)_t$ for $(t, \psi) \rightsquigarrow A^n(h)(\psi)_t$.

Proposition 4.2.6 Let C be an arbitrary subset of H. Suppose that a given countable class Ψ satisfies the assumptions (4.2.6), (4.2.7), (4.2.13) and (4.2.14), and Condition 4.2.3 for $P^n = P_0^n$, and introduce the Hilbert space \mathbb{H} given by (4.2.15). Suppose also that all assumptions of Proposition 4.2.5 are satisfied. Then, the random element $(t, \psi) \sim \widehat{A}^n(\psi)_t$ takes values in $\ell^{\infty}([0, \tau] \times \Psi)$, P_h^n -almost surely, for every $h \in C$, and it holds that

$$\sqrt{n}(\widehat{A}^n - A^n(h)) \xrightarrow{P_h^n} G \qquad in \ \ell^{\infty}([0,\tau] \times \Psi) \qquad \forall h \in C,$$

where $(t, \psi) \sim G_t^{\psi}$ is the process appearing in the limit of Theorem 4.2.4.

Proof. In view of the contiguity, a consequence of Proposition 4.2.5, all assumptions concerning convergence in P_0^n -probability hold also in P_h^n -probability for every $h \in C$. Thus the assertion can be proved in the same way as Theorem 4.2.4.

Notice that the assumptions (4.2.20) and (4.2.21) follow from that

$$\|\pi^n(h/y)y - h\|_{\mathbb{H}} \to 0 \qquad \forall h \in H.$$

Summarizing the above discussion, we can get the asymptotic efficiency of the estimator \widehat{A}^n in the same fashion as Corollary 4.1.5, under the assumption that C is a convex cone in \mathbb{H} such that its closed linear span coincides with \mathbb{H} .

4.3 Counting Processes with Non-linear Covariates

Let (E, \mathcal{E}) be a Blackwell space on which a σ -finite measure λ is defined; this is the state space of covariate processes in the following model. In the *n*-th statistical experiment, we consider k_n adapted point processes on \mathbb{R}_+ , namely $N^{n,i}$, $i = 1, ..., k_n$, defined on a filtered measurable space $(\Omega^n, \mathcal{F}^n, \mathbf{F}^n = (\mathcal{F}_t^n)_{t \in \mathbb{R}_+})$; we then denote $T_j^{n,i} = \inf\{t \in \mathbb{R}_+ : N_t^{n,i} = j\}$ for every $j \in \mathbb{N}$ (see page 34 of Jacod and Shiryaev (1987)). Suppose that the predictable compensator of $N^{n,i}$ under the probability measure P^n on $(\Omega^n, \mathcal{F}^n)$ is given by

$$\alpha(t, Z_t^{n,i})Y_t^{n,i}dt.$$

where $\alpha(t, z)$ is a $[0, \infty)$ -valued $\mathfrak{B}(\mathbb{R}_+) \otimes \mathcal{E}$ -measurable function, $t \rightsquigarrow Z_t^{n,i}$ is an E-valued predictable process, and $t \rightsquigarrow Y_t^{n,i}$ is a $\{0,1\}$ -valued predictable process. Let $\tau > 0$ be a constant, and suppose that we can observe the point processes, the processes $t \rightsquigarrow Y_t^{n,i}$, and the covariate processes $t \rightsquigarrow Z_t^{n,i}$ on the random sets $\{t \in [0,\tau] : Y_t^{n,i}(\omega) = 1\}$. The goal of this section is to estimate the unknown parameter $(t,\psi) \rightsquigarrow A(\psi)_t = \int_{[0,t]\times E} \psi(s,z)\alpha(s,z)ds\lambda(dz)$ where ψ 's are appropriate functions.

4. Integral Estimators

We analyze this problem by using the E-valued multivariate point processes

$$\mu^{n}(dt, dz) = \sum_{i=1}^{k_{n}} \sum_{j} \varepsilon_{(T_{j}^{n,i}, Z_{T_{j}^{n,i}}^{n,i})}(dt, dz)$$

which has the predictable compensator

$$\nu^n(dt,dz) = \sum_{i=1}^{k_n} \alpha(t,z) Y_t^{n,i} \varepsilon_{Z_t^{n,i}}(dz) dt.$$

Here, we suppose that $T_j^{n,i} \neq T_{j'}^{n,i'}$ whenever $i \neq i'$; then the basic requirement that $\mu^n(\{t\} \times E) \leq 1$ is indeed satisfied. The approach which we take here is quite close to that in the preceding section.

4.3.1 Preliminaries

In the same way as Section 4.2.1, we consider a partition $E = \bigcup_m E_m^n$ of \mathcal{E} -measurable sets, which is at most countable, such that $\sup_m \lambda(E_m^n) < \infty$ (recall (4.2.2)). Introduce \mathcal{E}^n , $\tilde{\mathcal{P}}^n$, $\tilde{\mathcal{Q}}^n$, Y^{n-} , I^n , $\overline{\alpha}^n$ and $\underline{\alpha}^n$ given there. We then have Lemmas 4.2.1 and 4.2.2. For a given $\tilde{\mathcal{P}}^n$ -measurable function W^n , we denote:

$$A(W^n)_t = \int_{[0,t]\times E} W^n(s,z,x)\alpha(s,z)ds\lambda(dz);$$

$$\overline{A}^n(W^n)_t = \int_{[0,t]\times E} W^n(s,z,x)\overline{\alpha}^n(s,z)ds\lambda(dz).$$

It follows from Lemma 4.2.1 that for any \tilde{Q}^n -measurable function H^n the following two inequalities hold provided the integrals are finite:

$$(4.3.1) \qquad \begin{aligned} \left| H^{n}Y^{n-} * \nu_{t}^{n} - A(H^{n}I^{n})_{t} \right| \\ \leq \int_{[0,t]\times E} \left| H^{n}(s,z) \right| (\overline{\alpha}^{n} - \underline{\alpha}^{n})(s,z)I^{n}(s,z)ds\lambda(dz); \\ (4.3.2) \qquad \qquad \left| H^{n}Y^{n-} * \nu_{t}^{n} \right| \leq \overline{A}^{n}(|H^{n}|I^{n})_{t}. \end{aligned}$$

Let Ψ be an arbitrary set. We will deal with a family $\{K^{n,\psi} : \psi \in \Psi\}$ of $\tilde{\mathcal{Q}}^n$ measurable functions which satisfies that

(4.3.3) the process
$$t \rightsquigarrow (|\overline{K}^n Y^{n-}| \lor |\overline{K}^n Y^{n-}|^2) * \nu_t^n$$
 is locally integrable

where

$$\overline{K}^{n} = \left[\sup_{\psi \in \Psi} |K^{n,\psi}|\right]_{\tilde{\mathcal{Q}}^{n}, M_{\nu^{n}}^{P^{n}}}$$
(cf., Definition 2.2.1). Then, for every $\psi \in \Psi$, the process $t \rightsquigarrow X_t^{n,\psi}$ defined by

(4.3.4)
$$X_t^{n,\psi} = K^{n,\psi}Y^{n-} * (\mu^n - \nu^n)_t$$

is an \mathbb{R} -valued locally square-integrable martingale with bounded variation. As an easy consequence of (4.3.2), we have that a sufficient condition for (4.3.3) is

(4.3.5) the process $t \rightsquigarrow \overline{A}^n (|\overline{K}^n| \lor |\overline{K}^n|^2 \cdot I^n)_t$ is locally integrable

(recall also (4.2.2)). The following lemma gives some tractable conditions to ensure the weak convergence of the $\ell^{\infty}([0,\tau] \times \Psi)$ -valued random element $(t,\psi) \rightsquigarrow X_t^{n,\psi}$ by using Theorem 3.2.4.

Lemma 4.3.1 Let $\tau > 0$ be a constant, and let S be a finite or dense subset of $[0, \tau]$ such that $\tau \in S$. Let $\{K^{n,\psi} : \psi \in \Psi\}$ be a family of $\tilde{\mathcal{Q}}^n$ -measurable functions satisfying (4.3.3), and consider $X^n = (X_t^{n,\psi} | (t,\psi) \in [0,\tau] \times \Psi)$ defined by (4.3.4). When $\operatorname{Card}(\Psi) < \infty$, suppose that the following (i), (ii) and (iii) and satisfied: when $\operatorname{Card}(\Psi) = \infty$, suppose that the following (i), (iii), (iii) and (iv) are satisfied:

 $\begin{array}{l} \text{(i)} \int_{[0,\tau]\times E} |\overline{K}^{n}(t,z)|^{2}Y^{n-}(t,z)(\overline{\alpha}^{n}-\underline{\alpha}^{n})(t,z)dt\lambda(dz) \xrightarrow{P^{n}} 0; \\ \text{(ii)} A\left(|\overline{K}^{n}|^{2}Y^{n-}1_{\{\overline{K}^{n}Y^{n-}>\varepsilon\}}\right)_{\tau} \xrightarrow{P^{n}} 0 \quad \forall \varepsilon > 0; \\ \text{(iii)} A(K^{n,\psi}K^{n,\phi}Y^{n-})_{t} \xrightarrow{P^{n}} C_{t}^{(\psi,\phi)} \quad \forall t \in S \ \forall (\psi,\phi) \in \Psi^{2}, \ where \ \{C_{t}^{(\psi,\phi)}\} \ is \ a \ family \ of \ constants \ satisfying \ (3.2.3); \end{array}$

(iv) there exists a DFP $\Pi = \{\Psi(\varepsilon; k) : 1 \le k \le N_{\Pi}(\varepsilon)\}_{\varepsilon \in (0,1]}$ of Ψ such that

$$\sup_{\varepsilon \in (0,1] \cap \mathbb{Q}} \max_{1 \le k \le N_{\Pi}(\varepsilon)} \frac{\overline{A}^{n}(|K^{n}(\Psi(\varepsilon;k))|^{2}Y^{n-})_{\tau}}{\varepsilon^{2}} = O_{P^{n}}(1) \quad and \quad \int_{0}^{1} \sqrt{\log N_{\Pi}(\varepsilon)} d\varepsilon < \infty,$$

where

$$K^{n}(\Psi') = \left[\sup_{\psi,\phi\in\Psi'} |K^{n,\psi} - K^{n,\phi}|\right]_{\tilde{\mathcal{Q}}^{n},\overline{A}^{n}(dt,dz)} \quad \forall \Psi' \subset \Psi.$$

Then, it holds that $X^n \stackrel{P^n}{\Longrightarrow} X$ in $\ell^{\infty}(S \times \Psi)$, where $(t, \psi) \sim X_t^{\psi}$ is a zero-mean Gaussian process such that $EX_t^{\psi}X_s^{\phi} = C_{t\wedge s}^{(\psi,\phi)}$. Furthermore, the formula (3.2.4) defines a semimetric ρ_2 on $S \times \Psi$ such that $S \times \Psi$ is totally bounded with respect to ρ_2 , and that almost all paths of X are uniformly ρ_2 -continuous. When S is dense in $[0, \tau]$, the space $S \times \Psi$ appearing in the conclusion can be replaced by $[0, \tau] \times \Psi$.

Proof. By using the inequality (4.3.1) and the assumption (i), we get the conditions **[L2]** and **[C2]** of Theorem 3.2.4 from (ii) and (iii), respectively. The condition **[PE]** is immediate from the inequality (4.3.2) and (iv).

4.3.2 Asymptotic Normality

Let a constant $\tau > 0$ be given, and let us consider the estimation problem of $(t, \psi) \rightsquigarrow A(\psi)_t$. Recall the notations \mathcal{L}^p and $\mathcal{L}^{p,n}$ given by (4.2.3) and (4.2.4), and the definition of π^n by (4.2.5), respectively. In the following, we will always assume:

(4.3.6)
$$\alpha_N^{\star} = \bigvee_{n \ge N} \overline{\alpha}^n \in \mathcal{L}^1 \text{ for some } N \in \mathbb{N}.$$

Thus $\alpha^{\star}_{N}(t,z)dt\lambda(dz)$ defines a finite measure on $[0,\tau] \times E$. We then denote

$$\mathcal{L}^p(\alpha^\star_N) = \mathcal{L}^p([0,\tau] \times E, \mathfrak{B}[\mathfrak{o},\tau] \otimes \mathcal{E}, \alpha^\star_{\mathfrak{N}}(\mathfrak{t},\mathfrak{z})\mathfrak{d}\mathfrak{t}\lambda(\mathfrak{d}\mathfrak{z})) \qquad \forall \mathfrak{p} \in [\mathfrak{1},\infty].$$

Furthermore, we denote by $\|\cdot\|_{\mathcal{L}^p(\alpha_N^\star)}$ the L^p -seminorm on $\mathcal{L}^p(\alpha_N^\star)$. These should not be confused with the notations \mathcal{L}^p and $\|\cdot\|_{\mathcal{L}^p}$ given by (4.2.3).

Let Ψ be a subset of $\mathcal{L}^{2p}(\alpha_N^*) \cap \mathcal{L}^1$ with an envelope function $\varphi \in \mathcal{L}^{2p}(\alpha_N^*) \cap \mathcal{L}^1$, where $p \in [1, \infty]$ should be specified in connection with another assumption in Condition 4.3.2. We propose the estimator $(t, \psi) \rightsquigarrow \widehat{\mathcal{A}}^n(\psi)_t$ defined by

$$\widehat{A}^{n}(\psi)_{t}(\omega) = (\pi^{n}(\psi)Y^{n-}) * \mu^{n}_{t}(\omega) \qquad \forall (t,\psi) \in [0,\tau] \times \Psi,$$

where π^n is the mapping defined by (4.2.5). We shall first consider the residual $\sqrt{n}(\widehat{A}^n - \widetilde{A}^n)$, where

$$\widetilde{A}^{n}(\psi)_{t}(\omega) = \int_{[0,t]\times E} \pi^{n}(\psi)(s,z)I(\omega,s,z)\alpha(s,z)ds\lambda(dz) \qquad \forall (t,\psi)\in[0,\tau]\times\Psi,$$

Condition 4.3.2 For some $p, q \in [1, \infty]$ such that (1/p) + (1/q) = 1, and a $\mathfrak{B}[\mathfrak{o}, \tau] \otimes \mathcal{E}$ measurable function y = y(t, z) on $[0, \tau] \times E$, which is bounded away from zero, it holds
that:

(4.3.7)
$$\varphi \in \mathcal{L}^{2p}(\alpha_N^{\star}) \cap \mathcal{L}^1 \quad and \quad \int_0^1 \sqrt{\log N_{[]}(\Psi, \|\cdot\|_{\mathcal{L}^{2p}(\alpha_N^{\star})}; \varepsilon)} d\varepsilon < \infty,$$

where the brackets should be constructed in $\mathcal{L}^{2p}(\alpha_N^{\star}) \cap \mathcal{L}^1;$

(4.3.8)
$$||nY^{n-}||_{\mathcal{L}^{q}(\alpha_{N}^{\star})} = O_{P^{n}}(1);$$

(4.3.9)
$$\left\| |\pi^{n}(\varphi)|^{2} \cdot |nY^{n-} - (1/y)| \right\|_{\mathcal{L}^{1}(\alpha_{N}^{\star})} = o_{P^{n}}(1);$$

(4.3.10)
$$\|\{\pi^n(\psi)\pi^n(\phi) - \psi\phi\} \cdot (\alpha/y)\|_{\mathcal{L}^1} \to 0 \qquad \forall \psi, \phi \in \Psi\}$$

(4.3.11)
$$\|\varphi \cdot (\overline{\alpha}^n - \underline{\alpha}^n)\|_{\mathcal{L}^1} = o(n^{-1/2}).$$

Theorem 4.3.3 Assume (4.3.6). Suppose that a given class Ψ satisfies Conditions 4.3.2. Then, it holds that $\sqrt{n}(\widehat{A}^n - \widetilde{A}^n) \stackrel{P^n}{\Longrightarrow} G$ in $\ell^{\infty}([0, \tau] \times \Psi)$, where $(t, \psi) \rightsquigarrow G_t^{\psi}$ is a zero-mean Gaussian process such that

$$EG_t^{\psi}G_s^{\phi} = \int_{[0,t\wedge s]\times E} \psi(u,z)\phi(u,z)\frac{\alpha(u,z)}{y(u,z)}dt\lambda(dz) \qquad \forall (t,\psi), (s,\phi) \in [0,\tau] \times \Psi$$

and that almost all paths are uniformly ρ_2 -continuous on Ψ , where

$$\rho_2((t,\psi),(s,\phi)) = \sqrt{E|G_t^{\psi} - G_s^{\phi}|^2} \qquad \forall (t,\psi), (s,\phi) \in [0,\tau] \times \Psi$$

Proof. It follows from (4.3.1) and (4.3.11) that

$$\sup_{(t,\psi)\in[0,\tau]\times\Psi} \left| (\pi^n(\psi)Y^{n-}) * \nu_t^n - A(\pi^n(\psi)I^n)_t \right| = o_{P^{n*}}(n^{-1/2})$$

Thus it suffices to derive the weak convergence of the sequence of processes $(t, \psi) \rightsquigarrow X_t^{n,\psi}$ defined by

$$X_t^{n,\psi} = \sqrt{n} (\pi^n(\psi) Y^{n-}) * (\mu^n - \nu^n)_t.$$

We will apply Lemma 4.3.1 with $K^{n,\psi} = \sqrt{n\pi^n(\psi)}$. The condition (4.3.5) for $n \ge N$ follows (4.3.6) and the fact $\varphi \in \mathcal{L}^2(\alpha_N^{\star})$. To show the condition (i), observe that

$$\begin{aligned} \left| n \int_{[0,\tau]\times E} |\pi^{n}(\varphi)(t,z)|^{2} Y^{n-}(t,z)(\overline{\alpha}^{n}-\underline{\alpha}^{n})(t,z)dt\lambda(dz) \right| \\ &\leq \left\| |\pi^{n}(\varphi)|^{2} \cdot |nY^{n-}-(1/y)| \cdot (\overline{\alpha}^{n}-\underline{\alpha}^{n}) \right\|_{\mathcal{L}^{1}} + \left\| |\pi^{n}(\varphi)|^{2} \cdot (1/y) \cdot (\overline{\alpha}^{n}-\underline{\alpha}^{n}) \right\|_{\mathcal{L}^{1}} \\ &\leq \left\| |\pi^{n}(\varphi)|^{2} \cdot |nY^{n-}-(1/y)| \right\|_{\mathcal{L}^{1}(\alpha_{N}^{\star})} + \|1/y\|_{\mathcal{L}^{\infty}} \cdot \left\| \varphi^{2} \cdot (\overline{\alpha}^{n}-\underline{\alpha}^{n}) \right\|_{\mathcal{L}^{1}}. \end{aligned}$$

The first term on the right hand side converges in P^n -probability to zero by (4.3.9). On the other hand, since it follows from (4.3.11) that $(\overline{\alpha}^n - \underline{\alpha}^n)(t, z) \to 0$ for $\varphi(t, z)dt\lambda(dz)$ almost all (t, z), the dominated convergence theorem yields that the second term also vanishes. Thus the condition (i) has been proved. Next, it follows from the same computation as (4.1.5) that

$$nA\left(|\pi^{n}(\varphi)|^{2}Y^{n-1}_{\{\sqrt{n}\pi^{n}(\varphi)Y^{n-}>\varepsilon\}}\right)$$

$$\leq \left\||\pi^{n}(\varphi)|^{2} \cdot nY^{n-} \cdot 1_{\{\sqrt{n}\pi^{n}(\varphi)Y^{n-}>\varepsilon\}}\right\|_{\mathcal{L}^{1}(\alpha_{N}^{\star})}$$

$$\leq \frac{1}{\sqrt{n}\varepsilon} \cdot \|\pi^{n}(\varphi)\|_{\mathcal{L}^{2p}(\alpha_{N}^{\star})}^{2} \cdot \|nY^{n-}\|_{\mathcal{L}^{q}(\alpha_{N}^{\star})},$$

which converges in P^n -probability to zero by (4.3.7) and (4.3.8). Thus the condition (ii) has been proved. The condition (iii) can be shown by using (4.3.9) and (4.3.10). This finishes the proof of the case where the set Ψ is finite.

On the other hand, it follows from Lemma 4.2.2 that, if $[l^k, u^k]$'s are $(\|\cdot\|_{\mathcal{L}^{2p}(\alpha_N^\star)}, \varepsilon)$ brackets in $\mathcal{L}^{2p}(\alpha_N^\star) \cap \mathcal{L}^1$ which cover the class Ψ , then $[\pi^n(l^k), \pi^n(u^k)]$'s are $(\|\cdot\|_{\mathcal{L}^{2p}(\alpha_N^\star)}, \varepsilon)$ brackets in $\mathcal{L}^{2p}(\alpha_N^\star) \cap \mathcal{L}^1$ which cover the class $\pi^n \Psi$. Hence, by using the Hölder inequality, it is shown that the DFP of Ψ induced from the minimum brackets satisfies the requirements of (iv) of Lemma 4.3.1. In order to derive the weak convergence of $\sqrt{n}(\widehat{A}^n - A)$ rather than $\sqrt{n}(\widehat{A}^n - \widetilde{A}^n)$, we have to show that $\sqrt{n}(\widetilde{A}^n - A) \stackrel{P^n}{\Longrightarrow} 0$ in $\ell^{\infty}([0, \tau] \times \Psi)$. For this purpose, it suffices to check the following:

(4.3.12)
$$\sup_{\psi \in \Psi} \| (\pi^n(\psi) - \psi) \cdot \alpha \|_{\mathcal{L}^1} = o(n^{-1/2});$$

(4.3.13)
$$\|\varphi \cdot (1 - I^n) \cdot \alpha\|_{\mathcal{L}^1} = o_{P^n}(n^{-1/2}).$$

See the discussion after Theorem 4.2.4 for getting simple sufficient conditions for all assumptions appearing above in the case where $E = [0, 1]^d$.

4.3.3 Asymptotic Efficiency

In order to discuss the asymptotic efficiency, we set:

$$(4.3.14) \qquad \mathbb{H} = L^2([0,\tau] \times E, \frac{\alpha(t,z)}{y(t,z)} dt\lambda(dz)); H = L^{\infty}([0,\tau] \times E, \mathfrak{B}[\mathfrak{o},\tau] \otimes \mathcal{E}, \mathfrak{d}t\lambda(\mathfrak{d}\mathfrak{z})); H^n = L^{\infty}([0,\tau] \times E, \mathfrak{B}[\mathfrak{o},\tau] \otimes \mathcal{E}^n, \mathfrak{d}t\lambda(\mathfrak{d}\mathfrak{z})).$$

Here, the function y is the one which appears in Condition 4.3.2. We equip \mathbb{H} with the usual L^2 -inner product $\langle \cdot, \cdot \rangle_{\mathbb{H}}$. Since we always assume (4.3.6), and since 1/y is assumed to be bounded, H is a linear subspace of the Hilbert space \mathbb{H} . Let $\mathbf{P}^n = \{P_h^n : h \in H\}$ be a family of probability measures on $(\Omega^n, \mathcal{F}^n)$ indexed by H. Define the mapping $\pi^n : H \to H^n$ by (4.2.5). Hereafter, we denote by $\pi^n(h)$ any function of the equivalent elements in H^n .

Suppose that the predictable compensator of $N^{n,i}$ under the probability measure P_h^n is given by

(4.3.15)
$$\alpha_h^n(t, Z_t^{n,i}) Y_t^{n,i} dt,$$

where $\alpha_h^n = \alpha_h^n(t, z)$ is defined by

(4.3.16)
$$\alpha_h^n = \left\{ 1 + n^{-1/2} \pi^n (h/2y) \right\}^2 \alpha$$

To see that the compensator (4.3.15) is well-defined, recall the discussion after (4.2.16) and (4.2.17). In particular, it holds that $\alpha_0^n(t, Z_t^{n,i}(\omega)) = \alpha(t, Z_t^{n,i}(\omega))$ for almost all t. The predictable compensator $\nu^{n,h}$ of μ^n under the probability measure P_h^n is then given by

$$\nu^{n,h}(dt,dz) = \alpha^n_h(t,z)\overline{\nu}^n(dt,dz)$$

where

$$\overline{\nu}^n(dt,dz) = \sum_{i=1}^{k_n} Y_t^{n,i} \varepsilon_{Z_t^{n,i}}(dz) dt.$$

STEP I: LOCAL ASYMPTOTIC NORMALITY. Assume $P_h^n \ll P_0^n$. It is well-known that, under some conditions, a version of the log-likelihood ratio is given by

(4.3.17)
$$\log \frac{dP_h^n | \mathcal{F}_{\tau}^n}{dP_0^n | \mathcal{F}_{\tau}^n} = 2 \left(\log \left| 1 + n^{-1/2} \pi^n (h/2y) \right| \right) * \mu_{\tau}^n - \left(\left| 1 + n^{-1/2} \pi^n (h/2y) \right|^2 - 1 \right) * \nu_{\tau}^{n,0}$$

(see, e.g., Theorem III.5.13 of Jacod and Shiryaev (1987)). This representation does not depend on the choice of a version of $\pi^n(h/2y) \in H^n$, because it holds that $\nu^{n,0}(\omega; B) = 0$ identically and that $P_0^n(\mu^n(B) = 0) = 1$ for any $B \in \mathfrak{B}[\mathfrak{o}, \tau] \otimes \mathcal{E}^n$ such that $\text{Leb} \otimes \lambda(B) = 0$.

Proposition 4.3.4 Let C be an arbitrary subset of H. Assume (4.3.6). Suppose that the function y appearing in Condition 4.3.2 satisfies the following:

$$y \in \mathcal{L}^{1}(\alpha_{N}^{\star}) \quad and \quad \left\| \frac{I^{n}}{nY^{n-}} - y \right\|_{\mathcal{L}^{1}(\alpha_{N}^{\star})} = o_{P_{0}^{n}}(1);$$
$$\left\| \left\{ \pi^{n}(h_{1}/2y)\pi^{n}(h_{2}/2y) - (h_{1}h_{2})/(4y^{2}) \right\} \cdot y \cdot \alpha \right\|_{\mathcal{L}^{1}} \to 0 \qquad \forall h_{1}, h_{2} \in C;$$
$$\| y \cdot (\overline{\alpha}^{n} - \underline{\alpha}^{n}) \|_{\mathcal{L}^{1}} \to 0.$$

Introduce the Hilbert space \mathbb{H} given by (4.3.14). Suppose also that $P_h^n \ll P_0^n$ and (4.3.17) hold for every $h \in C$. Then, it holds that for every $h \in C$

$$\log \frac{dP_h^n | \mathcal{F}_\tau^n}{dP_0^n | \mathcal{F}_\tau^n} = \Delta^n(h) - \frac{1}{2} \|h\|_{\mathbb{H}}^2 + \epsilon_n(h),$$

where

$$\Delta^{n}(h) = n^{-1/2} (2\pi^{n}(h/2y)) * (\mu^{n} - \nu^{n,0})_{\tau}$$

and $\epsilon_n(h) = o_{P_0^n}(1)$. Furthermore, it also holds that $(\Delta^n(h_1), ..., \Delta^n(h_d)) \xrightarrow{P_0^n} N(0, \Sigma)$ where $\Sigma_{ij} = \langle h_i, h_j \rangle_{\mathbb{H}}$.

Proof. Since $|\log(1+x) - x + \frac{x^2}{2}| \le \frac{8}{3}x^3$ for all $x \in [-\frac{1}{2}, \frac{1}{2}]$, we have that for any $\varepsilon \in (0, 1]$,

$$\left|\log\frac{dP_h^n|\mathcal{F}_\tau^n}{dP_0^n|\mathcal{F}_\tau^n} - \Delta^n(h) + D^n(h) + \tilde{D}^n(h)\right| \le \frac{16}{3}\varepsilon D^n(h)$$

whenever $\sup_{t,z} |n^{-1/2} \pi^n (h/2y)(t,z)| \le \varepsilon$, where:

$$D^{n}(h) = |n^{-1/2}\pi^{n}(h/2y)|^{2} * \mu_{\tau}^{n};$$

$$\tilde{D}^{n}(h) = |n^{-1/2}\pi^{n}(h/2y)|^{2} * \nu_{\tau}^{n,0}.$$

Since

$$\begin{split} \left\| \left(\left\| \pi^{n}(h/2y) \right\|^{2} \frac{I^{n}}{nY^{n-}} Y^{n-} \right) * \nu_{\tau}^{n,0} - A \left(\left\| \pi^{n}(h/2y) \right\|^{2} \frac{I^{n}}{nY^{n-}} \right)_{\tau} \right\| \\ & \leq \left\| \left\| \pi^{n}(h/2y) \right\|^{2} \cdot \frac{I^{n}}{nY^{n-}} \cdot \left(\overline{\alpha}^{n} - \underline{\alpha}^{n}\right) \right\|_{\mathcal{L}^{1}} \\ & \leq \left\| \left\| \pi^{n}(h/2y) \right\|^{2} \cdot \left\| \frac{I^{n}}{nY^{n-}} - y \right\| \cdot \left(\overline{\alpha}^{n} - \underline{\alpha}^{n}\right) \right\|_{\mathcal{L}^{1}} + \left\| \left\| \pi^{n}(h/2y) \right\|^{2} \cdot y \cdot \left(\overline{\alpha}^{n} - \underline{\alpha}^{n}\right) \right\|_{\mathcal{L}^{1}} \\ & \leq \left\| \pi^{n}(h/2y) \right\|_{\mathcal{L}^{\infty}}^{2} \left\{ \left\| \frac{I^{n}}{nY^{n-}} - y \right\|_{\mathcal{L}^{1}(\alpha_{N}^{\star})} + \left\| y \cdot \left(\overline{\alpha}^{n} - \underline{\alpha}^{n}\right) \right\|_{\mathcal{L}^{1}} \right\}, \end{split}$$

and since

$$A\left(|\pi^{n}(h/2y)|^{2} \left| \frac{I^{n}}{nY^{n-}} - y \right| \right)_{\tau} \leq \|\pi^{n}(h/2y)\|_{\mathcal{L}^{\infty}}^{2} \cdot \left\| \frac{I^{n}}{nY^{n-}} - y \right\|_{\mathcal{L}^{1}(\alpha_{N}^{\star})},$$

it holds that $\tilde{D}^n(h) \xrightarrow{P_0^n} \frac{1}{4} \|h\|_{\mathbb{H}}^2$. Also, using Lenglart's inequality, we have $D^n(h) - \tilde{D}^n(h) \xrightarrow{P_0^n} 0$. These facts imply the first assertion. Applying Lemma 4.3.1 to the family of \tilde{Q}^n -measurable functions $\{K^{n,h} : h \in C\}$ given by

$$K^{n,h} = n^{1/2} \cdot 2\pi^n (h/2y) \cdot \frac{I^n}{nY^{n-1}}$$

we can show the finite-dimensional convergence of $h \rightsquigarrow \Delta^n(h)$.

Discussion about STEP II and STEP III is similar to the preceding sections, hence is omitted.

4.A Notes

A part of the results in Section 4.1 was presented by Nishiyama (1997). As mentioned in the main text, a progress from the preceding work is the introduction of L^{2p} -bracketing entropies rather than the L^2 -bracketing one. This would be important also for other applications in non-i.i.d. settings; see, e.g., Chapter 6. The problems considered in Sections 4.2 and 4.3 were posed by McKeague and Utikal (1990) who treated the case where the state space of covariates is [0, 1]. Our way of constructing the estimator is motivated by their work. A moment assumption of theirs has been weakened.

Chapter 5

M-Estimators: General Criterion and Euclidean Parameters

5.1 General Criterion

The common structure of the models treated in this and next chapters is as follows.

Formulation 5.1.1 For every $n \in \mathbb{N}$, let $(\Omega^n, \mathcal{F}^n)$ be a measurable space and $\mathbf{P}^n = \{P_u^n : u \in U^n\}$ a family of probability measures on $(\Omega^n, \mathcal{F}^n)$ indexed by an arbitrary set U^n . For every $n \in \mathbb{N}$ and $u \in U^n$, let the following be given:

(i) a space Θ^n , a random point $\theta^n_u \in \Theta^n$, and a $[0,\infty)$ -valued stochastic process $\theta \sim d^n_u(\theta, \theta^n_u)$ with parameter in Θ^n ;

(ii) some stochastic processes $\theta \rightsquigarrow \Gamma^n_u(\theta)$ and $\theta \rightsquigarrow \gamma^n_u(\theta)$, with parameters in Θ^n .

We then denote $R_u^n(\delta) = \{\theta \in \Theta^n : (\delta/2) < d_u^n(\theta, \theta_u^n) \le \delta\}$ for every $\delta \in (0, \infty)$.

The $[0, \infty)$ -valued stochastic process $\theta \sim d_u^n(\theta, \theta_u^n)$ above is usually given by a (random) semimetric $d_u^n(\theta, \vartheta)$ on Θ^n and a (random) point θ_u^n which should be regarded as an (approximate) true point of unknown parameter. We refer the processes $\theta \sim \Gamma_u^n(\theta)$ and $\theta \sim \gamma_u^n(\theta)$ as the "criterion process" and the "contrast process", respectively; the latter is sometimes taken to be deterministic, and in that case it is referred as the "contrast function". The following result is a version of Theorem 3.4.1 of van der Vaart and Wellner (1996), into which contributions by some other authors in this area are condensed (see Notes at the end of this chapter).

Theorem 5.1.2 Consider Formulation 5.1.1 above. Suppose that the following M-CRITERION is satisfied for some $\delta_0 \in (0, \infty]$, p > 0, $a \in (0, p)$, not depending on nand u, some functions $\phi_u^n : (0, \delta_0) \to (0, \infty)$ such that $\delta \rightsquigarrow \delta^{-a} \phi_u^n(\delta)$ is decreasing, and some positive constants $r_{n,u}$ such that $r_{n,u}^{-1} \in (0, \delta_0)$ and that $\phi_u^n(r_{n,u}^{-1}) \leq r_{n,u}^{-p}$. *M*-CRITERION. For every $\varepsilon > 0$ there exists some $c_{\varepsilon}, C_{\varepsilon}, K_{\varepsilon} > 0$ and $n_{\varepsilon} \in \mathbb{N}$ such that: for every $n \ge n_{\varepsilon}$ and $u \in U^n$ there exists a set $B^n_u(\varepsilon) \subset \Omega^n$ such that

$$\gamma_u^n(\theta) - \gamma_u^n(\theta_u^n) \le -c_{\varepsilon}\delta^p \qquad \forall \theta \in R_u^n(\delta) \qquad on \ the \ set \ B_u^n(\varepsilon)$$

and

$$E_u^{n*} \sup_{\theta \in R_u^n(\delta)} |(\Gamma_u^n - \gamma_u^n)(\theta) - (\Gamma_u^n - \gamma_u^n)(\theta_u^n)| \mathbf{1}_{B_u^n(\varepsilon)} \le C_{\varepsilon} \phi_u^n(\delta)$$

whenever $\delta \in [K_{\varepsilon}r_{n,u}^{-1}, \delta_0)$, and that $P_u^{n*}(\Omega^n \setminus B_u^n(\varepsilon)) \leq \varepsilon$. Then, for any mappings $\widehat{\theta}_u^n : \Omega^n \to \Theta^n$ such that

(5.1.1)
$$\lim_{L \to \infty} \limsup_{n \to \infty} \sup_{u \in U^n} P_u^{n*} \left(\Gamma_u^n(\widehat{\theta}_u^n) < \Gamma_u^n(\theta_u^n) - Lr_{n,u}^{-p} \right) = 0$$

and that

(5.1.2)
$$\lim_{n \to \infty} \sup_{u \in U^n} P_u^{n*} \left(d_u^n(\widehat{\theta}_u^n, \theta_u^n) > \delta_0/2 \right) = 0,$$

it holds that

$$\lim_{L \to \infty} \limsup_{n \to \infty} \sup_{u \in U^n} P_u^{n*} \left(r_{n,u} d_u^n(\widehat{\theta}_u^n, \theta_u^n) > L \right) = 0.$$

When M-CRITERION is satisfied for $\delta_0 = \infty$, the assumption (5.1.2) is unnecessary.

Remark. In the sequel, we refer the first and the second displayed inequalities of M-CRITERION as the "FIRST INEQUALITY" and the "SECOND INEQUALITY", respectively.

Keeping a two-term Taylor expansion of the function $\theta \rightsquigarrow \gamma_u^n(\theta)$ in their mind, van der Vaart and Wellner (1996) presented some results of this fashion for the case of p = 2as their Theorems 3.2.5 and 3.4.1. The adaptation to the case of general p will be useful in Sections 5.2, 5.3 and 7.1. The truncation introduced in the SECOND INEQUALITY fits in our maximal inequalities based on the quadratic modulus. The last difference is the uniformity in the underlying probability measures (this is clear if the conclusion is given in the form of a probability inequality by using universal constants; see, e.g., Birgé and Massart (1993) and van de Geer (1995b)). Although the change of the proof is minor, we state the whole proof following exactly the same line as that of van der Vaart and Wellner (1996).

Proof of Theorem 5.1.2. Fix any $\varepsilon > 0$, and choose some constants $c_{\varepsilon}, C_{\varepsilon}, K_{\varepsilon} > 0$ and $n_{\varepsilon} \in \mathbb{N}$ according to *M*-CRITERION. In the following we will consider $n \ge n_{\varepsilon}$ only.

Now, fix any L > 0 for a while, and choose any $J \in \mathbb{N}$ such that $c_{\varepsilon} - 2^{-p(J-1)}L > 0$ and that $2^J \ge K_{\varepsilon}$. Put $J_u^n = \max\{j \in \mathbb{N} : 2^j r_{n,u}^{-1} < \delta_0\}$ (we have implicitly assumed $\delta_0 < \infty$,

but the case of $\delta_0 = \infty$ is easier; read the following argument replacing " $J \leq \forall j \leq J_u^n$ " by " $\forall j \geq J$ "). We denote:

$$\begin{split} S_u^n(j) &= \left\{ \omega \in \Omega^n : \widehat{\theta}_u^n(\omega) \in R_u^n(2^j r_{n,u}^{-1}) \right\} & J \leq \forall j \leq J_u^n; \\ \Omega_u^n(\varepsilon,L) &= \left\{ \omega \in \Omega^n : \Gamma_u^n(\widehat{\theta}_u^n(\omega))(\omega) - \Gamma_u^n(\theta_u^n)(\omega) \geq -Lr_{n,u}^{-p} \right\} \cap B_u^n(\varepsilon). \end{split}$$

Then it holds on the set $S^n_u(j) \cap \Omega^n_u(\varepsilon, L)$ that

$$\sup_{\theta \in R_u^n(2^j r_{n,u}^{-1})} \Gamma_u^n(\theta) - \Gamma_u^n(\theta_u^n) \ge -Lr_{n,u}^{-p}$$

and that

$$\inf_{\theta \in R^n_u(2^j r^{-1}_{n,u})} \gamma^n_u(\theta) - \gamma^n_u(\theta^n_u) \le -c_{\varepsilon} 2^{p(j-1)} r^{-p}_{n,u};$$

thus we have

$$(5.1.3) \sup_{\theta \in R_u^n(2^j r_{n,u}^{-1})} \{ (\Gamma_u^n - \gamma_u^n)(\theta) - (\Gamma_u^n - \gamma_u^n)(\theta_u^n) \} \ge (c_{\varepsilon} 2^{p(j-1)} - L) r_{n,u}^{-p} \\ \ge (c_{\varepsilon} - 2^{-p(J-1)} L) 2^{p(j-1)} r_{n,u}^{-p}.$$

Since $\{d_u^n(\widehat{\theta}_u^n, \theta_u^n) \leq \delta_0/2\} \subset \{d_u^n(\widehat{\theta}_u^n, \theta_u^n) \leq 2^{J_u^n} r_{n,u}^{-1}\}$ it holds that

$$P_u^{n*} \left(2^{J-1} r_{n,u}^{-1} < d_u^n(\widehat{\theta}_u^n, \theta_u^n) \le \delta_0/2, \ \Omega_u^n(\varepsilon, L) \right) \\ \le \sum_{J \le j \le J_u^n} P_u^{n*} \left(S_u^n(j) \cap \Omega_u^n(\varepsilon, L) \right),$$

where the summation with respect to j can be read as zero when $J > J_u^n$. If $J \leq J_u^n$, recalling (5.1.3), we obtain from the Markov inequality and the SECOND INEQUALITY of M-CRITERION that

$$\begin{split} \sum_{J \leq j \leq J_u^n} P_u^{n*} \left(S_u^n(j) \cap \Omega_u^n(\varepsilon, L) \right) &\leq \frac{C_{\varepsilon}}{c_{\varepsilon} - 2^{-p(J-1)}L} \sum_{J \leq j \leq J_u^n} \frac{\phi_u^n(2^j r_{n,u}^{-1})}{2^{p(j-1)} r_{n,u}^{-p}} \\ &\leq \frac{C_{\varepsilon}}{c_{\varepsilon} - 2^{-p(J-1)}L} \sum_{J \leq j \leq J_u^n} \frac{2^{aj} \phi_u^n(r_{n,u}^{-1})}{2^{p(j-1)} r_{n,u}^{-p}} \\ &\leq \frac{2^p C_{\varepsilon}}{c_{\varepsilon} - 2^{-p(J-1)}L} \sum_{j \geq J} 2^{(a-p)j}. \end{split}$$

Here we have also used the fact that $\phi_u^n(c\delta) \leq c^a \phi_u^n(\delta)$ for every c > 1.

Consequently we have

$$P_u^{n*}\left(r_{n,u}d_u^n(\widehat{\theta}_u^n, \theta_u^n) > 2^{J-1}\right) \leq P_u^{n*}(\Omega^n \setminus \Omega_u^n(\varepsilon, L)) + P_u^{n*}\left(d_u^n(\widehat{\theta}_u^n, \theta_u^n) > \delta_0/2\right) + \frac{2^p C_{\varepsilon}}{1 - 2^{(a-p)}} \cdot \frac{2^{(a-p)J}}{c_{\varepsilon} - 2^{-p(J-1)}L}.$$

This inequality holds also in the case of $\delta_0 = \infty$ by regarding the second term on the right hand side as zero. Notice that the last term on the right hand side does not depend on $n \in \mathbb{N}$ and $u \in U^n$ and converges to zero as $J \to \infty$ since a < p. To get the assertion, first choose large L > 0 according to the assumption (5.1.1), and next let $J \to \infty$. \Box

In the remaining sections of this chapter, we are concerned with some problems of estimating finite-dimensional parameters. Here, we sketch a procedure for deriving the asymptotic distribution of M-estimators based on a continuous mapping theorem for argmax functionals (Theorem 3.2.2 of van der Vaart and Wellner (1996)). In any case, we shall consider some rescaled criterion processes $h \rightsquigarrow \mathbb{M}^n(h)$ of the form

$$\mathbb{M}^n(h) = a_n \{ \Gamma^n(\theta_0 + r_n^{-1}h) - \Gamma^n(\theta_0) \},\$$

where r_n and a_n are some appropriate constants. Thus the first problem should be to find the "rate of convergence" r_n , and Theorem 5.1.2 is useful at this step. The constant a_n should be determined in connection with r_n . Next, according to Theorem 3.2.2 of van der Vaart and Wellner (1996), we shall show the following.

- (i) The uniform tightness of the local sequence $\hat{h}^n = r_n(\hat{\theta}^n \theta_0)$.
- (ii) The weak convergence of the process $h \rightsquigarrow \mathbb{M}^n(h)$ to a continuous process $h \rightsquigarrow$
- $\mathbb{M}(h)$ in $\ell^{\infty}(K)$, for every compact subset K of the space of local parameters.
- (iii) The existence of a unique maximum point \hat{h} of the path $h \rightsquigarrow \mathbb{M}(h)$.

Any Borel random variable on a Polish space is tight, hence so is \hat{h} . Thus a result of the form " $r_n(\hat{\theta}^n - \theta_0) \stackrel{P}{\Longrightarrow} \hat{h}$ " follows from the argmax continuous mapping theorem.

The reason why we are content with the case of finite-dimensional parameters in this approach is that the uniform tightness of the local sequence \hat{h}^n (the step (i) above) is equivalent to " $r_n |\hat{\theta}^n - \theta_0| = O_P(1)$ ", which is actually the consequence of Theorem 5.1.2. This is not always true when the parameter space is general, but Theorem 5.1.2 is still useful at least for deriving the rate of convergence as we see in Chapter 6. We will make use of the results given in Chapter 3 at step (ii). For simplicity, we will not discuss the uniformity in the underlying probability measures in Sections 5.2 and 5.3.

5.2 Gaussian White Noise Model

5.2.1 Criterion for Rate of Convergence

For every $n \in \mathbb{N}$, let $X^n = (X_t^n)_{t \in [0,1]}$ be a continuous stochastic process given by

(5.2.1)
$$dX_t^n = f(t)dt + n^{-1/2}dB_t, \qquad X_0^n = x_0 \in \mathbb{R},$$

where $f \in \mathcal{L}^2[0,1]$, and $B = (B_t)_{t \in [0,1]}$ is a standard Brownian motion on a stochastic basis $\mathbf{B} = (\Omega, \mathcal{F}, \mathbf{F} = (\mathcal{F}_t)_{t \in [0,1]}, P)$. Let (Θ, d) be a separable metric space; we will take it to be a Euclidean space later. Let $w = \{w_\theta : \theta \in \Theta\}$ be a class of elements of $\mathcal{L}^2[0,1]$. Equip Θ with the semimetric ρ_w given by

(5.2.2)
$$\rho_w(\theta,\vartheta) = \|w_\theta - w_\vartheta\|_{\mathcal{L}^2[0,1]} \quad \forall \theta,\vartheta \in \Theta.$$

We consider the criterion process $\theta \rightsquigarrow \Gamma^n(\theta)$ defined by

(5.2.3)
$$\Gamma^{n}(\theta) = \int_{0}^{1} w_{\theta}(t) dX_{t}^{n} \qquad \forall \theta \in \Theta,$$

and the contrast function $\theta \rightsquigarrow \gamma(\theta)$ defined by

(5.2.4)
$$\gamma(\theta) = \int_0^1 w_{\theta}(t) f(t) dt \qquad \forall \theta \in \Theta$$

The former is indeed well-defined as the stochastic integral with respect to the semimartingale $t \rightsquigarrow X_t^n$ (see, e.g., Theorem I.4.31 of Jacod and Shiryaev (1987)). When Θ is not countable, the process $\theta \rightsquigarrow \Gamma^n(\theta)$ is not unique even in the almost sure sense. However, the following argument holds for *any version* of the process, because we shall always consider a countable subset Θ^* of Θ when we apply Theorem 2.4.2 to the terminal variables of the continuous martingales

$$\Gamma^{n}(\theta) - \gamma(\theta) = n^{-1/2} \int_{0}^{1} w_{\theta}(t) dB_{t} \qquad \forall \theta \in \Theta^{*} \subset \Theta.$$

We denote by $B_{(\Theta,d)}(\theta; \delta)$ the closed ball in Θ with center $\theta \in \Theta$ and d-radius $\delta > 0$.

Theorem 5.2.1 Let (Θ, d) be a separable metric space. For a given class $w = \{w_{\theta} : \theta \in \Theta\} \subset \mathcal{L}^2[0, 1]$, introduce ρ_w , Γ^n and γ given by (5.2.2), (5.2.3) and (5.2.4). Suppose that there exists a countable, d-dense subset Θ^* of Θ such that ρ_w defines a proper metric on Θ^* . For a given point $\theta_0 \in \Theta$, suppose also that $\theta \rightsquigarrow \gamma(\theta)$ is d-continuous at θ_0 and that the following conditions are satisfied for some $\delta_0 \in (0, \infty]$, p > 0, $a \in (0, p)$, c > 0 and a function $\varphi : (0, \delta_0) \to (0, \infty)$ such that $\delta \rightsquigarrow \delta^{-a}\varphi(\delta)$ is decreasing:

(5.2.5)
$$\gamma(\theta) - \gamma(\theta_0) \le -cd(\theta, \theta_0)^p \qquad \forall \theta \in B_{(\Theta, d)}(\theta_0; \delta_0);$$

(5.2.6)
$$\sup_{\theta \in \Theta^*} \int_0^\infty \sqrt{\log N(B_{(\Theta,d)}(\theta;\delta),\rho_w;\varepsilon)} d\varepsilon \le \varphi(\delta) \qquad \forall \delta \in (0,\delta_0);$$

(5.2.7)
$$\sup_{\theta \in \Theta^*} \operatorname{Diam}(B_{(\Theta,d)}(\theta;\delta),\rho_w) \le \varphi(\delta) \qquad \forall \delta \in (0,\delta_0).$$

Choose any constants $r_n > 0$ such that $r_n^{-1} \in (0, \delta_0)$ and that $n^{-1/2}\varphi(r_n^{-1}) \leq r_n^{-p}$. Then, for any Θ^* -valued random sequence $\widehat{\theta}^n$ such that $d(\widehat{\theta}^n, \theta_0) = o_{P^*}(1)$ and that

$$\Gamma^{n}(\widehat{\theta}^{n}) \geq \sup_{\theta \in \Theta^{*}} \Gamma^{n}(\theta) - \epsilon_{n}$$

for some $\epsilon_n = O_{P^*}(r_n^{-p})$, it holds that $d(\widehat{\theta}^n, \theta_0) = O_{P^*}(r_n^{-1})$. When $\delta_0 = \infty$, the assumption " $d(\widehat{\theta}^n, \theta_0) = o_{P^*}(1)$ " is unnecessary. When $\theta_0 \in \Theta^*$, the assumption that $\theta \rightsquigarrow \gamma(\theta)$ is d-continuous at θ_0 is unnecessary.

Proof. We will apply Theorem 5.1.2. Since $\theta \rightsquigarrow \gamma(\theta)$ is *d*-continuous at θ_0 , we can choose a point $\theta_{\theta_0}^n \in \Theta^*$ such that $|\gamma(\theta_{\theta_0}^n) - \gamma(\theta_0)| \leq (c/2^{p+1}) \cdot r_n^{-p}$ and that $d(\theta_{\theta_0}^n, \theta_0) \leq r_n^{-1}$ (when $\theta_0 \in \Theta^*$, the choice $\theta_{\theta_0}^n = \theta_0$ satisfies these requirements, thus the assumption that $\theta \rightsquigarrow \gamma(\theta)$ is *d*-continuous is unnecessary). We then denote $R_{\theta_0}^n(\delta) = \{\theta \in \Theta^* : (\delta/2) < d(\theta, \theta_{\theta_0}^n) \leq \delta\}$ for every $\delta \in (0, \infty)$.

For any $\delta \in [r_n^{-1}, \delta_0)$, it holds that

$$\begin{split} \gamma(\theta) - \gamma(\theta_{\theta_0}^n) &= \gamma(\theta) - \gamma(\theta_0) + \gamma(\theta_0) - \gamma(\theta_{\theta_0}^n) \\ &\leq -cd(\theta, \theta_0)^p + \frac{c}{2^{p+1}} \cdot r_n^{-p} \\ &\leq -\frac{c}{2^p} \cdot \delta^p + \frac{c}{2^{p+1}} \cdot \delta^{-p} \quad \forall \theta \in R_{\theta_0}^n(\delta) \\ &\leq -\frac{c}{2^{p+1}} \cdot \delta^p. \end{split}$$

Thus the FIRST INEQUALITY of *M*-CRITERION is fulfilled. Next, to show the SECOND INEQUALITY, notice that for every $\delta \in (0, \delta_0)$

$$E \sup_{\substack{\theta, \vartheta \in R^{n}_{\theta_{0}}(\delta)}} \left| (\Gamma^{n} - \gamma)(\theta) - (\Gamma^{n} - \gamma)(\vartheta) \right|$$

$$\leq n^{-1/2} E \sup_{\substack{\theta, \vartheta \in B_{(\Theta,d)}(\theta^{n}_{\theta_{0}};\delta) \cap \Theta^{*}}} \left| \int_{0}^{1} w_{\theta}(t) dB_{t} - \int_{0}^{1} w_{\vartheta}(t) dB_{t} \right|.$$

Since the quadratic ρ_w -modulus of the family of continuous martingales

$$\left\{\int_0^{\cdot} w_{\theta}(t) dB_t : \theta \in \Theta^*\right\}$$

is bounded by 1, and since $\text{Diam}(B_{(\Theta,d)}(\theta_{\theta_0}^n; \delta), \rho_w) \leq \varphi(\delta)$, it follows from Theorem 2.4.2 and (2.1.1) that the right hand side is bounded by (up to a multiplicative universal constant)

$$n^{-1/2} \int_0^{\varphi(\delta)} \sqrt{\log(1 + N(B_{(\Theta,d)}(\theta_{\theta_0}^n; \delta), \rho_w; \varepsilon))} d\varepsilon$$

$$\leq n^{-1/2} \left\{ \varphi(\delta) \sqrt{\log 2} + \int_0^{\varphi(\delta)} \sqrt{\log N(B_{(\Theta,d)}(\theta_{\theta_0}^n;\delta),\rho_w;\varepsilon)} d\varepsilon \right\}$$

$$\leq \left\{ \sqrt{\log 2} + 1 \right\} \cdot n^{-1/2} \varphi(\delta).$$

Thus the SECOND INEQUALITY is satisfied with $\phi_{\theta_0}^n = n^{-1/2}\varphi$.

The assumptions (5.2.6) and (5.2.7) are analogous to an assumption of Theorem 3.2.10 of van der Vaart and Wellner (1996). Although the supremum with respect to δ comes out of the integral, this condition may still look awkward at first sight. Indeed, it requires a calculation of certain covering numbers of the sets $B_{(\Theta,d)}(\theta; \delta)$ for all sufficiently small $\delta > 0$. However, when the parameter space (Θ, d) is Euclidean, this condition can be replaced by a simple relationship between the two metrics d and ρ_w , as is given in the next theorem.

Theorem 5.2.2 Let Θ be a subset of a finite-dimensional Euclidean space with the usual metric d. For a given class $w = \{w_{\theta} : \theta \in \Theta\} \subset \mathcal{L}^2[0,1]$, introduce ρ_w , Γ^n and γ given by (5.2.2), (5.2.3) and (5.2.4). Suppose that there exists a countable, d-dense subset Θ^* of Θ such that ρ_w defines a proper metric on Θ^* . For a given point $\theta_0 \in \Theta$, suppose also that $\theta \rightsquigarrow \gamma(\theta)$ is d-continuous at θ_0 and that there exist some $\delta_0 \in (0, \infty]$ and some constants p > q > 0 and c, C > 0 such that:

(5.2.8)
$$\begin{cases} \gamma(\theta) - \gamma(\theta_0) \leq -cd(\theta, \theta_0)^p & \forall \theta \in B_{(\Theta, d)}(\theta_0; \delta_0); \\ \rho_w(\theta, \vartheta) \leq Cd(\theta, \vartheta)^q & \forall \theta, \vartheta \in \Theta. \end{cases}$$

Then, the same conclusion as Theorem 5.2.1 holds for $r_n = n^{1/2(p-q)}$.

Proof. It suffices to show that the conditions (5.2.6) and (5.2.7) of Theorem 5.2.1 are satisfied with $\varphi(\delta) = \text{const.}\delta^q$. First notice that for every $\delta > 0$

(5.2.9)
$$d(\theta, \vartheta) \le \varepsilon^{1/q} \delta \implies \rho_w(\theta, \vartheta) \le C \varepsilon \delta^q$$

Thus we have for every $\theta \in \Theta^*$

$$N(B_{(\Theta,d)}(\theta;\delta),\rho_w;C\varepsilon\delta^q) \le N(B_{(\Theta,d)}(\theta;\delta),d;\varepsilon^{1/q}\delta).$$

The right hand side is bounded by $\{(2\delta)/(\varepsilon^{1/q}\delta)+1\}^r$ for every $\varepsilon \in (0,1]$, where r is the dimension of Θ . Hence, noting also $N(B_{(\Theta,d)}(\theta;\delta),\rho_w;C\delta^q)=1$, we obtain

$$C^{-1}\delta^{-q} \int_{0}^{\infty} \sqrt{\log N(B_{(\Theta,d)}(\theta;\delta),\rho_{w};\varepsilon)} d\varepsilon$$

= $\int_{0}^{1} \sqrt{\log N(B_{(\Theta,d)}(\theta;\delta),\rho_{w};C\varepsilon\delta^{q})} d\varepsilon$
 $\leq \int_{0}^{1} \sqrt{r\log\{2\varepsilon^{-1/q}+1\}} d\varepsilon \ (=K_{0} \text{ say}) < \infty$

On the other hand, by putting $\varepsilon = 1$ in (5.2.9) we obtain $\text{Diam}(B_{(\Theta,d)}(\theta; \delta), \rho_w) \leq 2C\delta^q$. Hence, (5.2.6) and (5.2.7) are satisfied with $\varphi(\delta) = C \cdot (K_0 \vee 2) \cdot \delta^q$.

In so-called "regular" parametric models, the condition (5.2.8) is satisfied with p = 2 and q = 1, which leads to the "square root asymptotics". The "cube root asymptotics" investigated by Kim and Pollard (1990), whose origin goes back at least to Chernoff (1964), corresponds to the case of p = 2 and q = 1/2.

In both theorems, we have to show the consistency of estimators somehow. Thus let us state here a sufficient condition based on Corollary 3.2.3 of van der Vaart and Wellner (1996).

Proposition 5.2.3 Let (Θ, d) be a separable metric space. For a given class $w = \{w_{\theta} : \theta \in \Theta\} \subset \mathcal{L}^2[0, 1]$, introduce ρ_w , Γ^n and γ given by (5.2.2), (5.2.3) and (5.2.4). Suppose also that there exists a countable, d-dense subset Θ^* of Θ such that ρ_w defines a proper metric on Θ^* . Suppose that it holds for a given point $\theta_0 \in \Theta$ that

(5.2.10)
$$\gamma(\theta_0) > \sup_{\theta \notin G} \gamma(\theta)$$

for every d-open set $G \subset \Theta$ that contains θ_0 , and that

$$\int_0^\infty \sqrt{\log N(\Theta, \rho_w; \varepsilon)} d\varepsilon < \infty.$$

Then, for any Θ^* -valued random sequence $\widehat{\theta}^n$ such that

$$\Gamma^{n}(\widehat{\theta}^{n}) \geq \sup_{\theta \in \Theta^{*} \cup \{\theta_{0}\}} \Gamma^{n}(\theta) - \epsilon_{n}$$

for some $\epsilon_n = o_{P^*}(1)$, it holds that $d(\widehat{\theta}^n, \theta_0) = o_{P^*}(1)$.

Proof. We will apply (i) of Corollary 3.2.3 of van der Vaart and Wellner (1996) to $\Theta^* = \Theta^* \cup \{\theta_0\}$ (if $\rho_w(\theta_0^*, \theta_0) = 0$ for some $\theta_0^* \in \Theta^*$, set $\Theta^* = (\Theta^* \setminus \{\theta_0^*\}) \cup \{\theta_0\}$ to make (Θ^*, ρ_w) a proper metric space). Notice that the inequality (5.2.10) still holds for any $G \subset \Theta^*$ containing θ_0 which is open in the relative topology. Theorem 2.4.2 yields that

$$\begin{split} E \sup_{\theta \in \Theta^{\star}} |\Gamma^{n}(\theta) - \gamma(\theta)| &\leq C \cdot n^{-1/2} \int_{0}^{D} \sqrt{\log(1 + N(\Theta, \rho_{w}; \varepsilon))} d\varepsilon \\ &\leq C \cdot n^{-1/2} \left\{ D \sqrt{\log 2} + \int_{0}^{D} \sqrt{\log N(\Theta, \rho_{w}; \varepsilon)} d\varepsilon \right\}, \end{split}$$

where $D = \text{Diam}(\Theta, \rho_w) < \infty$ by our entropy condition, and C > 0 is a universal constant. This implies that

$$\sup_{\theta \in \Theta^{\star}} |\Gamma^{n}(\theta) - \gamma(\theta)| = o_{P}(1),$$

which completes the proof.

5.2.2 Examples

Example 1: Peak point of F

Let us consider estimating the value of

$$\theta_0 = \operatorname*{argmax}_{\theta \in [0,1]} F(\theta),$$

where $t \rightsquigarrow F(t)$ is the cumulative function of f defined by $F(t) = \int_0^t f(s) ds$. This problem can be treated in our general framework by setting

$$w_{\theta}(t) = \mathbb{1}_{[0,\theta]}(t) \qquad \forall \theta \in [0,1].$$

The criterion process and the contrast function, defined by (5.2.3) and (5.2.4), turn out to be $\Gamma^n(\theta) = X^n_{\theta}$ and $\gamma(\theta) = F(\theta)$, respectively.

We equip $\Theta = [0, 1]$ with the usual metric $d(\theta, \vartheta) = |\theta - \vartheta|$ to apply Theorem 5.2.2. It holds that $\rho_w(\theta, \vartheta) = \sqrt{|\theta - \vartheta|}$, and thus ρ_w is defines a proper metric on [0, 1]. The function $\theta \rightsquigarrow \gamma(\theta)$ is indeed continuous. Hence, if θ_0 is an inner point of [0, 1] and if there exist some constants $\delta_0, c > 0$ and p > 1/2 such that

(5.2.11)
$$F(\theta) - F(\theta_0) \le -c|\theta - \theta_0|^p \qquad \forall \theta \in [\theta_0 - \delta_0, \theta_0 + \delta_0]$$

then the same conclusion as Theorem 5.2.1 holds for $r_n = n^{1/(2p-1)}$.

To derive the asymptotic behavior of the rescaled residual $n^{1/(2p-1)}(\widehat{\theta}^n - \theta_0)$, let us introduce an assumption on the function $t \rightsquigarrow F(t)$.

Condition 5.2.4 Let $p \in \mathbb{N}$ be given. For a given point $\theta_0 \in (0,1)$, the function $t \rightsquigarrow F(t)$ is (p-1)-times continuously differentiable in a neighborhood of θ_0 with derivatives $F^{(m)}, m = 1, ..., p - 1$, and has p-th left- and right-derivatives $F_{-}^{(p)}$ and $F_{+}^{(p)}$ at θ_0 , respectively, which satisfy:

- when $p \ge 2$: $F^{(m)}(\theta_0) = 0$ for every m = 1, ..., p 1;
- when p is odd: $F_{-}^{(p)}(\theta_0) > 0 > F_{+}^{(p)}(\theta_0);$
- when p is even: $F_{-}^{(p)}(\theta_0) \vee F_{+}^{(p)}(\theta_0) < 0$.

The condition (5.2.11) follows from this assumption by a Taylor expansion. Moreover, we obtain the following result.

Proposition 5.2.5 Under Condition 5.2.4, for any $[0,1] \cap \mathbb{Q}$ -valued random sequence $\widehat{\theta}^n$ such that $\widehat{\theta}^n \xrightarrow{P^*} \theta_0$ and that

$$X_{\widehat{\theta}^n}^n \ge \sup_{\theta \in [0,1] \cap \mathbb{Q}} X_{\theta}^n - \epsilon_n$$

for some $\epsilon_n = o_{P^*}(n^{-p/(2p-1)})$, it holds that $n^{1/(2p-1)}(\widehat{\theta}^n - \theta_0) \stackrel{P}{\Longrightarrow} \operatorname{argmax}_{h \in \mathbb{R}} \{\mathbb{A}(h) + \mathbb{B}(h)\}$ in \mathbb{R} , where $h \rightsquigarrow \mathbb{A}(h)$ is the deterministic process given by

$$\mathbb{A}(h) = \begin{cases} h^p F_+^{(p)}(\theta_0)/p!, & \forall h \ge 0, \\ h^p F_-^{(p)}(\theta_0)/p!, & \forall h < 0, \end{cases}$$

and where $h \rightsquigarrow \mathbb{B}(h)$ is the two-sided Brownian motion, that is, a zero-mean, continuous Gaussian process such that $E|\mathbb{B}(h_1) - \mathbb{B}(h_2)|^2 = |h_1 - h_2|$ for every $h_1, h_2 \in \mathbb{R}$.

Proof. It has already been shown by Theorem 5.2.2 that the sequence $n^{1/(2p-1)}(\widehat{\theta}^n - \theta_0)$ is uniformly tight. Let us set:

$$H^{n} = \{h \in \mathbb{R} : \theta_{0} + n^{-1/(2p-1)}h \in [0,1]\};$$

$$H^{n*} = \{h \in \mathbb{R} : \theta_{0} + n^{-1/(2p-1)}h \in [0,1] \cap \mathbb{Q}\}.$$

We consider the stochastic process $h \rightsquigarrow \mathbb{M}^n(h)$, with parameter in H^n , defined by

$$\begin{split} \mathbb{M}^{n}(h) &= n^{p/(2p-1)} \left\{ \Gamma^{n}(\theta_{0} + n^{-1/(2p-1)}h) - \Gamma^{n}(\theta_{0}) \right\} \\ &= \mathbb{A}^{n}(h) + \mathbb{B}^{n}(h), \end{split}$$

where:

$$\mathbb{A}^{n}(h) = n^{p/(2p-1)} \int_{0}^{1} \left\{ w_{\theta_{0}+n^{-1/(2p-1)}h}(t) - w_{\theta_{0}}(t) \right\} f(t) dt;$$

$$\mathbb{B}^{n}(h) = n^{1/(4p-2)} \int_{0}^{1} \left\{ w_{\theta_{0}+n^{-1/(2p-1)}h}(t) - w_{\theta_{0}}(t) \right\} dB_{t}.$$

By Theorem 2.4.4, there exists a continuous version $h \rightsquigarrow \widetilde{\mathbb{B}}^n(h)$ of $h \rightsquigarrow \mathbb{B}^n(h)$. Thus we first consider $\widetilde{\mathbb{M}}^n = \mathbb{A}^n + \widetilde{\mathbb{B}}^n$ instead of \mathbb{M}^n . An easy computation gives that $\lim_{n\to\infty} \mathbb{A}^n(h) = \mathbb{A}(h)$ for every $h \in \mathbb{R}$. Furthermore, since $h \rightsquigarrow \mathbb{A}^n(h)$ and $h \rightsquigarrow \mathbb{A}(h)$ are continuous, this convergence is uniform on every compact set $K \subset \mathbb{R}$. On the other hand, it follows from Corollary 3.4.3 that $\widetilde{\mathbb{B}}^n \xrightarrow{P} \mathbb{B}$ in $\ell^{\infty}(K)$ for every compact set $K \subset \mathbb{R}$. Thus we have $\widetilde{\mathbb{M}}^n \xrightarrow{P} \mathbb{M} = \mathbb{A} + \mathbb{B}$ in $\ell^{\infty}(K)$ for every compact set $K \subset \mathbb{R}$. The existence and the uniqueness of the maximum point of $h \rightsquigarrow \mathbb{M}(h)$ follow from Khinchin's law of the iterated logarithm (see, e.g., page 61 of Hida (1980)) and Lemma 2.6 of Kim and Pollard (1990), respectively. Hence Theorem 3.2.2 of van der Vaart and Wellner (1996) yields the following CLAIM: for any uniformly tight sequence \hat{h}^n satisfying

$$\widetilde{\mathbb{M}}^{n}(\widehat{h}^{n}) \geq \sup_{h \in K} \widetilde{\mathbb{M}}^{n}(h) - \epsilon_{n,K}$$

for some $\epsilon_{n,K} = o_{P^*}(1)$ for each compact set $K \subset \mathbb{R}$, it holds that $\widehat{h}^n \stackrel{P}{\Longrightarrow} \operatorname{argmax}_{h \in \mathbb{R}} \mathbb{M}(h)$ in \mathbb{R} .

Now, for every $n \in \mathbb{N}$ such that $K \subset H^n$, using the continuity of $h \rightsquigarrow \widetilde{\mathbb{M}}^n(h)$, we have

$$\sup_{h \in K} \widetilde{\mathbb{M}}^{n}(h) = \sup_{h \in K \cap H^{n*}} \widetilde{\mathbb{M}}^{n}(h) \qquad P\text{-almost surely}$$
$$= \sup_{h \in K \cap H^{n*}} \mathbb{M}^{n}(h) \qquad P\text{-almost surely}$$
$$\leq \sup_{h \in H^{n*}} \mathbb{M}^{n}(h).$$

Hence, we can apply the above CLAIM to $\widehat{h}^n = n^{1/(2p-1)}(\widehat{\theta}^n - \theta_0)$, which takes values in H^{n*} identically, in order to obtain that $n^{1/(2p-1)}(\widehat{\theta}^n - \theta_0) \stackrel{P}{\Longrightarrow} \operatorname{argmax}_{h \in \mathbb{R}} \mathbb{M}(h)$ in \mathbb{R} . \Box

Since $\Gamma^n(\theta) = X^n_{\theta}$, and since $\theta \rightsquigarrow X^n_{\theta}$ is continuous, it is possible to apply Theorem 3.4.4 to \mathbb{B}^n , without introducing the continuous version $\widetilde{\mathbb{B}}^n$ by Theorem 2.4.4. However, since it is not always easy to show the ρ -separability of the original family of continuous local martingales $\{\int_0^{\cdot} w_{\theta}(t) dB_t\}$, we presented the above approach using $\widetilde{\mathbb{B}}^n$. This argument is indeed necessary for Example 3 given later.

Example 2: Steepest interval of F

Fix a constant $b \in (0, 1/2)$. We aim to estimate the value of

$$\theta_0 = \operatorname*{argmax}_{\theta \in \Theta} \int_{\theta-b}^{\theta+b} f(t) dt,$$

which is the center of the interval with length 2b where the function $t \rightsquigarrow F(t)$ increases most rapidly. This problem fits in our general framework by setting

$$w_{\theta}(t) = 1_{[\theta-b,\theta+b]}(t) \qquad \forall \theta \in [b, 1-b].$$

The criterion process and the contrast function, defined by (5.2.3) and (5.2.4), turn out to be $\Gamma^n(\theta) = X_{\theta+b}^n - X_{\theta-b}^n$ and $\gamma(\theta) = F(\theta+b) - F(\theta-b)$, respectively.

Here we make an assumption which is similar to Condition 5.2.4 in the preceding example.

Condition 5.2.6 Let an even integer $p \ge 2$ be given. For given $\theta_0 \in (b, 1 - b)$, the function $t \rightsquigarrow f(t)$ is (p - 1)-times continuously differentiable on an open set containing $\theta_0 - b$ and $\theta_0 + b$ with derivatives $f^{(m)}, m = 1, ..., p - 1$, satisfying:

- $f^{(m)}(\theta_0 b) = f^{(m)}(\theta_0 + b)$ for every m = 0, ..., p 2;
- $f^{(p-1)}(\theta_0 b) > f^{(p-1)}(\theta_0 + b).$

Proposition 5.2.7 Under Condition 5.2.6, for any $[b, 1-b] \cap \mathbb{Q}$ -valued random sequence $\widehat{\theta}^n$ such that $\widehat{\theta}^n \xrightarrow{P^*} \theta_0$ and that

$$X_{\widehat{\theta}^n+b}^n - X_{\widehat{\theta}^n-b}^n \ge \sup_{\theta \in [b,1-b] \cap \mathbb{Q}} \left\{ X_{\theta+b}^n - X_{\theta-b}^n \right\} - \epsilon_n$$

for some $\epsilon_n = o_{P^*}(n^{-p/(2p-1)})$, it holds that $n^{1/(2p-1)}(\widehat{\theta}^n - \theta_0) \stackrel{P}{\Longrightarrow} \operatorname{argmax}_{h \in \mathbb{R}} \{\mathbb{A}(h) + \mathbb{B}(h)\}$ in \mathbb{R} , where $h \rightsquigarrow \mathbb{A}(h)$ is the deterministic process given by

$$\mathbb{A}(h) = 2^{-1/2} h^p \{ f^{(p-1)}(\theta_0 + b) - f^{(p-1)}(\theta_0 - b) \} / p! \qquad \forall h \in \mathbb{R},$$

and where $h \rightsquigarrow \mathbb{B}(h)$ is the two-sided Brownian motion.

Proof. It follows from Condition 5.2.6 and a Taylor expansion that

$$\gamma(\theta) - \gamma(\theta_0) = \frac{f^{(p-1)}(\hat{\theta}_+) - f^{(p-1)}(\hat{\theta}_-)}{p!} (\theta - \theta_0)^p,$$

where $\tilde{\theta}_+$ (resp. $\tilde{\theta}_-$) is a point on the segment connecting $\theta + b$ and $\theta_0 + b$ (resp. $\theta - b$ and $\theta_0 - b$). Thus, since p is even, it holds that $\gamma(\theta) - \gamma(\theta_0) \leq -c|\theta - \theta_0|^p$ in a neighborhood of θ_0 for a constant c > 0. On the other hand, it is clear that $\rho_w(\theta, \vartheta) = \sqrt{2|\theta - \vartheta|}$, and thus ρ_w defines a proper metric on [b, 1 - b]. Hence Theorem 5.2.2 implies that $n^{1/(2p-1)}(\hat{\theta}^n - \theta_0)$ is uniformly tight. Repeating the same argument as that in the proof of Proposition 5.2.5 to the stochastic process $h \rightsquigarrow \mathbb{M}^n(h)$ defined by

$$\mathbb{M}^{n}(h) = 2^{-1/2} n^{p/(2p-1)} \left\{ (X^{n}_{\theta_{0}+b+n^{-1/(2p-1)}h} - X^{n}_{\theta_{0}+b}) - (X^{n}_{\theta_{0}-b+n^{-1/(2p-1)}h} - X^{n}_{\theta_{0}-b}) \right\},$$

we can obtain the assertion.

Example 3: Jump point of f

Let us introduce a model for the estimation problem of a jump point of f.

Condition 5.2.8 For an inner point θ_0 of [0,1], there exists a constant $a \in (0,1/2)$ such that the function $t \rightsquigarrow f(t)$ is càdlàg on the interval $[\theta_0 - a, \theta_0 + a]$ and that

$$D = (R_{\star} - L^{\star}) - (L^{\star} - L_{\star}) \lor (R^{\star} - R_{\star}) > 0$$

where

$$L^{\star} = \sup_{t \in [\theta_0 - a, \theta_0)} f(t), \qquad R^{\star} = \sup_{t \in [\theta_0, \theta_0 + a]} f(t),$$

$$L_{\star} = \inf_{t \in [\theta_0 - a, \theta_0)} f(t), \qquad R_{\star} = \inf_{t \in [\theta_0, \theta_0 + a]} f(t).$$

The constant a > 0 in the above assumption should be known to construct the estimator given later, but we do not specify any concrete shape of the function $t \rightsquigarrow f(t)$, even the value of the constant D > 0. Condition 5.2.8 means that the function $t \rightsquigarrow f(t)$ has a positive jump at θ_0 , namely $f(\theta_0) - f(\theta_0 -) \ge R_\star - L^\star$, which is the biggest one in the interval $[\theta_0 - a, \theta_0 + a]$. This interpretation shows how natural this assumption is in the present context.

Let the parameter space $\Theta = [a, 1 - a]$ be equipped with the Euclidean metric $d(\theta, \vartheta) = |\theta - \vartheta|$. Fixing a constant $b \in (0, a)$ we define

(5.2.12)
$$w_{\theta}(t) = k_b(t-\theta) \qquad \forall \theta \in [a, 1-a],$$

where

$$k_b(x) = \begin{cases} -x - b, & x \in [-b, 0), \\ -x + b, & x \in [0, b], \\ 0, & \text{otherwise.} \end{cases}$$

Proposition 5.2.9 Under Condition 5.2.8, consider the criterion process $\theta \rightsquigarrow \Gamma^n(\theta) = \int_0^1 w_\theta(t) dX_t^n$ with w_θ given by (5.2.12). For any $[a, 1 - a] \cap \mathbb{Q}$ -valued random sequence $\widehat{\theta}^n$ such that $\widehat{\theta}^n \xrightarrow{P^*} \theta_0$ and that

$$\Gamma^{n}(\widehat{\theta}^{n}) \geq \sup_{\theta \in [a,1-a] \cap \mathbb{Q}} \Gamma^{n}(\theta) - \epsilon_{n}$$

for some $\epsilon_n = o_{P^*}(n^{-1})$, it holds that $n(\widehat{\theta}^n - \theta_0) \stackrel{P}{\Longrightarrow} \operatorname{argmax}_{h \in \mathbb{R}} \{ \mathbb{A}(h) + \mathbb{B}(h) \}$ in \mathbb{R} , where $h \rightsquigarrow \mathbb{A}(h)$ is the deterministic process given by

$$\mathbb{A}(h) = \left\{ \begin{array}{ll} h \left\{ (2b)^{-1} \int_{\theta_0 - b}^{\theta_0 + b} f(t) dt - f(\theta_0) \right\}, & \forall h \ge 0\\ h \left\{ (2b)^{-1} \int_{\theta_0 - b}^{\theta_0 + b} f(t) dt - f(\theta_0 -) \right\}, & \forall h < 0 \end{array} \right.$$

and where $h \rightsquigarrow \mathbb{B}(h)$ is the two-sided Brownian motion.

Proof. It holds that for any $\theta \in [\theta_0, \theta_0 + a - b]$

$$\begin{split} \gamma(\theta) - \gamma(\theta_0) &\leq -(2b - |\theta - \theta_0|)R_\star |\theta - \theta_0| + |\theta - \theta_0|(R^\star + L^\star)b\\ &= -|\theta - \theta_0|\{b[(R_\star - L^\star) - (R^\star - R_\star)] - |\theta - \theta_0|R_\star\}\\ &\leq -|\theta - \theta_0|\{bD - |\theta - \theta_0|R_\star\} \end{split}$$

and that, in the same way, for any $\theta \in [\theta_0 - a + b, \theta_0)$

$$\gamma(\theta) - \gamma(\theta_0) \le -|\theta - \theta_0| \{ bD + |\theta - \theta_0| L^* \}.$$

Thus, choosing sufficiently small constants $\delta_0, c > 0$ we have $\gamma(\theta) - \gamma(\theta_0) \leq -c|\theta - \theta_0|$ for every $\theta \in [\theta_0 - \delta_0, \theta_0 + \delta_0]$. On the other hand, an easy computation implies that $\rho_w(\theta, \vartheta) \leq C\sqrt{|\theta - \vartheta|}$ with $C = \sqrt{4b^2 + 6b}$. Hence Theorem 5.2.2 yields that the rate of convergence in this model is $r_n = n$. Repeat the same argument as Proposition 5.2.5 to the stochastic process $h \rightsquigarrow \mathbb{M}^n(h)$ defined by $\mathbb{M}^n(h) = (2b)^{-1}n\{\Gamma^n(\theta_0 + n^{-1}h) - \Gamma^n(\theta_0)\}$ to get the assertion.

5.2.3 Remarks for Non-Gaussian Cases

Instead of the Gaussian white noise model (5.2.1), let us consider the following:

$$dX_t^n = f(t)dt + dM_t^n, \qquad X_0^n = x_0 \in \mathbb{R}$$

where $f \in \mathcal{L}^2[0,1]$ is as before, and $M^n = (M_t^n)_{t \in [0,1]}$ is a continuous local martingale, defined on a stochastic basis **B**, with the quadratic covariation given by $\langle M^n, M^n \rangle_t = \int_0^t g^n(s) ds$. Then, all results in Subsection 5.2.1 remain true whenever

$$\sup_{t \in [0,1]} g^n(t) = O_P(n^{-1}).$$

Furthermore, if

$$\sup_{t \in [0,1]} |ng^n(t) - 1| = o_P(1),$$

then all results in Subsection 5.2.2 also hold without change of limit distributions. More generally, if there exists $g \in \mathcal{L}^{\infty}[0, 1]$ such that

$$\sup_{t \in [0,1]} |ng^n(t) - g(t)| = o_P(1),$$

then one can get some results similar to those in Subsection 5.2.2, under some smoothness assumptions on g, with modification of limit distributions.

5.3 Multiplicative Intensity Model

Let μ^n be a 1-dimensional point process defined on a stochastic basis $(\Omega^n, \mathcal{F}^n, \mathbf{F}^n = (\mathcal{F}^n_t)_{t \in [0,1]}, P^n)$ with the predictable compensator ν^n given by

$$\nu^n(\omega; dt) = \alpha(t) Y_t^n(\omega) dt,$$

where α is a $[0, \infty)$ -valued measurable function on [0, 1], and $t \rightsquigarrow Y_t^n$ is a $[0, \infty)$ -valued predictable process. Let a class $\{w_\theta : \theta \in \Theta\}$ of bounded, measurable functions on [0, 1] be given. Assuming $\int_0^1 \alpha(t) dt < \infty$, we consider the contrast function given by

(5.3.1)
$$\gamma(\theta) = \int_0^1 w_{\theta}(t) \alpha(t) dt \qquad \forall \theta \in \Theta$$

and the criterion process given by

(5.3.2)
$$\Gamma^{n}(\theta) = (w_{\theta}Y^{n-}) * \mu_{1}^{n} \qquad \forall \theta \in \Theta,$$

where the generalized inverse process Y^{n-} of Y^n is defined by

$$Y_t^{n-}(\omega) = \begin{cases} \frac{1}{Y_t^{n}(\omega)} & \text{if } Y_t^{n}(\omega) \ge 1, \\ 0 & \text{otherwise.} \end{cases}$$

In this section, specializing the class w, we consider two kinds of estimation problems. In both problems, we will assume the following.

Condition 5.3.1 There exists a measurable function y on [0, 1], which is bounded away from zero, such that

$$\sup_{t \in [0,1]} \left| n^{-1} Y_t^n - y(t) \right| \xrightarrow{P^n} 0.$$

Now, setting

(5.3.3)
$$\Omega^{n}(L) = \left\{ \sup_{t \in [0,1]} \left| n^{-1} Y_{t}^{n} - y(t) \right| \le \frac{1}{L} \right\} \qquad L = \sup_{t \in [0,1]} \frac{2}{y(t)}$$

we have for all $n \ge L$ that

$$\Gamma^{n}(\theta) - \gamma(\theta) = (w_{\theta}Y^{n-}) * (\mu^{n} - \nu^{n})_{1} \quad \text{on the set } \Omega^{n}(L)$$

and that

$$\sup_{t\in[0,1]}nY_t^{n-}\leq L\qquad\text{on the set }\Omega^n(L).$$

Noting also that $\lim_{n\to\infty} P^n(\Omega^n \setminus \Omega^n(L)) = 0$, we will use these facts to establish the SECOND INEQUALITY of Theorem 5.1.2.

Problem 1: Peak point of α

Let us consider estimating the location of (approximate) peak of the function $t \rightsquigarrow \alpha(t)$. This problem can be treated by setting

$$\gamma(\theta) = \int_0^1 \mathbf{1}_{[\theta-b,\theta+b]}(t)\alpha(t)dt \qquad \forall \theta \in (b, 1-b),$$

where $b \in (0, 1/2)$ is a given constant.

Proposition 5.3.2 Suppose that Conditions 5.3.1 and 5.2.6 with f replaced by α are satisfied, and that the function $t \rightsquigarrow y(t)$ appearing in the former is continuous at points $\theta_0 - b$ and $\theta_0 + b$ appearing in the latter. Define $\Gamma^n(\theta)$ by (5.3.2) for the class $w = \{\mathbf{1}_{[\theta-b,\theta+b]}: \theta \in [b, 1-b]\}$. Then, for any [b, 1-b]-valued random sequence $\widehat{\theta}^n$ such that $\widehat{\theta}^n \xrightarrow{P^{n*}} \theta_0$ and that

$$\Gamma^{n}(\widehat{\theta}^{n}) \geq \sup_{\theta \in [b,1-b]} \Gamma^{n}(\theta) - \epsilon_{n}$$

for some $\epsilon_n = o_{P^{n*}}(n^{-p/(2p-1)})$, it holds that $n^{1/(2p-1)}(\widehat{\theta}^n - \theta_0) \stackrel{P^n}{\Longrightarrow} \operatorname{argmax}_{h \in \mathbb{R}} \{\mathbb{A}(h) + \mathbb{B}(h)\}$ in \mathbb{R} , where $h \rightsquigarrow \mathbb{A}(h)$ is the deterministic function given by

$$\mathbb{A}(h) = h^{p} \cdot \frac{\alpha^{(p-1)}(\theta_{0}+b) - \alpha^{(p-1)}(\theta_{0}-b)}{p!} \left| \frac{\alpha(\theta_{0}+b)}{y(\theta_{0}+b)} + \frac{\alpha(\theta_{0}-b)}{y(\theta_{0}-b)} \right|^{-1/2}$$

and where $h \rightsquigarrow \mathbb{B}(h)$ is the two-sided Brownian motion.

Proof. First, let us apply Theorem 5.1.2 to get $|\hat{\theta}^n - \theta_0| = O_{P^{n*}}(n^{-1/(2p-1)})$. The FIRST INEQUALITY of *M*-CRITERION can be shown in the same way as Proposition 5.2.7. To show the SECOND INEQUALITY, we apply (ii) of Theorem 2.2.3 to the class

$$\mathcal{W}^{n}_{\delta} = \{\mathbf{1}_{[\theta-b,\theta+b]}Y^{n-} : \theta \in \Theta_{\delta}\}$$

where $\Theta_{\delta} = [\theta_0 - \delta, \theta_0 + \delta] \cap [b, 1 - b]$. Notice that there exist some constants $\delta_0, M > 0$ such that $|\alpha(\theta + b) + \alpha(\theta - b)| \leq M$ for all $\theta \in \Theta_{\delta_0}$. For every $\delta \in (0, \delta_0)$, we construct a DFP $\Pi_{\delta} = {\Pi_{\delta}(\varepsilon)}_{\varepsilon \in (0,1]}$ of Θ_{δ} as follows: for every $\varepsilon \in (0,1]$ we divide the interval Θ_{δ} , with length less than 2δ , into $N_{\Pi_{\delta}}(\varepsilon)$ sub-intervals with length at most $2\delta\varepsilon^2$; this can be done with $N_{\Pi_{\delta}}(\varepsilon) \leq \frac{1}{\varepsilon^2} + 1$. Then it holds that

$$\|\mathcal{W}_{\delta}^{n}\|_{\Pi_{\delta},1} \leq \sqrt{2\delta \cdot M \cdot n^{-1}L}$$
 on the set $\Omega^{n}(L)$

and that

 $|\overline{W}^n_\delta|^2*\nu_1^n\leq 2\delta\cdot M\cdot n^{-1}L\qquad\text{on the set }\Omega^n(L),$

where the constant L > 0 and the set $\Omega^n(L)$ is given by (5.3.3). Thus we have for all $n \ge L$ that

$$E^{n*} \sup_{\theta \in \Theta_{\delta}} |(\Gamma^{n} - \gamma)(\theta) - (\Gamma^{n} - \gamma)(\theta_{0})| \mathbf{1}_{\Omega^{n}(L)}$$

$$\lesssim n^{-1/2} \delta^{1/2} \left\{ \sqrt{2ML} \int_{0}^{1} \sqrt{\log\left(\frac{1}{\varepsilon^{2}} + 2\right)} d\varepsilon + 1 \right\}$$

Since Condition 5.3.1 implies that $\lim_{n\to\infty} P^n(\Omega^n \setminus \Omega^n(L)) = 0$, we can deduce from Theorem 5.1.2 with $\phi_u^n(\delta) = n^{-1/2} \delta^{1/2}$ that $|\hat{\theta}^n - \theta_0| = O_{P^{n*}}(n^{-1/(2p-1)})$.

Next, let us derive the asymptotic behavior of the sequence of rescaled processes

$$\mathbb{M}^{n}(h) = n^{p/(2p-1)} C_0 \left\{ \Gamma^{n}(\theta_0 + n^{-1/(2p-1)}h) - \Gamma^{n}(\theta_0) \right\}$$

where

$$C_0 = \left| \frac{\alpha(\theta_0 + b)}{y(\theta_0 + b)} + \frac{\alpha(\theta_0 - b)}{y(\theta_0 - b)} \right|^{-1/2}$$

Let us decompose $\mathbb{M}^n(h) = \mathbb{A}^n(h) + \mathbb{B}^n(h)$, where

$$\begin{aligned} \mathbb{A}^n(h) &= W^{n,h} * \nu_1^n, \\ \mathbb{B}^n(h) &= W^{n,h} * (\mu^n - \nu^n)_1 \end{aligned}$$

with

$$W^{n,h} = n^{p/(2p-1)} C_0(\mathbf{1}_{[\theta_0 + n^{-1/(2p-1)}h - b, \theta_0 + n^{-1/(2p-1)}h + b]} - \mathbf{1}_{[\theta_0 - b, \theta_0 + b]}) Y^{n-1}$$

An easy calculation implies that $\mathbb{A}^n(h) \xrightarrow{P^n} \mathbb{A}(h)$ for every $h \in \mathbb{R}$. Furthermore, since $h \rightsquigarrow \mathbb{A}^n(h)$ and $h \rightsquigarrow \mathbb{A}(h)$ are continuous, this convergence is uniform on every compact set $K \subset \mathbb{R}$. On the other hand, Theorem 3.2.4 yields that $\mathbb{B}^n \xrightarrow{P^n} \mathbb{B}$ in $\ell^{\infty}(K)$ for every compact set $K \subset \mathbb{R}$. Hence, the same argument as Proposition 5.2.5 implies the conclusion.

Problem 2: Jump point of α

The second problem is concerned with a jump point of the hazard function $t \sim \alpha(t)$.

Proposition 5.3.3 Assume Conditions 5.3.1 and 5.2.8 with f replaced by α . Fixing a constant $b \in (0, a)$, define $\Gamma^n(\theta) = (w_{\theta}Y^{n-}) * \mu_1^n$, where $w_{\theta} = k_b(t - \theta)$ is given by (5.2.12). Then, for any [a, 1 - a]-valued random sequence $\widehat{\theta}^n$ such that $\widehat{\theta}^n \xrightarrow{P^{n*}} \theta_0$ and that

$$\Gamma^{n}(\widehat{\theta}^{n}) \geq \sup_{\theta \in [a,1-a]} \Gamma^{n}(\theta) - \epsilon_{n}$$

$$\epsilon_{n} = o_{P^{n*}}(n^{-1}), \text{ it holds that } |\widehat{\theta}^{n} - \theta_{0}| = O_{P^{n*}}(n^{-1}).$$

Proof. The FIRST INEQUALITY of *M*-CRITERION of Theorem 5.1.2 with p = 1 follows from the same computation as Proposition 5.2.9. Using Theorem 2.2.3 again, we can show that the SECOND INEQUALITY holds with the truncation by the set $\Omega^n(L)$ defined by (5.3.3) and with $\phi_u^n(\delta) = n^{-1/2} \delta^{1/2}$.

We do not derive the asymptotic behavior of a sequence of "rescaled processes" as in the proof of Proposition 5.3.2; the Lindeberg condition is not satisfied in the present situation, thus no result in Section 3.2 works well. It is conjectured that the limit of $n(\hat{\theta}^n - \theta_0)$ would be the argmax of a process with jumps.

5.A Notes

The general criterion Theorem 5.1.2 is an adaptation of Theorem 3.4.1 of van der Vaart and Wellner (1996) which is a fruit of recent works by some other authors, including Birgé and Massart (1993), van de Geer (1995a) and Wong and Shen (1995). The two inequalities of *M*-CRITERION are from, e.g., (1.2) and (1.3) of van de Geer (1995a), respectively.

The problem of estimating the mode of a density function, which is the motivation of Propositions 5.2.7 and 5.3.2, was studied by Chernoff (1964). It is treated also by Kim and Pollard (1990) in their systematic study of the cube root asymptotics. A progress of our results is that the smoothness around the mode has come into our scope; this is possible also in the i.i.d. case although we did not state in the main text. We continue this study further in Section 7.2.

Related to Propositions 5.2.9 and 5.3.3, let us briefly review some known results of jump point estimation. The asymptotic distribution of the maximum likelihood estimator $\hat{\theta}^n$ of a jump point θ_0 of $t \rightsquigarrow f(t)$ in the Gaussian white noise model (5.2.1) can be found in Ibragimov and Has'minskii (1981, Section VII.2). More precisely, they derived the asymptotic behavior of $n(\hat{\theta}^n - \theta_0)$ when the function f is of the form $f_{\theta}(t) = S(t - \theta)$ with S being a known function, along the approach of finite-dimensional parametric estimation. Korostelev (1987) showed the rate of convergence is still order n in a certain non-parametric model. Wang (1995) considered a broader model, including not only jumps but also cusps, and derived that the rate of convergence of a jump point estimator is $n |\log n|^{-\eta}$ with any constant $\eta > 0$, which is quite close to the best rate. Our setting is more general than that of Korostelev (1987), but does not contain that of Wang (1995). The point of Proposition 5.2.7 is that we have gotten the asymptotic distribution result of the rate n. By contrary, Proposition 5.3.3 gives the rate n only. See also Müller and Wang (1990) who considered estimating the point where a hazard function changes most rapidly.

Chapter 6

Non-parametric Maximum Likelihood Estimators

6.1 Gaussian White Noise Model

Let $\tau > 0$ be a fixed constant, and Θ a subset of $\mathcal{L}^2[0, \tau]$. For every $n \in \mathbb{N}$, let $t \rightsquigarrow X_t^n$ be a continuous, adapted process on a filtered measurable space $(\Omega^n, \mathcal{F}^n, \mathbf{F}^n = (\mathcal{F}_t^n)_{t \in [0, \tau]})$, and let $\mathbf{P}^n = \{P_{\theta}^n : \theta \in \Theta\}$ be a family of probability measures on $(\Omega^n, \mathcal{F}^n)$ indexed by Θ . Suppose that X^n is a special semimartingale under P_{θ}^n with the canonical decomposition

$$dX_t^n = \theta(t)dt + n^{-1/2}dB_t^{n,\theta}, \qquad X_0^n = x_0 \in \mathbb{R},$$

where $t \sim B^{n,\theta}$ is a standard Brownian motion defined on $(\Omega^n, \mathcal{F}^n, \mathbf{F}^n, P_{\theta}^n)$. It is well-known that, under some conditions, the log-likelihood ratio is given by

$$(6.1.1)\log\frac{P_{\theta}^{n}|\mathcal{F}_{\tau}^{n}}{P_{\vartheta}^{n}|\mathcal{F}_{\tau}^{n}} = \int_{0}^{\tau} \{\theta(t) - \vartheta(t)\} dX_{t}^{n} - \frac{1}{2} \left\{ \|\theta\|_{\mathcal{L}^{2}[0,1]}^{2} - \|\vartheta\|_{\mathcal{L}^{2}[0,1]}^{2} \right\} \quad \forall \theta, \vartheta \in \Theta$$

(see e.g. Theorem III.5.34 of Jacod and Shiryaev (1987)). Thus the maximum likelihood estimator is the maximizer of the criterion process $\theta \rightsquigarrow \Gamma^n(\theta)$ defined by

(6.1.2)
$$\Gamma^{n}(\theta) = \int_{0}^{\tau} \theta(t) dX_{t}^{n} - \frac{1}{2} \|\theta\|_{\mathcal{L}^{2}[0,\tau]}^{2} \qquad \forall \theta \in \mathcal{L}^{2}[0,\tau].$$

For every $\theta_0 \in \Theta$, the corresponding contrast function $\theta \rightsquigarrow \gamma_{\theta_0}(\theta)$ under $P_{\theta_0}^n$ turns out to be

(6.1.3)
$$\gamma_{\theta_0}(\theta) = \int_0^\tau \theta(t)\theta_0(t)dt - \frac{1}{2} \|\theta\|_{\mathcal{L}^2[0,\tau]}^2 \quad \forall \theta \in \mathcal{L}^2[0,\tau]$$
$$= -\frac{1}{2} \|\theta - \theta_0\|_{\mathcal{L}^2[0,\tau]}^2 + \frac{1}{2} \|\theta_0\|_{\mathcal{L}^2[0,\tau]}^2.$$

We shall *not* use the fact that (6.1.1) gives the log-likelihood ratio, and thus $\Gamma^{n}(\theta)$ can be thought just as a random variable defined by (6.1.2). Moreover, we have defined $\Gamma^{n}(\theta)$ and $\gamma_{\theta_0}(\theta)$ for all elements θ of $\mathcal{L}^2[0,\tau]$ in order to consider "sieved" maximum likelihood estimators. In view of (6.1.3) and the FIRST INEQUALITY of *M*-CRITERION of Theorem 5.1.2, it is natural to adopt the L^2 -semimetric as the canonical semimetric d_u^n on the parameter space Θ , that is, $d_u^n(\theta, \vartheta) = \|\theta - \vartheta\|_{\mathcal{L}^2[0,\tau]}$.

Let U be a subset of Θ (one may take $U = \Theta$ or $\{\theta_0\}$ for instance). We now introduce a *local* entropy condition on a "sieve" $\Theta^n \subset \mathcal{L}^2[0,\tau]$, which need not be contained in Θ , uniformly over U. Throughout this section, we denote by $B(\theta; \delta)$ the closed ball in $\mathcal{L}^2[0,\tau]$ with center θ and $\|\cdot\|_{\mathcal{L}^2[0,\tau]}$ -radius δ .

Condition 6.1.1 Let a set $U \subset \Theta \subset \mathcal{L}^2[0,\tau]$ and some countable sets $\Theta^n \subset \mathcal{L}^2[0,\tau]$ be given. For every $n \in \mathbb{N}$ and $\theta_0 \in U$, there exist a proper metric $\hat{\rho}_{\theta_0}^n$ on Θ^n such that $\|\cdot\|_{\mathcal{L}^2[0,\tau]} \leq \hat{\rho}_{\theta_0}^n$, and a function $\varphi_{\theta_0}^n : (0,\infty) \to (0,\infty)$ such that $\delta \rightsquigarrow \delta^{-1}\varphi_{\theta_0}^n(\delta)$ is decreasing and that

(6.1.4)
$$\int_0^{\delta} \sqrt{\log(1 + N(\Theta^n \cap B(\theta_0; \delta), \widehat{\rho}^n_{\theta_0}; \varepsilon))} d\varepsilon \le \varphi^n_{\theta_0}(\delta) \qquad \forall \delta \in (0, \infty).$$

Then, choose some positive constants r_{n,θ_0} such that $n^{-1/2}\varphi_{\theta_0}^n(r_{n,\theta_0}^{-1}) \leq r_{n,\theta_0}^{-2}$.

Theorem 6.1.2 Let a set $U \subset \Theta \subset \mathcal{L}^2[0,\tau]$ and some countable sets $\Theta^n \subset \mathcal{L}^2[0,\tau]$ be given. Suppose that Condition 6.1.1 is satisfied, and choose some constants r_{n,θ_0} described there. Suppose also that there exists a constant M > 0 such that: for every $n \in \mathbb{N}$ and $\theta_0 \in U$ there exists $\theta_{\theta_0}^n \in \Theta^n$ such that

(6.1.5)
$$\|\theta_0 - \theta_{\theta_0}^n\|_{\mathcal{L}^2[0,\tau]} \le M r_{n,\theta_0}^{-1}.$$

Then, for any mapping $\widehat{\theta}^n : \Omega^n \to \Theta^n$ such that

(6.1.6)
$$\Gamma^{n}(\widehat{\theta}^{n}) \geq \sup_{\theta \in \Theta^{n}} \Gamma^{n}(\theta) - r_{n}^{-2} \quad with \quad r_{n} = \sup_{\theta_{0} \in U} r_{n,\theta_{0}},$$

it holds that

$$\lim_{L \to \infty} \limsup_{n \to \infty} \sup_{\theta_0 \in U} P_{\theta_0}^{n*} \left(r_{n,\theta_0} \| \widehat{\theta}^n - \theta_0 \|_{\mathcal{L}^2[0,\tau]} > L \right) = 0.$$

It is trivial that the assumption (6.1.5) is not a real restriction when $\Theta^n = \Theta$; it is satisfied with $\theta_{\theta_0}^n = \theta_0$. On the other hand, it should be noted that the positive constant $M = M_U$ appearing there may depend on U; the case of $U = \Theta$ leads to the rate of convergence uniformly in the true parameter θ_0 , while the case of $U = \{\theta_0\}$ implies the point-wise assertion only. However, from the practical point of view we should choose a sieve Θ^n satisfying $\Theta \subset \bigcup_{\theta \in \Theta^n} B(\theta; M^* r_n^{-1})$ with a positive constant M^* not depending on U even in the case of $U = \{\theta_0\}$, because statisticians do not know which is the true point θ_0 .

Proof. We will apply Theorem 5.1.2. Formulation 5.1.1 should be read as follows: for every $\theta_0 \in U(=U^n)$,

- (i) the semimetric space $(\Theta^n, \|\cdot\|_{\mathcal{L}^2[0,\tau]})$ and the point $\theta^n_{\theta_0} \in \Theta^n$;
- (ii) the criterion processes $\theta \rightsquigarrow \Gamma^n(\theta)$ and the contrast functions $\theta \rightsquigarrow \gamma_{\theta_0}(\theta)$, with parameters in Θ^n , defined by (6.1.2) and (6.1.3), respectively.

We then denote $R_{\theta_0}^n(\delta) = \left\{ \theta \in \Theta^n : (\delta/2) < \|\theta - \theta_{\theta_0}^n\|_{\mathcal{L}^2[0,\tau]} \leq \delta \right\}$ for every $\delta \in (0,\infty)$. To show the FIRST INEQUALITY of *M*-CRITERION with p = 2, first observe that

$$\begin{aligned} \|\theta_{\theta_0}^n - \theta_0\|_{\mathcal{L}^2[0,\tau]} &\leq Mr_{n,\theta_0}^{-1} \\ &\leq \frac{1}{4} \cdot \frac{\delta}{2} \quad \text{whenever } \delta \geq 8Mr_{n,\theta_0}^{-1} \\ &\leq \frac{1}{4} \|\theta - \theta_{\theta_0}^n\|_{\mathcal{L}^2[0,\tau]} \quad \text{whenever } \theta \in R_{\theta_0}^n(\delta). \end{aligned}$$

Thus we have for every $\delta \geq 8Mr_{n,\theta_0}^{-1}$ and every $\theta \in R_{\theta_0}^n(\delta)$

$$\begin{split} \gamma_{\theta_{0}}(\theta) - \gamma_{\theta_{0}}(\theta_{\theta_{0}}^{n}) &= \frac{1}{2} \left\{ -\|\theta - \theta_{0}\|_{\mathcal{L}^{2}[0,\tau]}^{2} + \|\theta_{\theta_{0}}^{n} - \theta_{0}\|_{\mathcal{L}^{2}[0,\tau]}^{2} \right\} \\ &\leq \frac{1}{2} \left\{ -\|\theta - \theta_{\theta_{0}}^{n}\|_{\mathcal{L}^{2}[0,\tau]}^{2} + 2\|\theta - \theta_{\theta_{0}}^{n}\|_{\mathcal{L}^{2}[0,\tau]} \cdot \|\theta_{\theta_{0}}^{n} - \theta_{0}\|_{\mathcal{L}^{2}[0,\tau]} \right\} \\ &\leq \frac{1}{2} \left\{ -\|\theta - \theta_{\theta_{0}}^{n}\|_{\mathcal{L}^{2}[0,\tau]}^{2} + 2\|\theta - \theta_{\theta_{0}}^{n}\|_{\mathcal{L}^{2}[0,\tau]} \cdot \frac{\|\theta - \theta_{\theta_{0}}^{n}\|_{\mathcal{L}^{2}[0,\tau]}}{4} \right\} \\ &= -\frac{1}{4} \|\theta - \theta_{\theta_{0}}^{n}\|_{\mathcal{L}^{2}[0,\tau]}^{2} \\ &\leq -\frac{1}{16} \delta^{2}. \end{split}$$

This means that the FIRST INEQUALITY holds for $c_{\varepsilon} = \frac{1}{16}$ and $K_{\varepsilon} = 8M$ (with $\delta_0 = \infty$). On the other hand, by (6.1.5) we have $R_{\theta_0}^n(\delta) \subset \Theta^n \cap B(\theta_0; \frac{9}{8}\delta)$ whenever $\delta \in [8Mr_{n,\theta_0}^{-1}, \infty)$. Thus Theorem 2.4.2 implies that

$$\begin{split} E^{n*}_{\theta_0} \sup_{\theta \in R^n_{\theta_0}(\delta)} |(\Gamma^n - \gamma)(\theta) - (\Gamma^n - \gamma)(\theta^n_{\theta_{\theta_0}})| \\ \lesssim n^{-1/2} \int_0^{\delta} \sqrt{\log(1 + N(\Theta^n \cap B(\theta_0; \frac{9}{8}\delta), \widehat{\rho}^n_{\theta_0}; \varepsilon))} d\varepsilon \\ \leq n^{-1/2} \varphi^n_{\theta_0}(\frac{9}{8}\delta) \leq \frac{9}{8} \cdot n^{-1/2} \varphi^n_{\theta_0}(\delta), \end{split}$$

which means that the SECOND INEQUALITY holds with $\phi_{\theta_0}^n = n^{-1/2} \varphi_{\theta_0}^n$. Hence Theorem 5.1.2 yields that

$$\lim_{L\to\infty}\limsup_{n\to\infty}\sup_{\theta_0\in U}P_{\theta_0}^{n*}\left(r_{n,\theta_0}\|\widehat{\theta}^n-\theta_{\theta_0}^n\|_{L^2[0,1]}>L\right)=0.$$

Recalling (6.1.5), we obtain the assertion.

Example 1: Monotone functions

Let us set Θ to be the class of monotone functions $\theta : [0, \tau] \to [0, 1]$. Then it follows from Theorem 2.7.5 of van der Vaart and Wellner (1996) that

$$\int_0^\delta \sqrt{\log N(\Theta, \|\cdot\|_{\mathcal{L}^2[0,\tau]}; \varepsilon)} d\varepsilon \le \text{const.} \delta^{1/2} \qquad \forall \delta > 0.$$

This suggests that, by choosing a sieve Θ^n appropriately, Condition 6.1.1 should be fulfilled with $\varphi_{\theta_0}^n(\delta) = \text{const.}(\delta^{1/2} \vee \delta)$, which leads to the rate $r_n = n^{1/3}$, not depending on $\theta_0 \in U = \Theta$.

Proposition 6.1.3 Choosing any grids $0 = t_0^n < t_1^n < \cdots < t_{k_n}^n = \tau$ such that $t_i^n - t_{i-1}^n \leq n^{-2/3}$, define Θ^n as the class of monotone functions $\theta : [0, \tau] \to V^n$ which are piecewise constant on each interval $[t_{i-1}^n, t_i^n)$, where $V^n = \{j \cdot n^{-2/3} : j \in \mathbb{Z}\} \cap [0, 1]$. Then, it holds that $\Theta \subset \bigcup_{\theta \in \Theta^n} B(\theta; \sqrt{\tau + 1n^{-1/3}})$.

Proof. Fix any $\theta \in \Theta$. Let us choose $\theta^u, \theta^l \in \Theta^n$ given by

$$\begin{cases} \theta^u(t) &= u_i \\ \theta^l(t) &= l_i \end{cases} \quad \text{for } t \in [t_{i-1}^n, t_i^n), \ i = 1, \dots, k_n, \end{cases}$$

where

$$u_i = \min\left\{ y \in V^n : \sup_{s \in [t_{i-1}^n, t_i^n)} \theta(s) \le y \right\},\$$
$$l_i = \max\left\{ y \in V^n : \inf_{s \in [t_{i-1}^n, t_i^n)} \theta(s) \ge y \right\},\$$

and $\theta^{u}(\tau) = \theta^{l}(\tau) = 0$. If the function $t \rightsquigarrow \theta(t)$ is increasing, then $u_{i} = l_{i+1} + n^{-2/3}$. Thus we have $\|\theta - \theta^{l}\|_{\mathcal{L}^{2}[0,\tau]}^{2} \leq \|\theta^{u} - \theta^{l}\|_{\mathcal{L}^{2}[0,\tau]}^{2} \leq \|\theta^{u} - \theta^{l}\|_{\mathcal{L}^{1}[0,\tau]}^{2} \leq (\tau+1)n^{-2/3}$. The case of $t \rightsquigarrow \theta(t)$ being decreasing is also shown in the same way. \Box

Since $\Theta^n \subset \Theta$, Condition 6.1.1 is indeed satisfied for $\hat{\rho}_{\theta_0}^n = \|\cdot\|_{\mathcal{L}^2[0,\tau]}$ and $\varphi_{\theta_0}^n(\delta) = \text{const.}(\delta^{1/2} \vee \delta)$. The above proposition says that the assumption (6.1.5) is also fulfilled. Consequently, it holds for any $\hat{\theta}^n$ satisfying (6.1.6) with Θ^n given in Proposition 6.1.3 that

$$\lim_{L \to \infty} \limsup_{n \to \infty} \sup_{\theta_0 \in \Theta} P_{\theta_0}^{n*} \left(n^{1/3} \| \widehat{\theta}^n - \theta_0 \|_{\mathcal{L}^2[0,\tau]} > L \right) = 0.$$

It should be noted that grids of order $n^{-2/3}$ is sufficient to get this rate, and the discrete observation of the process $t \rightsquigarrow X_t^n$ only on the grids is enough to compute the estimator.

Example 2: Smooth functions

Let some constants $\alpha > 1/2$ and H > 0 be given. Let us consider the class $\Theta = C_H^{\alpha}([0, \tau])$ defined in (11) of Section 1.2. Recall that

$$\int_0^\delta \sqrt{\log N(C_H^\alpha([0,\tau]), \|\cdot\|_\infty; \varepsilon)} d\varepsilon \le \text{const.} \delta^{1-(1/2\alpha)}.$$

This suggests that Condition 6.1.1 should be fulfilled with $\varphi_{\theta_0}^n(\delta) = \text{const.}(\delta^{1-(1/2\alpha)} \vee \delta)$, which leads to the rate $r_n = n^{\alpha/(2\alpha+1)}$.

Choosing any grids $0 = t_0^n < t_1^n < \cdots < t_{k_n}^n = \tau$ such that $t_i^n - t_{i-1}^n \leq n^{-\alpha/(2\alpha+1)}$, define the mapping $\pi^n : \Theta \to \ell^{\infty}[0,\tau]$ by

$$\pi^n \theta(t) = \sum_{i=1}^{k_n} \frac{1_{(t_{i-1}^n, t_i^n]}(t)}{t_i^n - t_{i-1}^n} \int_{t_{i-1}^n}^{t_i^n} \theta(s) ds \qquad \forall t \in [0, \tau].$$

This mapping π^n is a special case of (4.2.5). Using also Lemma 4.2.2, we have:

$$\|\pi^{n}\theta - \pi^{n}\vartheta\|_{\mathcal{L}^{2}[0,\tau]} \leq \|\theta - \vartheta\|_{\mathcal{L}^{2}[0,\tau]} \leq \tau\|\theta - \vartheta\|_{\infty} \quad \forall \theta, \vartheta \in \Theta;$$
$$\|\theta - \pi^{n}\theta\|_{\mathcal{L}^{2}[0,\tau]} \leq \tau\|\theta - \pi^{n}\theta\|_{\infty} \leq \tau H n^{-\alpha/(2\alpha+1)} \quad \forall \theta \in \Theta.$$

Hence, if we choose a sequence of countable subsets Θ^n of $\pi^n \Theta$ such that $\pi^n \Theta \subset \bigcup_{\theta \in \Theta^n} B(\theta; Mn^{-\alpha/(2\alpha+1)})$ for a constant M > 0 not depending on n, then Condition 6.1.1 and the assumption (6.1.5) are satisfied for $U = \Theta$, $\hat{\rho}_{\theta_0}^n = \|\cdot\|_{\infty}$ and $\varphi_{\theta_0}^n(\delta) = \text{const.}(\delta^{1-(1/2\alpha)} \vee \delta)$. Thus, the assertion of Theorem 6.1.2 holds for such a sieve Θ^n , with $U = \Theta$.

Similarly to the preceding example, this result says that taking some grids of order $n^{-\alpha/(2\alpha+1)}$ is enough to get the convergence rate $r_n = n^{\alpha/(2\alpha+1)}$.

6.2 Multiplicative Intensity Model

Let (E, \mathcal{E}) be a Blackwell space on which a measure λ is defined, and let $\tau > 0$ be a fixed constant. Let us denote:

$$\mathcal{L}^p_+ = \{ f \in \mathcal{L}^p([0,\tau] \times E, \mathfrak{B}[\mathfrak{o},\tau] \otimes \mathcal{E}, \mathfrak{dt}\lambda(\mathfrak{dg})) : \mathfrak{f}(\mathfrak{t},\mathfrak{g}) \ge \mathfrak{o} \} \qquad \forall p \in [1,\infty].$$

We also denote by $\|\cdot\|_{\mathcal{L}^p}$ the L^p -seminorm on $\mathcal{L}^p([0,\tau] \times E, \mathfrak{B}[\mathfrak{o},\tau] \otimes \mathcal{E}, \mathfrak{dt}\lambda(\mathfrak{dg})).$

For every $n \in \mathbb{N}$, let μ^n be an *E*-valued multivariate point process on a filtered measurable space $(\Omega^n, \mathcal{F}^n, \mathbf{F}^n = (\mathcal{F}^n_t)_{t \in [0,\tau]})$. Let $\mathbf{P}^n = \{P^n_\theta : \theta \in \Theta\}$ be a family of probability measures on $(\Omega^n, \mathcal{F}^n)$ indexed by a subset Θ of \mathcal{L}^p_+ for some $p \in [1, \infty]$ specified later. We suppose that the predictable compensator of μ^n with respect to the probability measure P_{θ}^n is given by

$$\nu^{n,\theta}(\omega; dt, dz) = \theta(t,z)Y^n(\omega,t,z)dt\lambda(dz),$$

where Y^n is a $[0, \infty)$ -valued predictable function. It is well-known that, under some conditions, the log-likelihood ratio is given by

$$\log \frac{dP_{\theta}^{n} | \mathcal{F}_{\tau}^{n}}{dP_{\vartheta}^{n} | \mathcal{F}_{\tau}^{n}} = l^{n}(\theta) - l^{n}(\vartheta) \qquad \forall \theta, \vartheta \in \Theta,$$

where

(6.2.1)
$$l^{n}(\theta) = (\log \theta) * \mu_{\tau}^{n} - \theta * \overline{\nu}_{\tau}^{n}$$

with

(6.2.2)
$$\overline{\nu}^n(\omega; dt, dz) = Y^n(\omega, t, z) dt\lambda(dz).$$

However, as in the preceding section, we shall not use the fact that $l^n(\theta)$ above is a component of the log-likelihood ratio; it may be regarded just as a $[-\infty, \infty)$ -valued random variable defined by the right hand side of (6.2.1). As a matter of fact, in the following we will define $l^n(\theta)$ for all elements θ of \mathcal{L}^p_+ in order to discuss some "sieved" maximum likelihood estimators, again.

Condition 6.2.1 For some $p, q \in [1, \infty]$ such that (1/p) + (1/q) = 1, it holds that

$$\Theta \subset \mathcal{L}^p_+ \qquad and \qquad Y^n(\omega, \cdot, \cdot) \in \mathcal{L}^q_+ \quad \forall \omega \in \Omega^n.$$

Furthermore, for a given subset U of Θ , it holds that

$$\lim_{L\to\infty}\limsup_{n\to\infty}\sup_{\theta_0\in U}P_{\theta_0}^{n*}(\Omega^n\setminus\Omega_q^n(L))=0,$$

where

(6.2.3)
$$\Omega_q^n(L) = \left\{ \omega \in \Omega^n : \sqrt{n^{-1} \|Y^n\|_{\mathcal{L}^q}(\omega)} \le L \right\} \qquad \forall L > 0.$$

A typical case, considered by van de Geer (1995b), is p = 1 and $q = \infty$. This choice is optimal in the context of censoring models, where it indeed holds that $Y^n \leq n$. For our discussion of rate of convergence, we adopt the random Hellinger semimetric ρ^n defined by

$$\begin{split} \varrho^{n}(\theta,\vartheta) &= \sqrt{\frac{1}{n} \left| \sqrt{\theta} - \sqrt{\vartheta} \right|^{2} * \overline{\nu}_{\tau}^{n}} \\ &= \sqrt{\frac{1}{n} \int_{[0,\tau] \times E} \left| \sqrt{\theta(t,z)} - \sqrt{\vartheta(t,z)} \right|^{2} Y^{n}(t,z) dt \lambda(dz)} \qquad \forall \theta, \vartheta \in \mathcal{L}_{+}^{p}. \end{split}$$

Due to the first requirement of Condition 6.2.1, the above formula indeed defines a random semimetric. On the other hand, an entropy condition on the sieve should be given in terms of the L^{2p} -Hellinger semimetric ρ_{2p} defined by

$$\rho_{2p}(\theta,\vartheta) = \left\| \sqrt{\theta} - \sqrt{\vartheta} \right\|_{\mathcal{L}^{2p}}$$
$$= \left(\int_{[0,\tau] \times E} \left| \sqrt{\theta(t,z)} - \sqrt{\vartheta(t,z)} \right|^{2p} dt \lambda(dz) \right)^{1/2p} \quad \forall \theta, \vartheta \in \mathcal{L}^{p}_{+}$$

Then it follows from the Hölder inequality that $\rho^n \leq L\rho_{2p}$ on the set $\Omega_q^n(L)$ given by (6.2.3). Throughout this section, we denote by $B_{\rho_{2p}}(\theta;\varepsilon)$ the closed ball in \mathcal{L}_+^p with center θ and ρ_{2p} -radius ε , and by $B_{\rho^n}(\theta;\varepsilon)$ the (random) closed ball in \mathcal{L}_+^p with center θ and ρ^n -radius ε . We consider a sequence Θ^n of subsets of \mathcal{L}_+^p which satisfies the following condition.

Condition 6.2.2 Let $U \subset \Theta \subset \mathcal{L}^p_+$ be given, where $p \in [1, \infty]$. For every $n \in \mathbb{N}$ and $\theta_0 \in U$, there exist a function $\varphi_{\theta_0}^n : (0, \infty) \to (0, \infty)$ and some sets $\Theta^n(\theta_0; \delta) \subset \Theta^n \subset \mathcal{L}^p_+$ for $\delta \in (0, \infty)$ such that $\delta \rightsquigarrow \delta^{-1} \varphi_{\theta_0}^n(\delta)$ is decreasing and that

(6.2.4)
$$\int_0^{\delta} \sqrt{\log(1+N_{[]}(\Theta^n(\theta_0;\delta),\rho_{2p};\varepsilon))} d\varepsilon \le \varphi_{\theta_0}^n(\delta) \qquad \forall \delta \in (0,\infty).$$

Then, choose some positive constants r_{n,θ_0} such that $n^{-1/2}\varphi_{\theta_0}^n(r_{n,\theta_0}^{-1}) \leq r_{n,\theta_0}^{-2}$.

The subsets $\Theta^n(\theta_0; \delta)$ of Θ^n have to be chosen to satisfy not only (6.2.4) but also (6.2.6) below. It can be taken to be $\Theta^n \cap B_{\rho_{2p}}(\theta_0; \delta)$ if the random semimetric ρ^n is "asymptotically equivalent" to the semimetric ρ_{2p} (i.e., the assumption (6.2.8) below). Generally speaking, a smaller choice of $\Theta^n(\theta_0; \delta)$'s makes it easy to check the entropy condition (6.2.4), but does it difficult to check the condition (6.2.6). If we choose $\Theta^n(\theta_0; \delta) = \Theta^n$, the condition (6.2.6) is always satisfied; thus this choice is wise when it does not affect the inequality (6.2.4).

Theorem 6.2.3 Let $U \subset \Theta \subset \mathcal{L}^p_+$ be given, where $p \in [1, \infty]$. Suppose that Conditions 6.2.1 and 6.2.2 are satisfied for some $\Theta^n(\theta_0; \delta) \subset \Theta^n \subset \mathcal{L}^p_+$, and choose some constants r_{n,θ_0} described there. Suppose also that there exists a constant M > 0 such that: for every $n \in \mathbb{N}$ and $\theta_0 \in U$ there exists $\theta^n_{\theta_0} \in \Theta^n$ such that

(6.2.5) $\theta_0 \le M\theta_{\theta_0}^n \qquad and \qquad \rho_{2p}(\theta_0, \theta_{\theta_0}^n) \le Mr_{n,\theta_0}^{-1},$

and that

 $\lim_{K \to \infty} \limsup_{n \to \infty} \sup_{\theta_0 \in U} P_{\theta_0}^{n*}(\Omega^n \setminus \Omega_{\theta_0}^n(K)) = 0$

$$(6.2.6) \quad where \quad \Omega_{\theta_0}^n(K) = \left\{ R_{\theta_0}^n(\delta) \cup \{\theta_{\theta_0}^n\} \subset \Theta^n(\theta_0; K\delta) \quad \forall \delta \in [Kr_{n,\theta_0}^{-1}, \infty) \right\}$$
$$with \quad R_{\theta_0}^n(\delta) = \left\{ \theta \in \Theta^n : (\delta/2) < \varrho^n(\theta, \theta_{\theta_0}^n) \le \delta \right\}.$$

Then, for any mapping $\widehat{\theta}^n : \Omega^n \to \Theta^n$ such that

(6.2.7)
$$l^{n}(\widehat{\theta}^{n}) \geq \sup_{\theta \in \Theta^{n}} l^{n}(\theta) - nr_{n}^{-2} \quad with \quad r_{n} = \sup_{\theta_{0} \in U} r_{n,\theta_{0}},$$

it holds that

$$\lim_{L\to\infty}\limsup_{n\to\infty}\sup_{\theta_0\in U}P^n_{\theta_0}\left(r_{n,\theta_0}\varrho^n(\widehat{\theta}^n,\theta_0)>L\right)=0.$$

When $\Theta^n(\theta_0; \delta) = \Theta^n$, the assumption (6.2.6) is unnecessary. When $\Theta^n(\theta_0; \delta) = \Theta^n \cap B_{\rho_{2p}}(\theta_0; \delta)$, the assumption (6.2.6) is satisfied if

(6.2.8)
$$\lim_{K \to \infty} \limsup_{n \to \infty} \sup_{\theta_0 \in U} P_{\theta_0}^{n*} \left(\sup_{\substack{\theta, \vartheta \in \Theta^n \\ \varrho^n(\theta, \vartheta) > Kr_{n, \theta_0}^{-1}}} \frac{\rho_{2p}(\theta, \vartheta)}{\varrho^n(\theta, \vartheta)} > K \right) = 0.$$

It is trivial that the assumption (6.2.5) is not a real restriction when $\Theta^n = \Theta$; it is satisfied with M = 1 and $\theta^n_{\theta_0} = \theta_0$. Recall also the remark following Theorem 6.1.2.

To prove the above result, we shall apply Theorem 5.1.2 not to the naive criterion process $\theta \rightsquigarrow \Gamma^n(\theta) = n^{-1}l^n(\theta)$ but to the process $\theta \rightsquigarrow \Gamma^n_{\theta_0}(\theta)$ given by

(6.2.9)
$$\Gamma_{\theta_0}^n(\theta) = \frac{1}{n} \left\{ \left(\log \frac{\theta + \theta_{\theta_0}^n}{2\theta_{\theta_0}^n} \right) * \mu_\tau^n - \left(\left(\frac{\theta + \theta_{\theta_0}^n}{2\theta_{\theta_0}^n} - 1 \right) \theta_{\theta_0}^n \right) * \overline{\nu}_\tau^n \right\}$$

where $\theta_{\theta_0}^n$ is an element of Θ^n satisfying (6.2.5). Then it is natural to introduce the process $\theta \rightsquigarrow \gamma_{\theta_0}^n(\theta)$ given by

$$(6.2.10) \quad \gamma_{\theta_0}^n(\theta) = \frac{1}{n} \left\{ \left(\log \frac{\theta + \theta_{\theta_0}^n}{2\theta_{\theta_0}^n} \right) * \nu_{\tau}^{n,\theta_0} - \left(\left(\frac{\theta + \theta_{\theta_0}^n}{2\theta_{\theta_0}^n} - 1 \right) \theta_{\theta_0}^n \right) * \overline{\nu}_{\tau}^n \right\}$$
$$= \frac{1}{n} \left(\left(\log \frac{\theta + \theta_{\theta_0}^n}{2\theta_{\theta_0}^n} \right) \theta_0 - \left(\frac{\theta + \theta_{\theta_0}^n}{2\theta_{\theta_0}^n} - 1 \right) \theta_{\theta_0}^n \right) * \overline{\nu}_{\tau}^n,$$

which can be approximated by

$$\frac{1}{n} \left(\left(\log \frac{\theta + \theta_{\theta_0}^n}{2\theta_{\theta_0}^n} - \frac{\theta + \theta_{\theta_0}^n}{2\theta_{\theta_0}^n} + 1 \right) \theta_{\theta_0}^n \right) * \overline{\nu}_{\tau}^n$$

if $\theta_{\theta_0}^n$ is "close" to θ_0 . We should have noticed in advance that

(6.2.11)
$$\frac{\theta + \theta_{\theta_0}^n}{2\theta_{\theta_0}^n}$$
 should be read as $\frac{(\theta + \theta_{\theta_0}^n)\mathbf{1}_{\{\theta_{\theta_0}^n > 0\}}}{2\theta_{\theta_0}^n}$ (where $0/0 = 1$).

However, it follows from the assumption (6.2.5) that $\{\theta_{\theta_0}^n = 0\} \subset \{\theta_0 = 0\}$, and we also have $P_{\theta_0}^n(\mu^n(\{\theta_0 = 0\}) = 0) = 1$. These facts allow us to adopt the notational convention (6.2.11). The merit of these "modified" processes is that

(6.2.12)
$$\frac{\theta + \theta_{\theta_0}^n}{2\theta_{\theta_0}^n} \ge \frac{1}{2}$$

To justify the "modification", we should first see the following.

Lemma 6.2.4 Under the first requirement of (6.2.5), for any mapping $\widehat{\theta}^n : \Omega^n \to \Theta^n$ satisfying (6.2.7), it holds that

$$\Gamma^n_{\theta_0}(\widehat{\theta}^n) \ge \Gamma^n_{\theta_0}(\theta^n_{\theta_0}) - \frac{1}{2}r_n^{-2} \qquad P^n_{\theta_0} \text{-almost surely.}$$

Proof. Since the function $x \rightarrow \log(x)$ is concave, it holds that

$$\begin{split} n\Gamma_{\theta_0}^n(\widehat{\theta}^n) &= \left(\log\frac{\widehat{\theta}^n + \theta_{\theta_0}^n}{2\theta_{\theta_0}^n}\right) * \mu_{\tau}^n - \frac{1}{2}(\widehat{\theta}^n - \theta_{\theta_0}^n) * \overline{\nu}_{\tau}^n \\ &\geq \frac{1}{2}\left(\log\frac{\widehat{\theta}^n}{\theta_{\theta_0}^n}\right) * \mu_{\tau}^n - \frac{1}{2}(\widehat{\theta}^n - \theta_{\theta_0}^n) * \overline{\nu}_{\tau}^n \quad \text{ on the set } \{\mu^n(\{\theta_{\theta_0}^n = 0\}) = 0\} \\ &\geq \frac{1}{2}(l^n(\widehat{\theta}^n) - l^n(\theta_{\theta_0}^n)) \\ &\geq -\frac{1}{2}nr_n^{-2}. \end{split}$$

Since $\Gamma_{\theta_0}^n(\theta_{\theta_0}^n) = 0$, and since $P_{\theta_0}^n(\mu^n(\{\theta_{\theta_0}^n = 0\}) = 0) \ge P_{\theta_0}^n(\mu^n(\{\theta_0 = 0\}) = 0) = 1$, we obtain the assertion.

For computation of the Hellinger semimetric, we will use the following inequalities:

(6.2.13)
$$\frac{1}{2}\left|\sqrt{x} - \sqrt{y}\right| \le \left|\sqrt{\frac{x+y}{2}} - \sqrt{y}\right| \le \left|\sqrt{x} - \sqrt{y}\right| \qquad \forall x, y \in [0,\infty)$$

(6.2.14) $\left|\sqrt{x+a} - \sqrt{y+a}\right| \le \left|\sqrt{x} - \sqrt{y}\right| \qquad \forall x, y, a \in [0,\infty).$

Proof of Theorem 6.2.3. We will apply Theorem 5.1.2. Formulation 5.1.1 should be read as follows: for every $\theta_0 \in U(=U^n)$:

(i) the random semimetric space (Θ^n, ϱ^n) and the point $\theta^n_{\theta_0}$ satisfying (6.2.5) and (6.2.6);

(ii) the stochastic processes $\theta \rightsquigarrow \Gamma^n_{\theta_0}(\theta)$ and $\theta \rightsquigarrow \gamma^n_{\theta_0}(\theta)$, with parameters in Θ^n , defined by (6.2.9) and (6.2.10), respectively.

As in (6.2.6), we denote $R_{\theta_0}^n(\delta) = \left\{ \theta \in \Theta^n : (\delta/2) < \varrho^n(\theta, \theta_{\theta_0}^n) \le \delta \right\}$ for every $\delta \in (0, \infty)$. Notice that $\Gamma_{\theta_0}^n(\theta_{\theta_0}^n) = \gamma_{\theta_0}^n(\theta_{\theta_0}^n) = 0$.

To show the FIRST INEQUALITY of *M*-CRITERION, let us write $\gamma_{\theta_0}^n(\theta) = (I) + (II)$, where:

$$(I) = \frac{1}{n} \left(\left(\log \frac{\theta + \theta_{\theta_0}^n}{2\theta_{\theta_0}^n} - \frac{\theta + \theta_{\theta_0}^n}{2\theta_{\theta_0}^n} + 1 \right) \theta_{\theta_0}^n \right) * \overline{\nu}_{\tau}^n;$$

$$(II) = \frac{1}{n} \left(\left(\log \frac{\theta + \theta_{\theta_0}^n}{2\theta_{\theta_0}^n} \right) (\theta_0 - \theta_{\theta_0}^n) \right) * \overline{\nu}_{\tau}^n.$$

Since $\log x - x + 1 \le -|\sqrt{x} - 1|^2$ for all x > 0, we have

$$(I) \leq -\frac{1}{n} \left(\left| \sqrt{\frac{\theta + \theta_{\theta_0}^n}{2\theta_{\theta_0}^n}} - 1 \right|^2 \theta_{\theta_0}^n \right) * \overline{\nu}_{\tau}^n$$
$$= -\left| \varrho^n \left(\frac{\theta + \theta_{\theta_0}^n}{2}, \theta_{\theta_0}^n \right) \right|^2$$
$$\leq -\frac{1}{4} \left| \varrho^n(\theta, \theta_{\theta_0}^n) \right|^2 \quad \text{by (6.2.13)}$$
$$\leq -\frac{1}{16} \delta^2 \quad \text{whenever } \theta \in R_{\theta_0}^n(\delta).$$

On the other hand, since $|\log x| \le c_2 |\sqrt{x} - 1|$ for all $x \ge 1/2$ where $c_2 = (2 + \sqrt{2}) \log 2$, it follows from (6.2.12) that

$$\begin{aligned} |(II)| &\leq \frac{c_2}{n} \left(\left| \sqrt{\frac{\theta + \theta_{\theta_0}^n}{2\theta_{\theta_0}^n}} - 1 \right| \left| \sqrt{\theta_{\theta_0}^n} + \sqrt{\theta_0} \right| \left| \sqrt{\theta_{\theta_0}^n} - \sqrt{\theta_0} \right| \right) * \overline{\nu}_{\tau}^n \\ &\leq \frac{c_2(1 + \sqrt{M})}{n} \left(\left| \sqrt{\frac{\theta + \theta_{\theta_0}^n}{2\theta_{\theta_0}^n}} - 1 \right| \sqrt{\theta_{\theta_0}^n} \cdot \left| \sqrt{\theta_{\theta_0}^n} - \sqrt{\theta_0} \right| \right) * \overline{\nu}_{\tau}^n \\ &\leq c_2(1 + \sqrt{M}) \varrho^n \left(\frac{\theta + \theta_{\theta_0}^n}{2}, \theta_{\theta_0}^n \right) \varrho^n (\theta_{\theta_0}^n, \theta_0) \\ &\leq c_2(1 + \sqrt{M}) \varrho^n (\theta, \theta_{\theta_0}^n) \varrho^n (\theta_{\theta_0}^n, \theta_0) \quad \text{by } (6.2.13) \\ &\leq c_2(1 + \sqrt{M}) \cdot \delta \cdot Mr_{n,\theta_0}^{-1} \quad \text{whenever } \theta \in R_{\theta_0}^n(\delta). \end{aligned}$$

Thus, we have

$$\gamma_{\theta_0}^n(\theta) - \gamma_{\theta_0}^n(\theta_{\theta_0}^n) \le -\frac{1}{32}\delta^2 \qquad \forall \theta \in R_{\theta_0}^n(\delta)$$

whenever $\delta \ge Kr_{n,\theta_0}^{-1}$ with any $K \ge 32 \cdot c_2(1+\sqrt{M})M$.

To show the SECOND INEQUALITY, observe that

$$(\Gamma_{\theta_0}^n - \gamma_{\theta_0}^n)(\theta) - (\Gamma_{\theta_0}^n - \gamma_{\theta_0}^n)(\theta_{\theta_0}^n) = \frac{1}{n} \cdot W^{n,\theta} * (\mu^n - \nu^{n,\theta_0})_{\tau} \qquad \forall \theta \in \Theta^n,$$

where $W^{n,\theta}$'s are defined by

$$W^{n,\theta} = \log\left(\frac{\theta + \theta_{\theta_0}^n}{2\theta_{\theta_0}^n}\right) \qquad \forall \theta \in \mathcal{L}^p_+$$

(we have extended the parameter space Θ^n to \mathcal{L}^p_+ in the latter). For a given $\epsilon > 0$, choose some $L_{\epsilon}, K_{\epsilon} \geq 1$ and $n_{\epsilon} \in \mathbb{N}$ such that $P^n_{\theta_0}(\Omega^n \setminus \Omega^n_q(L_{\epsilon}) \cap \Omega^n_{\theta_0}(K_{\epsilon})) \leq \epsilon$ holds for all $n \geq n_{\epsilon}$, where the sets $\Omega^n_q(L_{\epsilon})$ and $\Omega^n_{\theta_0}(K_{\epsilon})$ are given by (6.2.3) and (6.2.6), respectively. For every $n \geq n_{\epsilon}$ and $\theta_0 \in U$, fix any $\delta \in [r^{-1}_{n,\theta_0}, \infty)$, and we will apply Theorem 2.2.8 to the class $\mathcal{W} = \{W^{n,\theta} : \theta \in \Theta^n(\theta_0; K_{\epsilon}\delta)\}$ of predictable functions with a = 1. Indeed, since $\nu^n(\omega; [0, \tau] \times E) < \infty$ for all $\omega \in \Omega^n$, we have that $\exp(|\overline{W}|) * \nu_{\tau}^n(\omega) < \infty$ by using also (6.2.5). For every $\varepsilon > 0$, choose (ρ_{2p}, ε) -brackets $[l^{\varepsilon,k}, u^{\varepsilon,k}]$, $k = 1, ..., N_{[]}(\Theta^n(\theta_0; K_{\epsilon}\delta), \rho_{2p}; \varepsilon)$, in \mathcal{L}^p_+ , which cover the class $\Theta^n(\theta_0; K_{\epsilon}\delta)$. Construct a NFP Π of $\Theta^n(\theta_0; K_{\epsilon}\delta)$ from this series of brackets. We have that

$$|W^{n,\theta} - W^{n,\vartheta}| \le W^{n,u^{\varepsilon,k}} - W^{n,l^{\varepsilon,k}} \quad \text{if } l^{\varepsilon,k} \le \theta, \vartheta \le u^{\varepsilon,k}$$

and that, since $0 \le 2(x - 1 - \log x) \le |x - 1|^2$ for all $x \ge 1$,

$$\begin{split} \frac{1}{4} \left(\mathcal{E}_{2}(|W^{n,u^{\varepsilon,k}} - W^{n,l^{\varepsilon,k}}|) \right) * \nu_{\tau}^{n,\theta_{0}} \\ &= 2 \left(\exp\left(\frac{1}{2}|W^{n,u^{\varepsilon,k}} - W^{n,l^{\varepsilon,k}}|\right) - 1 - \frac{1}{2}|W^{n,u^{\varepsilon,k}} - W^{n,l^{\varepsilon,k}}| \right) * \nu_{\tau}^{n,\theta_{0}} \\ &= 2 \left(\left(\sqrt{\frac{u^{\varepsilon,k} + \theta_{\theta_{0}}^{n}}{l^{\varepsilon,k} + \theta_{\theta_{0}}^{n}}} - 1 - \log\sqrt{\frac{u^{\varepsilon,k} + \theta_{\theta_{0}}^{n}}{l^{\varepsilon,k} + \theta_{\theta_{0}}^{n}}} \right) \theta_{0} \right) * \overline{\nu}_{\tau}^{n} \\ &\leq 2M \left(\left(\sqrt{\frac{u^{\varepsilon,k} + \theta_{\theta_{0}}^{n}}{l^{\varepsilon,k} + \theta_{\theta_{0}}^{n}}} - 1 - \log\sqrt{\frac{u^{\varepsilon,k} + \theta_{\theta_{0}}^{n}}{l^{\varepsilon,k} + \theta_{\theta_{0}}^{n}}} \right) \theta_{\theta_{0}}^{n} \right) * \overline{\nu}_{\tau}^{n} \\ &\leq M \left(\left| \sqrt{\frac{u^{\varepsilon,k} + \theta_{\theta_{0}}^{n}}{l^{\varepsilon,k} + \theta_{\theta_{0}}^{n}}} - 1 \right|^{2} \theta_{\theta_{0}}^{n} \right) * \overline{\nu}_{\tau}^{n} \\ &\leq M \left(\left| \sqrt{u^{\varepsilon,k} + \theta_{\theta_{0}}^{n}} - \sqrt{l^{\varepsilon,k} + \theta_{\theta_{0}}^{n}} \right|^{2} \right) * \overline{\nu}_{\tau}^{n} \\ &\leq M n |\varrho^{n} (u^{\varepsilon,k} + \theta_{\theta_{0}}^{n}, l^{\varepsilon,k} + \theta_{\theta_{0}}^{n})|^{2} \\ &\leq M n |\varrho^{n} (u^{\varepsilon,k}, l^{\varepsilon,k})|^{2} \qquad \text{by } (6.2.14) \\ &\leq L_{\epsilon}^{2} M n |\rho_{2p} (u^{\varepsilon,k}, l^{\varepsilon,k})|^{2} \qquad \text{on the set } \Omega_{q}^{n} (L_{\epsilon}) \end{aligned}$$

where the set $\Omega_q^n(L_{\epsilon})$ is given by (6.2.3). Thus we obtain $\|\mathcal{W}\|_{\Pi,\tau}^{\mathcal{E}_2} \leq \sqrt{4Mn}L_{\epsilon}$ on the set $\Omega_q^n(L_{\epsilon})$. Likewise, it holds that

$$\sqrt{\mathcal{E}_2(|W^{n,\theta} - W^{n,\vartheta}|) * \nu_{\tau}^{\theta_0}} \le \sqrt{4Mn}\varrho^n(\theta,\vartheta) \qquad \forall \theta, \vartheta \in \Theta^n;$$

this can be shown by computing on the sets $\{(t,z) : \theta(t,z) \leq \vartheta(t,z)\}$ and $\{(t,z) : \theta(t,z) > \vartheta(t,z)\}$ separately. This suggests that we should apply Theorem 2.2.8 to the random metric $\varrho = \sqrt{4Mn}\varrho^n$. Notice that $\|\sqrt{4Mn}\varrho^n\|_{\Pi} \leq \sqrt{4Mn}L_{\epsilon}$ on the set $\Omega_q^n(L_{\epsilon})$. Hence, applying the theorem to $K = \sqrt{4Mn}L_{\epsilon}K_{\epsilon}$ and $\delta > 0$ we obtain

(6.2.15)
$$E_{\theta_0}^n \sup_{\substack{\theta,\vartheta \in \Theta^n(\theta_0:K_\epsilon\delta)\\ q^n(\theta,\vartheta) \le \delta}} |(W^{n,\theta} - W^{n,\vartheta}) * (\mu^n - \nu^{n,\theta_0})_\tau| \mathbf{1}_{\Omega^n_q(L_\epsilon)}$$

6. Non-parametric Maximum Likelihood Estimators

$$\lesssim \sqrt{4Mn} L_{\epsilon} K_{\epsilon} \int_{0}^{\delta} \sqrt{\log(1 + N_{[]}(\Theta^{n}(\theta_{0}; K_{\epsilon}\delta), \rho; \varepsilon))} d\varepsilon$$

$$\leq \sqrt{4Mn} L_{\epsilon} K_{\epsilon} \varphi^{n}_{\theta_{0}}(K_{\epsilon}\delta) \leq \sqrt{4Mn} L_{\epsilon} K^{2}_{\epsilon} \varphi^{n}_{\theta_{0}}(\delta)$$

whenever δ satisfies the restriction that $\varphi_{\theta_0}^n(K_\epsilon\delta) \leq \sqrt{4MnL_\epsilon K_\epsilon} \cdot \delta^2$; assuming $M \geq 1/4$ without loss of generality, this restriction is weaker than $n^{-1/2}\varphi_{\theta_0}^n(\delta) \leq \delta^2$. Thus the above inequality holds for any $\delta \in [r_n^{-1}, \infty)$. Now, recall that $R_{\theta_0}^n(\delta) \cup \{\theta_{\theta_0}^n\} \subset \Theta^n(\theta_0; K_\epsilon\delta)$ for all $\delta \in [K_\epsilon r_{n,\theta_0}^{-1}, \infty)$ on the set $\Omega_{\theta_0}^n(K_\epsilon)$ given by (6.2.6). Multiplying the both sides of (6.2.15) by n^{-1} , we have that

$$E_{\theta_0}^n \sup_{\theta \in R_{\theta_0}^n(\delta)} |(\Gamma_{\theta_0}^n - \gamma_{\theta_0}^n)(\theta) - (\Gamma_{\theta_0}^n - \gamma_{\theta_0}^n)(\theta_{\theta_{\theta_0}}^n)| 1_{\Omega^n(L_{\epsilon}) \cap \Omega_{\theta_0}^n(K_{\epsilon})} \lesssim \sqrt{4M} L_{\epsilon} K_{\epsilon}^2 n^{-1/2} \varphi_{\theta_0}^n(\delta)$$

for any $\delta \in [K_{\epsilon}r_n^{-1}, \infty)$. The SECOND INEQUALITY has been established.

6.3 Counting Processes with Non-linear Covariates

Let (E, \mathcal{E}) be a Blackwell space on which a σ -finite measure λ is defined; this is the state space of covariate processes in the following model. Let $\tau > 0$ be a constant. We define the notations \mathcal{L}^p_+ and $\|\cdot\|_{\mathcal{L}^p}$ in the same way as the first paragraph of Section 6.2.

In the *n*-th statistical experiment, we consider k_n adapted point processes on $[0, \tau]$, namely $N^{n,i}$, $(i = 1, ..., k_n)$, defined on a filtered measurable space $(\Omega^n, \mathcal{F}^n, \mathbf{F}^n = (\mathcal{F}_t^n)_{t \in [0,\tau]})$; we then denote $T_j^{n,i} = \inf\{t \in [0,\tau] : N_t^{n,i} = j\}$ for every $j \in \mathbb{N}$ (see page 34 of Jacod and Shiryaev (1987)). Let $\mathbf{P}^n = \{P_{\theta}^n : \theta \in \Theta\}$ be a family of probability measures on $(\Omega^n, \mathcal{F}^n)$ indexed by a subset Θ of \mathcal{L}_+^p for some $p \in [1,\infty]$ specified later. Suppose that the predictable compensator of $N^{n,i}$ with respect to the probability measure P_{θ}^n is given by

$$\theta(t, Z_t^{n,i})Y_t^{n,i}dt,$$

where $t \rightsquigarrow Z_t^{n,i}$ is an *E*-valued predictable process and $t \rightsquigarrow Y_t^{n,i}$ is a $[0,\infty)$ -valued predictable process. It is implicitly assumed that $\int_0^\tau \theta(t, Z_t^{n,i}(\omega))Y_t^{n,i}(\omega)dt < \infty$ for every $\omega \in \Omega^n$. Suppose that we can observe the processes $N^{n,i}$, $Y^{n,i}$ and $Z^{n,i}$ on the random sets $\{t \in [0,\tau] : Y_t^{n,i}(\omega) > 0\}$. The goal of this section is to derive the rate of convergence of sieved maximum likelihood estimators for θ .

We analyze this problem by using the E-valued multivariate point processes

$$\mu^{n}(dt, dz) = \sum_{i=1}^{k_{n}} \sum_{j} \varepsilon_{(T_{j}^{n,i}, Z_{T_{j}^{n,i}}^{n,i})}(dt, dz).$$

100
Here, we suppose that $T_j^{n,i} \neq T_{j'}^{n,i'}$ whenever $i \neq i'$; then all requirement for μ^n to be an E-valued multivariate point process, including that $\mu^n(\{t\} \times E) \leq 1$, are indeed satisfied (see Definition III.1.23 of Jacod and Shiryaev (1987)). The predictable compensator of μ^n with respect to the probability measure P_{θ}^n is given by

$$\nu^{n,\theta}(dt,dz) = \theta(t,z)\overline{\nu}^n(dt,dz)$$

where

(6.3.1)
$$\overline{\nu}^n(dt,dz) = \sum_{i=1}^{k_n} Y_t^{n,i} \varepsilon_{Z_t^{n,i}}(dz) dt$$

Under some conditions, the log-likelihood ratio is given by

$$\log \frac{dP_{\theta}^{n} | \mathcal{F}_{\tau}^{n}}{dP_{\vartheta}^{n} | \mathcal{F}_{\tau}^{n}} = l^{n}(\theta) - l^{n}(\vartheta) \qquad \forall \theta, \vartheta \in \Theta.$$

where

$$l^{n}(\theta) = (\log \theta) * \mu_{\tau}^{n} - \theta * \overline{\nu}_{\tau}^{n},$$

although we shall not require any property of the log-likelihood ratio.

For our discussion of rate of convergence, we adopt the random Hellinger "semimetric" ρ^n which is "formally" defined by

(6.3.2)
$$\varrho^{n}(\theta,\vartheta) = \sqrt{\frac{1}{n} \left| \sqrt{\theta} - \sqrt{\vartheta} \right|^{2} * \overline{\nu}_{\tau}^{n}} \\ = \sqrt{\frac{1}{n} \sum_{i=1}^{k_{n}} \int_{0}^{\tau} \left| \sqrt{\theta(t,Z_{t}^{n,i})} - \sqrt{\vartheta(t,Z_{t}^{n,i})} \right|^{2} Y_{t}^{n,i} dt}$$

for every $\theta, \vartheta \in \mathcal{L}^p_+$. The meaning of the quotation marks is that $\varrho^n(\theta, \vartheta) < \infty$ may not hold, although it has been implicitly assumed at least for $\theta, \vartheta \in \Theta$. On the other hand, an entropy condition on the sieve should be given in terms of the L^{2p} -Hellinger semimetric ρ_{2p} defined by

$$\rho_{2p}(\theta,\vartheta) = \left\| \sqrt{\theta} - \sqrt{\vartheta} \right\|_{\mathcal{L}^{2p}}$$
$$= \left(\int_{[0,\tau] \times E} \left| \sqrt{\theta(t,z)} - \sqrt{\vartheta(t,z)} \right|^{2p} dt \lambda(dz) \right)^{1/2p} \quad \forall \theta, \vartheta \in \mathcal{L}^{2p}_{+}.$$

A main difficulty in this model is that the random measure $\overline{\nu}^n(dt, dz)$ on $[0, \tau] \times E$ defined by (6.3.1) is not dominated by the measure $dt\lambda(dz)$; compare (6.2.2) and (6.3.1). Hence, in the present situation, an entropy condition in terms of ρ_{2p} is not directly translated into that in terms of ϱ^n as in the multiplicative intensity model where the relation $\varrho^n \leq L\rho_{2p}$ holds on the set $\Omega_q^n(L)$ given by (6.2.3). To solve this problem, we will take the same approach as Sections 4.2 and 4.3. Let $E = \bigcup_m E_m^n$ be a partition of \mathcal{E} measurable sets, which is at most countable, such that $\lambda(E_m^n) \in (0, \infty)$. Set $\mathcal{E}^n = \sigma\{E_m^n : m = 1, 2, ...\}$, and denote by $\mathcal{L}_+^{p,n}$ the space of elements f of \mathcal{L}_+^p that are $\mathfrak{B}[\mathfrak{o}, \tau] \otimes \mathcal{E}^n$ measurable; it is trivial that $\mathcal{L}_+^{p,n} \subset \mathcal{L}_+^p$. We define the predictable function Y^n on $\Omega^n \times [0, \tau] \times E$ by

(6.3.3)
$$Y^{n}(\omega, t, z) = \sum_{m} \sum_{i=1}^{k_{n}} Y^{n,i}_{t}(\omega) \mathbb{1}_{\{Z^{n,i}_{t}(\omega) \in E^{n}_{m}\}} \frac{\mathbb{1}_{\{z \in E^{n}_{m}\}}}{\lambda(E^{n}_{m})}$$

(do not confuse this $Y^n(\omega, t, z)$ and the original $Y_t^{n,i}(\omega)$'s). Then, it holds for any $f \in \mathcal{L}^{p,n}_+$ that

(6.3.4)
$$f * \overline{\nu}_{\tau}^{n} = \int_{[0,\tau] \times E} f(t,z) Y^{n}(\cdot,t,z) dt \lambda(dz)$$

if the integral is finite. We thus introduce the following condition.

Condition 6.3.1 For some $p, q \in [1, \infty]$ such that (1/p) + (1/q) = 1, it holds that

(6.3.5) $\Theta \subset \mathcal{L}^p_+$ and $Y^n(\omega, \cdot, \cdot) \in \mathcal{L}^{q,n}_+ \quad \forall \omega \in \Omega^n.$

Furthermore, for a given subset U of Θ , it holds that

$$\lim_{L\to\infty}\limsup_{n\to\infty}\sup_{\theta_0\in U}P^{n*}_{\theta_0}(\Omega^n\setminus\Omega^n_q(L))=0,$$

where

(6.3.6)
$$\Omega_q^n(L) = \left\{ \omega \in \Omega^n : \sqrt{n^{-1} \|Y^n\|_{\mathcal{L}^q}} \le L \right\} \qquad \forall L > 0$$

and where Y^n is defined by (6.3.3).

Under (6.3.5), the equation (6.3.4) does hold for all $f \in \mathcal{L}^{p,n}_+$ by the Hölder inequality, and thus the same relation as (6.2.2) is fulfilled on the smaller σ -field $\mathfrak{B}[\mathfrak{o},\tau] \otimes \mathcal{E}^{\mathfrak{n}}$. In this case, the formula (6.3.2) indeed defines a random semimetric on $\mathcal{L}^{p,n}_+ \cup \Theta$. In particular, we have that

$$\varrho^{n}(\theta,\vartheta) = \sqrt{\frac{1}{n} \int_{[0,\tau] \times E} \left| \sqrt{\theta(t,z)} - \sqrt{\vartheta(t,z)} \right|^{2} Y^{n}(t,z) dt \lambda(dz)} \qquad \forall \theta, \vartheta \in \mathcal{L}^{p,r}_{+}$$

and that

 $\varrho^{n}(\theta,\vartheta) \leq L\rho_{2p}(\theta,\vartheta) \qquad \forall \theta,\vartheta \in \mathcal{L}^{p,n}_{+} \quad \text{on the set } \Omega^{n}_{q}(L).$

Hence it would be convenient for computation of entropy to construct a sieve $\{\Theta^n\}$ of subsets of $\mathcal{L}^{p,n}_+$'s rather than \mathcal{L}^p_+ . To do it, we introduce the mapping $\pi^n : \mathcal{L}^p_+ \to \mathcal{L}^{p,n}_+$ defined by

$$\pi^n \theta(t,z) = \sum_m \frac{1}{\lambda(E_m^n)^2} \left| \int_{E_m^n} \sqrt{\theta(t,w)} \lambda(dw) \right|^2 \mathbb{1}_{\{z \in E_m^n\}}$$

(notice that this is different from that of Section 4.2). Then we have the following:

Lemma 6.3.2 (i) If $\theta \leq \vartheta$ where $\theta, \vartheta \in \mathcal{L}^p_+$, then $\pi^n \theta \leq \pi^n \vartheta$. (ii) If f is a $[0,\infty)$ -valued $\mathfrak{B}[\mathfrak{o},\tau] \otimes \mathcal{E}^n$ -measurable function, then for every $\theta, \vartheta \in \mathcal{L}^p_+$

$$\begin{split} &\int_{[0,\tau]\times E} \left| \sqrt{\pi^n \theta(t,z)} - \sqrt{\pi^n \vartheta(t,z)} \right|^{2p} f(t,z) dt \lambda(dz) \\ &\leq \int_{[0,\tau]\times E} \left| \sqrt{\theta(t,z)} - \sqrt{\vartheta(t,z)} \right|^{2p} f(t,z) dt \lambda(dz), \end{split}$$

provided the integrals are finite.

Proof. The assertion (i) is trivial, and (ii) follows from that

$$\begin{split} \left| \sqrt{\pi^n \theta(t,z)} - \sqrt{\pi^n \vartheta(t,z)} \right|^{2p} \\ &= \sum_m \frac{1}{\lambda(E_m^n)^{2p}} \left| \int_{E_m^n} \left(\sqrt{\theta(t,w)} - \sqrt{\vartheta(t,w)} \right) \lambda(dw) \right|^{2p} \mathbf{1}_{\{z \in E_m^n\}} \\ &\leq \sum_m \frac{1}{\lambda(E_m^n)} \int_{E_m^n} \left| \sqrt{\theta(t,w)} - \sqrt{\vartheta(t,w)} \right|^{2p} \lambda(dw) \mathbf{1}_{\{z \in E_m^n\}}. \end{split}$$

Consequently, we obtain that: if we choose (ρ_{2p}, ε) -brackets in \mathcal{L}^p_+ , namely $[l^{\varepsilon,k}, u^{\varepsilon,k}]$'s, which cover the class Θ , then it holds on the set $\Omega^n_q(L)$ that $[\pi^n l^{\varepsilon,k}, \pi^n u^{\varepsilon,k}]$'s form an $(\varrho^n, L\varepsilon)$ -brackets in $\mathcal{L}^{p,n}_+$ which cover the class $\pi^n \Theta$. This allows us to make an entropy condition with respect to the non-random semimetric ρ_{2p} rather than ϱ^n . Hereafter, we denote by $B_{\rho_{2p}}(\theta; \varepsilon)$ the closed ball in \mathcal{L}^p_+ with center θ and ρ_{2p} -radius ε .

Condition 6.3.3 There exists a function $\varphi : (0, \infty) \to (0, \infty)$ such that $\delta \rightsquigarrow \delta^{-1} \varphi(\delta)$ is decreasing and that

(6.3.7)
$$\int_0^{\delta} \sqrt{\log(1+N_{[}](\Theta,\rho_{2p};\varepsilon))} d\varepsilon \le \varphi(\delta) \qquad \forall \delta \in (0,\infty).$$

Then, choose some positive constants r_n such that $n^{-1/2}\varphi(r_n^{-1}) \leq r_n^{-2}$.

Although we have stated the version corresponding to the case $\Theta^n(\theta_0; \delta) = \Theta^n$ of Condition 6.2.2 only, it is also possible to replace Θ of the entropy assumption (6.3.7) by the local ball $\Theta \cap B_{\rho_{2p}}(\theta_0; \delta)$ when we can show an "asymptotic equivalence" of ϱ^n and ρ_{2p} (i.e., (6.2.8)).

Theorem 6.3.4 Let $U \subset \Theta \subset \mathcal{L}^p_+$ for some $p \in [1, \infty]$. Assume Conditions 6.3.1 and 6.3.3, and choose a sequence r_n described there. For a given sequence of subsets

 $\Theta^n \subset \pi^n \Theta$, suppose that there exists a constant M > 0 such that: for every $n \in \mathbb{N}$ and $\theta_0 \in U$ there exists $\theta_{\theta_0}^n \in \Theta^n$ such that

(6.3.8)
$$\theta_0 \le M\theta_{\theta_0}^n \qquad and \qquad \rho_{2p}(\theta_0, \theta_{\theta_0}^n) \le Mr_n^{-1}.$$

Then, for any mapping $\widehat{\theta}^n : \Omega^n \to \Theta^n$ such that

$$l^{n}(\widehat{\theta}^{n}) \geq \sup_{\theta \in \Theta^{n}} l^{n}(\theta) - nr_{n}^{-2},$$

it holds that

$$\lim_{L \to \infty} \limsup_{n \to \infty} \sup_{\theta_0 \in U} P_{\theta_0}^n \left(r_n \varrho^n(\widehat{\theta}^n, \theta_0) > L \right) = 0.$$

This result can be proved exactly in the same way as Theorem 6.2.3 by means of Lemma 6.2.4; those proofs can be read with little change of notation (notice that " $\theta \in \Theta^n$ " there should be read as " $\pi^n \theta \in \Theta^n$ " here).

The next problem we should consider is how to check the assumption (6.3.8) for the sieve Θ^n given as a subset of $\pi^n \Theta$. When Θ is a class of "smooth" functions, the assumption is always satisfied if we use a slightly different sieve; a part of the idea has already appeared in Example 2 of Section 6.1. Define the mapping $\pi^n_+ : \mathcal{L}^p_+ \to \mathcal{L}^{p,n}_+$ by

(6.3.9)
$$\pi^{n}_{+}\theta(t,z) = \sum_{m} \frac{1}{\lambda(E^{n}_{m})^{2}} \left| \int_{E^{n}_{m}} \sqrt{\theta(t,w) + r_{n}^{-1}} \lambda(dw) \right|^{2} \mathbb{1}_{\{z \in E^{n}_{m}\}}.$$

It is easy to show the same facts as Lemma 6.3.2 with π^n replaced by π^n_+ . Thus we have:

Corollary 6.3.5 Let $U \subset \Theta \subset \mathcal{L}^p_+$. Assume Conditions 6.3.1 and 6.3.3, and choose a sequence r_n described there. Define $\Theta^n = \pi^n_+ \Theta$. If there exists a constant M > 0 such that

$$\sup_{t\in[0,\tau]} \sup_{m} \left| \sup_{z\in E_m^n} \sqrt{\theta_0(t,z)} - \inf_{z\in E_m^n} \sqrt{\theta_0(t,z)} \right| \le Mr_n^{-1} \qquad \forall \theta_0 \in U, \ \forall n \in \mathbb{N},$$

and if $\lambda(E) < \infty$, then the same conclusion as Theorem 6.3.4 holds.

Proof. Notice that: if we choose (ρ_{2p}, ε) -brackets in \mathcal{L}^p_+ which cover the class Θ , namely $[l^{\varepsilon,k}, u^{\varepsilon,k}]$'s, then it holds on the set $\Omega^n_q(L)$ that $[\pi^n_+ l^{\varepsilon,k} - r_n^{-1}, \pi^n_+ u^{\varepsilon,k} - r_n^{-1}]$'s form $(\varrho^n, L\varepsilon)$ -brackets in $\mathcal{L}^{p,n}_+$ which cover $\Theta^n = \pi^n_+ \Theta$. Hence it suffices to prove that (6.3.8) is satisfied for the present sieve $\Theta^n = \pi^n_+ \Theta$; we will show it for $\theta^n_{\theta_0} = \pi^n_+ \theta_0$. First observe that

$$\sqrt{\frac{\theta_0(t,z)}{\pi_+^n \theta_0(t,z)}} \le \frac{\inf_{z \in E_m^n} \sqrt{\theta_0(t,z)} + Mr_n^{-1}}{\inf_{z \in E_m^n} \sqrt{\theta_0(t,z)} + r_n^{-1}} \le 1 + M.$$

Next, since

$$\inf_{z \in E_m^n} \sqrt{\theta_0(t,z)} \le \sqrt{\pi_+^n \theta_0(t,z)} - r_n^{-1} \le \sup_{z \in E_m^n} \sqrt{\theta_0(t,z)}$$

we have

$$\sup_{(t,z)\in[0,\tau]\times E} \left| \sqrt{\theta_0(t,z)} - \sqrt{\pi_+^n \theta_0(t,z)} \right| \le (1+M)r_n^{-1}$$

Thus it holds that $\rho_{2p}(\theta_0, \pi_+^n \theta_0) \le (1+M)\tau\lambda(E) \cdot r_n^{-1}$.

Example 1: Smooth functions

Let us take $E = [0,1]^d$ endowed with the Lebesgue measure. Let $\alpha > (d+1)/2$ and H > 0, and we consider the class $C_H^{\alpha} = C_H^{\alpha}([0,\tau] \times [0,1]^d)$ given in (11) of Section 1.2. We set $\Theta = \{\theta \in \mathcal{L}_+^{\infty} : \sqrt{\theta} \in C_H^{\alpha}\}$. Then, Condition 6.3.3 is satisfied with $U = \Theta$ and $\varphi(\delta) = \text{const.}(\delta^{1-(d+1)/2\alpha} \vee \delta)$; thus we set $r_n = n^{\alpha/(2\alpha+d+1)}$. Let $[0,1]^d = \bigcup_m E_m^n$ be a partition of Borel measurable subsets of $[0,1]^d$ such that $\text{Diam}(E_m^n) \leq r_n^{-1}$. Then, the displayed assumption of Corollary 6.3.5 is satisfied also with $U = \Theta$. Hence, we can get the conclusion of Theorem 6.3.4 if Condition 6.3.1 for $p = \infty$ and q = 1 is satisfied. A sufficient condition for this is that $k_n = n$ and the processes $Y^{n,i}$ take values only in a bounded set [0, K]. As in the next example, the state space of the process $Y^{n,i}$ is typically $\{0, 1\}$ in the context of survival analysis, hence the result above is always valid.

Example 2: Lexis diagram

Let us discuss the Lexis diagram which is an important method describing models in survival analysis (see Keiding (1990) or Chapter X of Andersen et al. (1993) for the details). Let $E = [0, \tau]$ and $k_n = n$. We suppose that the covariate process $Z^{n,i}$ is given by

$$Z_t^{n,i} = (t - e^{n,i}) \lor 0,$$

where $e^{n,i}$ is a $[0, \tau]$ -valued random variable representing the *entry time* of individual *i*; then $Z_t^{n,i}$ is considered to represent the *age* or *duration* of the individual *i* at *calendar time* $t \in [0, \tau]$. We also suppose that the process $Y^{n,i}$ is given by

$$Y_t^{n,i} = 1_{\{e^{n,i} \le t, \ Z_t^{n,i} < U^{n,i}\}},$$

where $U^{n,i}$ is a $[0, \infty]$ -valued random variable; it is typically of the form $U^{n,i} = (T_1^{n,i} - e^{n,i}) \wedge C^{n,i}$ where $C^{n,i}$ is a $[0, \infty]$ -valued random variable representing the *censoring time* of the individual *i*. To guarantee the predictability of the processes $Y^{n,i}$ and $Z^{n,i}$, the random variables $e^{n,i}$ and $e^{n,i} + U^{n,i}$ are assumed to be stopping times.

We consider the parameter space Θ given in the preceding example with d = 1 (and E = [0, 1] is replaced by $[0, \tau]$), and thus we set $r_n = n^{\alpha/(2\alpha+2)}$. If we take a sequence of partitions $[0, \tau] = \bigcup_m E_m^n$ of Borel measurable sets E_m^n such that $\text{Diam}(E_m^n) \leq r_n^{-1}$, the conclusion of Theorem 6.3.4 holds for the sieve $\Theta^n = \pi^n_+ \Theta$ given by (6.3.9).

6.4 Diffusion-type Processes

Let us consider the stochastic differential equation

(6.4.1) $dX_t = \theta(t, X)dt + n^{-1/2}dB_t, \qquad X_0 = x_0 \in \mathbb{R},$

where $t \rightsquigarrow B_t$ is a standard Brownian motion defined on a stochastic basis $(\Omega, \mathcal{F}, \mathbf{F} = (\mathcal{F}_t)_{t \in [0,\tau]}, P)$ and $\tau > 0$ is a fixed constant. The functional θ appearing above should satisfy some appropriate properties described as follows. We equip $C[0, \tau]$, the canonical space of sample paths, with the σ -field $\mathcal{H}_t = \sigma\{x_s : s \leq t\}$ for every $t \in [0, \tau]$.

Definition 6.4.1 We denote by \mathcal{A} the set of functionals $\theta : [0, \tau] \times C[0, \tau] \to \mathbb{R}$ such that:

(i) $x \rightsquigarrow \theta(t, x)$ is \mathcal{H}_t -measurable for every $t \in [0, \tau]$; (ii) $\sup_{t \in [0, \tau]} \sup_{x \in C[0, \tau]} |\theta(t, x)| < \infty$.

Definition 6.4.2 For a given constant H > 0, we denote by \mathcal{L}_H the set of functionals $\theta \in \mathcal{A}$ such that

$$|\theta(t,x) - \theta(t,y)| \le H \sup_{s \in [0,t]} |x_s - y_s| \qquad \forall x,y \in C[0,\tau], \ \forall t \in [0,\tau].$$

It is well-known that the stochastic differential equation (6.4.1) has a unique strong solution $X^{n,\theta} = (X_t^{n,\theta})_{t \in [0,\tau]}$ whenever $\theta \in \mathcal{L}_H$ for some H > 0 (see e.g. Theorem 13.1 of Rogers and Williams (1987)). We denote by $x^{\theta} = (x_t^{\theta})_{t \in [0,\tau]}$ the solution of the ordinary differential equation

(6.4.2)
$$dx_t = \theta(t, x)dt, \qquad x_0 \in \mathbb{R}.$$

Fix any H > 0. Let us introduce three notations. For every $n \in \mathbb{N}$ and $\theta_0 \in \mathcal{L}_H$ we define the random semimetric $\varrho_{\theta_0}^n$ on \mathcal{A} by

$$\varrho_{\theta_0}^n(\theta,\vartheta) = \sqrt{\frac{1}{\tau} \int_0^\tau |\theta(t, X^{n,\theta_0}) - \vartheta(t, X^{n,\theta_0})|^2 dt} \qquad \forall \theta, \vartheta \in \mathcal{A}$$

For $\theta_0 \in \mathcal{L}_H$ we define the semimetric ρ_{θ_0} on \mathcal{A} by

$$\rho_{\theta_0}(\theta,\vartheta) = \sqrt{\frac{1}{\tau} \int_0^\tau |\theta(t,x^{\theta_0}) - \vartheta(t,x^{\theta_0})|^2 dt} \qquad \forall \theta, \vartheta \in \mathcal{A}.$$

For every $\theta_0 \in \mathcal{L}_H$ and $\varepsilon > 0$, we define

$$\mathcal{L}_{H}^{\star}(\rho_{\theta_{0}};\varepsilon) = \{ \theta \in \mathcal{A} : \exists \theta_{1}, \theta_{2} \in \mathcal{L}_{H} \text{ such that } \theta_{1} \leq \theta \leq \theta_{2} \text{ and } \rho_{\theta_{0}}(\theta_{1},\theta_{2}) \leq \varepsilon \}.$$

It is clear that $\mathcal{L}_H \subset \mathcal{L}_H^{\star}(\rho_{\theta_0}; \varepsilon) \subset \mathcal{A}$.

We denote by $\mathbb{P}^{n,\theta}$ the induced probability distribution of $X^{n,\theta}$ on the canonical space $C[0,\tau]$. Then, the family $\{\mathbb{P}^{n,\theta} : \theta \in \mathcal{L}_H\}$ is equivalent, and the log-likelihood ratio is given by

$$\log \frac{\mathbb{P}_{\theta}^{n}}{\mathbb{P}_{\vartheta}^{n}}(X) = n\{l(\theta; X) - l(\vartheta; X)\} \qquad \forall \theta, \vartheta \in \mathcal{L}_{H}$$

where

(6.4.3)
$$l(\theta; X) = \int_0^\tau \theta(t, X) dX_t - \frac{1}{2} \int_0^\tau |\theta(t, X)|^2 dt$$

(see, e.g., page 29 of Kutoyants (1994)). Although the representation of log-likelihood ratio relies on the existence of unique strong solution of the stochastic differential equation (6.4.1), the formula (6.4.3) itself is well-defined for all $\theta \in \mathcal{A}$. We then consider a maximum likelihood estimator (MLE) on the sieve $\Theta^n \subset \mathcal{A}$

$$\widehat{\theta}^n(X) = \operatorname*{argmax}_{\theta \in \Theta^n} l(\theta; X).$$

The precise description will be given in the main theorem below.

The condition which we shall assume is as follows; we denote by $B_{\rho_{\theta_0}}(\theta; \delta)$ the closed ball in \mathcal{A} with center θ and ρ_{θ_0} -radius δ .

Condition 6.4.3 Let $U \subset \Theta \subset \mathcal{L}_H$ and $\Theta^n \subset \mathcal{A}$ be given. For every $n \in \mathbb{N}$ and $\theta_0 \in U$, there exist a proper metric $\widehat{\rho}_{\theta_0}^n$ on Θ^n such that $\rho_{\theta_0} \leq \widehat{\rho}_{\theta_0}^n$, and a function $\varphi_{\theta_0}^n : (0,\infty) \to (0,\infty)$ such that $\delta \rightsquigarrow \delta^{-1} \varphi_{\theta_0}^n(\delta)$ is decreasing and that

(6.4.4)
$$\int_0^{\delta} \sqrt{\log(1 + N(\Theta^n \cap B_{\rho_{\theta_0}}(\theta_0; \delta), \widehat{\rho}_{\theta_0}^n; \varepsilon))} d\varepsilon \le \varphi_{\theta_0}^n(\delta) \qquad \forall \delta \in (0, \infty).$$

Then, choose some constants $r_{n,\theta_0} \in (0, n^{1/2}]$ such that $n^{-1/2}\varphi_{\theta_0}^n(r_{n,\theta_0}^{-1}) \leq r_{n,\theta_0}^{-2}$.

Theorem 6.4.4 Let $U \subset \Theta \subset \mathcal{L}_H$ for a constant H > 0. Suppose that Condition 6.4.3 is satisfied for some countable sets $\Theta^n \subset \mathcal{A}$, and choose some constants r_{n,θ_0} described there. Suppose also that there exists a constant M > 0 such that: for every $n \in \mathbb{N}$ and $\theta_0 \in U$

$$\Theta^n \subset \mathcal{L}^{\star}_H(\rho_{\theta_0}; Mr_{n,\theta_0}^{-1}) \qquad and \qquad \rho_{\theta_0}(\theta_0, \theta_{\theta_0}^n) \leq Mr_{n,\theta_0}^{-1} \quad for \ some \ \theta_{\theta_0}^n \in \Theta^n.$$

Then, for any mapping $\widehat{\theta}^n : C[0,\tau] \to \Theta^n$ such that

(6.4.5)
$$l(\widehat{\theta}^n(X);X) \ge \sup_{\theta \in \Theta^n} l(\theta;X) - r_n^{-2} \quad with \quad r_n = \sup_{\theta_0 \in U} r_{n,\theta_0},$$

it holds that

$$\lim_{L \to \infty} \limsup_{n \to \infty} \sup_{\theta_0 \in U} P^* \left(r_{n,\theta_0} \rho_{\theta_0}(\widehat{\theta}^n(X^{n,\theta_0}), \theta_0) > L \right) = 0;$$
$$\lim_{L \to \infty} \limsup_{n \to \infty} \sup_{\theta_0 \in U} P^* \left(r_{n,\theta_0} \varrho_{\theta_0}^n(\widehat{\theta}^n(X^{n,\theta_0}), \theta_0) > L \right) = 0.$$

In order to prove the result above, we will apply Theorem 5.1.2 to the processes $\theta \rightsquigarrow \Gamma^n_{\theta_0}(\theta)$ and $\theta \rightsquigarrow \gamma^n_{\theta_0}(\theta)$ given by:

(6.4.6)
$$\Gamma_{\theta_{0}}^{n}(\theta) = l(\theta; X^{n,\theta_{0}})$$

$$= \int_{0}^{\tau} \theta(t, X^{n,\theta_{0}}) dX^{n,\theta_{0}} - \frac{1}{2} \int_{0}^{\tau} |\theta(t, X^{n,\theta_{0}})|^{2} dt;$$
(6.4.7)
$$\gamma_{\theta_{0}}^{n}(\theta) = \int_{0}^{\tau} \theta(t, X^{n,\theta_{0}}) \theta_{0}(t, X^{n,\theta_{0}}) dt - \frac{1}{2} \int_{0}^{\tau} |\theta(t, X^{n,\theta_{0}})|^{2} dt$$

$$= -\frac{1}{2} \varrho_{\theta_{0}}^{n}(\theta, \theta_{0})^{2} + \frac{1}{2} \int_{0}^{\tau} |\theta_{0}(t, X^{n,\theta_{0}})|^{2} dt.$$

The key point of the proof is that

(6.4.8)
$$\Gamma^n_{\theta_0}(\theta) - \gamma^n_{\theta_0}(\theta) = n^{-1/2} \int_0^\tau \theta(t, X^{n, \theta_0}) dB_t.$$

First, let us investigate the relationship between ρ_{θ}^{n} and $\varrho_{\theta_{0}}^{n}$.

Lemma 6.4.5 Let H > 0 and $\varepsilon > 0$ be arbitrary constants. For every $\theta_0 \in \mathcal{L}_H$ and every $\theta, \vartheta \in \mathcal{L}_H^*(\rho_{\theta_0}, \varepsilon)$ it holds that

$$|\varrho_{\theta_0}^n(\theta,\vartheta) - \rho_{\theta_0}(\theta,\vartheta)|^2 \le n^{-1} \cdot 24H^2 \tau e^{2H\tau} \sup_{t \in [0,\tau]} |B_t|^2 + 48\varepsilon^2$$

P-almost surely.

Proof. Observe that

$$\begin{split} |\varrho_{\theta_0}^n(\theta,\vartheta) - \rho_{\theta_0}(\theta,\vartheta)|^2 \\ &\leq \int_0^\tau \left| \{\theta(X^{n,\theta_0}) - \vartheta(X^{n,\theta_0})\} - \{\theta(x^{\theta_0}) - \vartheta(x^{\theta_0})\} \right|^2 dt \\ &\leq 2\int_0^\tau \left| \theta(X^{n,\theta_0}) - \theta(x^{\theta_0}) \right|^2 dt + 2\int_0^\tau \left| \vartheta(X^{n,\theta_0}) - \vartheta(x^{\theta_0}) \right|^2 dt. \end{split}$$

Here, for given $\theta \in \mathcal{L}_{H}^{\star}(\rho_{\theta_{0}}, \varepsilon)$ choose some $\theta_{1}, \theta_{2} \in \mathcal{L}_{H}$ such that $\theta_{1} \leq \theta \leq \theta_{2}$ and that $\rho_{\theta_{0}}(\theta_{1}, \theta_{2}) \leq \varepsilon$. Then we have

$$\begin{aligned} |\theta(t, X^{n,\theta_0}) - \theta(t, x^{\theta_0})| &\leq |\theta_1(t, X^{n,\theta_0}) - \theta_2(t, x^{\theta_0})| + |\theta_2(t, X^{n,\theta_0}) - \theta_1(t, x^{\theta_0})| \\ &\leq |\theta_1(t, X^{n,\theta_0}) - \theta_1(t, x^{\theta_0})| + |\theta_2(t, X^{n,\theta_0}) - \theta_2(t, x^{\theta_0})| \\ &+ 2|\theta_1(t, x^{\theta_0}) - \theta_2(t, x^{\theta_0})|, \end{aligned}$$

thus

$$|\theta(t, X^{n,\theta_0}) - \theta(t, x^{\theta_0})|^2 \le 6H^2 \sup_{s \in [0,t]} |X^{n,\theta_0}_s - x^{\theta_0}_s|^2 + 12|\theta_1(t, x^{\theta_0}) - \theta_2(t, x^{\theta_0})|^2 \le 6H^2 |x^{\theta_0}|^2 + 12|\theta_1(t, x^{\theta_0}) - \theta_2(t, x^{\theta_0})|^2 \le 6H^2 |x^{\theta_0}|^2 + 12|\theta_1(t, x^{\theta_0}) - \theta_2(t, x^{\theta_0})|^2 \le 6H^2 |x^{\theta_0}|^2 + 12|\theta_1(t, x^{\theta_0}) - \theta_2(t, x^{\theta_0})|^2 \le 6H^2 |x^{\theta_0}|^2 + 12|\theta_1(t, x^{\theta_0}) - \theta_2(t, x^{\theta_0})|^2 \le 6H^2 |x^{\theta_0}|^2 + 12|\theta_1(t, x^{\theta_0}) - \theta_2(t, x^{\theta_0})|^2 \le 6H^2 |x^{\theta_0}|^2 + 12|\theta_1(t, x^{\theta_0}) - \theta_2(t, x^{\theta_0})|^2 \le 6H^2 |x^{\theta_0}|^2 + 12|\theta_1(t, x^{\theta_0}) - \theta_2(t, x^{\theta_0})|^2 \le 6H^2 |x^{\theta_0}|^2 + 12|\theta_1(t, x^{\theta_0}) - \theta_2(t, x^{\theta_0})|^2 \le 6H^2 |x^{\theta_0}|^2 + 12|\theta_1(t, x^{\theta_0}) - \theta_2(t, x^{\theta_0})|^2 \le 6H^2 |x^{\theta_0}|^2 + 12|\theta_1(t, x^{\theta_0}) - \theta_2(t, x^{\theta_0})|^2 \le 6H^2 |x^{\theta_0}|^2 + 12|\theta_1(t, x^{\theta_0}) - \theta_2(t, x^{\theta_0})|^2 \le 6H^2 |x^{\theta_0}|^2 + 12|\theta_1(t, x^{\theta_0}) - \theta_2(t, x^{\theta_0})|^2 \le 6H^2 |x^{\theta_0}|^2 + 12|\theta_1(t, x^{\theta_0}) - \theta_2(t, x^{\theta_0})|^2 \le 6H^2 |x^{\theta_0}|^2 + 12|\theta_1(t, x^{\theta_0}) - \theta_2(t, x^{\theta_0})|^2 \le 6H^2 |x^{\theta_0}|^2 + 12|\theta_1(t, x^{\theta_0}) - \theta_2(t, x^{\theta_0})|^2 \le 6H^2 |x^{\theta_0}|^2 + 12|\theta_1(t, x^{\theta_0}) - \theta_2(t, x^{\theta_0})|^2 \le 6H^2 |x^{\theta_0}|^2 + 12|\theta_1(t, x^{\theta_0}) - \theta_2(t, x^{\theta_0})|^2 \le 6H^2 |x^{\theta_0}|^2 + 12|\theta_1(t, x^{\theta_0}) - \theta_2(t, x^{\theta_0})|^2 \le 6H^2 |x^{\theta_0}|^2 + 12|\theta_1(t, x^{\theta_0}) - \theta_2(t, x^{\theta_0})|^2 \le 6H^2 |x^{\theta_0}|^2 + 12|\theta_1(t, x^{\theta_0}) - \theta_2(t, x^{\theta_0})|^2 \le 6H^2 |x^{\theta_0}|^2 + 12|\theta_1(t, x^{\theta_0}) - \theta_2(t, x^{\theta_0})|^2 \le 6H^2 |x^{\theta_0}|^2 + 12|\theta_1(t, x^{\theta_0}) - \theta_2(t, x^{\theta_0})|^2 \le 6H^2 |x^{\theta_0}|^2 + 12|\theta_1(t, x^{\theta_0}) - \theta_2(t, x^{\theta_0})|^2 \le 6H^2 |x^{\theta_0}|^2 + 12|\theta_1(t, x^{\theta_0}) - \theta_2(t, x^{\theta_0})|^2 \le 6H^2 |x^{\theta_0}|^2 + 12|\theta_1(t, x^{\theta_0})|^2 \le 6H^2 |x^{\theta_0}|^2 + 12$$

It follows from the Grownwall inequality that

$$\sup_{t \in [0,\tau]} |X_t^{n,\theta_0} - x_t^{\theta_0}| \le e^{H\tau} \sup_{t \in [0,\tau]} |n^{-1/2}B_t|,$$

hence we obtain

$$\int_0^\tau |\theta(t, X^{n, \theta_0}) - \theta(t, x^{\theta_0})|^2 dt \le 6H^2 \tau e^{2H\tau} \sup_{t \in [0, \tau]} |n^{-1/2} B_t|^2 + 12\varepsilon^2.$$

Those inequalities imply the assertion.

We will use the above lemma in the following form.

Lemma 6.4.6 Let U be an arbitrary subset of \mathcal{L}_H for some H > 0, and let M > 0 be an arbitrary constant. For every $n \in \mathbb{N}$ and $\theta_0 \in U$, let r_{n,θ_0} be some positive constants such that $r_{n,\theta_0} \leq n^{1/2}$, and let Θ^n a countable subset of the set $\bigcap_{\theta_0 \in U} \mathcal{L}_H^*(\rho_{\theta_0}, Mr_{n,\theta_0}^{-1})$. Then, it holds that:

$$\lim_{L \to \infty} \sup_{n \in \mathbb{N}} \sup_{\theta_0 \in U} P\left(\sup_{\theta, \vartheta \in \Theta^n} |\varrho_{\theta_0}^n(\theta, \vartheta) - \rho_{\theta_0}(\theta, \vartheta)| > Lr_{n, \theta_0}^{-1}\right) = 0;$$
$$\lim_{K \to \infty} \sup_{n \in \mathbb{N}} \sup_{\theta_0 \in U} P\left(\sup_{\theta, \vartheta \in \Theta^n \atop \rho_{\theta_0}(\theta, \vartheta) > Kr_{n, \theta_0}^{-1}} \frac{\rho_{\theta_0}(\theta, \vartheta)}{\varrho_{\theta_0}^n(\theta, \vartheta)} > 2\right) = 0.$$

Proof. It follows from Lemma 6.4.5 that for every $\varepsilon > 0$ there exists a constant $L_{\varepsilon} > 0$ such that $\sup_{n \in \mathbb{N}} \sup_{\theta_0 \in \Theta} P(\Omega \setminus \Omega^n_{\theta_0}(\varepsilon)) \leq \varepsilon$, where

$$\Omega^n_{\theta_0}(\varepsilon) = \left\{ \sup_{\theta, \vartheta \in \Theta^n} |\varrho^n_{\theta_0}(\theta, \vartheta) - \rho_{\theta_0}(\theta, \vartheta)| \le L_{\varepsilon} r_{n, \theta_0}^{-1} \right\}.$$

The first assertion is nothing else than this fact. On the other hand, it holds on the set $\Omega_{\theta_0}^n(\varepsilon)$ that

$$\sup_{\substack{\theta,\vartheta\in\Theta^{n}\\\rho_{\theta_{0}}(\theta,\vartheta)>Kr_{n,\theta_{0}}^{-1}}}\frac{\rho_{\theta_{0}}(\theta,\vartheta)}{\varrho_{\theta_{0}}^{n}(\theta,\vartheta)} \leq \sup_{\substack{\theta,\vartheta\in\Theta^{n}\\\rho_{\theta_{0}}(\theta,\vartheta)>Kr_{n,\theta_{0}}^{-1}}}\frac{\rho_{\theta_{0}}(\theta,\vartheta)}{\rho_{\theta_{0}}(\theta,\vartheta)-L_{\varepsilon}r_{n,\theta_{0}}^{-1}}$$

$$\leq \sup_{\substack{\theta,\vartheta\in\Theta^{n}\\\rho_{\theta_{0}}(\theta,\vartheta)>Kr_{n,\theta_{0}}^{-1}}}\frac{\rho_{\theta_{0}}(\theta,\vartheta)}{\rho_{\theta_{0}}(\theta,\vartheta)-\frac{1}{2}\rho_{\theta_{0}}(\theta,\vartheta)} \quad \text{whenever } K \geq 2L_{\varepsilon}$$

$$= 2.$$

This implies the second assertion.

Proof of Theorem 6.4.4. We will apply Theorem 5.1.2. Formulation 5.1.1 should be read as follows: for every $\theta_0 \in U(=U^n)$,

- (i) the random semimetric space $(\Theta^n, \varrho^n_{\theta_0})$ and the point $\theta^n_{\theta_0} \in \Theta^n$;
- (ii) the stochastic processes $\theta \rightsquigarrow \Gamma^n_{\theta_0}(\theta)$ and $\theta \rightsquigarrow \gamma^n_{\theta_0}(\theta)$ with parameters in Θ^n , given
- by (6.4.6) and (6.4.7), respectively.

We then denote $R_{\theta_0}^n(\delta) = \{\theta \in \Theta^n : (\delta/2) < \varrho_{\theta_0}^n(\theta, \theta_{\theta_0}^n) \le \delta\}$ for every $\delta \in (0, \infty)$.

First of all, it follows from Lemma 6.4.6 that for every $\varepsilon > 0$ there exists a constant $K_{\varepsilon} > 0$ such that

$$\sup_{n\in\mathbb{N}}\sup_{\theta_0\in U}P(\Omega\setminus\Omega^n_{\theta_0}(K_\varepsilon))\leq\varepsilon,$$

where

$$\Omega^n_{\theta_0}(K_{\varepsilon}) = \left\{ \varrho^n_{\theta_0}(\theta^n_{\theta_0}, \theta_0) \le K_{\varepsilon} r_{n,\theta_0}^{-1} \right\} \cap \bigcap_{\substack{\theta, \vartheta \in \Theta^n \\ \rho_{\theta_0}(\theta, \vartheta) > K_{\varepsilon} r_{n,\theta_0}^{-1}}} \left\{ \frac{1}{2} \le \frac{\varrho^n_{\theta_0}(\theta, \vartheta)}{\rho_{\theta_0}(\theta, \vartheta)} \le 2 \right\}.$$

To show the FIRST INEQUALITY of M-CRITERION, observe that for any $L \ge 16L_{\varepsilon}$

$$\begin{array}{ll} \varrho_{\theta_0}^n(\theta_{\theta_0}^n,\theta_0) &\leq K_{\varepsilon}r_{n,\theta_0}^{-1} \quad \text{on the set } \Omega_{\theta_0}^n(K_{\varepsilon}) \\ &\leq \frac{1}{4}\cdot\frac{\delta}{2} \quad \text{whenever } \delta \geq 8K_{\varepsilon}r_{n,\theta_0}^{-1} \\ &\leq \frac{1}{4}\varrho_{\theta_0}^n(\theta,\theta_{\theta_0}^n) \quad \text{whenever } \theta \in R_{\theta_0}^n(\delta). \end{array}$$

Hence, we have for every $\delta \geq 8K_{\varepsilon}r_{n,\theta_0}^{-1}$ and $\theta \in R_{\theta_0}^n(\delta)$ that

$$\begin{split} \gamma_{\theta_{0}}^{n}(\theta) - \gamma_{\theta_{0}}^{n}(\theta_{\theta_{0}}^{n}) &= \frac{1}{2} \left\{ -\varrho_{\theta_{0}}^{n}(\theta,\theta_{0})^{2} + \varrho_{\theta_{0}}^{n}(\theta_{\theta_{0}}^{n},\theta_{0})^{2} \right\} \\ &\leq \frac{1}{2} \left\{ -\varrho_{\theta_{0}}^{n}(\theta,\theta_{\theta_{0}}^{n})^{2} + 2\varrho_{\theta_{0}}^{n}(\theta,\theta_{\theta_{0}}^{n})\varrho_{\theta_{0}}^{n}(\theta_{\theta_{0}}^{n},\theta_{0}) \right\} \\ &\leq \frac{1}{2} \left\{ -\varrho_{\theta_{0}}^{n}(\theta,\theta_{\theta_{0}}^{n})^{2} + 2\varrho_{\theta_{0}}^{n}(\theta,\theta_{\theta_{0}}^{n}) \frac{\varrho_{\theta_{0}}^{n}(\theta,\theta_{\theta_{0}}^{n})}{4} \right\} \\ &= -\frac{1}{4} \varrho_{\theta_{0}}^{n}(\theta,\theta_{\theta_{0}}^{n})^{2} \\ &\leq -\frac{1}{16} \delta^{2} \end{split}$$

on the set $\Omega_{\theta_0}^n(K_{\varepsilon})$. This means that the FIRST INEQUALITY holds for p = 2. Next, notice that whenever $K_{\varepsilon} \geq 2$

$$\Omega^n_{\theta_0}(K_{\varepsilon}) \subset \left\{ R^n_{\theta_0}(\delta) \cup \{\theta^n_{\theta_0}\} \subset \Theta^n \cap B_{\rho_{\theta_0}}(\theta^n_{\theta_0}; K_{\varepsilon}\delta) \quad \forall \delta \in [K_{\varepsilon}r_{n,\theta_0}^{-1}, \infty) \right\}.$$

Since $\Theta^n \cup B_{\rho_{\theta_0}}(\theta^n_{\theta_0}; K_{\varepsilon}\delta)$ is contained in $\Theta^n \cup B_{\rho_{\theta_0}}(\theta_0; (K_{\varepsilon} + M)\delta)$, we can deduce from Theorem 2.4.5 that for every $\delta \geq K_{\varepsilon} r_{n,\theta_0}^{-1}$

$$\begin{split} E^* \sup_{\theta \in R^n_{\theta_0}(\delta)} |(\Gamma^n_{\theta_0} - \gamma^n_{\theta_0})(\theta) - (\Gamma^n_{\theta_0} - \gamma^n_{\theta_0})(\theta^n_{\theta_0})| 1_{\Omega^n_{\theta_0}(K_{\varepsilon})} \\ &\lesssim K_{\varepsilon} \cdot n^{-1/2} \int_0^{\delta} \sqrt{\log(1 + N(\Theta^n \cap B_{\rho_{\theta_0}}(\theta_0; (K_{\varepsilon} + M)\delta), \widehat{\rho}^n_{\theta_0}; \varepsilon))} d\varepsilon \\ &\leq K_{\varepsilon} \cdot n^{-1/2} \varphi^n_{\theta_0}((K_{\varepsilon} + M)\delta) \\ &\leq K_{\varepsilon}(K_{\varepsilon} + M) \cdot n^{-1/2} \varphi^n_{\theta_0}(\delta). \end{split}$$

Thus the SECOND INEQUALITY is fulfilled with $\phi_{\theta_0}^n = n^{-1/2} \varphi_{\theta_0}^n$.

Hence it follows from Theorem 5.1.2 that

$$\lim_{L \to \infty} \limsup_{n \to \infty} \sup_{\theta_0 \in U} P^* \left(r_{n,\theta_0} \rho_{\theta_0}(\widehat{\theta}^n(X^{n,\theta_0}), \theta_{\theta_0}^n) > L \right) = 0.$$

Noting $\rho_{\theta_0}(\theta_{\theta_0}^n, \theta_0) \leq Mr_{n,\theta_0}^{-1}$, we get the first conclusion of the theorem. The second one follows from the first and Lemma 6.4.6.

Example: Markovian case

Consider the stochastic differential equation

$$dX_t = \theta(X_t)dt + n^{-1/2}dB_t, \qquad X_0 = x_0 \in \mathbb{R},$$

where $\theta : \mathbb{R} \to \mathbb{R}$ is a bounded, Lipschitz continuous function. Then, the unique strong solution $X^{n,\theta} = (X_t^{n,\theta})_{t \in [0,\tau]}$ is a time-homogeneous Markov process. In the same way as the general case, we denote by $x^{\theta} = (x_t^{\theta})_{t \in [0,\tau]}$ the solution of the ordinary differential equation

$$dx_t = \theta(x_t)dt, \qquad x_0 \in \mathbb{R}$$

For a given bounded, Lipschitz continuous function $\theta_0 : \mathbb{R} \to \mathbb{R}$, we define

$$\varrho_{\theta_0}^n(\theta,\vartheta) = \sqrt{\frac{1}{\tau} \int_0^\tau |\theta(X_t^{n,\theta_0}) - \vartheta(X_t^{n,\theta_0})|^2 dt} \qquad \forall \theta, \vartheta \in \mathcal{A}$$

and

$$\rho_{\theta_0}(\theta,\vartheta) = \sqrt{\frac{1}{\tau} \int_0^\tau |\theta(x_t^{\theta_0}) - \vartheta(x_t^{\theta_0})|^2 dt} \qquad \forall \theta, \vartheta \in \mathcal{A},$$

where \mathcal{A} denotes the space of bounded measurable functions on \mathbb{R} .

Let some constants $\alpha \geq 1$ and H > 0 be given. We consider the class $C_H^{\alpha}(\mathbb{R})$ given in (11) of Section 1.2. Due to the fact that $t \rightsquigarrow x_t^{\theta}$ takes values only in $[x_0 - H\tau, x_0 + H\tau]$ whenever $\theta \in C_H^{\alpha}(\mathbb{R})$, and recalling also Lemma 6.4.5, we define the parameter space

$$\Theta = \{ \theta \in C_H^\alpha : x \rightsquigarrow \theta(x) \text{ is constant on } E^c \}.$$

where $E = [x_0 - H\tau - 1, x_0 + H\tau + 1]$. We denote by $\|\cdot\|_{\infty}$ the supremum norm on $\ell^{\infty}(\mathbb{R})$; notice that $\rho_{\theta_0} \leq \|\cdot\|_{\infty}$. Since we have

$$\log N(\Theta, \|\cdot\|_{\infty}; \varepsilon) \le \text{const.} \varepsilon^{-1/\alpha},$$

Condition 6.4.3 is satisfied with $\varphi(\delta) = \text{const.}(\delta^{1-(1/2\alpha)} \vee \delta)$ whenever $\Theta^n \subset \Theta$. In this case, we get the rate $r_n = n^{\alpha/(2\alpha+1)}$. Consequently, if we choose a countable subset Θ^n of Θ such that $\Theta \subset \bigcup_{\theta \in \Theta^n} B_{(\Theta, \|\cdot\|_{\infty})}(\theta; Mn^{-\alpha/(2\alpha+1)})$ for some M > 0 not depending on n, then it holds for any Θ^n -sieved MLE $\hat{\theta}^n$ that:

$$\lim_{L \to \infty} \limsup_{n \to \infty} \sup_{\theta_0 \in \Theta} P^* \left(n^{\alpha/(2\alpha+1)} \rho_{\theta_0}(\widehat{\theta}^n(X^{n,\theta_0}), \theta_0) > L \right) = 0;$$
$$\lim_{L \to \infty} \limsup_{n \to \infty} \sup_{\theta_0 \in \Theta} P^* \left(n^{\alpha/(2\alpha+1)} \varrho_{\theta_0}^n(\widehat{\theta}^n(X^{n,\theta_0}), \theta_0) > L \right) = 0.$$

6.A Notes

The rate of convergence of infinite-dimensional M-estimators has been studied vigorously by Birgé and Massart (1993), van de Geer (1990, 1993, 1995a, 1995b), Wong and Severini (1991) and Wong and Shen (1995); see also Chapters 3.2 and 3.4 of van der Vaart and Wellner (1996) and the bibliographical Notes there. Among the preceding works, van de Geer (1995b) is a unique paper that deals with dependent data. Based on her general result for counting processes, she considered non-parametric maximum likelihood estimators in the multiplicative intensity model (without marks); it should be emphasized that, although there are some differences, a major part of Section 6.2 has been already known through her work. However, the marks and the discussion about sieves that have been newly added there are useful for analyzing the non-linear covariate model in Section 6.3.

Some M-estimation problems in finite-dimensional parametric models of diffusiontype processes have been studied by Lánska (1979), Genon-Catalot (1990), Yoshida (1990, 1992) and Kutoyants (1994, Chapter 7); see also the references therein. The results in Section 6.4 seem to be the first attempt in the infinite-dimensional model.

Chapter 7

Miscellanies

7.1 Local Random Fields of Kernel Estimators

It is well-known that kernel density estimators for i.i.d. data have point-wise asymptotic normality. However, since the density f is originally defined as a Radon-Nikodym derivative with respect to Lebesgue measure, the value f(x) at each point x does not intrinsically make sense. Thus, an assertion in some functional sense is preferable in order for, e.g., the construction of confidence intervals.

The purpose of this section is to extend the asymptotic normality of kernel density estimators to the functional sense with respect to a local parameter. The localizing constants should be chosen to be the same as the bandwidth. Further, in Subsection 7.1.2, we apply it to the estimation problem of the mode of f using also Theorem 5.1.2. The generalizations of those results to some dependent cases are discussed in Subsection 7.1.3.

7.1.1 I.I.D. Case

Let $\{X_i\}_{i\in\mathbb{N}}$ be an i.i.d. sequence of \mathbb{R}^d -valued random variables with Lebesgue density f. Let $x_0 \in \mathbb{R}^d$ be a fixed point, and let $\{b_n\}_{n\in\mathbb{N}}$ be a sequence of positive constants such that $b_n \downarrow 0$ as $n \to \infty$. We are interested in estimating the local function $u \rightsquigarrow f(x_0+b_n u)$, where the parameter u runs through a subset U of \mathbb{R}^d . We consider the kernel estimator

$$\widehat{f}_n(x_0 + b_n u) = \frac{1}{nb_n^d} \sum_{i=1}^n K\left(\frac{X_i - x_0}{b_n} - u\right) \qquad \forall u \in U,$$

where K(x) is a kernel function on \mathbb{R}^d . Throughout this section, the notation $x^{(p)}$ means the *p*-th component of a vector $x \in \mathbb{R}^d$. We make two kinds of conditions either of which the kernel function should satisfy.

Condition 7.1.1 (smooth kernel) The function $K : \mathbb{R}^d \to \mathbb{R}$ satisfies that:

7. Miscellanies

(i) $\int_{\mathbb{R}^d} K(x) dx = 1$, K(x) = K(-x) for every $x \in \mathbb{R}^d$, and K has a compact support; (ii) there exist $\alpha \in (0,1]$ and L > 0 such that $|K(x) - K(y)| \leq L|x - y|^{\alpha}$ for every $x, y \in \mathbb{R}^d$.

Condition 7.1.2 (monotone kernel) The function $K : \mathbb{R}^d \to \mathbb{R}$ is of the product form $K(x) = \prod_{p=1}^d K_p(x^{(p)})$ of some functions $K_p : \mathbb{R} \to \mathbb{R}$, p = 1, ..., d. The functions K_p need not be the same, but each of them satisfies: (i) $\int_{\mathbb{R}} K_p(x) dx = 1$, $K_p(x) = K_p(-x)$ for every $x \in \mathbb{R}$, and K_p has a compact support; (ii) the function $x \to K_p(x)$ is decreasing on $[0, \infty)$.

We aim to derive the asymptotic behavior of the sequence of (normalized) residual processes $R^n = (R^n(u)|u \in U)$ defined by

$$R^{n}(u) = \sqrt{nb_{n}^{d}} \left\{ \widehat{f_{n}}(x_{0} + b_{n}u) - f(x_{0} + b_{n}u) \right\} \qquad \forall u \in U.$$

The key point is to investigate the processes $Z^n = (Z^n(u) | u \in U)$ given by

$$Z^{n}(u) = \sqrt{nb_{n}^{d}} \left\{ \widehat{f}_{n}(x_{0} + b_{n}u) - \widetilde{f}_{n}(x_{0} + b_{n}u) \right\} \qquad \forall u \in U,$$

where

$$\begin{aligned} \widetilde{f}_n(x_0 + b_n u) &= \frac{1}{b_n^d} \int_{\mathbb{R}^d} K\left(\frac{x - x_0}{b_n} - u\right) f(x) dx \\ &= \int_{\mathbb{R}^d} K(y) f(x_0 + b_n(u + y)) dy \quad \forall u \in U. \end{aligned}$$

Notice that the processes \mathbb{R}^n and \mathbb{Z}^n are not necessarily continuous in the case of a monotone kernel, and thus we treat them as $\ell^{\infty}(U)$ -valued random elements. This is natural especially in the multi-dimensional case.

Proposition 7.1.3 Choose a kernel function $K : \mathbb{R}^d \to \mathbb{R}$ satisfying either Condition 7.1.1 or 7.1.2, and let $\{b_n\}$ be a sequence of positive constants such that $b_n \downarrow 0$ and that $nb_n^d \uparrow \infty$ as $n \to \infty$. If f is continuous at x_0 , and if U is bounded, then it holds that $Z^n \Longrightarrow Z$ in $\ell^{\infty}(U)$, where $u \rightsquigarrow Z(u)$ a zero-mean, continuous Gaussian process such that

(7.1.1)
$$E(Z(u_1)Z(u_2)) = f(x_0) \int_{\mathbb{R}^d} K(x-u_1)K(x-u_2)dx \quad \forall u_1, u_2 \in U.$$

Remark. The continuity of the limit process $u \rightsquigarrow Z(u)$ is considered with respect to the Euclidean metric.

7.1. Local Random Fields of Kernel Estimators

Let $U_1 \subset U_2 \subset \cdots$ be a sequence of bounded subsets of \mathbb{R}^d such that $\bigcup_{i=1}^{\infty} U_i = \mathbb{R}^d$. We denote by $\ell_{\text{loc}}^{\infty}(\mathbb{R}^d)$ the set of all functions $z : \mathbb{R}^d \to \mathbb{R}$ that are bounded on every U_i , and equip it with the local uniform metric d defined by

$$d(z_1, z_2) = \sum_{i=1}^{\infty} \left(\sup_{u \in U_i} |z_1(u) - z_2(u)| \wedge 1 \right) 2^{-i}.$$

Using Theorem 1.6.1 of van der Vaart and Wellner (1996), we obtain the following.

Theorem 7.1.4 Choose a kernel function $K : \mathbb{R}^d \to \mathbb{R}$ and a sequence of constants $\{b_n\}$ as in Proposition 7.1.3.

(i) If f is continuous at x_0 , then it holds that $Z^n \Longrightarrow Z$ in $\ell^{\infty}_{loc}(\mathbb{R}^d)$, where $u \rightsquigarrow Z(u)$ a zero-mean, continuous Gaussian process whose covariance $E(Z(u_1)Z(u_2))$ is given by (7.1.1) for every $u_1, u_2 \in \mathbb{R}^d$.

(ii) If f is twice continuously differentiable in a neighborhood of x_0 , and if

$$\lim_{n \to \infty} n b_n^{5d} = h < \infty,$$

then it holds that $R^n \Longrightarrow z_0 + Z$ in $\ell^{\infty}_{loc}(\mathbb{R}^d)$, where

(7.1.2)
$$z_0 = \frac{\sqrt{h}}{2} \sum_{p=1}^d \sum_{q=1}^d \int_{\mathbb{R}^d} y^{(p)} y^{(q)} K(y) dy \left. \frac{\partial^2 f(x)}{\partial x^{(p)} \partial x^{(q)}} \right|_{x=x_0}$$

This result can be applied to construct a confidence band, substituting estimators for $f(x_0)$ in the covariance of the limit process Z and for the second derivatives of f at x_0 in the constant z_0 . Another application is given in Subsection 7.1.2. Notice that the assumptions appearing above are exactly the same as those in the context of point-wise asymptotic normality, and thus are quite reasonable. Our conclusion is that the local smoothness of the density f implies not only the point-wise asymptotic normality but also the weak convergence of local residual processes \mathbb{R}^n .

Proof of Proposition 7.1.3. We can write $Z^n(u) = \sum_{i=1}^n \xi_i^n(u)$ where

$$\xi_i^n(u) = \frac{1}{\sqrt{nb_n^d}} \left\{ K\left(\frac{X_i - x_0}{b_n} - u\right) - \int_{\mathbb{R}^d} K\left(\frac{x - x_0}{b_n} - u\right) f(x) dx \right\}.$$

We will check the conditions of Theorem 3.3.2. For every $u_1, u_2 \in U$, since

$$\begin{split} E\xi_{i}^{n}(u_{1})\xi_{i}^{n}(u_{2}) &= \frac{1}{nb_{n}^{d}} \left\{ \int_{\mathbb{R}^{d}} K\left(\frac{x-x_{0}}{b_{n}}-u_{1}\right) K\left(\frac{x-x_{0}}{b_{n}}-u_{2}\right) f(x)dx \\ &- \int_{\mathbb{R}^{d}} K\left(\frac{x-x_{0}}{b_{n}}-u_{1}\right) f(x)dx \int_{\mathbb{R}^{d}} K\left(\frac{x-x_{0}}{b_{n}}-u_{2}\right) f(x)dx \right\} \\ &= \frac{1}{n} \left\{ \int_{\mathbb{R}^{d}} K(y-u_{1}) K(y-u_{2}) f(x_{0}+b_{n}y)dy \\ &- b_{n}^{d} \int_{\mathbb{R}^{d}} K(y-u_{1}) f(x_{0}+b_{n}y)dy \int_{\mathbb{R}^{d}} K(y-u_{2}) f(x_{0}+b_{n}y)dy \right\} \end{split}$$

we easily obtain

$$\lim_{n \to \infty} \sum_{i=1}^{n} E\xi_{i}^{n}(u_{1})\xi_{i}^{n}(u_{2}) = f(x_{0}) \int_{\mathbb{R}^{d}} K(y - u_{1})K(y - u_{2})dy.$$

The Lindeberg condition $[\mathbf{L2'}]$ follows from the assumption $nb_n^d \uparrow \infty$. In the following, we will show that $[\mathbf{PE'}]$ of Theorem 3.3.2 is satisfied under either Condition 7.1.1 or 7.1.2, and that the limit process $u \rightsquigarrow Z(u)$ is continuous with respect to the Euclidean metric.

[The case of smooth kernel.] Assume Condition 7.1.1. First notice that for any $u_1, u_2 \in U$

(7.1.3)
$$|K(y - u_1) - K(y - u_2)| \le L|u_1 - u_2|^{\alpha} \quad \forall y \in \mathbb{R}^d.$$

We can take a compact set S which is a common support of the functions $y \rightsquigarrow K(y-u)$ for all $u \in U$. Now, for every $\varepsilon > 0$, choose a finite partition $\Pi(\varepsilon) = \{U(\varepsilon; k) : 1 \le k \le N_{\Pi}(\varepsilon)\}$ of U such that the diameter of each partitioning set is not bigger than $\varepsilon^{1/\alpha}$. This can be done with $N_{\Pi}(\varepsilon) \le \text{const. } \varepsilon^{-d/\alpha}$; thus it holds that $\int_0^1 \sqrt{\log N_{\Pi}(\varepsilon)} d\varepsilon < \infty$. On the other hand, it follows from (7.1.3) that if $|u_1 - u_2| \le \varepsilon^{1/\alpha}$ then

$$\left|\xi_i^n(u_1) - \xi_i^n(u_2)\right| \le \frac{L\varepsilon}{\sqrt{nb_n^d}} \left\{ 1_S\left(\frac{X_i - x_0}{b_n}\right) + \int_{\mathbb{R}^d} 1_S\left(\frac{x - x_0}{b_n}\right) f(x)dx \right\}.$$

We thus have

$$\begin{aligned} \|\xi^n\|_{\Pi}^2 &\leq \frac{L^2}{nb_n^d} \sum_{i=1}^n E \left| \mathbf{1}_S \left(\frac{X_i - x_0}{b_n} \right) + \int_{\mathbb{R}^d} \mathbf{1}_S \left(\frac{x - x_0}{b_n} \right) f(x) dx \right|^2 \\ &\leq \frac{4L^2}{b_n^d} \int_{\mathbb{R}^d} \mathbf{1}_S \left(\frac{x - x_0}{b_n} \right) f(x) dx \\ &= 4L^2 \int_{\mathbb{R}^d} \mathbf{1}_S(y) f(x_0 + b_n y) dy \\ &\leq 4L^2 \cdot \operatorname{Leb}(S) \cdot \sup_{x \in N} f(x) \quad \text{for all sufficiently large } n \in \mathbb{N}, \end{aligned}$$

where N is a neighborhood of x_0 . The condition $[\mathbf{PE'}]$ of Theorem 3.3.2 has been established.

[The case of monotone kernel.] Assume Condition 7.1.2. For every p = 1, ..., d, choose a constant $c_p > 0$ such that $[-c_p, c_p]$ is a support of K_p and that $U \subset \prod_{p=1}^d (-c_p, c_p]$.

For every $\varepsilon > 0$ and every p = 1, ..., d, we introduce a finite partition $(-c_p, c_p] = \bigcup_{k_p=1}^{N_p(\varepsilon)} I_p(\varepsilon; k_p)$ where $I_p(\varepsilon; k_p) = (\gamma_p(\varepsilon; k_p - 1), \gamma_p(\varepsilon; k_p)]$, such that

$$0 < \gamma_p(\varepsilon; k_p) - \gamma_p(\varepsilon; k_p - 1) \le \varepsilon^2, \qquad k_p = 1, \dots, N_p(\varepsilon).$$

This can be done with $N_p(\varepsilon) \leq [2c_p\varepsilon^{-2}] + 1$. Now, to check the condition [**PE**'] of Theorem 3.3.2, we consider the DFP $\Pi = {\Pi(\varepsilon)}_{\varepsilon \in (0,1]}$ of U given by

$$\Pi(\varepsilon) = \left\{ U \cap \prod_{p=1}^{d} I_p(\varepsilon; k_p) : \quad 1 \le k_p \le N_p(\varepsilon), \ 1 \le p \le d \right\}.$$

Then, since

$$N_{\Pi}(\varepsilon) = \#\Pi(\varepsilon) \le \prod_{p=1}^{d} N_p(\varepsilon) \le \prod_{p=1}^{d} \left(\left[\frac{2c_p}{\varepsilon^2} \right] + 1 \right)$$

we have $\int_0^1 \sqrt{\log N_{\Pi}(\varepsilon)} d\varepsilon < \infty$.

Next, for every $u^{(p)} \in (-c_p, c_p]$ we define

$$K_p^{n,u^{(p)}}(x^{(p)}) = \frac{1}{\sqrt{b_n}} K_p\left(\frac{x^{(p)} - x_0^{(p)}}{b_n} - u^{(p)}\right), \qquad \forall x^{(p)} \in \mathbb{R}.$$

Then it holds that

(7.1.4)
$$|K_{p}^{n,u_{1}^{(p)}} - K_{p}^{n,u_{2}^{(p)}}| \le \overline{K}_{p}^{n,\varepsilon,k_{p}}$$
 whenever $u_{1}^{(p)}, u_{2}^{(p)} \in I_{p}(\varepsilon;k_{p}),$

where

$$\overline{K}_p^{n,\varepsilon,k_p}(x^{(p)}) = \begin{cases} \frac{K_p(0)}{\sqrt{b_n}}, & \text{if } \frac{x^{(p)} - x_0^{(p)}}{b_n} \in I_p(\varepsilon;k_p), \\ |K_p^{n,\gamma_p(\varepsilon;k_p)} - K_p^{n,\gamma_p(\varepsilon;k_p-1)}|(x^{(p)}), & \text{otherwise.} \end{cases}$$

The key points are the following:

(7.1.5)
$$\int_{\mathbb{R}} |\overline{K}_{p}^{n,\varepsilon,k_{p}}(x_{0}^{(p)}+b_{n}y^{(p)})|^{2}dy^{(p)} \leq \frac{3|K_{p}(0)|^{2}\varepsilon^{2}}{b_{n}}$$

(7.1.6)
$$\operatorname{Support}(\overline{K}_p^{n,\varepsilon,k_p}) \subset [x_0^{(p)} - 2b_n c_p, x_0^{(p)} + 2b_n c_p];$$

(7.1.7)
$$f_0 = \limsup_{n \to \infty} \sup_{y \in S} f(x_0 + b_n y) < \infty$$
, where $S = \prod_{p=1}^{a} [-2c_p, 2c_p]$.

The fact (7.1.5) will be proved later, while (7.1.6) and (7.1.7) are trivial.

Let us proceed with the main part of the proof. It follows from (7.1.4) that

$$\frac{1}{\sqrt{b_n^d}} \left| K\left(\frac{x - x_0}{b_n} - u_1\right) - K\left(\frac{x - x_0}{b_n} - u_2\right) \right|$$
$$= \left| \prod_{p=1}^d K_p^{n, u_1^{(p)}}(x^{(p)}) - \prod_{p=1}^d K_p^{n, u_2^{(p)}}(x^{(p)}) \right|$$

7. Miscellanies

$$= \left| \sum_{p=1}^{d} \left(\prod_{q=1}^{p-1} K_{q}^{n,u_{2}^{(q)}}(x^{(q)}) \right) \left(\prod_{q=p+1}^{d} K_{q}^{n,u_{1}^{(q)}}(x^{(q)}) \right) \left\{ K_{p}^{n,u_{1}^{(p)}}(x^{(p)}) - K_{p}^{n,u_{2}^{(p)}}(x^{(p)}) \right\} \right|$$

$$\leq \sum_{p=1}^{d} \left(\prod_{q \neq p} \frac{K_{q}(0)}{\sqrt{b_{n}}} \right) \left| K_{p}^{n,\varepsilon,k_{p}}(x^{(p)}) - K_{p}^{n,u_{2}^{(p)}}(x^{(p)}) \right|$$

$$\leq \sum_{p=1}^{d} \left(\prod_{q \neq p} \frac{K_{q}(0)}{\sqrt{b_{n}}} \right) \overline{K}_{p}^{n,\varepsilon,k_{p}}(x^{(p)}), \quad \text{if } u_{1}, u_{2} \in \prod_{p=1}^{d} I_{p}(\varepsilon;k_{p}).$$

Here, for every p = 1, ..., d, we obtain from (7.1.5), (7.1.6) and (7.1.7) that for all sufficiently large $n \in \mathbb{N}$

$$\begin{split} \left(\prod_{q\neq p} \frac{K_q(0)}{\sqrt{b_n}}\right)^2 \int_{\mathbb{R}^d} |\overline{K}_p^{n,\varepsilon,k_p}(x^{(p)})|^2 f(x) dx \\ &= \left(\prod_{q\neq p} |K_q(0)|^2\right) b_n \int_{\mathbb{R}^d} |\overline{K}_p^{n,\varepsilon,k_p}(x_0^{(p)} + b_n y^{(p)})|^2 f(x_0 + b_n y) dy \\ &\leq \left(\prod_{q\neq p} |K_q(0)|^2\right) b_n (f_0 + 1) \cdot \int_S |\overline{K}_p^{n,\varepsilon,k_p}(x_0^{(p)} + b_n y^{(p)})|^2 dy \\ &\leq \left(\prod_{q\neq p} |K_q(0)|^2\right) b_n (f_0 + 1) \cdot \left(\prod_{q\neq p}^d 4c_q\right) \frac{3|K_p(0)|^2 \varepsilon^2}{b_n} \\ &= \varepsilon^2 D_p, \quad \text{where} \quad D_p = \left(\prod_{q=1}^d 4c_q |K_q(0)|^2\right) \frac{3(f_0 + 1)}{4c_p}, \end{split}$$

which implies that

$$\int_{\mathbb{R}^d} \left| \sum_{p=1}^d \left(\prod_{q \neq p} \frac{K_q(0)}{\sqrt{b_n}} \right) \overline{K}_p^{n,\varepsilon,k_p}(x^{(p)}) \right|^2 f(x) dx \le \varepsilon^2 d \sum_{p=1}^d D_p.$$

We therefore have

$$\limsup_{n \to \infty} \|\xi^n\|_{\Pi} \le \sqrt{4d \sum_{p=1}^d D_p}.$$

It remains to prove (7.1.5). Observe that

$$\int_{\mathbb{R}} |\overline{K}_{p}^{n,\varepsilon,k_{p}}(x_{0}^{(p)}+b_{n}y^{(p)})|^{2} dy^{(p)} = (I) + \frac{|K_{p}(0)|^{2}}{b_{n}}\varepsilon^{2} + (II).$$

where:

$$(I) = \frac{1}{b_n} \int_{-\infty}^{\gamma_p(\varepsilon;k_p-1)} \left| K_p(y^{(p)} - \gamma_p(\varepsilon;k_p)) - K_p(y^{(p)} - \gamma_p(\varepsilon;k_p-1)) \right|^2 dy^{(p)};$$

$$(II) = \frac{1}{b_n} \int_{\gamma_p(\varepsilon;k_p)}^{\infty} \left| K_p(y^{(p)} - \gamma_p(\varepsilon;k_p)) - K_p(y^{(p)} - \gamma_p(\varepsilon;k_p-1)) \right|^2 dy^{(p)}.$$

Further, it holds that

$$(II) = \frac{1}{b_n} \int_0^\infty \left| K_p(y^{(p)}) - K_p(y^{(p)} + \gamma_p(\varepsilon, k_p) - \gamma_p(\varepsilon, k_p - 1)) \right|^2 dy^{(p)}$$

$$\leq \frac{1}{b_n} \int_0^\infty \left| K_p(y^{(p)}) - K_p(y^{(p)} + \varepsilon^2) \right|^2 dy^{(p)}$$

$$\leq \frac{K_p(0)}{b_n} \int_0^\infty \left| K_p(y^{(p)}) - K_p(y^{(p)} + \varepsilon^2) \right| dy^{(p)}$$

$$= \frac{K_p(0)}{b_n} \int_{-\varepsilon^2}^0 K_p(y^{(p)} + \varepsilon^2) dy^{(p)}$$

$$\leq \frac{K_p(0)}{b_n} \cdot K_p(0)\varepsilon^2.$$

Since the same bound holds also for (I), we get (7.1.5).

[Continuity of the limit process.] Theorem 3.3.2 says that the process $u \rightsquigarrow Z(u)$ is continuous with respect to the pseudo-metric ρ on U defined by

$$\rho(u_1, u_2) = \sqrt{\int_{\mathbb{R}^d} |K(x - u_1) - K(x - u_2)|^2 dx} \qquad \forall u_1, u_2 \in U.$$

Hence it suffices to show that $u_1 \rightsquigarrow \rho(u_1, u_2)$ is continuous at u_2 with respect to the Euclidean metric for every $u_2 \in U$. This is immediate from (7.1.3) in the case of a smooth kernel. On the other hand, in the case of a monotone kernel, the claim follows from the inequality

$$|K(x - u_1) - K(x - u_2)| \le \sum_{p=1}^d \left(\prod_{q \ne p} K_q(0)\right) \left| K_p(x^{(p)} - u_1^{(p)}) - K_p(x^{(p)} - u_2^{(p)}) \right|$$

which can be easily shown by the same argument as above.

Proof of Theorem 7.1.4. The assertion (i) is immediate from Proposition 7.1.3 and Theorem 1.6.1 of van der Vaart and Wellner (1996). Next, observe that

$$\widetilde{f}_n(x_0 + b_n u) - f(x_0 + b_n u) = \int_{\mathbb{R}^d} K(y) \left\{ f(x_0 + b_n(u + y)) - f(x_0 + b_n u) \right\} dy$$

and that

$$\begin{split} f(x_0 + b_n(u + y)) - f(x_0 + b_n u) &= b_n^d \sum_{p=1}^d y^{(p)} \left. \frac{\partial f(x)}{\partial x^{(p)}} \right|_{x = x_0 + b_n u} \\ &+ \frac{b_n^{2d}}{2} \sum_{p=1}^d \sum_{q=1}^d y^{(p)} y^{(q)} \left. \frac{\partial^2 f(x)}{\partial x^{(p)} \partial x^{(q)}} \right|_{x = \tilde{x}_n}, \end{split}$$

where \tilde{x}_n is a point on the segment connecting $x_0 + b_n u$ and $x_0 + b_n (u + y)$. We can obtain the assertion (ii) using the assumption that the kernel function K is symmetric.

7.1.2 Estimation of Mode

In this subsection, we consider the 1-dimensional case only. We are interested in estimating the mode of a density f of an i.i.d. data, namely, $x_0 = \operatorname{argmax}_{x \in \mathbb{R}} f(x)$. A natural estimator would be $\hat{\theta}^n = \operatorname{argmax}_{x \in \mathbb{R}} \hat{f}_n(x)$, where

$$\widehat{f}_n(x) = \frac{1}{nb_n} \sum_{i=1}^n K\left(\frac{X_i - x}{b_n}\right) \qquad \forall x \in \mathbb{R}.$$

Here, b_n is a vanishing sequence of positive constants, and K is a kernel function on \mathbb{R} . We now introduce a condition on f in a neighborhood of $x_0 \in \mathbb{R}$ (we do not assume that x_0 is the maximum point over the whole line \mathbb{R} ; the point x_0 should be regarded as a local mode of f).

Condition 7.1.5 For an even integer $p \ge 2$, the function $x \rightsquigarrow f(x)$ is p-times continuously differentiable in a neighborhood N of x_0 with derivatives $f^{(m)}$, m = 1, ..., p such that:

- $f^{(m)}(x_0) = 0$ for every m = 1, ..., p 1;
- $\sup_{x \in N} f^{(p)}(x) < 0.$

Theorem 7.1.6 For a given point $x_0 \in \mathbb{R}$, suppose that Condition 7.1.5 is satisfied for an even integer $p \geq 2$. Put the bandwidth $b_n = n^{-1/(2p+1)}$, and choose a kernel function K on \mathbb{R} following either of Condition 7.1.1 with $\alpha = 1$ or Condition 7.1.2. Then, for any \mathbb{R} -valued random sequence $\widehat{\theta}^n$ such that $\widehat{\theta}^n \xrightarrow{P} x_0$ and that

$$\widehat{f}_n(\widehat{\theta}^n) \ge \widehat{f}_n(x_0) - \epsilon_n$$

for some $\epsilon_n = O_{P^*}(n^{-p/(2p+1)})$, it holds that $|\widehat{\theta}^n - x_0| = O_{P^*}(n^{-1/(2p+1)})$.

Proof. We will check the conditions of Theorem 5.1.2 for $r_{n,u}^{-1} = b_n = n^{-1/(2p+1)}$. Formulation 5.1.1 should be as follows: (Θ^n, d_u^n) is the Euclidean space \mathbb{R} , and $\theta_u^n = x_0$; let $\Gamma^n(x) = \widehat{f_n}(x)$ and $\gamma^n(x) = f_n(x)$, where

$$f_n(x) = \frac{1}{b_n} \int_{\mathbb{R}} K\left(\frac{y-x}{b_n}\right) f(y) dy$$
$$= \int_{\mathbb{R}} K(y) f(x+b_n y) dy.$$

7.1. Local Random Fields of Kernel Estimators

(

We then denote $R(\delta) = \{x \in \mathbb{R} : (\delta/2) < |x - x_0| \le \delta\}$ for every $\delta \in (0, \infty)$.

To show the first inequality of M-criterion, we denote

$$c_p = \inf_{x \in N} |f^{(p)}(x)|$$
 and $C_p = \sup_{x \in N} |f^{(p)}(x)|.$

It follows from a *p*-term Taylor expansion of f around $x_0 + b_n y$ that

$$\gamma^{n}(x) - \gamma^{n}(x_{0}) = \int_{\mathbb{R}} K(y) \{ f(x + b_{n}y) - f(x_{0} + b_{n}y) \} dy$$

= (I) + (II),

where

$$(I) = \sum_{m=1}^{p-1} \frac{(x-x_0)^m}{m!} \int_{\mathbb{R}} K(y) f^{(m)}(x_0 + b_n y) dy$$

and $\left(a_{1}, a_{2}, a_{3} \right)$

$$II) \leq -\frac{|x-x_0|^p}{p!} \int_{\mathbb{R}} K(y) c_p dy$$

$$\leq -\frac{c_p}{p! 2^p} \delta^p \quad \text{whenever } x \in R(\delta).$$

Furthermore, since $f^{(k)}(x_0) = 0$ for k = 1, ..., p - 1, it follows from a (p - m)-term Taylor expansion of $f^{(m)}$ around x_0 that

$$f^{(m)}(x_0 + b_n y) = \frac{(b_n y)^{p-m}}{(p-m)!} f^{(p)}(\widetilde{x}_n) \quad \text{for some } \widetilde{x}_n \in N,$$

for every m = 1, ..., p - 1. Thus it holds that if $\delta \in [Lb_n, \infty)$ then

$$\begin{aligned} |(I)| &\leq \sum_{m=1}^{p-1} \frac{|x-x_0|^m}{m!} \cdot b_n^{p-m} \cdot \frac{C_p}{(p-m)!} \int_{\mathbb{R}} |y|^{p-m} K(y) dy \\ &\leq \delta^p \sum_{m=1}^{p-1} \frac{C_p}{m!(p-m)! L^{p-m}} \int_{\mathbb{R}} |y|^{p-m} K(y) dy \qquad \text{whenever } x \in R(\delta). \end{aligned}$$

Thus, choosing a sufficiently large L > 0 we can conclude that the FIRST INEQUALITY is satisfied.

To check the SECOND INEQUALITY, we will apply Theorem 2.3.3 (ii) to $\{\xi_i\}_{i\in\mathbb{N}} = \{(\xi_i^{n,x}|x\in\Psi)\}_{i\in\mathbb{N}}, \text{ where } \Psi \text{ is a subset of } \mathbb{R} \text{ and }$

$$\xi_i^{n,x} = \frac{1}{n} \left\{ \frac{1}{b_n} K\left(\frac{X_i - x}{b_n}\right) - f_n(x) \right\}.$$

Notice that

$$\begin{aligned} |\xi_i^{n,x} - \xi_i^{n,y}| &\leq \frac{1}{nb_n} \left| K\left(\frac{X_i - x}{b_n}\right) - K\left(\frac{X_i - y}{b_n}\right) \right| \\ &+ \frac{1}{nb_n} \int_{\mathbb{R}} \left| K\left(\frac{z - x}{b_n}\right) - K\left(\frac{z - y}{b_n}\right) \right| f(z) dz \end{aligned}$$

We may choose a sufficiently small $\delta_0 > 0$ so that $[x_0 - 2\delta_0, x_0 + 2\delta_0] \subset N$; then we have $f^* = \sup_{z \in [x_0 - 2\delta_0, x_0 + 2\delta_0]} f(z) < \infty$. We discuss the cases of smooth and monotone kernels, separately; in both cases, let L > 0 be a constant such that [-L, L] is a support of K. In each case, for every $\delta \in (0, \delta_0)$ we will construct a DFP $\Pi_{\delta} = {\Pi_{\delta}(\varepsilon)}_{\varepsilon \in (0,1]}$ of $\Psi = \Psi_{\delta} = [x_0 - \delta, x_0 + \delta]$ such that

(7.1.8)
$$\sup_{\delta \in (0,\delta_0)} \int_0^1 \sqrt{\log(1 + N_{\Pi_\delta}(\varepsilon))} d\varepsilon < \infty$$

and that

 $\|\xi^n\|_{\Pi_{\delta}} \leq \operatorname{const.}\varphi_1^n(\delta) \qquad \forall \delta \in [Lb_n, \delta_0), \ \forall n \in \mathbb{N}$

for some appropriate functions $\delta \rightsquigarrow \varphi_1^n(\delta)$ indexed by $n \in \mathbb{N}$. Notice that we have in both cases that with $\Psi_{\delta} = [x_0 - \delta, x_0 + \delta]$

$$\sum_{i=1}^{n} E |\overline{\xi}_{i}^{n}|^{2} \leq n \cdot \left| \frac{\sup_{x} K(x)}{nb_{n}} \right|^{2} \cdot (2\delta + 2Lb_{n}) \cdot f^{*}$$
$$\leq D \cdot \varphi_{2}^{n}(\delta) \quad \text{whenever } \delta \in [Lb_{n}, \delta_{0})$$

where $D = 4 \sup_x |K(x)|^2 f^*$ and $\varphi_2^n(\delta) = n^{-1} b_n^{-2} \delta$. Then, Theorem 2.3.3 (ii) yields that

$$E \sup_{x \in R(\delta)} |(\Gamma^n - \gamma^n)(x) - (\Gamma^n - \gamma^n)(x_0)| \le \text{const.}\phi^n(\delta) \qquad \forall \delta \in [Lb_n, \delta_0)$$

where $\phi^n(\delta) = \varphi_1^n(\delta) \vee (\varphi_2^n(\delta)/\varphi_1^n(\delta)).$

[The case of smooth kernel.] For every $\varepsilon \in (0, 1]$, we make a finite partition $(x_0 - \delta, x_0 + \delta] = \bigcup_{k=1}^{N_{\Pi_{\delta}}(\varepsilon)} (u_{k-1}, u_k]$ such that $u_k - u_{k-1} \leq \varepsilon \delta$. This can be done with $N_{\Pi_{\delta}}(\varepsilon) \leq [2\varepsilon^{-1}] + 1$, thus (7.1.8) is satisfied. On the other hand, if $x, y \in [u_{k-1}, u_k]$ with $u_k - u_{k-1} \leq \varepsilon \delta$, then

$$\left| K\left(\frac{z-x}{b_n}\right) - K\left(\frac{z-y}{b_n}\right) \right| \le \frac{L\varepsilon\delta}{b_n} \cdot \mathbf{1}_{[u_{k-1}-Lb_n, u_k+Lb_n]}(z) \qquad \forall z \in \mathbb{R},$$

where L > 0 is a constant appearing Condition 7.1.1. Thus it holds that if $Lb_n \leq \delta < \delta_0$ then

$$\begin{aligned} \|\xi^n\|_{\Pi_{\delta}} &\leq \sqrt{4n \cdot \frac{1}{n^2 b_n^2} \cdot \left|\frac{L\delta}{b_n}\right|^2 \cdot (\delta + 2Lb_n) \cdot f^*} \\ &\leq \sqrt{12L^2 f^*} \cdot \varphi_1^n(\delta) \end{aligned}$$

where $\varphi_1^n(\delta) = n^{-1/2} b_n^{-2} \delta^{3/2}$. We thus have $\phi^n(\delta) = (n^{-1/2} b_n^{-2} \delta^{3/2}) \vee (n^{-1/2} \delta^{-1/2})$. The relation $\phi^n(b_n) = b_n^p$ holds since we put $b_n = n^{-1/(2p+1)}$.

[The case of monotone kernel.] For every $\varepsilon \in (0,1]$, we make a finite partition $(x_0 - \delta, x_0 + \delta] = \bigcup_{k=1}^{N_{\Pi_{\delta}}(\varepsilon)} (u_{k-1}, u_k]$ such that $u_k - u_{k-1} \leq \varepsilon^2 \delta$. This can be done with

 $N_{\Pi_{\delta}}(\varepsilon) \leq [2\varepsilon^{-2}] + 1$, thus (7.1.8) is satisfied. On the other hand, if $x, y \in [u_{k-1}, u_k]$ then

$$\left| K\left(\frac{z-x}{b_n}\right) - K\left(\frac{z-y}{b_n}\right) \right| \le \left| K\left(\frac{z-u_{k-1}}{b_n}\right) - K\left(\frac{z-u_k}{b_n}\right) \right| \qquad \forall z \in \mathbb{R}.$$

Here notice that if $u_k - u_{k-1} \leq \varepsilon^2 \delta$ then

$$\begin{split} \int_{\mathbb{R}} \left| K\left(\frac{z-u_{k-1}}{b_n}\right) - K\left(\frac{z-u_k}{b_n}\right) \right|^2 dz &= \int_{\mathbb{R}} \left| K\left(\frac{z}{b_n}\right) - K\left(\frac{z-\varepsilon^2\delta}{b_n}\right) \right|^2 dz \\ &\leq K(0) \int_{\mathbb{R}} \left| K\left(\frac{z}{b_n}\right) - K\left(\frac{z-\varepsilon^2\delta}{b_n}\right) \right| dz \\ &\leq 3|K(0)|^2 \varepsilon^2 \delta. \end{split}$$

Thus it holds that if $Lb_n \leq \delta < \delta_0$ then

$$\begin{aligned} \|\xi^{n}\|_{\Pi_{\delta}} &\leq \sqrt{4n \cdot \frac{1}{n^{2}b_{n}^{2}} \cdot 3|K(0)|^{2}\delta \cdot f^{*}} \\ &\leq \sqrt{12|K(0)|^{2}f^{*}} \cdot \varphi_{1}^{n}(\delta) \end{aligned}$$

where $\varphi_1^n(\delta) = n^{-1/2} b_n^{-1} \delta^{1/2}$. We thus have $\phi^n(\delta) = n^{-1/2} b_n^{-1} \delta^{1/2}$. The relation $\phi^n(b_n) = b_n^p$ holds since we put $b_n = n^{-1/(2p+1)}$.

Corollary 7.1.7 For a given point $x_0 \in \mathbb{R}$, suppose that Condition 7.1.5 is satisfied for an even integer $p \geq 2$. Put the bandwidth $b_n = n^{-1/(2p+1)}$, and set $K(x) = \frac{1}{2} \cdot 1_{[-1,1]}(x)$. Then, for any \mathbb{R} -valued random sequence $\widehat{\theta}^n$ such that $\widehat{\theta}^n \xrightarrow{P^*} x_0$ and that

$$\widehat{f}_n(\widehat{\theta}^n) \ge \sup_{x \in \mathbb{R}} \widehat{f}_n(x) - \epsilon_n$$

for some $\epsilon_n = o_{P^*}(n^{-p/(2p+1)})$, it holds that $n^{1/(2p+1)}(\widehat{\theta}^n - x_0) \stackrel{P}{\Longrightarrow} \operatorname{argmax}_{h \in \mathbb{R}} \{\mathbb{A}(h) + \mathbb{B}(h+1) - \mathbb{B}(h-1)\}$, where the deterministic process $h \rightsquigarrow \mathbb{A}(h)$ is given by

$$\mathbb{A}(h) = \frac{f^{(p)}(x_0)}{\sqrt{f(x_0)}p!} \int_{-1}^{1} (h+y)^p dy$$

and $h \rightsquigarrow \mathbb{B}(h)$ is the two-sided Brownian motion.

Proof. Theorem 7.1.6 asserts that the sequence $b_n^{-1}(\widehat{\theta}^n - x_0)$ is uniformly tight. Let us consider the stochastic process $h \rightsquigarrow \mathbb{M}^n(h)$ defined by

$$\mathbb{M}^{n}(h) = b_{n}^{-p} \left\{ \widehat{f}_{n}(x_{0} + b_{n}h) - f(x_{0}) \right\}$$

= $Y^{n}(h) + Z^{n}(h),$

where:

$$Y^{n}(h) = b_{n}^{-p} \{ f_{n}(x_{0} + b_{n}h) - f(x_{0}) \} ;$$

$$Z^{n}(h) = b_{n}^{-p} \{ \widehat{f}_{n}(x_{0} + b_{n}h) - f_{n}(x_{0} + b_{n}h) \} .$$

Noting that $\sqrt{nb_n} = b_n^{-p}$, we obtain from Proposition 7.1.3 that $Z^n \stackrel{P}{\Longrightarrow} Z$ in $\ell^{\infty}(K)$ for any compact set $K \subset \mathbb{R}$, where $Z(h) = \sqrt{f(x_0)} \{\mathbb{B}(h+1) - \mathbb{B}(h-1)\}/2$. On the other hand, an easy computation shows that $\lim_{n\to\infty} Y^n(h) = Y(h) = \sqrt{f(x_0)}\mathbb{A}(h)/2$ for every $h \in \mathbb{R}$. Since $h \rightsquigarrow Y^n(h)$ and $h \rightsquigarrow Y(h)$ are continuous, this convergence is uniform on every compact set $K \subset \mathbb{R}$. Hence, by the same argument as the last part of the proof of Proposition 5.2.5, we can obtain the assertion. \Box

7.1.3 Remarks for Non-I.I.D. Cases

Gaussian White Noise Model

For every $n \in \mathbb{N}$, let $X^n = (X_t^n)_{t \in [0,1]}$ be a continuous stochastic process given by

$$dX_t^n = f(t)dt + n^{-1/2}dB_t,$$

where $f \in \mathcal{L}^2[0,1]$, and $B = (B_t)_{t \in [0,1]}$ is a standard Brownian motion. Let $t_0 \in (0,1)$ be a fixed point, and let b_n be a vanishing sequence of positive constants. In order to estimate the local function $u \rightsquigarrow f(t_0 + b_n u)$, a natural estimator would be

$$\widehat{f}_n(t_0 + b_n u) = \frac{1}{nb_n} \int_0^1 K\left(\frac{t - t_0}{b_n} - u\right) dX_t^n$$

where K is a kernel function on \mathbb{R} satisfying either of Condition 7.1.1 or 7.1.2 with d = 1. Then, we can get the same conclusions as Theorem 7.1.4 by using Theorem 3.4.4.

Multiplicative Intensity Model

For every $n \in \mathbb{N}$, let μ^n be an \mathbb{R}^d -valued multivariate point process on a stochastic basis $\mathbf{B}^n = (\Omega^n, \mathcal{F}^n, \mathbf{F}^n, P^n)$; μ^n can be identified with an \mathbb{R}^d -marked point process $\{(T_i^n, Z_i^n); i \in \mathbb{N}\}$ through the equality

$$\mu^{n}(\omega; dt, dz) = \sum_{i} \varepsilon_{(T_{i}^{n}(\omega), Z_{i}^{n}(\omega))}(dt, dz),$$

where $0 < T_1^n < T_2^n < \cdots$ and each Z_i^n is an \mathbb{R}^d -valued random variable.

We assume that the predictable compensator ν^n of μ^n is given by

$$\nu^{n}(\omega; dt, dz) = \alpha(t, z)Y^{n}(\omega, t, z)dtdz,$$

where $\alpha(t, z)$ is a $[0, \infty)$ -valued measurable function on $\mathbb{R}_+ \times \mathbb{R}^d$, and $Y^n(\omega, t, z)$ is a $[0, \infty)$ -valued predictable function on $\Omega^n \times \mathbb{R}_+ \times \mathbb{R}^d$.

Let $(t_0, z_0) \in (0, \infty) \times \mathbb{R}^d$ be a fixed point, and let b_n and c_n be vanishing sequences of positive constants. In order to estimate the local function $(u, v) \rightsquigarrow \alpha(t_0 + b_n u, z_0 + c_n v)$, a natural estimator would be

$$\widehat{\alpha}_{n}(t_{0}+b_{n}u,z_{0}+c_{n}v) = \frac{1}{nb_{n}c_{n}^{d}} \int_{\mathbb{R}_{+}\times\mathbb{R}^{d}} K\left(\frac{t-t_{0}}{b_{n}}-u,\frac{z-z_{0}}{c_{n}}-v\right) Y^{n-}(\cdot,t,z)\mu^{n}(\cdot;dt,dz),$$

where K is a kernel function on \mathbb{R}^{d+1} satisfying either of Condition 7.1.1 or 7.1.2 with "d" replaced by "d + 1", and where Y^{n-} is the generalized inverse of Y^n .

Assume the "local" version of Condition 5.3.1, that is:

Condition 7.1.8 There exists a measurable function y = y(t, z) on a neighborhood N of (t_0, z_0) such that

$$\sup_{(t,z)\in N} \left| n^{-1} Y^n(\cdot,t,z) - y(t,z) \right| \xrightarrow{P^{n*}} 0.$$

Then, under some conditions of smoothness of the functions $(t, z) \rightsquigarrow \alpha(t, z)$ and $(t, z) \rightsquigarrow y(t, z)$, we can derive some conclusions about the residual processes $(u, v) \rightsquigarrow R^n(u, v)$ given by

$$R^{n}(u,v) = \sqrt{nb_{n}c_{n}^{d}} \{ \widehat{\alpha}_{n}(t_{0} + b_{n}u, z_{0} + c_{n}v) - \alpha(t_{0} + b_{n}u, z_{0} + c_{n}v) \}$$

similarly to those of Theorem 7.1.4 by using Theorem 3.2.4; the term " $f(x_0)$ " in (7.1.1) is replaced by " $\alpha(t_0, z_0)/y(t_0, z_0)$ ", while the change of (7.1.2) is clear.

7.2 Log-likelihood Ratio Random Fields

7.2.1 Results

For every $n \in \mathbb{N}$, let $\mathbf{B}^n = (\Omega^n, \mathcal{F}^n, \mathbf{F}^n = \{\mathcal{F}^n_i\}_{i \in \mathbb{N}_0}, P^n)$ be a discrete-time stochastic basis. Let $\mathbf{P}^n = \{P^{n,\psi} : \psi \in \Psi\}$ be a family of probability measures on $(\Omega^n, \mathcal{F}^n)$, indexed by an arbitrary set Ψ , such that $P^{n,\psi} \ll P^n$ for every $\psi \in \Psi$. We denote

$$Z_i^{n,\psi} = \frac{dP_i^{n,\psi}}{dP_i^n},$$

where $P_i^{n,\psi}$ (resp. P_i^n) is the restriction of $P^{n,\psi}$ (resp. P^n) on the σ -field \mathcal{F}_i^n . We assume $P_0^{n,\psi} = P_0^n$ for every $\psi \in \Psi$, hence we can set $Z_0^{n,\psi} = 1$. For a given finite stopping time σ^n on \mathbf{B}^n , and we suppose also that the random element $\log Z_{\sigma^n}^n = (\log Z_{\sigma^n}^{n,\psi} | \psi \in \Psi)$ takes values in $\ell^{\infty}(\Psi)$. Here we set

$$\zeta_i^{n,\psi} = \sqrt{\frac{Z_{i\wedge\sigma^n}^{n,\psi}}{Z_{(i-1)\wedge\sigma^n}^{n,\psi}}} - 1 \qquad \forall i \in \mathbb{N} \ \forall \psi \in \Psi.$$

(b) $\sum_{i=1}^{\sigma^n} E_{i-1}^n |\overline{\zeta_i^n}|^2 \mathbf{1}_{\{|\overline{\zeta_i^n}| > \varepsilon\}} \xrightarrow{P^n} 0$ for every $\varepsilon > 0$;

(c) there exists a DFP Π of Ψ such that

$$\|\zeta^n\|_{\Pi,\sigma^n} = O_{P^n}(1) \quad and \quad \int_0^{\Delta_{\Pi}} H_{\Pi}(\varepsilon) d\varepsilon < \infty.$$

Then, it holds that $\log Z_{\sigma^n}^n \stackrel{P^n}{\Longrightarrow} X$ in $\ell^{\infty}(\Psi)$, where $X(\psi) = -\frac{1}{2}C(\psi,\psi) + G(\psi)$ and $\psi \rightsquigarrow G(\psi)$ is a zero-mean Gaussian process such that $EG(\psi)G(\phi) = C(\psi,\phi)$. Furthermore, the formula

$$\rho(\psi,\phi) = \sqrt{C(\psi,\psi) + C(\phi,\phi) - 2C(\psi,\phi)} \qquad \forall \psi,\phi \in \Psi$$

defines a semimetric on Ψ such that (Ψ, ρ) is totally bounded and that almost all paths of X are uniformly ρ -continuous.

Remark. If a version of the conditional expectation $E_{i-1}^n \zeta_i^{n,\psi} \zeta_i^{n,\phi}$ satisfies the assumption (a₁), then so does any version. However, this is not true in (a₂); the assumption means that there exist *some* versions of $E_{i-1}^n |\zeta_i^{n,\psi}|^2$'s which satisfy the requirement.

Example: Ergodic Markov chains

Let $\{X_i\}_{i\in\mathbb{N}_0}$ be an ergodic Markov chain, defined on a probability space (Ω, \mathcal{F}, P) , with values in an arbitrary state space (E, \mathcal{E}) . Let $\mu(dx)$ denote the initial distribution, p(x, dy) the transition distribution, and $\pi(dx)$ the invariant distribution. Let us equip the space $\mathcal{L}^2 = \mathcal{L}^2(E \times E, \pi(dx)p(x, dy))$ with the "inner product" given by

$$\langle h_1, h_2 \rangle_{\mathcal{L}^2} = \int_{E \times E} h_1(x, y) h_2(x, y) \pi(dx) p(x, dy) \qquad \forall h_1, h_2 \in \mathcal{L}^2$$

The meaning of the quotation marks is that $||h||_{\mathcal{L}^2} = \sqrt{\langle h, h \rangle_{\mathcal{L}^2}}$ is merely a "semi-" norm. Next we define the subset \mathcal{L}^2_0 of \mathcal{L}^2 by

$$\mathcal{L}_0^2 = \left\{ h \in \mathcal{L}^2 : \quad \int_E h(x, y) p(x, dy) = 0 \quad \forall x \in E \quad \text{and} \quad h > -1 \right\}$$

Fix a subset $\mathcal{H} \subset \mathcal{L}_0^2$. For every $n \in \mathbb{N}$, let us consider a family of probability measures $\mathbf{P}^n = \{P^{n,h} : h \in \mathcal{H}\}$ on (Ω, \mathcal{F}) such that: under $P^{n,h}$, the process $\{X_i\}_{i \in \mathbb{N}_0}$ is the Markov chain with initial distribution μ and transition distribution $p^{n,h}$ given by

$$p^{n,h}(x,dy) = \left(1 + \frac{h(x,y)}{\sqrt{n}}\right)p(x,dy).$$

7.2. Log-likelihood Ratio Random Fields

Here we set $\mathcal{F}_i = \sigma\{X_0, ..., X_i\}$. Then it holds that

$$Z_i^{n,h} = \frac{dP_i^{n,h}}{dP_i} = \prod_{j=1}^i \left(1 + \frac{h(X_{j-1}, X_j)}{\sqrt{n}} \right).$$

We need some more notations to state the following result, which concerns the asymptotic behavior of the process $\log Z_n^n = (\log Z_n^{n,h} | h \in \mathcal{H})$. For a given $K \in \mathcal{L}^2(E, \pi(dx))$ we define the semimetric ρ_K on \mathcal{L}^2 by

$$\rho_K(h_1, h_2) = \sup_{x \in E} \frac{\rho_x(h_1, h_2)}{|K(x)| \vee 1} \qquad \forall h_1, h_2 \in \mathcal{L}^2,$$

where

$$\rho_x(h_1, h_2) = \sqrt{\int_E |h_1(x, y) - h_2(x, y)|^2 p(x, dy)} \qquad \forall x \in E$$

Proposition 7.2.2 Let $\{X_i\}_{i\in\mathbb{N}_0}$, $(\Omega, \mathcal{F}, \mathbf{F} = \{\mathcal{F}_i\}_{i\in\mathbb{N}_0}, P)$ and $\mathbf{P}^n = \{P^{n,h} : h \in \mathcal{H}\}$ as above be given. Suppose that there exists $h^* \in \mathcal{L}^4(E \times E, \pi(dx)p(x, dy))$ such that $\sup_{h\in\mathcal{H}} |h| \leq h^*$, and also that there exists $K \in \mathcal{L}^2(E, \pi(dx))$ such that

$$\int_0^1 \sqrt{\log N_{[]}(\mathcal{H}, \rho_K; \varepsilon)} d\varepsilon < \infty$$

Then, it holds that $\log Z_n^n \stackrel{P}{\Longrightarrow} X$ in $\ell^{\infty}(\mathcal{H})$, where $X(h) = -\frac{1}{2} \|h\|_{\mathcal{L}^2}^2 + G(h)$ and $h \rightsquigarrow G(h)$ is a zero-mean Gaussian process such that $EG(h_1)G(h_2) = \langle h_1, h_2 \rangle_{\mathcal{L}^2}$. Furthermore, almost all paths of X are uniformly $\|\cdot\|_{\mathcal{L}^2}$ -continuous.

This result is easily derived from the ergodic theorem and Theorem 7.2.1, hence the proof is omitted. Here we give a statistical application. Fix a subset $\mathcal{H} \subset \mathcal{L}_0^2$ such that $\|h\|_{\mathcal{L}^2} > 0$ for every $h \in \mathcal{H}$. Let us consider the testing problem:

hypothesis
$$H_0$$
: p
against H_1^n : $p^{n,h}$ for some $h \in \mathcal{H}$

We propose the test statistics

$$S^{n} = \sup_{h \in \mathcal{H}} \left| \frac{1}{2} \|h\|_{\mathcal{L}^{2}}^{2} + \log Z_{n}^{n,h} \right|.$$

Assume the same conditions as in Proposition 7.2.2. Then, it holds that

$$S^n \xrightarrow{p^{n,u}} \sup_{h \in \mathcal{H}} |\langle h, u \rangle_{\mathcal{L}^2} + G(h)| \quad \text{in } \mathbb{R} \quad \forall u \in \{0\} \cup \mathcal{H},$$

where the process $h \rightsquigarrow G(h)$ is as above. This fact follows from Proposition 7.2.2 that implies local asymptotic normality and contiguity, together with Le Cam's third lemma and the continuous mapping theorem. In view of Anderson's lemma (e.g., Lemma 3.11.4 of van der Vaart and Wellner (1996)), the statistics S^n seems reasonable.

7.2.2 Proofs

Let us denote:

$$\begin{split} \tilde{Z}_{i}^{n,\psi} &= \frac{Z_{i\wedge\sigma^{n}}^{n,\psi}}{Z_{(i-1)\wedge\sigma^{n}}^{n,\psi}} \quad \forall i \in \mathbb{N} \; \forall \psi \in \Psi; \\ \lambda_{i}^{n,a,\psi} &= \log \tilde{Z}_{i}^{n,\psi} \mathbf{1}_{\{\overline{\zeta}_{i}^{n} \leq a\}} \quad \forall i \in \mathbb{N} \; \forall \psi \in \Psi \; \forall a > 0. \end{split}$$

The process $\psi \rightsquigarrow \log Z_{\sigma^n}^{n,\psi} = \sum_{i=1}^{\sigma^n} \log \tilde{Z}_i^{n,\psi}$ can be well-approximated by the process $\psi \rightsquigarrow \Lambda^{n,a,\psi} = \sum_{i=1}^{\sigma^n} \lambda_i^{n,a,\psi}$. As a matter of fact, it holds that

$$\sup_{\psi \in \Psi} \mathbf{1}_{\{ \log Z_{\sigma^n}^{n,\psi} \neq \Lambda^{n,a,\psi} \}} \leq \sum_{i=1}^{\sigma^n} \mathbf{1}_{\{\overline{\zeta}_i^n > a\}}$$
$$\leq \frac{1}{a^2} \sum_{i=1}^{\sigma^n} |\overline{\zeta}_i^n|^2 \mathbf{1}_{\{\overline{\zeta}_i^n > a\}}$$

hence using also Lenglart's inequality we obtain

$$\sup_{\psi \in \Psi} |\log Z^{n,\psi}_{\sigma^n} - \Lambda^{n,a,\psi}| \xrightarrow{P^{n*}} 0.$$

We consider the decomposition

(7.2.1)
$$\Lambda^{n,a,\psi} = \sum_{i=1}^{\sigma^n} E_{i-1}^n \lambda_i^{n,a,\psi} + \sum_{i=1}^{\sigma^n} \left\{ \lambda_i^{n,a,\psi} - E_{i-1}^n \lambda_i^{n,a,\psi} \right\}.$$

We will derive the uniform convergence of the first term in (outer) probability, and apply Theorem 3.3.2 to the martingale difference array $\{\xi_i^n\}_{i\in\mathbb{N}_0}$ of the second term, that is, $\xi_i^{n,\psi} = \lambda_i^{n,a,\psi} - E_{i-1}^n \lambda_i^{n,a,\psi}$. We use the following lemma which will be proved later.

Lemma 7.2.3 For every $a \in (0,1)$, there exist some versions of the conditional expectations $E_{i-1}^n \lambda_i^{n,a,\psi}$ such that:

(i) if
$$\sup_{\psi \in \Psi} C(\psi, \psi) < \infty$$
 then $\sup_{\psi \in \Psi} \left| \sum_{i=1}^{\sigma^n} E_{i-1}^n \lambda_i^{n,a,\psi} + \frac{1}{2} C(\psi, \psi) \right| \xrightarrow{P^n} 0;$
(ii) $\sum_{i=1}^{\sigma^n} E_{i-1}^n \lambda_i^{n,a,\psi} \lambda_i^{n,a,\phi} \xrightarrow{P^n} C(\psi, \phi)$ for every $\psi, \phi \in \Psi;$
(iii) $\sum_{i=1}^{\sigma^n} |E_{i-1}^n \lambda_i^{n,a,\psi}|^2 \xrightarrow{P^n} 0$ for every $\psi \in \Psi.$

Remark. (i) We will see later that the conditions of the theorem actually implies that $\sup_{\psi \in \Psi} C(\psi, \psi) < \infty$. (ii) The choice of versions of the conditional expectations $E_{i-1}^n \lambda_i^{n,a,\psi} \lambda_i^{n,a,\phi}$ is not important.

Let us proceed with the main part of the proof. It is clear that there exists a constant $\delta \in (0,1)$ such that $|\log x - \log y| \le 2|\sqrt{x} - \sqrt{y}|$ whenever $x, y \in [1-\delta, 1+\delta]$. We consider

the decomposition (7.2.1) for $a = \sqrt{1+\delta} - 1$; then it holds that $\{x : |\sqrt{x} - 1| \le a\} \subset \{x : |x - 1| \le \delta\}.$

First we show the weak convergence of the second term of the decomposition (7.2.1). The condition $[\mathbf{C2'}]$ is direct from (ii) and (iii) of Lemma 7.2.3. It is also easy to see that the assumption (b) implies the Lindeberg condition $[\mathbf{L2'}]$. Finally, recalling the choice of δ and the relationship between a and δ , we have for any subset $\Psi' \subset \Psi$

$$\begin{split} E_{i-1}^{n} \left[\sup_{\psi,\phi\in\Psi'} |\lambda_{i}^{n,a,\psi} - \lambda_{i}^{n,a,\phi}| \right]_{\mathcal{F}_{i}^{n},P^{n}}^{2} \\ &= E_{i-1}^{n} \left[\sup_{\psi,\phi\in\Psi} |\log\tilde{Z}_{i}^{n,\psi} - \log\tilde{Z}_{i}^{n,\phi}| \mathbf{1}_{\{\overline{\zeta}_{i}^{n}\leq a\}} \right]_{\mathcal{F}_{i}^{n},P^{n}}^{2} \\ &\leq E_{i-1}^{n} \left[\sup_{\psi,\phi\in\Psi'} 2|\zeta_{i}^{n,\psi} - \zeta_{i}^{n,\phi}| \right]_{\mathcal{F}_{i}^{n},P^{n}}^{2} . \end{split}$$

Thus the assumption (c) implies the condition [**PE**']. Consequently, Theorem 3.3.2 yields $\sum_{i}^{n} \xi_{i}^{n} \stackrel{P^{n}}{\Longrightarrow} G$ in $\ell^{\infty}(\Psi)$.

Next we consider the first term of the decomposition. Observe that

$$\begin{split} \sqrt{C(\psi,\psi)} &= \sqrt{E|G(\psi)|^2} \\ &\leq \sqrt{E|G(\psi) - G(\phi)|^2} + \sqrt{E|G(\phi)|^2} = \rho(\psi,\phi) + \sqrt{C(\psi,\phi)}. \end{split}$$

The inequality above and the total boundedness of (Ψ, ρ) , a consequence of Theorem 3.3.2, imply that $\sup_{\psi \in \Psi} C(\psi, \psi) < \infty$. Hence (i) of Lemma 7.2.3 works to show the uniform convergence of the first term of (7.2.1). Also, it is trivial from the above inequality that $\psi \sim \sqrt{C(\psi, \psi)}$ is uniformly ρ -continuous, thus so is $\psi \sim C(\psi, \psi)$.

Proof of Lemma 7.2.3. For every $\varepsilon > 0$ we denote:

$$B^{n,\varepsilon}(\psi) = \sum_{i=1}^{\sigma^n} E_{i-1}^n \lambda_i^{n,\varepsilon,\psi};$$
$$C^{n,\varepsilon}(\psi,\phi) = \sum_{i=1}^{\sigma^n} E_{i-1}^n \lambda_i^{n,\varepsilon,\psi} \lambda_i^{n,\varepsilon,\phi}.$$

[STEP 1] First we prove the following facts: for given $a \in (0,1)$ there exist constants $K_1, K_2, K_3 > 0$ such that for every $\varepsilon \in (0, a]$

(7.2.2)
$$\sup_{\psi \in \Psi} \left| B^{n,\varepsilon}(\psi) + \frac{1}{2}C(\psi,\psi) \right| \le \varepsilon K_1 + o_{P^n}(1),$$

(7.2.3)
$$|C^{n,\varepsilon}(\psi,\phi) - C(\psi,\phi)| \le \varepsilon K_2 + o_{P^n}(1) \quad \forall \psi,\phi \in \Psi,$$

7. Miscellanies

(7.2.4)
$$\sum_{i=1}^{\sigma^n} |E_{i-1}^n \lambda_i^{n,\varepsilon,\psi}| \le \frac{1}{2} C(\psi,\psi) + \varepsilon K_3 + o_{P^n}(1) \qquad \forall \psi \in \Psi.$$

In order to show (7.2.2), first notice that there exists a constant K > 0 such that $|\log x - (x - 1) + 2(\sqrt{x} - 1)^2| \le K|\sqrt{x} - 1|^3$ whenever $|\sqrt{x} - 1| \le a$. Hence, for fixed $\varepsilon \in (0, a]$ we obtain

$$\left| B^{n,\varepsilon}(\psi) + 2\sum_{i=1}^{\sigma^n} E_{i-1}^n |\zeta_i^{n,\psi}|^2 \mathbf{1}_{\{\overline{\zeta}_i^n \le \varepsilon\}} \right|$$

$$\leq \varepsilon K \sum_{i=1}^{\sigma^n} E_{i-1}^n |\zeta_i^{n,\psi}|^2 \mathbf{1}_{\{\overline{\zeta}_i^n \le \varepsilon\}} + \left| \sum_{i=1}^{\sigma^n} E_{i-1}^n (\tilde{Z}_i^{n,\psi} - 1) \mathbf{1}_{\{\overline{\zeta}_i^n \le \varepsilon\}} \right|$$

almost surely. Since $E_{i-1}^n(\tilde{Z}_i^{n,\psi}-1)=0$ almost surely, the last term on the right hand side equals to

$$\begin{aligned} \left| \sum_{i=1}^{\sigma^{n}} E_{i-1}^{n} (\tilde{Z}_{i}^{n,\psi} - 1) \mathbf{1}_{\{\overline{\zeta}_{i}^{n} > \varepsilon\}} \right| \\ &\leq \sum_{i=1}^{\sigma^{n}} E_{i-1}^{n} |\tilde{Z}_{i}^{n,\psi} - 1| \mathbf{1}_{\{\overline{\zeta}_{i}^{n} > \varepsilon\}} \\ &\leq \frac{\varepsilon + 2}{\varepsilon} \sum_{i=1}^{\sigma^{n}} E_{i-1}^{n} |\zeta_{i}^{n,\psi}|^{2} \mathbf{1}_{\{\overline{\zeta}_{i}^{n} > \varepsilon\}} \end{aligned}$$

almost surely. Thus we obtain

(7.2.5)
$$\begin{vmatrix} B^{n,\varepsilon}(\psi) + \frac{1}{2}C(\psi,\psi) \end{vmatrix}$$
$$\leq (2+\varepsilon K) \left| \sum_{i=1}^{\sigma^n} E_{i-1}^n |\zeta_i^{n,\psi}|^2 - \frac{1}{4}C(\psi,\psi) \right| + \varepsilon K \cdot \frac{1}{4}C(\psi,\psi)$$
$$+ \left(2 + \frac{\varepsilon+2}{\varepsilon}\right) \sum_{i=1}^{\sigma^n} E_{i-1}^n |\overline{\zeta}_i^n|^2 \mathbf{1}_{\{\overline{\zeta}_i^n > \varepsilon\}}$$

almost surely. In order to get the estimate for all $\omega \in \Omega^n$, we can choose the versions of conditional expectations as follows: first, we may without loss of generality choose a version of $E_{i-1}^n |\overline{\zeta}_i^n|^2 \mathbf{1}_{\{\overline{\zeta}_i^n > \varepsilon\}}$ which is non-negative identically; next, on the union of all exceptional sets for the estimates appeared above, we define the values of all other conditional expectations as zero. Then, the inequality (7.2.5) holds *identically* for all $\psi \in \Psi$. By taking the supremum of (7.2.5) with respect to $\psi \in \Psi$, and letting $n \to \infty$, we obtain the assertion (7.2.2).

A similar argument yields (7.2.4). In fact, it is much easier than (7.2.2), because the assertion of (7.2.4) is ψ -wise, for which we do not need any argument about versions of

130

conditional expectations. Also, it is easy to show (7.2.3) if we notice the following fact: for given $a \in (0, 1)$ there exists a constant K > 0 such that $|\log x \cdot \log y - 4(\sqrt{x}-1)(\sqrt{y}-1)| \le K \max\{|\sqrt{x}-1|^3, |\sqrt{y}-1|^3\}$ whenever $\max\{|\sqrt{x}-1|, |\sqrt{y}-1|\} \le a$.

[STEP 2] Next we prove the following facts:

(7.2.6)
$$\sup_{\psi \in \Psi} |B^{n,a}(\psi) - B^{n,\varepsilon}(\psi)| \xrightarrow{P^{n*}} 0 \qquad \forall \varepsilon \in (0,a);$$

(7.2.7)
$$|C^{n,a}(\psi,\phi) - C^{n,\varepsilon}(\psi,\phi)| \xrightarrow{P^n} 0 \qquad \forall \psi,\phi \in \Psi \ \forall \varepsilon \in (0,a)$$

In order to show (7.2.6), notice that for given $a \in (0, 1)$ there exists a constant K > 0such that $|\log x| \le K |\sqrt{x} - 1|^2$ whenever $|\sqrt{x} - 1| \le a$. For every $\varepsilon \in (0, a)$ it holds that

(7.2.8)

$$\begin{aligned} \left| \sum_{i=1}^{\sigma^{n}} E_{i-1}^{n} \lambda_{i}^{n,a,\psi} - \sum_{i=1}^{\sigma^{n}} E_{i-1}^{n} \lambda_{i}^{n,\varepsilon,\psi} \right| \\
&= \left| \sum_{i=1}^{\sigma^{n}} E_{i-1}^{n} \log \tilde{Z}_{i}^{n,\psi} \mathbf{1}_{\{\varepsilon < \overline{\zeta}_{i}^{n} \le a\}} \right| \\
&\leq K \sum_{i=1}^{\sigma^{n}} E_{i-1}^{n} |\zeta_{i}^{n,\psi}|^{2} \mathbf{1}_{\{\varepsilon < |\zeta_{i}^{n}| \le a\}} \\
&\leq K \sum_{i=1}^{\sigma^{n}} E_{i-1}^{n} |\overline{\zeta}_{i}^{n}|^{2} \mathbf{1}_{\{\overline{\zeta}_{i}^{n} > \varepsilon\}},
\end{aligned}$$

almost surely. We can choose some versions of conditional expectations such that the estimate above holds identically for all $\psi \in \Psi$, in the same way as in the proof of (7.2.2). Take the supremum of (7.2.8) with respect to $\psi \in \Psi$, and let $n \to \infty$, then we get (7.2.6). A similar computation yields (7.2.7).

[STEP 3] Now it is easy to see that (7.2.2) and (7.2.6) imply the assertion (i), and that (7.2.3) and (7.2.7) do the assertion (ii); first choose $\varepsilon > 0$ small enough, and then, let $n \to \infty$. In order to show the assertion (iii), notice that for any $\varepsilon \in (0, a)$

$$\begin{vmatrix} \sum_{i=1}^{\sigma^n} |E_{i-1}^n \lambda_i^{n,a,\psi}|^2 - \sum_{i=1}^{\sigma^n} |E_{i-1}^n \lambda_i^{n,\varepsilon,\psi}|^2 \\ = \left| \sum_{i=1}^{\sigma^n} E_{i-1}^n (\lambda_i^{n,a,\psi} + \lambda_i^{n,\varepsilon,\psi}) E_{i-1}^n (\lambda_i^{n,a,\psi} - \lambda_i^{n,\varepsilon,\psi}) \right| \\ \leq 2|\log(1-a^2)| \sum_{i=1}^{\sigma^n} \left| E_{i-1}^n \log \tilde{Z}_i^{n,\psi} \mathbf{1}_{\{\varepsilon < \overline{\zeta}_i^n \le a\}} \right| = o_{P^n}(1), \end{aligned}$$

hence

$$\sum_{i=1}^{\sigma^n} |E_{i-1}^n \lambda_i^{n,a,\psi}|^2 = \sum_{i=1}^{\sigma^n} |E_{i-1}^n \lambda_i^{n,\varepsilon,\psi}|^2 + o_{P^n}(1)$$

7. Miscellanies

$$\leq |\log(1-\varepsilon^2)| \sum_{i=1}^{\sigma^n} |E_{i-1}^n \lambda_i^{n,\varepsilon,\psi}| + o_{P^n}(1).$$

We therefore obtain (iii) by virtue of (7.2.4); first choose $\varepsilon > 0$ small enough, and then let $n \to \infty$.

7.3 Model Checking for a Non-linear Times Series

Let us consider the \mathbb{R} -valued time series $\{X_i\}_{i\in\mathbb{Z}}$ given by

$$X_i = \psi(X_{i-1}) + \varepsilon_i,$$

where ψ is an \mathbb{R} -valued function on \mathbb{R} and ε_i is an \mathbb{R} -valued random variable such that

$$P(\varepsilon_i \le 0 | \mathcal{F}_{i-1}) = \frac{1}{2}$$
 almost surely,

where $\mathcal{F}_i = \sigma\{X_j : j \leq i\}.$

Let $K : \mathbb{R} \to [0, \infty)$ be a kernel function with a compact support, and let $\{b_n\}_{n \in \mathbb{N}}$ be a sequence of positive constants such that $b_n \downarrow 0$ as $n \to \infty$. We introduce the stochastic process $V^n = (V^n(x) | x \in \mathbb{R})$ given by

$$V^n(x) = \frac{1}{\sqrt{n}} \sum_{i=1}^n Y_i^n(x) Z_i,$$

where

$$Y_i^n(x) = \frac{1}{b_n} \int_{-\infty}^x K\left(\frac{X_{i-1} - u}{b_n}\right) du$$

and

$$Z_i = \operatorname{sign}\left(X_i - \psi(X_{i-1})\right),\,$$

with

$$\operatorname{sign}(x) = \begin{cases} -1 & \text{for } x \leq 0, \\ 1 & \text{for } x > 0. \end{cases}$$

The process V^n is a "smoothed" version of that of Erlenmaier (1997) who considered a process V^n with $Y_i^n(x)$ replaced by $1_{(-\infty,x]}(X_{i-1})$.

In order to state some conditions which we shall assume, we denote by \widehat{F}_n and \widehat{f}_n the empirical distribution function and the empirical density function associated to the kernel function K and the bandwidth b_n , of the data $X_0, ..., X_{n-1}$: that is,

$$\widehat{F}_{n}(x) = \frac{1}{n} \sum_{i=1}^{n} \mathbb{1}_{(-\infty,x]}(X_{i-1});$$

$$\widehat{f}_{n}(x) = \frac{1}{nb_{n}} \sum_{i=1}^{n} K\left(\frac{X_{i-1}-x}{b_{n}}\right)$$

132

We are ready to present our result.

Theorem 7.3.1 Assume the following conditions (i) and (ii): (i) $\widehat{F}_n(x) \xrightarrow{P} F(x)$ for every $x \in \mathbb{R}$, where F is a continuous distribution function on \mathbb{R} ; (ii) there exists a function $g \in \mathcal{L}^p(\mathbb{R})$ such that $\|\widehat{f}_n/g\|_{\mathcal{L}^q(\mathbb{R})} = O_P(1)$ for some $p, q \in (1, \infty)$ such that (1/p) + (1/q) = 1. Then, it holds that $V^n \xrightarrow{P} V$ in $\ell^{\infty}(\mathbb{R})$ where $V(x) = B_{F(x)}$ and $t \rightsquigarrow B_t$ is a standard Brownian motion on [0, 1].

Notice that, when the time series takes values only in a bounded subset of \mathbb{R} , a sufficient condition for (ii) is that $\|\widehat{f}_n\|_{\mathcal{L}^q(\mathbb{R})} = O_P(1)$ for an arbitrary q > 1. The result above, together with the continuous mapping theorem, yields the following.

Corollary 7.3.2 Define $S^n = \sup_{x \in \mathbb{R}} |V^n(x)|$. Under the conditions (i) and (ii) of Theorem 7.3.1, it holds that $S^n \stackrel{P}{\Longrightarrow} \sup_{t \in [0,1]} |B_t|$ in \mathbb{R} , where $t \rightsquigarrow B_t$ is a standard Brownian motion on [0,1].

We have thus obtained the asymptotically distribution-free test statistics S^n . Notice that the time series $\{X_i\}_{i\in\mathbb{Z}}$ need not be Markovian (the noise ε_i may depend on the whole past).

Proof of Theorem 7.3.1. We can write $V^n(x) = \sum_{i=1}^n \xi_i^{n,x}$, where $\xi_i^{n,x} = \frac{1}{\sqrt{n}} Y_i^n(x) Z_i.$

It is clear that $\{\xi_i^n\}_{i\in\mathbb{N}} = \{(\xi_i^{n,x}|x\in\mathbb{R})\}_{i\in\mathbb{N}}$ is an $\ell^{\infty}(\mathbb{R})$ -valued martingale difference array on the discrete-time stochastic basis $(\Omega, \mathcal{F}, \{\mathcal{F}_i\}_{i\in\mathbb{N}_0}, P)$, where $\mathcal{F}_i = \sigma\{X_j : j \leq i\}$. We will check the conditions of Theorem 3.3.2.

To check **[PE']**, for every $\varepsilon > 0$, choose some finite points $\{x_{\varepsilon,k} : 1 \le k \le N(\varepsilon) - 1\}$ of \mathbb{R} such that

$$x_{\varepsilon,0} = -\infty < x_{\varepsilon,1} < x_{\varepsilon,2} < \dots < x_{\varepsilon,N(\varepsilon)-1} < \infty = x_{\varepsilon,N(\varepsilon)}$$

and that

$$\int_{x_{\varepsilon,k-1}}^{x_{\varepsilon,k}} |g(x)|^p dx \le \varepsilon^{2p} \qquad \forall k = 1, ..., N(\varepsilon).$$

This can be done with $N(\varepsilon) \leq \text{const.}\varepsilon^{-2p}$, thus it holds that $\int_0^1 \sqrt{\log N(\varepsilon)} d\varepsilon < \infty$. On the other hand, it holds that

$$\begin{aligned} |\xi_i^{n,x} - \xi_i^{n,y}| &\leq \frac{1}{\sqrt{n}b_n} \int_{x \wedge y}^{x \vee y} K\left(\frac{X_{i-1} - u}{b_n}\right) du \\ &\leq \frac{1}{\sqrt{n}b_n} \int_{x_{\varepsilon,k-1}}^{x_{\varepsilon,k}} K\left(\frac{X_{i-1} - u}{b_n}\right) du \qquad \text{if } x, y \in (x_{\varepsilon,k-1}, x_{\varepsilon,k}]. \end{aligned}$$

So we have

$$\begin{split} \sum_{i=1}^{n} E_{i-1} \left[\sup_{x,y \in (x_{\varepsilon,k-1}, x_{\varepsilon,k}]} |\xi_{i}^{n,x} - \xi_{i}^{n,y}| \right]_{\mathcal{F}_{i}}^{2} \\ &\leq \left. \frac{1}{n} \sum_{i=1}^{n} \left| \frac{1}{b_{n}} \int_{x_{\varepsilon,k-1}}^{x_{\varepsilon,k}} K\left(\frac{X_{i-1} - u}{b_{n}}\right) du \right|^{2} \\ &\leq \left. \frac{1}{n} \sum_{i=1}^{n} \frac{1}{b_{n}} \int_{x_{\varepsilon,k-1}}^{x_{\varepsilon,k}} K\left(\frac{X_{i-1} - u}{b_{n}}\right) du \\ &= \int_{\mathbb{R}} 1_{(x_{\varepsilon,k-1}, x_{\varepsilon,k}]}(u) \widehat{f}_{n}(u) du \\ &= \left(\int_{\mathbb{R}} |1_{(x_{\varepsilon,k-1}, x_{\varepsilon,k}]}(u) g(u)|^{p} du \right)^{1/p} \cdot \left(\int_{\mathbb{R}} |\widehat{f}_{n}(u)/g(u)|^{q} du \right)^{1/q} \\ &\leq \varepsilon^{2} \cdot \|\widehat{f}_{n}/g\|_{\mathcal{L}^{q}(\mathbb{R})}. \end{split}$$

Hence the condition (ii) implies $[\mathbf{PE'}]$.

To check [C2'], first observe that

(7.3.1)
$$\int_{-\infty}^{x} \widehat{f}_{n}(u) du \xrightarrow{P} F(x) \qquad \forall x \in \mathbb{R}.$$

This fact is proved as follows. Since K has a compact support and since $b_n \downarrow 0$, it holds that for any $\varepsilon > 0$

$$\widehat{F}_n(x-\varepsilon) \le \frac{1}{nb_n} \sum_{i=1}^n \int_{-\infty}^x K\left(\frac{X_{i-1}-u}{b_n}\right) du \le \widehat{F}_n(x+\varepsilon)$$

for all sufficiently large $n \in \mathbb{N}$. Due to (i), the left and the right hand side converge in probability to $F(x - \varepsilon)$ and $F(x + \varepsilon)$, respectively. The claim (7.3.1) follows from the assumption that $x \rightsquigarrow F(x)$ is continuous.

Let us now turn to the convergence of $\sum_{i=1}^{n} E_{i-1} \xi_i^{n,x} \xi_i^{n,y}$'s. In case of $x \neq y$, we have that

$$\sum_{i=1}^{n} E_{i-1}\xi_{i}^{n,x}\xi_{i}^{n,y} = \frac{1}{n}\sum_{i=1}^{n}Y_{i}^{n}(x)Y_{i}^{n}(y)$$
$$= \frac{1}{n}\sum_{i=1}^{n}Y_{i}^{n}(x \wedge y) \quad \text{for all sufficiently large } n \in \mathbb{N}$$
$$= \int_{-\infty}^{x \wedge y}\widehat{f}_{n}(u)du,$$

which converges in probability to $F(x \wedge y)$ due to (7.3.1). In case of x = y, observe that

$$\sum_{i=1}^{n} E_{i-1}\xi_{i}^{n,x-\varepsilon}\xi_{i}^{n,x} \le \sum_{i=1}^{n} E_{i-1}|\xi_{i}^{n,x}|^{2} \le \sum_{i=1}^{n} E_{i-1}\xi_{i}^{n,x}\xi_{i}^{n,x+\varepsilon} \qquad \forall \varepsilon > 0,$$

and that the left and the right hand side converge in probability to $F(x - \varepsilon)$ and F(x), respectively. Thus it follows from the continuity of $x \rightsquigarrow F(x)$ that the middle of the above inequalities converges in probability to F(x). We therefore have shown that

$$\sum_{i=1}^{n} E_{i-1}\xi_{i}^{n,x}\xi_{i}^{n,y} \xrightarrow{P} F(x \wedge y) \qquad \forall x, y \in \mathbb{R}.$$

It is trivial that the Lindeberg condition $[\mathbf{L2'}]$ is satisfied, and all conditions of Theorem 3.3.2 have been established.

7.A Notes

The bandwidth processes or/and deviation processes of kernel density estimators as random elements taking values in the space C were studied by Krieger and Pickands, III (1981), Müller and Prewitt (1992, 1993), and Müller and Wang (1990). Theorem 7.1.4 could be obtained from a general study of "local empirical processes" by Einmahl and Mason (1997) combined with a uniform Donsker theorem by Sheehy and Wellner (1992). Although the notion of "local empirical process" is more general than the local kernel estimators, their approach is essentially based on the i.i.d. setup.

The asymptotic behavior of the log-likelihood ratio random fields in finite-dimensional parametric models has been studied by many authors including Le Cam (1970), Inagaki and Ogata (1975), Ogata and Inagaki (1977), Ibragimov and Has'minskii (1981), Kutoyants (1984) and Vostrikova (1987). Although no result for infinite-dimensional cases seems to have been presented in the literature so far, some results in i.i.d. cases are immediate from the Donsker theorems for empirical processes. Theorem 7.2.1 seems the first to consider the general statistical experiment with abstract parameters.

The problem considered in Section 7.3 and the basic idea of the test statistics were posed by Erlenmaier (1997), who obtained the same conclusion as Theorem 7.3.1 for a slightly different statistics in a Markovian case under an explicit assumption on the transition kernel.

Bibliography

- Andersen, P.K., Borgan, Ø., Gill, R.D. and Keiding, N. (1993). Statistical Models Based on Counting Processes. Springer-Verlag, New York.
- [2] Andersen, N.T., Giné, E., Ossiander, M. and Zinn, J. (1988). The central limit theorem and the law of iterated logarithm for empirical processes under local conditions. *Probab. Theory Relat. Fields* 77 271-305.
- [3] Arcones, M.A. and Yu, B. (1994). Central limit theorems for empirical and U-processes of stationary mixing sequences. J. Theoret. Probab. 7 47-71.
- [4] Bae, J. (1995). An empirical CLT for stationary martingale differences. J. Korean Math. Soc. 32 427-446.
- [5] Bae, J. and Levental, S. (1995a). Uniform CLT for Markov chains and its invariance principle: a martingale approach. J. Theoret. Probab. 8 549-570.
- [6] Bae, J. and Levental, S. (1995b). A uniform CLT for continuous martingales. J. Korean Statist. Soc. 24 225-231.
- [7] Birgé, L. and Massart, P. (1993). Rates of convergence for minimum contrast estimators. Probab. Theory Relat. Fields 97 113-150.
- [8] Chernoff, H. (1964). Estimation of the mode. Ann. Inst. Statist. Math. 16 31-41.
- [9] Doukhan, P., Massart, P. and Rio, E. (1995). Invariance principles for absolutely regular empirical processes. Ann. Inst. Henri Poincaré Probab. Statist. 31 393-427.
- [10] Dudley, R.M. (1978). Central limit theorems for empirical measures. Ann. Probab.
 6 899-929. Correction, 7 909-911.
- [11] Einmahl, U. and Mason, D.M. (1997). Gaussian approximation of local empirical processes indexed by functions. *Probab. Theory Relat. Fields* 107 283-311.
- [12] Erlenmaier, U. (1997). A new criterion for tightness of stochastic processes and an application to Markov processes. *Manuscript*.
- [13] Freedman, D.A. (1975). On tail probabilities for martingales. Ann. Probab. 3 100-118.
- [14] Genon-Catalot, V. (1990). Maximum contrast estimation for diffusion processes from discrete observations. *Statistics* 21 99-116.
- [15] Gill, R.D. (1980). Censoring and Stochastic Integrals. Mathematical Centre Tracts 124. Mathematisch Centrum, Amsterdam.
- [16] Hida, T. (1980). Brownian Motion. Springer-Verlag, New York.
- [17] Ibragimov, I.A. and Has'minskii, R.Z. (1981). Statistical Estimation. Springer-Verlag, New York.
- [18] Inagaki, N. and Ogata, Y. (1975). The weak convergence of likelihood ratio random fields and its applications. Ann. Inst. Statist. Math. 27 391-419.
- [19] Jacod, J. and Shiryaev, A.N. (1987). Limit Theorems for Stochastic Processes. Springer-Verlag, Berlin Heidelberg.
- [20] Jain, N. and Marcus, M. (1975). Central limit theorems for C(S)-valued random variables. J. Funct. Anal. 19 216-231.
- [21] Kim, J. and Pollard, D. (1990). Cube root asymptotics. Ann. Statist. 18 191-219.
- [22] Korostelev, A. (1987). On minimax estimation of discontinuous signal. Theory Probab. Appl. 32 727-730.
- [23] Krieger, A.M. and Pickands, J., III (1981). Weak convergence and efficient density estimation at a point. Ann. Statist. 9 1066-1078.
- [24] Kutoyants, Yu. (1984). Parameter Estimation for Stochastic Processes. Heldermann Verlag, Berlin.
- [25] Kutoyants, Yu. (1994). Identification of Dynamical Systems with Small Noise. Kluwer Academic Publishers, Dordrecht.
- [26] Lánska, V. (1979). Minimum contrast estimation in diffusion processes. J. Appl. Probab. 16 65-75.

- [27] Le Cam, L. (1970). On the assumptions used to prove asymptotic normality of maximum likelihood estimates. Ann. Math. Statist. 41 802-828.
- [28] Ledoux, M. and Talagrand, M. (1991). Probability in Banach Spaces. Springer-Verlag, Berlin Heidelberg.
- [29] Levental, S. (1989). A uniform CLT for uniformly bounded families of martingale differences. J. Theoret. Probab. 2 271-287.
- [30] Liptser, R.S. and Shiryayev A.N. (1989). Theory of Martingales. Kluwer Academic Publishers, Dordrecht.
- [31] McKeague, I.W. and Utikal, K.J. (1990). Identifying nonlinear covariate effects in semimartingale regression models. *Probab. Theory Relat. Fields* 87 1-25.
- [32] Müller, H.G. and Prewitt, K.A. (1992). Weak convergence and adaptive peak estimation for spectral densities. Ann. Statist. 20 1329-1349.
- [33] Müller, H.G. and Prewitt, K.A. (1993). Multiparameter bandwidth processes and adaptive surface smoothing. J. Multivariate Anal. 47 1-21.
- [34] Müller, H.G. and Wang, J.L. (1990). Locally adaptive hazard smoothing. Probab. Theory Relat. Fields 85 523-538.
- [35] Nishiyama, Y. (1997). Some central limit theorems for ℓ[∞]-valued semimartingales and their applications. Probab. Theory Relat. Fields 108 459-494.
- [36] Nishiyama, Y. (1999a). A maximal inequality for continuous martingales and Mestimation in a Gaussian white noise model. Ann. Statist. 27 675-696.
- [37] Nishiyama, Y. (1999b). Weak convergence of some classes of martingales with jumps. To appear in Ann. Probab.
- [38] Ogata, Y. and Inagaki, N. (1977). The weak convergence of the likelihood ratio random fields for Markov observations. Ann. Inst. Statist. Math. 29 165-187.
- [39] Ossiander, M. (1987). A central limit theorem under metric entropy with L_2 bracketing. Ann. Probab. 15 897-919.
- [40] Pollard, D. (1982). A central limit theorem for empirical processes. J. Austral. Math. Soc. Ser. A 33 235-248.

- [41] Pollard, D. (1989). Asymptotics via empirical processes (with discussion). Statistical Science 4 341-366.
- [42] Pollard, D. (1990). Empirical Processes: Theory and Applications. Inst. Math. Statist., Hayward.
- [43] Rogers, L.C.G. and Williams, D. (1987). Diffusions, Markov Processes, and Martingales. Volume 2: Itô Calculus. John Wiley & Sons, Chichester New York Brisbane Tronto Singapore.
- [44] Sheehy, A. and Wellner, J.A. (1992). Uniform Donsker classes of functions. Ann. Probab. 20 1983-2030.
- [45] van de Geer, S. (1990). Estimating a regression function. Ann. Statist. 18 907-924.
- [46] van de Geer, S. (1993). Hellinger-consistency of certain nonparametric maximum likelihood estimators. Ann. Statist. 21 14-44.
- [47] van de Geer, S. (1995a). The method of sieves and minimum contrast estimators. Math. Methods Statist. 1 20-38.
- [48] van de Geer, S. (1995b). Exponential inequalities for martingales, with application to maximum likelihood estimation for counting processes. Ann. Statist. 23 1779-1801.
- [49] van de Geer, S. (1997). Empirical Processes and the Quality of Estimators in Complex Parameter Space. Manuscript for forthcoming Lecture Notes.
- [50] van der Vaart, A.W. and Wellner, J.A. (1996). Weak Convergence and Empirical Processes: With Applications to Statistics. Springer-Verlag, New York.
- [51] Vostrikova, L. (1987). On the weak convergence of likelihood ratio processes of general statistical parametric models. *Stochastics* 23 277-298.
- [52] Wang, Y. (1995). Jump and sharp cusp detection by wavelets. Biometrika 82 385-397.
- [53] Wong, W.H. and Severini, T.A. (1991). On maximum likelihood estimation in infinite dimensional parameter spaces. Ann. Statist. 19 603-632.
- [54] Wong, W.H. and Shen, X. (1995). Probability inequalities for likelihood ratios and convergence rates of sieve MLEs. Ann. Statist. 23 339-362.

- [55] Yoshida, N. (1990). Asymptotic behavior of *M*-estimator and related random field for diffusion process. *Ann. Inst. Statist. Math.* **42** 221-251.
- [56] Yoshida, N. (1992). Estimation for diffusion processes from discrete observation. J. Multivariate Anal. 41 220-242.

CWI TRACTS

- 1 D.H.J. Epema. Surfaces with canonical hyperplane sections. 1984.
- 2 J.J. Dijkstra. Fake topological Hilbert spaces and characterizations of dimension in terms of negligibility. 1984.
- 3 A.J. van der Schaft. System theoretic descriptions of physical systems. 1984.
- 4 J. Koene. Minimal cost flow in processing networks, a primal approach. 1984.
- 5 B. Hoogenboom. Intertwining functions on compact Lie groups. 1984.
- 6 A.P.W. Böhm. Dataflow computation. 1984.
- 7 A. Blokhuis. Few-distance sets. 1984.
- 8 M.H. van Hoorn. Algorithms and approximations for queueing systems. 1984.
- 9 C.P.J. Koymans. *Models of the lambda calculus.* 1984.
- 10 C.G. van der Laan, N.M. Temme. Calculation of special functions: the gamma function, the exponential integrals and error-like functions. 1984.
- 11 N.M. van Dijk. Controlled Markov processes; time-discretization. 1984.
- 12 W.H. Hundsdorfer. The numerical solution of nonlinear stiff initial value problems: an analysis of one step methods. 1985.
- 13 D. Grune. On the design of ALEPH. 1985.
- 14 J.G.F. Thiemann. Analytic spaces and dynamic programming: a measure theoretic approach. 1985.
- 15 F.J. van der Linden. Euclidean rings with two infinite primes. 1985.
- 16 R.J.P. Groothuizen. Mixed elliptic-hyperbolic partial differential operators: a case-study in Fourier integral operators. 1985.
- 17 H.M.M. ten Eikelder. Symmetries for dynamical and Hamiltonian systems. 1985.
- 18 A.D.M. Kester. Some large deviation results in statistics. 1985.
- 19 T.M.V. Janssen. Foundations and applications of Montague grammar, part 1: Philosophy, framework, computer science. 1986.
- 20 B.F. Schriever. Order dependence. 1986.
- 21 D.P. van der Vecht. Inequalities for stopped Brownian motion. 1986.
- 22 J.C.S.P. van der Woude. *Topological dynamix.* 1986.
- 23 A.F. Monna. *Methods, concepts and ideas in mathematics: aspects of an evolution.* 1986.
- 24 J.C.M. Baeten. Filters and ultrafilters over definable subsets of admissible ordinals. 1986.
- 25 A.W.J. Kolen. Tree network and planar rectilinear location theory. 1986.
- 26 A.H. Veen. The misconstrued semicolon: Reconciling imperative languages and dataflow machines. 1986.
- 27 A.J.M. van Engelen. Homogeneous zerodimensional absolute Borel sets. 1986.
- 28 T.M.V. Janssen. Foundations and applications of Montague grammar, part 2: Applications to natural language. 1986.
- 29 H.L. Trentelman. Almost invariant subspaces and high gain feedback. 1986.
- 30 A.G. de Kok. Production-inventory control models: approximations and algorithms. 1987.

- 31 E.E.M. van Berkum. Optimal paired comparison designs for factorial experiments. 1987.
- 32 J.H.J. Einmahl. *Multivariate empirical processes*. 1987.
- 33 O.J. Vrieze. Stochastic games with finite state and action spaces. 1987.
- 34 P.H.M. Kersten. Infinitesimal symmetries: a computational approach. 1987.
- 35 M.L. Eaton. Lectures on topics in probability inequalities. 1987.
- 36 A.H.P. van der Burgh, R.M.M. Mattheij (editors). Proceedings of the first international conference on industrial and applied mathematics (ICIAM 87). 1987.
- 37 L. Stougie. Design and analysis of algorithms for stochastic integer programming. 1987.
- 38 J.B.G. Frenk. On Banach algebras, renewal measures and regenerative processes. 1987.
- 39 H.J.M. Peters, O.J. Vrieze (eds.). Surveys in game theory and related topics. 1987.
- 40 J.L. Geluk, L. de Haan. *Regular variation, extensions and Tauberian theorems.* 1987.
- 41 Sape J. Mullender (ed.). The Amoeba distributed operating system: Selected papers 1984-1987. 1987.
- 42 P.R.J. Asveld, A. Nijholt (eds.). Essays on concepts, formalisms, and tools. 1987.
- 43 H.L. Bodlaender. Distributed computing: structure and complexity. 1987.
- 44 A.W. van der Vaart. Statistical estimation in large parameter spaces. 1988.
- 45 S.A. van de Geer. *Regression analysis and empirical processes*. 1988.
- 46 S.P. Spekreijse. Multigrid solution of the steady Euler equations. 1988.
- 47 J.B. Dijkstra. Analysis of means in some nonstandard situations. 1988.
- 48 F.C. Drost. Asymptotics for generalized chisquare goodness-of-fit tests. 1988.
- 49 F.W. Wubs. Numerical solution of the shallowwater equations. 1988.
- 50 F. de Kerf. Asymptotic analysis of a class of perturbed Korteweg-de Vries initial value problems. 1988.
- 51 P.J.M. van Laarhoven. Theoretical and computational aspects of simulated annealing. 1988.
- 52 P.M. van Loon. *Continuous decoupling transformations for linear boundary value problems.* 1988.
- 53 K.C.P. Machielsen. Numerical solution of optimal control problems with state constraints by sequential quadratic programming in function space. 1988.
- 54 L.C.R.J. Willenborg. Computational aspects of survey data processing. 1988.
- 55 G.J. van der Steen. A program generator for recognition, parsing and transduction with syntactic patterns. 1988.
- 56 J.C. Ebergen. Translating programs into delayinsensitive circuits. 1989.
- 57 S.M. Verduyn Lunel. Exponential type calculus for linear delay equations. 1989.
- 58 M.C.M. de Gunst. A random model for plant cell population growth. 1989.
- 59 D. van Dulst. Characterizations of Banach spaces not containing l¹. 1989.

- 60 H.E. de Swart. Vacillation and predictability properties of low-order atmospheric spectral models. 1989.
- 61 P. de Jong. Central limit theorems for generalized multilinear forms. 1989.
- 62 V.J. de Jong. A specification system for statistical software. 1989.
- 63 B. Hanzon. Identifiability, recursive identification and spaces of linear dynamical systems, part I. 1989.
- 64 B. Hanzon. Identifiability, recursive identification and spaces of linear dynamical systems, part II. 1989.
- 65 B.M.M. de Weger. Algorithms for diophantine equations. 1989.
- 66 A. Jung. Cartesian closed categories of domains. 1989.
- 67 J.W. Polderman. Adaptive control & identification: Conflict or conflux?. 1989.
- 68 H.J. Woerdeman. *Matrix and operator extensions*. 1989.
- 69 B.G. Hansen. Monotonicity properties of infinitely divisible distributions. 1989.
- 70 J.K. Lenstra, H.C. Tijms, A. Volgenant (eds.). Twenty-five years of operations research in the Netherlands: Papers dedicated to Gijs de Leve. 1990.
- 71 P.J.C. Spreij. Counting process systems. Identification and stochastic realization. 1990.
- 72 J.F. Kaashoek. Modeling one dimensional pattern formation by anti-diffusion. 1990.
- 73 A.M.H. Gerards. Graphs and polyhedra. Binary spaces and cutting planes. 1990.
- 74 B. Koren. Multigrid and defect correction for the steady Navier-Stokes equations. Application to aerodynamics. 1991.
- 75 M.W.P. Savelsbergh. Computer aided routing. 1992.
- 76 O.E. Flippo. Stability, duality and decomposition in general mathematical programming. 1991.
- 77 A.J. van Es. Aspects of nonparametric density estimation. 1991.
- 78 G.A.P. Kindervater. Exercises in parallel combinatorial computing. 1992.
- 79 J.J. Lodder. Towards a symmetrical theory of generalized functions. 1991.
- 80 S.A. Smulders. *Control of freeway traffic flow.* 1996.
- 81 P.H.M. America, J.J.M.M. Rutten. A parallel object-oriented language: design and semantic foundations. 1992.
- 82 F. Thuijsman. *Optimality and equilibria in stochastic games.* 1992.
- 83 R.J. Kooman. Convergence properties of recurrence sequences. 1992.
- 84 A.M. Cohen (ed.). Computational aspects of Lie group representations and related topics. Proceedings of the 1990 Computational Algebra Seminar at CWI, Amsterdam. 1991.
- 85 V. de Valk. One-dependent processes. 1994.
- 86 J.A. Baars, J.A.M. de Groot. On topological and linear equivalence of certain function spaces. 1992.
- 87 A.F. Monna. The way of mathematics and mathematicians. 1992.
- 88 E.D. de Goede. Numerical methods for the threedimensional shallow water equations. 1993.

- 89 M. Zwaan. Moment problems in Hilbert space with applications to magnetic resonance imaging. 1993.
- 90 C. Vuik. The solution of a one-dimensional Stefan problem. 1993.
- 91 E.R. Verheul. Multimedians in metric and normed spaces. 1993.
- 92 J.L.M. Maubach. Iterative methods for non-linear partial differential equations. 1994.
- 93 A.W. Ambergen. Statistical uncertainties in posterior probabilities. 1993.
- 94 P.A. Zegeling. Moving-grid methods for timedependent partial differential equations. 1993.
- 95 M.J.C. van Pul. Statistical analysis of software reliability models. 1993.
- 96 J.K. Scholma. A Lie algebraic study of some integrable systems associated with root systems. 1993.
- 97 J.L. van den Berg. Sojourn times in feedback and processor sharing queues. 1993.
- 98 A.J. Koning. Stochastic integrals and goodnessof-fit tests. 1993.
- 99 B.P. Sommeijer. *Parallelism in the numerical integration of initial value problems.* 1993.
- 100 J. Molenaar. Multigrid methods for semiconductor device simulation. 1993.
- 101 H.J.C. Huijberts. Dynamic feedback in nonlinear synthesis problems. 1994.
- 102 J.A.M. van der Weide. *Stochastic processes and point processes of excursions*. 1994.
- 103 P.W. Hemker, P. Wesseling (eds.). Contributions to multigrid. 1994.
- 104 I.J.B.F. Adan. A compensation approach for queueing problems. 1994.
- 105 O.J. Boxma, G.M. Koole (eds.). Performance evaluation of parallel and distributed systems solution methods. Part 1. 1994.
- 106 O.J. Boxma, G.M. Koole (eds.). Performance evaluation of parallel and distributed systems solution methods. Part 2. 1994.
- 107 R.A. Trompert. Local uniform grid refinement for time-dependent partial differential equations. 1995.
- 108 M.N.M. van Lieshout. *Stochastic geometry models in image analysis and spatial statistics.* 1995.
- 109 R.J. van Glabbeek. Comparative concurrency semantics and refinement of actions. 1996.
- 110 W. Vervaat, H. Holwerda (ed.). *Probability and lattices.* 1997.
- 111 I. Helsloot. Covariant formal group theory and some applications. 1995.
- 112 R.N. Bol. Loop checking in logic programming. 1995.
- 113 G.J.M. Koole. Stochastic scheduling and dynamic programming. 1995.
- 114 M.J. van der Laan. Efficient and inefficient estimation in semiparametric models. 1995.
- 115 S.C. Borst. Polling models. 1996.
- 116 G.D. Otten. Statistical test limits in quality control. 1996.
- 117 K.G. Langendoen. Graph reduction on sharedmemory multiprocessors. 1996.
- 118 W.C.A. Maas. Nonlinear \mathcal{H}_{∞} control: the singular case. 1996.
- 119 A. Di Bucchianico. *Probabilistic and analytical aspects of the umbral calculus.* 1997.
- 120 M. van Loon. Numerical methods in smog prediction. 1997.

- 121 B.J. Wijers. Nonparametric estimation for a windowed line-segment process. 1997.
- 122 W.K. Klein Haneveld, O.J. Vrieze, L.C.M. Kallenberg (editors). *Ten years LNMB – Ph.D. research and graduate courses of the Dutch Network of Operations Research*. 1997.
- 123 R.W. van der Hofstad. One-dimensional random polymers. 1998.
- 124 W.J.H. Stortelder. Parameter estimation in nonlinear dynamical systems. 1998.
- 125 M.H. Wegkamp. Entropy methods in statistical estimation. 1998.
- 126 K. Aardal, J.K. Lenstra, F. Maffioli, D.B. Shmoys (eds.) Selected publications of Eugene L. Lawler. 1999.
- 127 E. Belitser. *Minimax estimation in regression and random censorship models.* 2000.
- 128 Y. Nishiyama. Entropy methods for martingales. 2000.

- 129 J.A. van Hamel. Algebraic cycles and topology of real algebraic varieties. 2000.
- 130 P.J. Oonincx. Mathematical signal analysis: wavelets, Wigner distribution and a seismic application. 2000.
- 131 M. Ruzhansky. *Regularity theory of Fourier integral operators with complex phases and singularities of affine fibrations.* 2001.
- 132 J.V. Stokman. Multivariable orthogonal polynomials and quantum Grassmannians. 2001.
- 133 N.R. Bruin. Chabauty methods and covering techniques applied to generalised Fermat equations. 2002.
- 134 E.H. van Brummelen. Numerical methods for steady viscous free-surface flows. 2003.
- 135 K. Dajani, J. von Reis (eds.). European Women in Mathematics – Marseille 2003. Proceedings of the 11th conference of EWM. 2005.